

Emotion Expression on Social Networking Sites:

Exploring Mood Profiles and Depression

Elizabeth Mary Seabrook

Bachelor of Psychology (Honours)

A thesis submitted for the degree of Doctor of Philosophy at

Monash University in 2018

Monash Institute of Cognitive and Clinical Neurosciences

School of Psychological Sciences

-Page intentionally left blank-

Table of	of Cor	ntents
----------	--------	--------

LIST OF TABLES	v
LIST OF FIGURES	VIII
COPYRIGHT NOTICES	X
ABSTRACT	XI
GENERAL DECLARATION	XIII
ACKNOWLEDGEMENTS	XVI
CHAPTER 1 – GENERAL INTRODUCTION	1
1.0 OVERVIEW AND CONTRIBUTION	1
1.1 INTRODUCTION	1
1.1.1 Depression and Emotion Dysregulation	2
1.1.2 Monitoring Depression: Social Media for Passive Data Collection	4
1.1.3 Research Rationale	5
1.2 RESEARCH QUESTIONS	6
1.3 STRUCTURE OF THE THESIS	6
CHAPTER 2 – SOCIAL NETWORKING SITES, DEPRESSION, AND ANXIETY: A S REVIEW	SYSTEMATIC
2.1 DEFAMILE TO THE SYSTEMATIC DEVIEW	0
2.2 Social Networking sites dedression and anyiety: A systematic deview	o
2.3 CONCLUDING REMARKS	60
CHAPTER 3 – GENERAL METHODOLOGY	62
3.0 INTRODUCTION	62 SAMPLING
METHODS	
3.2 THE DEVELOPMENT OF <i>MOODPRISM</i>	
3.2.1 Participants	
5.2.2 Privacy and Etnical Considerations	
5.2.5 Medsures	
3.2.3.1 Social media data collection by Mood Filsm	
3.2.5.2.1 Sychological measures and daily mood report items from Moodi rism	
3 3 CONCLUDING REMARKS	88
CHAPTER 4 – PREDICTING DEPRESSION FROM LANGUAGE-BASED EMOTION DYNAMICS: LONGITUDINAL ANALYSIS OF FACEBOOK AND TWITTER STAT UPDATES	N US 89
4.1 PREAMBLE TO EMPIRICAL PAPER 1	
4.2 DEPRESSION IS PREDICTED BY EMOTIONAL INSTABILITY ON FACEBOOK, BUT BY REDUCI	ED EMOTION
VARIABILITY ON TWITTER	90
4.3 CONCLUDING REMARKS	137
CHAPTER 5– DO PEOPLE FEEL WHAT THEY TWEET: A CASE-SERIES EXPLO DAILY MOOD RATINGS AND THE EMOTIONS EXPRESSED ON TWITTER	RATION OF
	120
5.1 PREAMBLE TO EMPIRICAL PAPER 2	138 ATINGS AND
THE EMOTIONS EXPRESSED ON TWITTER	139
5.3 CONCLUDING REMARKS	

CHAPTER 6 – GENERAL DISCUSSION AND CONCLUSIONS	187
6.0 Introduction	187
6.1 SUMMARY OF THE MAIN FINDINGS	187
6.1.1 Systematic Review of the Literature	187
6.1.2 Examination of Emotion Dynamics in Social Media posts in Relation to a Static Depression	
Severity Index	191
6.1.3 Emotion Language Expressed on Social Media and Concurrent Self-Reported Emotion: The	
Impact of Depression Severity	192
6.2 CONTRIBUTIONS OF THIS THESIS	193
6.2.1 Mood Profile Features in Status Updates Predict Depression Severity	193
6.2.2 The Language-Mood Gap: Implications for Depression Prediction from Social Media Conte	nt . 195
6.3 PRACTICAL IMPLICATIONS	198
6.3.1 For Social Media Users	198
6.3.2 For Researchers	199
6.3.3 For Social Media Providers and Mental Health Services	200
6.4 STRENGTHS AND LIMITATIONS	201
6.5 FUTURE RESEARCH DIRECTIONS	206
6.6 CONCLUDING REMARKS	208
REFERENCES	210
APPENDIX A – DEVELOPMENT OF A MOBILE PHONE APP TO SUPPORT SELF- MONITORING OF EMOTIONAL WELL-BEING: A MENTAL HEALTH DIGITAL INNOVATION	253
APPENDIX B – HUMAN ETHICS CERTIFICATE OF APPROVAL	280
APPENDIX C – VICTORIAN GOVERNMENT DEPARTMENT OF EDUCATION AND TRAI RESEARCH PERMISSION	NING 281
APPENDIX D –PARTICIPANT INFORMATION SHEET AND CONSENT PROCESS	283
APPENDIX E – SUPPLEMENTARY POSITIVE AND NEGATIVE EMOJIS AND INTERNET SLANG FOR THE <i>LIWC 2007</i>	287

List of Tables

Chapter 2: Soc	cial Networking Sites, Depression, and Anxiety: A Systematic Review	V
Table 2.1	Broad functions of social networking site use and example	13
	behaviours.	
Table 2.2	Multimedia Appendix 1: Summary of studies included in the	28
	systematic review.	
Table 2.3	Multimedia Appendix 2: Main results table: associations	31
	between depression, anxiety, and social networking site	
	outcomes across the 70 reviewed studies.	
Chapter 3: Gei	neral Methodology	
Table 3.1	Frequency counts, means, and standard deviations of	69-71
	participants who opted-in and opted-out of contributing social	
	media data to MoodPrism.	
Table 3.2	Raw social media data collected by MoodPrism via the	76
	Facebook Graph API and Twitter REST API.	
Table 3.3	Measures delivered at baseline only by MoodPrism.	82
Table 3.4	Measures delivered at baseline and follow-up by MoodPrism.	83
Table 3.5	Items in the daily mood reports delivered by MoodPrism.	84
Chapter 4: Pre	edicting Depression from Language-Based Emotion Dynamics: Long	itudinal
Analysis of Fac	cebook and Twitter Status Updates	
Table 4.1	Definitions and conceptual overlap of variability, instability,	95
	and inertia.	
Table 4.2	Descriptive statistics of the PHQ-9, status update frequency,	105
	and the emotion features expressed in status updates on	
	Facebook ($n = 29$) and Twitter ($n = 49$).	

Table 4.3	Spearman's rho correlation analyses between depression	106
	severity (as rated by the PHQ-9) and the positive emotion	
	features expressed in status updates on Facebook $(n = 29)$ and	
	Twitter ($n = 49$).	
Table 4.4	Spearman's rho correlation analyses between depression	107
	severity (as rated by the PHQ-9) and the negative emotion	
	features expressed in status updates on Facebook $(n = 29)$ and	
	Twitter $(n = 49)$.	
Appendix 1	Emoji & internet slang additions to the LIWC 2007 emotion	130
	dictionaries.	
Appendix 2	Means and standard deviations of psychological	133
	characteristics in the Facebook and Twitter samples.	
Chapter 5: Do people	feel what they Tweet? A case-series exploration of daily mood rati	ings
Chapter 5: Do people and the emotions exp	feel what they Tweet? A case-series exploration of daily mood rati ressed on Twitter	ings
Chapter 5: Do people and the emotions exp Table 5.1	feel what they Tweet? A case-series exploration of daily mood rati ressed on Twitter Demographic characteristics of participants/cases.	ings 146
Chapter 5: Do people and the emotions exp Table 5.1 Table 5.2	feel what they Tweet? A case-series exploration of daily mood ration ressed on Twitter Demographic characteristics of participants/cases. Depression scores (T1, T2), and Tweet and daily mood report	ings 146 152
Chapter 5: Do people and the emotions exp Table 5.1 Table 5.2	feel what they Tweet? A case-series exploration of daily mood ration ressed on Twitter Demographic characteristics of participants/cases. Depression scores (T1, T2), and Tweet and daily mood report activity across the 30-day study period (d1-d30).	ings 146 152
Chapter 5: Do people and the emotions exp Table 5.1 Table 5.2 Table 5.3	feel what they Tweet? A case-series exploration of daily mood rational research on Twitter Demographic characteristics of participants/cases. Depression scores (T1, T2), and Tweet and daily mood report activity across the 30-day study period (d1-d30). Spearman ranked correlation between SR- and T-Positive (and	ings 146 152 157
Chapter 5: Do people and the emotions exp Table 5.1 Table 5.2 Table 5.3	feel what they Tweet? A case-series exploration of daily mood rational research on Twitter Demographic characteristics of participants/cases. Depression scores (T1, T2), and Tweet and daily mood report activity across the 30-day study period (d1-d30). Spearman ranked correlation between SR- and T-Positive (and Negative) mood profiles by participant.	ings 146 152 157
Chapter 5: Do people and the emotions exp Table 5.1 Table 5.2 Table 5.3	feel what they Tweet? A case-series exploration of daily mood rational research on Twitter Demographic characteristics of participants/cases. Depression scores (T1, T2), and Tweet and daily mood report activity across the 30-day study period (d1-d30). Spearman ranked correlation between SR- and T-Positive (and Negative) mood profiles by participant. Descriptive mood profile features from self-report and Twitter	ings 146 152 157 159
Chapter 5: Do people and the emotions exp Table 5.1 Table 5.2 Table 5.3 Table 5.4	feel what they Tweet? A case-series exploration of daily mood rations ressed on Twitter Demographic characteristics of participants/cases. Depression scores (T1, T2), and Tweet and daily mood report activity across the 30-day study period (d1-d30). Spearman ranked correlation between SR- and T-Positive (and Negative) mood profiles by participant. Descriptive mood profile features from self-report and Twitter positive and negative mood data.	ings 146 152 157 159
Chapter 5: Do people and the emotions exp Table 5.1 Table 5.2 Table 5.3 Table 5.4	feel what they Tweet? A case-series exploration of daily mood rations ressed on Twitter Demographic characteristics of participants/cases. Depression scores (T1, T2), and Tweet and daily mood report activity across the 30-day study period (d1-d30). Spearman ranked correlation between SR- and T-Positive (and Negative) mood profiles by participant. Descriptive mood profile features from self-report and Twitter positive and negative mood data. Salient differences by feature and by participant between the	ings 146 152 157 159 163

vi

Twitter.

Appendix A	Emoji & internet slang additions to the LIWC 2007 emotion	181
	dictionaries.	
Appendix B.1	Raw score values for descriptive mood profile features from	182
	self-report and Twitter positive and negative mood data.	

List of Figures

Chapter 2: Social Net	tworking Sites, Depression, and Anxiety: A Systematic Review	
Figure 2.1	Overview of search strategy and selection process for the	11
	systematic review.	
Figure 2.2	Publication frequency of research into well-being, depression	12
	or anxiety only, and depression or anxiety with well-being	
	from 2005 to June 2016.	
Chapter 3: General M	Aethodology	
Figure 3.1	Visualisation of the geopolitical discrepancies between the (a)	64
	Facebook and (b) UnFacebook world.	
Figure 3.2	Participant selection flow chart.	68
Figure 3.3	MoodPrism screenshots showing the participant's experience	87
	through the general research procedure.	
Chapter 4: Predicting	g Depression from Language-Based Emotion Dynamics: Longitudin	al
Analysis of Facebook	and Twitter Status Updates	
Figure 4.1	A simulated time-series showing the proportion of negative	103
	emotion words used in status updates of 14 days.	
Figure 4.2	Graphs showing the proportion of negative emotion words	109
	used in individual status updates on Facebook across 35 days.	
Figure 4.3	Graphs showing the proportion of negative emotion words	111
	used in individual status updates across (a) 160 and (b) 182	
	days.	
Appendix 3	Example participant data showing instability under fixed	136

variability (*iSD*) conditions.

Chapter 5: Do people feel what they Tweet? A case-series exploration of daily mood ratings

and the emotions expressed on Twitter

Figure 5.1	MoodPrism procedure for Twitter word count extraction.	148
Figure 5.2	Panel plots of z-transformed positive and negative mood per	154-156
	day from MoodPrism self-report and Twitter emotion word	
	use.	
Figure 5.3	Acute change between mood observations on self-report and	161
	Twitter time-series.	
Appendix B.1	Raw score dual axis plots of self-reported daily mood and	183-185
	daily emotion words in Tweets across the 30-day study.	

Chapter 6: General Discussion

Figure 6.1	The <i>MoodPrism</i> compliance curve across the 30-day study.	204
1 15010 0.1	The mooth risk compliance carve across the 50 day stady.	204

Copyright notices

© The author (2018). Except as provided in the Copyright Act 1968, this thesis may not be reproduced in any form without the written permission of the author.

I certify that I have made all reasonable efforts to secure copyright permissions for third-party content included in this thesis and have not knowingly added copyright content to my work without the owner's permission.

Abstract

This thesis explores the use of emotion language on social media and its relationship with depression. Previous research has shown that depression is visible on social media through the language used in status updates. Typically, using negative emotion words over time is thought to reflect depression symptoms such as persistent low mood. While the detection of depression from social media language generally performs well, the predictive performance of language models seeking to identify depressed individuals range from .30 to around .80 accuracy, suggesting that both sensitivity and specificity of depression identification from these models could be improved. In addition, the research to date has primarily utilized depression assessed at a single time-point to infer depression status of individuals, with little verification that the content of social media posts is reflective of how an individual is feeling in real time. Considering the time-sensitive data available from social media, it is likely that the dynamic patterns of negative emotion words across status updates will provide more sensitive identification and earlier prediction of mental health status as a reflection of underlying emotion processes.

The research reported in this thesis aimed to address two primary questions:

(1) What emotion dynamic features in the mood profiles generated across status updates are associated with depression severity? and;

(2) Do the emotions expressed on social media through language accurately reflect subjective daily mood?

A systematic review of the literature was conducted to explore the intersections between mental health and social media use which identified several behaviors, including language use, that were consistently linked to poorer mental health outcomes. This review also indicated the need for a data collection method that better integrated social media data with the psychological information of participants. *MoodPrism*, a smartphone experience sampling app collecting psychological,

xi

contextual, social media, and daily mood data was developed and utilized to meet this methodological need.

Examination of the patterns of emotion word use across status updates revealed that negative affect instability was visible on the social media platform Facebook and that it was associated with greater depression severity. On Twitter, greater variability of negative emotion word use was predictive of lesser depression symptom severity. A further detailed exploration of daily mood and social media language was conducted with five participants through case studies revealing that the emotion words used on Twitter do not correspond well with subjective daily mood, indicating that language may not reliably sample experienced mood over time. This thesis demonstrates the need to move beyond static measure of language alone in identifying individuals at risk of depression on social media. It shows that by considering indices of variability and instability more sensitive depression prediction can be achieved. It also highlights the need to take into account factors other than language in collecting reliable and valid data for depression. The practical extensions for language-based depression detection discussed in this thesis may improve the specificity of the automated monitoring of depression risk on social media. Integrating accurate and automated language-models with tailored advertising and push notifications on social media will greatly improve the speed with which support, and resources can be delivered to at risk individuals in the community.

General Declaration

Publications during enrolment

- Rickard, N., Arjmand, H.-A., Bakker, D., & Seabrook, E. (2016). Development of a mobile phone app to support self-monitoring of emotional well-being: A mental health digital innovation. *JMIR Mental Health*, *3*(4), e49. doi: 10.2196/mental.6202
- Seabrook, E. M., Kern, M. L., & Rickard, N. S. (2016). Social networking sites, depression, and anxiety: A systematic review. *JMIR Mental Health*, *3*(4), e50. doi: 10.2196/mental.5842
- Seabrook, E. M., Kern, M. L., Fulcher, B. D., & Rickard, N. S. (2018). Predicting depression from language-based emotion dynamics: Longitudinal analysis of Facebook and Twitter status updates. *JMIR (forthcoming)*. doi:10.2196/jmir.9267
- Seabrook, E. M., Kern, M. L., Fulcher, B. D., & Rickard, N. S. (2018). Do people Tweet what they feel? A case-series exploration of daily mood ratings and the emotions expressed on Twitter Submitted for publication.

I hereby declare that this thesis contains no material which has been accepted for the award of any other degree or diploma at any university or equivalent institution and that, to the best of my knowledge and belief, this thesis contains no material previously published or written by another person, except where due reference is made in the text of the thesis.

This thesis includes *two* original papers published in peer reviewed journals and *one* submitted publication. The core theme of the thesis is emotion expression on social media and depression. The ideas, development and writing up of all the papers in the thesis were the principal responsibility of myself, the student, working within the School of Psychological Sciences, Monash University under the supervision of Adjunct Associate Professor Nikki Rickard, Dr Peggy Kern, and Dr Ben Fulcher.

The inclusion of co-authors reflects the fact that the work came from active collaboration between researchers and acknowledges input into team-based research.

Thesis	Publication Title	Status	Nature and % of student	Co-author name(s)
Chapter			contribution	Nature and % of Co-
				author's contribution*
Chapter 2	Social networking sites, depression, and anxiety: A systematic review	Published	70% Conceptualisation, investigation and data collection, writing-original draft, writing-review and editing.	 Nikki Rickard 15% Peggy Kern 15% Supervisory guidance, framing, inter-rater relatibility, and editing.
Chapter 4	Predicting depression from language-based emotion dynamics: Longitudinal analysis of Facebook and Twitter status updates	Published	70% Conceptualisation, investigation and data collection, methodology, data curation, data analysis, writing-original draft, writing-review and editing.	1) Nikki Rickard 10% 2) Peggy Kern 10% 3) Ben Fulcher 10% Supervisory guidance, project oversight, contributions to data analysis, manuscript editing.

In the case of chapters 2, 4, and 5 my contribution to the work involved the following:

Thesis	Publication Title	Status	Nature and % of student	Co-author name (s)
Chapter			contribution	Nature and % of Co-
				author's contribution*
				1) Nikki Rickard 10%
			70%	2) Peggy Kern 10%
	De noorde feel whet		Conceptualisation,	3) Ben Fulcher 10%
	Do people feel what	nat use- n of Submitted gs ns	investigation and data	
they Chapter 5 series daily and expre	they I weet? A case-		collection, methodology,	Supervisory guidance,
	series exploration of		data curation, data	project oversight,
	daily mood ratings		analysis, writing-original	conceptualisation,
	and the emotions		draft, writing-review and	contributions to
	expressed on Twitter		editing.	methodology, manuscript
				editing.

I have not renumbered sections of submitted or published papers to generate a consistent presentation within the thesis.

The undersigned hereby certify that the above declaration correctly reflects the nature and extent of the student's and co-authors' contributions to this work. In instances where I am not the responsible author I have consulted with the responsible author to agree on the respective contributions of the authors.

Student signature: Date: 28/01/2018 Main Supervisor signature: Date: 28/01/2018

Acknowledgements

I would like to take this opportunity to extend my thanks and gratitude to the many people who have supported me in completing this research. I am deeply thankful for the time, advice, and support of my supervisors Nikki Rickard, Peggy Kern, and Ben Fulcher. I want to thank you all for introducing me to new ways of thinking, pushing me to pursue opportunities outside of my comfort zone, and for the incredibly valuable feedback you have all provided across all aspects of this research. Thank you.

I would also like to thank the staff at TwoBulls for their hours of hard work developing *MoodPrism*. From the Monash Academy for Cross & Interdisciplinary Mathematical Applications I would also like to thank Lito Cruz and Vincent Ssemaganda. This project would not have been possible without your technical assistance in organising our back-end database. Similarly, I am grateful for the statistical guidance of Dr Mehmet Özmen.

I would like to acknowledge my lab mates working on the *MoodPrism* app David Bakker and Abdullah Arjmand for their years of friendship, advice, and collaboration. I also extend my deepest thanks to my partner Ed Hamley, family (including Basil and Daisy), and friends for their endless love and support during some very stressful moments (and the less stressful ones too).

Finally, I would like to thank the Australian Government for supporting this work through an Australian Government Research Training Program (RTP) Scholarship.

xvi

Chapter 1

General Introduction

1.0 Overview and Contribution

This thesis provides a novel exploration of the way emotion language is expressed on social media over time and how patterns of emotion expression are associated with depression. The major contributions of this research to the field of mental health include: the development and application of an integrated experience sampling approach to collecting social media and daily mood data; an improved understanding of the way social media behaviours may signal depression risk; and the application of emotion dynamic features to the language data from social media to predict depression status. In this General Introduction, the research rationale is developed, and the thesis structure is provided.

1.1 Introduction

The use of social media is intertwined with mental health. Social media platforms, like Facebook, Twitter, and Weibo, are used worldwide and are a common component of many people's daily lives. Psychological well-being can be bolstered through the creation and maintenance of friendships, positive online social interactions, and the availability of social support (Best, Manktelow, & Taylor, 2014; Davila et al., 2012a; Ellison, Vitak, Gray, & Lampe, 2014; Verduyn, Ybarra, Résibois, Jonides, & Kross, 2017). However, negative online social interactions and addictive social media use may perpetuate or contribute to psychopathology such as depression (Koc & Gulyagci, 2013; Landoll, La Greca, Lai, Chan, & Herge, 2015; Radovic, Gmelin, Stein, & Miller, 2017; Shensa et al., 2017).

Social media is, in part, defined by the user-generated content individuals produce and consume (boyd & Ellison, 2007; Ellison & boyd, 2013). As such, observable behaviours such as the content produced in status updates, have been shown to be useful in detecting the depression status of social media users (Bazarova, Choi, Whitlock, Cosley, & Sosik, 2017; Guntuku, Yaden, Kern, Ungar, & Eichstaedt, 2017; Moreno et al., 2012). Emotion language, specifically, the frequent use of negative emotion words, has emerged as one of the key features in status updates that can provide insight into the presence of depression (De Choudhury, Gamon, Counts, & Horvitz, 2013; Settanni & Marengo, 2015). However, emotion language expressed on social media has predominantly been utilised as a static (trait) construct for depression prediction; little has been reported on how patterns of emotion language across status updates may indicate changes relevant to the evolution of depression in an individual. As emotion dysregulation is a predisposing factor for depression (Gotlib & Joormann, 2010), uncovering patterns of emotion language and how well these patterns might act as an indicator of emotion experiences is critical to utilising the naturally occurring behavioural traces from social media for the early detection of depression.

1.1.1 Depression and emotion dysregulation.

Major depressive disorder (MDD) and dysthymia impact on mood, with major symptoms including persistent sadness or depressed mood, and loss of interest or pleasure (American Psychiatric Association, 2013; World Health Organization, 2017). Depression is a leading cause of disability worldwide and recent estimates suggest the global prevalence to be 4.4% (World Health Organization, 2017). Further, depression is underreported and underdiagnosed (Cepoiu et al., 2008; Collins, Westra, Dozois, & Burns, 2004). There are multiple factors that determine help-seeking behaviour, and at an individual level this may be hampered by viewing depression symptoms as temporary or a lack of insight into their severity (Collins et al., 2004; Magaard, Seeralan, Schulz, & Brütt, 2017). The challenge of identifying and delivering resources to at-risk individuals who may not be visible to mental health providers prioritises the need for development of early detection strategies for depression that do not require professional assessment. Behavioural markers that indicate early risk factors, without the need for an individual to seek help or self-report their mood, may be useful in this regard.

Emotion dysregulation is a key risk factor for depression that is visible prior to depression onset and persists through remission (Gotlib & Joormann, 2010; Wichers, 2014). Recent work has suggested that depression can be reliably predicted by the way emotion changes over time, or by

dynamic emotion features (Houben, Van Den Noortgate, & Kuppens, 2015; Koval, Pe, Meers, & Kuppens, 2013; Wichers, 2014). Kuppens and Verduyn (2017) suggest that emotion change is governed by four intrinsic processes: (1) that emotions occur in response to internal or external events (principle of contingency), (2) emotions are resistant to change and 'carry over' from moment to moment (principle of inertia), (3) emotions are regulated to reach an optimal fit for a given context (principle of regulation), and (4) emotions interact with each other over time (principle of interaction). When these processes interact in a maladaptive way for extended periods of time, the risk of macro-level psychopathology like depression is increased (Kuppens & Verduyn, 2017; Wichers, 2014).

Cognitive theories of depression argue that biased cognitive processing and emotion dysregulation interact to constitute the vulnerability factors for depression (Beck, 1974; Gotlib & Joormann, 2010). For example, cognitive biases such as rumination – a persistent negative selffocus – contribute to prolonged experiences of negative affect, suggesting that the negative emotions experienced in depression are resistant to change over time (Gotlib & Joormann, 2010; Nolen-Hoeksema, Wisco, & Lyubomirsky, 2008). Depression is also associated with a reduced ability to inhibit negative stimuli (Joormann & Gotlib, 2010) and increased exposure to stressful life events (Connolly, Eberhart, Hammen, & Brennan, 2010). Ineffective emotion regulation in these contexts may contribute to more unstable daily experiences of negative emotion (Thompson et al., 2012).

Over different time periods and in response to different stimuli, depression manifests observable dynamic emotion patterns, which include stability (i.e., persistent or sustained periods of negative affect), and large fluctuations (i.e., frequent and intense experiences of negative affect in response to stressors; Koval et al., 2013; Thompson et al., 2012). The ability to monitor emotion dynamic processes at a large scale may be instrumental in developing approaches for the early detection of depression. However, sampling emotion data is challenging, particularly where active individual input is required. With over 2 billion Facebook users alone, social media is a rich data

source for passively sampling momentary data where expressions of emotion occur frequently (Kramer, 2012; Manago, Taylor, & Greenfield, 2012; Nowak, 2017).

1.1.2 Monitoring depression: Social media for passive data collection.

Language sits at the intersection of cognition and emotion, playing a role in the way emotion is experienced, interpreted, and regulated (Lindquist, Gendron, & Satpute, in press). Depression has been associated with unique linguistic markers that manifest in both verbal and written communication (Pennebaker, Mehl, & Niederhoffer, 2003; Pulverman, Lorenz, & Meston, 2015; Rude, Gortner, & Pennebaker, 2004; Segrin & Flora, 1998). In this way, social media is uniquely positioned as a medium through which rich psychologically relevant language data may be sampled in a non-invasive, ongoing, and temporally sensitive manner (Kern et al., 2016; Schwartz et al., 2013).

Numerous studies have indicated that depression can be reliably predicted by the language used in status updates (Guntuku et al., 2017). Depressive symptoms are disclosed in status updates more frequently by those with more severe depression symptoms (Moreno et al., 2012), and the frequent use of negative emotion words has emerged across many language prediction models as a key language feature associated with depression status (Guntuku et al., 2017). While at a population level, diurnal and seasonal variation in depression severity has been observed in social media language (De Choudhury et al., 2013; Schwartz et al., 2014), the individual path of emotion expression over time as a potential indicator of emotion dysregulation has not been explored.

1.1.2.1 The caveats.

It is also important to note here, some prominent caveats to the use of social media language for depression prediction. First, there are social media specific contexts that may influence the way people express themselves online. For example, status updates are broadcast publicly to an individual's social network (Ellison & boyd, 2013). Audience composition (close or weak ties) may influence the likelihood to disclose emotion and the valence expressed (Lin, Tov, & Qiu, 2014). The norms and expectations of interactions between social media platforms also differ and may

have an impact on language use (Waterloo, Baumgartner, Peter, & Valkenburg, 2017). Second, individual characteristics, like personality, are related to emotion traits (e.g., neuroticism and frequent negative emotional experiences), posting behaviours on social media (e.g. conscientiousness inhibiting impulsive disclosures), and to the use of emotion language (e.g., extraversion is association with the use of positive emotion words; Gosling, Augustine, Vazire, Holtzman, & Gaddis, 2011; Park et al., 2015; Seidman, 2013; Verduyn & Brans, 2012). Together, these confounding factors influence a person's idiolect (their specific way of using language; Dittmar, 1996) on social media which may obscure the detection of depression for some individuals. This has implications for both the patterns of emotion expressed on social media, and the reliability with which social media posts can serve as a proxy for self-reported or experienced emotion.

1.1.3 Research rationale.

The language expressed on social media is likely to reflect some of the cognitive and affective processes involved in depression. While emotion language can differentiate between depressed and non-depressed social media users, little is known about the emotion dynamics expressed through language and if, over status updates, they signal depression risk. As emotion dynamic features often precede depression onset, detecting these processes through language may signal depression risk earlier than looking at the average emotion language used in status updates alone. Further, no clear link has been established between the emotion expressed on social media and the subjective emotional experience of the individual. Understanding how reliably emotion language taps into the emotional experience of social media users is critical for developing a clear understanding of when negative emotion words signal depression symptoms or when other drivers of language use (e.g., personality, gender, age) may be prominent. Taken together, it is important to examine how, at the individual level, the emotion visible on social media unfolds over time to reveal patterns of change that signal depression status or onset.

1.2 Research Questions

This thesis has the broad aim of investigating the link between depression and the emotion expressed on social media. To achieve this aim, it focuses on exploring the positive and negative mood profiles of social media users across their status updates and advances a methodology that integrates social media data collection with a complementary experience sampling method delivered via smartphone. Specifically, this work addresses two major research questions:

(1) What emotion dynamic features in the mood profiles generated across status updates are associated with depression severity? and;

(2) Do the emotions expressed on social media through language accurately reflect subjective daily mood?

Methodologically, the research reported in this thesis involved the development and testing of the social media data capabilities of the experience sampling smartphone application (app), *MoodPrism* (introduced in Chapter 3), and aimed to demonstrate the feasibility of this method through its practical application in this research.

1.3 Structure of the Thesis

This thesis comprises six chapters, including this introduction. Chapter 2 presents a systematic review of the literature considering social networking site use and its association with depression and anxiety. It includes discussion of potential moderators and mediators to these relationships and, additionally, highlights the research that has included well-being variables as a part of its investigation. This chapter provides a broad context for the work presented here and was instrumental in the development of the research questions addressed in the empirical papers presented in this thesis and the rationale for the methodology introduced in Chapter 3.

Chapter 3 provides a detailed description of the research methodology and links the gaps in the literature identified in Chapter 2 to the development of the data collection method used in this research, a smartphone app – *MoodPrism*. *MoodPrism* integrates the capacity to collect social

media data and deliver a self-report experience sampling method for collecting daily mood, mental health, and well-being data. Specific focus is given to the social media data components of this research with further detail on the *MoodPrism* app presented in Appendix A "*Development of a mobile phone app to support self-monitoring of emotional well-being: A mental health digital innovation*" (Rickard, Arjmand, Bakker, & Seabrook, 2016). Brief discussion is also provided on the ethical considerations of conducting social media research and how these considerations were addressed by the approach taken here.

Chapter 4 presents the first empirical paper that has been submitted for publication. This paper introduces the observation of temporal patterns in the expression of emotion across Facebook and Twitter status updates as potential indicators of depression. This chapter also presents cross-platform comparisons of the emotion patterns in language, and supplementary materials explore differences in individual characteristics between platforms that may impact on emotion expression and depression prediction.

Chapter 5 presents the second empirical paper that has been submitted for publication. This paper provides a detailed account of emotion language expressed on Twitter and its similarity and dissimilarities to self-reported daily mood. In this case-series analysis, mood profile features (the average, variability, instability, and probability of acute change) are compared between Twitter and self-report, and this is discussed with reference to the impact this may have on detecting depression status through language.

The final chapter (Chapter 6) consists of an overall integrated discussion, bringing together the findings from Chapters 2, 4, and 5. This chapter provides a brief overview of the findings, then discusses implications, directions for future research, and a conclusion.

Chapter 2

Social Networking Sites, Depression, and Anxiety: A Systematic Review 2.1 Preamble to the Systematic Review

This chapter presents a systematic review titled "Social Networking Sites, Depression, and Anxiety: A Systematic Review". This paper reviews the literature from 2005 to 2016, providing an up-to-date account of the research investigating social media use and depression or anxiety. Recommendations for future research are provided and limitations of the literature are discussed. Importantly, this paper provides the context for the development of the research questions introduced in Chapter 1 and the empirical investigation conducted in Chapters 4 and 5.

The following paper was published in the peer-reviewed journal, the *Journal of Medical Internet Research – Mental Health* in November 2016 and is formatted in accordance with the journal requirements. References are provided in the style of the *American Medical Association* (10th edition).

Review

Social Networking Sites, Depression, and Anxiety: A Systematic Review

Elizabeth M Seabrook1, BPsych (Hons); Margaret L Kern2, PhD; Nikki S Rickard1.2, PhD

¹Monash Institute of Cognitive and Clinical Neurosciences, School of Psychological Sciences, Monash University, Clayton, Australia
²Center for Positive Psychology, Melbourne School of Graduate Education, University of Melbourne, Melbourne, Australia

Corresponding Author:

Elizabeth M Seabrook, BPsych (Hons) Monash Institute of Cognitive and Clinical Neurosciences School of Psychological Sciences Monash University 18 Innovation Walk Monash University Clayton, 3800 Australia Phone: 61 413170668 Fax: 61 399053948 Email: <u>elizabeth.seabrook@monash.edu</u>

Abstract

Background: Social networking sites (SNSs) have become a pervasive part of modern culture, which may also affect mental health.

Objective: The aim of this systematic review was to identify and summarize research examining depression and anxiety in the context of SNSs. It also aimed to identify studies that complement the assessment of mental illness with measures of well-being and examine moderators and mediators that add to the complexity of this environment.

Methods: A multidatabase search was performed. Papers published between January 2005 and June 2016 relevant to mental illness (depression and anxiety only) were extracted and reviewed.

Results: Positive interactions, social support, and social connectedness on SNSs were consistently related to lower levels of depression and anxiety, whereas negative interaction and social comparisons on SNSs were related to higher levels of depression and anxiety. SNS use related to less loneliness and greater self-esteem and life satisfaction. Findings were mixed for frequency of SNS use and number of SNS friends. Different patterns in the way individuals with depression and individuals with social anxiety engage with SNSs are beginning to emerge.

Conclusions: The systematic review revealed many mixed findings between depression, anxiety, and SNS use. Methodology has predominantly focused on self-report cross-sectional approaches; future research will benefit from leveraging real-time SNS data over time. The evidence suggests that SNS use correlates with mental illness and well-being; however, whether this effect is beneficial or detrimental depends at least partly on the quality of social factors in the SNS environment. Understanding these relationships will lead to better utilization of SNSs in their potential to positively influence mental health.

(JMIR Ment Health 2016;3(4):e50) doi:10.2196/mental.5842

KEYWORDS

depression; anxiety; social media; social networking; review, systematic; mental health; well-being

Introduction

Background

Social networking sites (SNSs) are Web-based platforms on which individuals connect with other users to generate and maintain social connections [1]. Considerable disagreement

http://mental.jmir.org/2016/4/e50/

exists as to associations that SNS use may have with depression and anxiety [2,3]. On the one hand, SNSs may protect from mental illness, as they support and enable social interaction and connection [1,4], and allow users to reflect aspects of their identity and express emotion that may be relevant to their lived experience [5]. On the other hand, there are many opportunities

> JMIR Ment Health 2016 | vol. 3 | iss. 4 | e50 | p.1 (page number not for citation purposes)

for miscommunications and mismanaged expectations, and maladaptive tendencies can be exaggerated, leaving individuals feeling a greater sense of isolation [2,6]. As a whole, the SNS environment may be just as complex as face-to-face interactions. As SNS membership continues to rise [7], it is becoming increasingly important to address the possible benefits and detriments the use of SNSs may have on mental health.

Affective disorders such as depression and anxiety have been shown to have bidirectional interactions with the social environment that influence the path of illness onset and maintenance [8]. Depression and anxiety have an approximate prevalence of 4.7% and 7.3%, respectively, in the global population [9,10]. These disorders have high levels of comorbidity [11] and impact the quality of social relationships [12,13]. Depression and anxiety may be implicated in determining the size and structure of an individual's social network [12], the quality of interactions within these networks, and how effectively social capital may be leveraged or developed to provide an individual with social support [8,14].

The social characteristics (both qualitative and structural) affected by depression or anxiety are also relevant to one's sense of well-being. Current mental health theories suggest that the presence of well-being is not the same as the absence of mental illness; a complete model of mental health requires not just the absence of psychopathology, but also a focus on positive indices of functioning such as subjective well-being [15]. This is particularly pertinent when exploring how the social environment may affect an individual, as such environments may simultaneously confer a number of benefits to the individual and exaggerate deficits [16-18].

Social aspects of the Internet have been argued to augment social relationships and support mental health. SNSs in particular connect us to friends, family, colleagues, strangers, and celebrities and can help users to maintain and make new friendships, express thoughts and feelings, and express identity [1,4,19]. The primary social functions that SNSs perform may augment the benefits of engaging in face-to-face interaction by extending the reach and accessibility of our social networks [20]. Indeed, SNS use is associated with lower levels of loneliness and greater feelings of belonging (social connectedness), social capital, and actual and perceived access to social support and is generally associated with higher levels of life satisfaction and self-esteem [6,21-26].

As a whole, the positive social components of SNS use suggest a protective role against depression and anxiety. For instance, higher levels of self-esteem and life satisfaction may aid in attenuating depressive symptoms [27]. Kraut et al [28] found that frequent general Internet use did not increase depression over time, and, in a second study, communication activities on the Internet were shown to be associated with lower levels of depressive symptoms [29]. Computer-mediated communication (CMC; eg, email, instant messaging) allows users to express and interpret emotion in a similar way to face-to-face interaction [17]. CMC may therefore be beneficial for emotion regulation as has been demonstrated for offline forms of written emotional expression [30,31]. However, for individuals with depression or anxiety, the interpretation and frequent exposure to this emotion may have a negative impact [13]. SNS use may increase an individual's exposure to negative social interactions (eg, cyberbullying), which may negatively impact mood and mental health [2]. For example, negative interaction quality was associated with decreases in self-esteem and life satisfaction [32]. Even passive exposure to the language used in SNS posts has been shown to influence the emotive language subsequently expressed by the receiving SNS user, where positive or negative emotions are argued to transfer via contagion [33-35]. As SNSs explicitly support a number of social features, the relationships and interactions between the user, their emotional experience, and Web-based technology are likely to be complex and may even accentuate differences between those who are doing well in life and those who are struggling.

Cognitive and social factors frequently emerge as both moderators and mediators of the relationships between offline social interactions or events and depression [36-38] and might also occur in Web-based environments. For instance, self-esteem mediates the pathway between relationship interactions and depressive symptoms [39], but it might also moderate how a person uses and is affected by the SNS. Rumination, a response style where an individual maintains a passive and repetitive focus on their distress [40], is one mechanism linking stressful life events and the development or maintenance of depression [41], and the SNS environment provides opportunity for a person to both internally ruminate on bad events and have an entire social network further accentuate shortcomings. Social support has additionally been shown to moderate relationships between stress and depression, with greater levels of social support acting as a buffer to depressive symptoms [42]. This is pertinent to SNSs as they present a potential intervention opportunity for developing and strengthening supportive social networks for vulnerable individuals.

Objective

Since the advent of SNSs, a number of articles have been published examining the relationship between SNS use and depression and anxiety. The interaction between SNSs and our mental health and well-being is clearly varied and complex. The objective of this paper was to provide a systematic review of literature examining SNSs and their relationship with depression and anxiety. It also considers links with well-being, as well as potential mediators and moderators to these relationships.

Methods

Search Strategy

Figure 1 summarizes the search strategy and article selection. A multidatabase search identified studies conducted between January 2005 and June 2016. The databases included were PsycINFO, MEDLINE (Ovid), Scopus, IEEE Xplore, CINAHL (Cumulative Index to Nursing and Allied Health Literature), Education Resources Information Center, Social Sciences Citation Index, and Communication and Mass Media Complete. The inclusion of conference papers accessed through IEEE Xplore was intended to capture the research within the computer

sciences and engineering fields that may have been relevant to the psychological literature.

Search terms were selected in order to comprehensively capture the various ways mental health, mental illness, subjective well-being, and SNSs have been defined and explored in the existing literature.

SNSs were defined as conceptualized by Ellison and Boyd [1] as sites that are a Web-based communication platform with 3 distinct characteristics: (1) user profiles are unique and created through user-provided content and content provided by other users, (2) the network connections between individuals are visible and can be navigated through by other users, and (3) individuals can broadcast content and consume and interact with content contributed by others in a continuous stream of information. Prototypical examples of SNSs include Facebook, Twitter, Myspace, and Instagram.

For mental health, search terms specifically focused on depression and anxiety, as well as overall well-being (eg, subjective well-being, psychological well-being, wellness; see Figure 1 for full list of search terms).

Figure 1. Overview of search strategy and selection process for the systematic review.



Inclusion and Exclusion Criteria

Studies were included if they had a primary focus on SNS use as a behavior. As such, studies that referred to SNSs as a recruitment method only or used SNSs as a means for intervention delivery were excluded.

Articles were included if they provided results addressing anxiety or depression directly and were excluded if they were only referred to in the context of general psychological distress (or similar). As the primary focus of the review was on depression and anxiety, not the broader well-being construct, articles addressing well-being were only included if they also included specific reference to anxiety or depression. The search was limited to articles published after 2005 to capture research on the prototypical examples of SNSs that include the basic features of modern networks. Studies that had a primary focus on the Internet, chat rooms, or online support forums were also excluded; although they may contain some of the features of SNSs, differences in the function they perform for users may exist [19].

Additionally, articles were restricted to English language, peer-reviewed journal or conference proceedings, and quantitative or mixed methodologies. Gray literature, commentary and editorial, qualitative research, literature reviews, and descriptive case studies were excluded.

Data Extraction and Data Synthesis

Two raters (the first author and a trained research assistant) reviewed all abstracts returned from the literature search and selected abstracts for full-text reading based on the inclusion and exclusion criteria. All articles that included measurement of depression, anxiety, or well-being were retained. The selected full-text articles were downloaded and reviewed by the first and third authors.

To provide some preliminary evaluation of the strength of the research, three risk of bias indicators were adapted from the Cochrane bias tool (Cochrane Handbook for Systematic Reviews of Interventions [43]), which classifies methodology that may limit replicability or generalizability. Studies were rated to indicate whether the study (1) included psychometrically reliable and valid measures, (2) used an external measurement criterion for mental health, and (3) provided description of the sample demographics including some SNS activity statistics (eg, number of friends and/or use frequency). These were rated by the first and third authors from "0=No bias," "1=Unclear risk of bias," and "2=High risk of bias" and were summed to create a final score between 0 and 6. A linear weighted kappa statistic for interrater reliability (.78, SE=.06) indicated that there was very good agreement in applying the bias criteria. Consensus was reached on all ratings. Articles with a rating of 3 or above were excluded [44-52], resulting in the final set of 70 studies, as presented in Multimedia Appendix 1.

From each article, the year of study, population of interest, type of SNS, and variables used (anxiety, depression, well-being) were noted, along with whether or not any formal mediators or moderators of these relationships were indicated. Information was then qualitatively synthesized to identify common themes.

Results

Description of Studies

Figure 2 indicates the number of articles addressing SNSs, depression and anxiety, and well-being from 2005 through 2016, based on the 302 full-text articles initially reviewed. There were considerably more articles addressing well-being alone than articles only addressing depression and anxiety. Only 15 articles included both positive and negative aspects of mental health. This review includes the 70 articles that include depression or anxiety only or depression or anxiety and well-being.

A total of 22 studies addressed potential moderators or mediators in SNSs' relationship with depression or anxiety (see Multimedia Appendix 1). Most articles obtained a bias rating of 0 to 1. Ratings of 1 or above were primarily due to the limited focus on reporting SNS activity statistics, such as the number of friends or average frequency of use, which help characterize the average SNS user in each sample. Facebook was the most commonly explored SNS followed by the measurement of SNS use as a general category (ie, no specific platform explored). The majority of studies examined young adults (late teens or early 20s).

Figure 2. Publication frequency of research into well-being, depression or anxiety only, and depression or anxiety with well-being from 2005 to June 2016, based on the initial 302 full text articles reviewed, which included quantitative findings. Case studies, editorials, literature reviews, and gray literature were excluded.



Year of Publication

http://mental.jmir.org/2016/4/e50/

JMIR Ment Health 2016 | vol. 3 | iss. 4 | e50 | p.4 (page number not for citation purposes)

Depression, Anxiety, and Social Networking Sites: Summary of Findings

Across the 70 articles, several general themes were apparent: frequency of use, size and structure of the SNS, language features and observable SNS activities, self-disclosure and expression, quality of interactions, social support, social connectivity, social comparison, addictive and problematic behaviors, and physiological associations. Findings are summarized in Multimedia Appendix 2 and are described below, with particular attention to moderators and possible mechanisms involved in the associations. As some articles were relevant to multiple themes, these articles appear in multiple sections. Studies that included well-being are also highlighted.

Frequency of Social Networking Site Use

Overall, total frequency or time spent on SNSs had mixed associations with depression and anxiety. Of the 30 studies examining these variables (see Multimedia Appendix 2) [53-81], 8 studies found a direct positive association with depression and 16 found a nonsignificant association. For anxiety (and social anxiety), 3 studies found direct positive associations and 7 found nonsignificant associations. With the exception of 1 study showing a significant negative association between Facebook-specific social anxiety and the frequency of SNS use [80], no studies supported an association between the frequent use of SNSs and a lower level of anxiety or depressive symptoms.

Several moderators appeared. In one study, the number of strangers followed moderated frequent Instagram use and greater depressive symptoms, where a significant relationship only occurred for those with high proportions of strangers in their social networks [68]. Similarly, time spent on Facebook was only a predictor of depression and anxiety for those individuals who have higher motives to use the site for social connection [73].

Table 1. Broad functions of social networking site use and example behaviors.

	Passive use	Active social use	
	(alpha=.7788) ^a [75,78]	(alpha=.8386) ^a [69,76]	
		Content production (public)	Interactive communication
		(alpha=.52) ^a [75]	(alpha=.80) ^a [75]
Example	Checking or reading friends' profiles or posts	Status updates	Chatting in messages (private)
behaviors	Browsing the newsfeed	Updating profile pictures	Posting on friends' walls (public)
		Image management (maintaining profile information)	Posting comments on statuses (public)

^aCronbach alphas indicating the internal consistency of measures defining functions of social networking site use as defined in the reviewed literature.

In general, passive uses of SNSs was not directly related to depression and anxiety, but there may be differential behavioral patterns for individuals high in depression or social anxiety [75,78]. Higher levels of social anxiety were significantly related to passive uses of Facebook but not to content production uses of Facebook [75]. Brooding, or anxious rumination, emerged as a mediator of the relationship between passive Facebook use and social anxiety and may be a cognitive risk factor for increasing social anxiety symptoms where passive Facebook use is frequent. Tandoc et al [78] found that Facebook envy

mediated frequent passive Facebook use and depression, where lower levels of Facebook envy resulted in a direct effect of passive Facebook use reducing depressive symptoms and higher levels of envy led to greater depressive symptoms.

Active uses of SNSs demonstrate a more complex relationship. Shaw et al [75] found that depressive symptoms positively correlated with more frequent content production and interactive communications. McCord et al [69] showed that the frequency of social Facebook use did not predict social anxiety in the entire

Associations may be affected by the study design. Studies utilizing an experience sampling method (ESM) to collect SNS use frequency over 1 to 2 weeks found no significant associations between SNS use frequency and depressive symptoms over time [61,63,77]. Indeed, across 2 studies, while Steers et al [77] found a positive association between the time spent on Facebook and depression when using a retrospective survey, this effect was nonsignificant when participants completed daily ESM diaries. In addition, 2 studies [54,56] conducted a 3-week follow-up and demonstrated no change in depressive or anxiety symptoms over time as a function of SNS use frequency.

Tendencies toward depressive rumination and corumination did not moderate associations, suggesting that the frequency of SNS use may not be a significant risk factor for depression even across varying cognitive styles [54]. Kross et al [63] additionally included depression as a moderator of the relationship between the frequency of daily SNS use and affective well-being (ratings of negative affect) and cognitive well-being (life satisfaction). Although more frequent SNS use was associated with more negative affect and lower life satisfaction across a 2-week period, depression did not moderate these associations.

A number of studies have made a more nuanced consideration of SNS use frequency by looking at the different functions of use of SNSs [54,56,69,74-76,78]. Table 1 presents how these broad functions have been defined in the literature and presents some example behaviors. It also provides the Cronbach alphas that have been reported for the measures of each function. The table shows a distinction between passive and active use (broad-level functions). Active use may further be divided into content production and interactive communication functions. The table also shows where behaviors may be enacted in public (entire SNS friend network audience) or in private (dyads or small selected audience).

sample but was positively correlated with anxiety for a high anxiety group only.

Simoncic et al [76] suggested that personality and gender moderate the association of frequent active uses of Facebook (content production and interactive communication) and depression and may be protective. The study found a three-way interaction between gender, Facebook active uses, and neuroticism, such that lower depressive symptoms occurred in females who were high in neuroticism and actively used Facebook.

Size and Structure of Social Network on Social Networking Sites

The size of the SNS friendship network and its association with depression and anxiety has similarly yielded mixed findings. Fernandez et al [57] and Weidmann and Levinson [82] found significant negative relationships between social anxiety and the number of friends, and Park et al [83], Park et al [84], Rae and Lonborg [73], and Rosen et al [74] found this same relationship direction when examining depression. Rae and Lonborg [73] found that a greater number of friends on Facebook was associated with higher general positive affect and life satisfaction, when use of the site was motivated by maintaining friendships. The remaining studies demonstrated no significant relationship between the number of SNS friends, depression, or anxiety [53,57,64,67,71,73,78,79,85,86].

Specific friend categories have also been examined. Tsai et al [87] found that users accepting the friend request of an ex-partner tend to have higher levels of trait anxiety and depression severity than those who reject the request. Mota-Pereira [88] demonstrated that for individuals with treatment-resistant major depressive disorder (MDD) also currently taking antidepressants, the use of Facebook over a 3-month period significantly reduced depressive symptoms, compared with a no-Facebook control, and the addition of a "psychiatrist as a friend" showed significantly faster improvement in depressive symptoms. Such findings suggest a broad beneficial impact of SNS use when treatment is augmented by friends from a user's network.

The structure of the network itself may make a difference. For instance, Homan et al [89] revealed significant differences in the network structures of individuals with depression and those without on an LGBTQ (lesbian, gay, bisexual, transgender, and queer) support SNS, TrevorSpace. Individuals without depression had significantly more integrated friendship networks on the SNS compared with depressed individuals, with their friends being more likely to know each other and also having a higher proportion of friends who do not know each other. For the depressed group this could indicate they have less diverse social networks. Peer-selected groups have the potential to offer social support to depressed individuals, whereas groups over which the user had less control may contribute further exposure to psychological distress [90].

Language Features and Observable Social Networking Site Activity

A number of articles have examined the language features in SNS posts, with the potential for identifying individuals with

http://mental.jmir.org/2016/4/e50/

depression. SNS users with depression differ from users without depression in that they express negative affect more frequently, use more personal pronouns, and generally have lower frequencies of interaction with others in their SNS network [91,92]. Park et al [93] have shown that individuals with a diagnosis of MDD more frequently post negative sentiment than those who are not depressed, and Moreno and colleagues [85,94] demonstrated that depression could be identified in the language used in the Facebook posts of college students based on the *Diagnostic and Statistical Manual of Mental Disorders* (Fourth Edition) criteria for MDD.

Settani and Marengo [95] directly examined the expressed emotion in participant status updates and generated an automated word count from the emotion dictionaries of the Italian version of *Linguistic Inquiry and Word Count (LIWC*), which was also supplemented with emoticons. Providing face validity, the frequency of word use from the negative emotion and sadness *LIWC* subscales positively correlated with depression, while the anger subscale positively correlated with anxiety. Positive emotion was unrelated to depression or anxiety scores. Interestingly, only the relationship between the sadness subscale and anxiety remained statistically significant when examining individuals older than 25 years.

In addition to language features, the time of posting, relative volume of posts, and reciprocity (likes and comments, tweets and retweets) may also aid in describing individuals with and without depression, with depression correlating with more night activity and less volume and reciprocity than nondepressed peers [84,91,96]. Over multiple weeks, there may also be subtle variation across time [96]. Park et al [84] provided evidence indicating that, for individuals experiencing acute depression (or a relative increase in their symptom severity), there is an increase in their posting frequency over a 6-month period. This is consistent with Shaw and colleagues' [75] findings indicating those with higher depressive symptoms engage in content production features on Facebook frequently.

The number of identity items on SNS users' profile page have also been associated with both depression and social anxiety scores [57,82,97]. For example, listing a "Single" relationship status relates to higher levels of social anxiety [82]. This related to the quantity of information provided in specific areas of a user's profile information (eg, TV, Books, Quotes, Music; [57]). Although some of the specific findings are mixed [57,82,98], studies generally suggest that social anxiety may be visible on SNSs through compensatory behaviors (increases in information disclosure) or through relative inactivity or social withdrawal [57,82].

Social Networking Sites for Self-Disclosure and Expression

At a broad level, it has been suggested that users of Facebook have lower levels of social anxiety than nonusers, suggesting that there might be a selection effect, such that SNS activities are unattractive to individuals high in social anxiety [99]. However, this depends on the social media platform. Baker and Moore [100] showed that, for new Myspace users, those who intended to use the site for blogging had higher mean depression and anxiety ratings than those who did not intend to blog. These

> JMIR Ment Health 2016 | vol. 3 | iss. 4 | e50 | p.6 (page number not for citation purposes)

individuals were also more likely than nonbloggers to feel dissatisfaction with their social networks and had a greater likelihood to use self-blame and venting coping strategies. Average levels of depression and anxiety among the bloggers were maintained across a 2-month period, although there was a trend in some symptoms being reduced and a significant increase in feelings of social integration and satisfaction with online and offline friendships [101]. Similarly, große Deters and Mehl [102] found that depressive symptoms remained stable through an intervention, although loneliness decreased via feelings of social connectedness.

Social anxiety is associated with an increased preference for SNS-mediated communication [103] and relates to differences in the depth of self-disclosure via public (status updates) or private (eg, messages) communication on SNSs. For individuals with higher levels of social anxiety, greater importance is placed on the need for reduced social cues and increased controllability of communication [59,104]. This leads to greater disinhibition and Facebook self-disclosure for private SNS communication only and not for public SNS communication [59]. Green et al [59] suggest that this may be related to the trust, audience size, and privacy differences between private and public communication on SNSs, which may position private SNS communication as more attractive and accessible for individuals high in social anxiety. Similarly, Baker and Jeske [80] suggested that assertiveness on Facebook (the ease with which an individual offers opinion or interacts with others) is lower for individuals high in social anxiety compared with those low in social anxiety.

A potential explanation for the self-disclosure activities of individuals with high social anxiety on SNSs may be related to motivations or perceived pressure to present an idealized self-image or to avoid presenting a negative image on SNSs [86,105,106]. Motivations to avoid presenting a negative self-image have been found to be a greater concern for individuals who had experienced high social anxiety the previous day and does not vary according to levels of perceived social competence [105]. Similarly, frequent impression management (including updating profile information) on SNSs is positively related with depression [74].

Frequently expressing positive or negative affect (emotional valence) in SNS status updates has also been shown to relate to depression and may be mediated by rumination [67]. In contrast, positive and negative expression appears to be unrelated to social anxiety [98]. Positive and negative self-disclosures may, instead, impact the quantity of social reciprocity an individual with social anxiety receives [98]. For example, when individuals higher in social anxiety post positive status updates, this generates more pronounced increase in social feedback (likes) than when positive posts are made by those low in social anxiety or when posts have low positive content [98].

Quality of Interactions

Considerable evidence suggests a link between the quality of interactions on SNSs and mental health. Studies have operationalized SNS interaction quality as either the perceived (when self-rated) or observed (when coded by experimenters) valence of interactions between friends and the user on SNSs.

http://mental.jmir.org/2016/4/e50/

Items often refer to a global estimate of "How positive [or negative] are your interactions with people on Facebook" [54] or, where coded, the frequency of positive or negative sentiment expressed in comments on posts [103]. This differs from the frequency of social or interactive communication on SNSs, discussed above, which refers to the estimated frequency or total time spent engaging in these activities.

Depression is generally associated with fewer positive interactions and more negative interactions on SNSs [54,56,103,107,108]. Social and global anxiety similarly relate to the perception of negative quality interactions on SNSs [56,107]. Depressed individuals may use SNSs in a more problematic manner than do anxious individuals [56], thus creating negative interactions. For instance, symptoms recorded at the age of 13 years significantly predicted a reduced likelihood of receiving comments that contained deviancy talk from SNS peers at the age of 20 years; however, symptoms at the age of 20 years predicted a greater instance of verbally abusive comments from peers [103]. The findings of Frison et al [81] also suggest that depressive symptoms are a risk factor for peer victimization on Facebook. Moberg and Anestis [108] have additionally shown that, when controlling for the influence of depressive symptoms on perceived negative interactions on SNSs, greater ratings of negative interactions predict feelings of thwarted belongingness (disconnection), a potential risk factor for suicidal desire.

Depressive rumination and corumination may moderate associations between the perception of SNS interaction quality and depression. In 2 studies, Davila et al [54] showed that those with higher levels of depressive rumination exhibited a stronger relationship between the frequency of perceived negative interactions on SNSs and greater depressive symptoms. Although corumination (ie, "excessive discussion of problems within friendships"; [54] p73) did not emerge as a significant moderator, it did yield a number of relationships with other variables, notably, feeling down or depressed after interactions on SNSs and a greater frequency of SNS use. The quality of use also relates to intentions for continued SNS use. Belief that online communities are dangerous, including concerns about privacy and the potential to encounter hostile or negative interactions, has been shown to be a potential antecedent of online and general social anxiety and their link to reduced continuance intention of using Facebook for social communication [109].

Associations may depend in part on the methodologies used. When researchers have directly observed and coded the language of comments made to an SNS user by their friends, it has been shown that a greater level of social anxiety at age 20 years was a significant predictor of more positive supportive comments from SNS friends and fewer negative peer interactions [103]. This is in contrast with the research utilizing self-report survey methods that show more frequent reporting of negative interactions for those with high levels of depression and anxiety symptoms [54,56,107]. This discrepancy suggests there may be a role for perceptual bias in a participant's interpretation of the quality of interactions to which they are exposed on SNSs. In this light, individuals with higher levels of depression and anxiety may be more inclined to interpret or perceive SNS

> JMIR Ment Health 2016 | vol. 3 | iss. 4 | e50 | p.7 (page number not for citation purposes)

interaction as more negative regardless of the communication content exchanged between users. The potential for such a perceptual bias in interpreting SNS interactions has also been suggested in reference to social support perceptions and is further discussed below (see Park et al [93]).

Social Support

Social support plays a mixed and varied role within the SNS environment. Studies suggest that individuals with higher depressive symptoms perceive their SNS friend networks as providing them with less social support than they actually receive [93] and that SNS social support seeking may exacerbate depressed mood for some individuals [110]. Perception of support appears to be more important than actual support. Across 2 studies, Park et al [93] showed that in the general population greater depressive symptoms were associated with more actual social support on status updates that contained negative emotion. In contrast, perceived support was negatively associated with depression, and higher depressive symptoms were associated with a greater discrepancy between actual and perceived social support. Frison and Eggermont [110] similarly found that depressed mood increased in adolescents when social support was sought on Facebook but perceived to not occur. Other research has also demonstrated the protective role of perceived social support in ameliorating the impact of SNS peer victimization on depression [81].

For anxiety, social support provided on SNSs may play a protective role. Indian and Grieve [111] found that perceptions of Facebook social support were only predictive of subjective well-being for individuals with high levels of social anxiety and not for those reporting low levels of social anxiety. Furthermore, in the high social anxiety group, perceived Facebook social support was the only significant predictor of subjective well-being, suggesting that SNS social support may provide unique benefits to individuals with high levels of social anxiety.

The nature of seeking social support on SNSs may differ from traditional face-to-face approaches [110,112]. Some evidence suggests that emotional support provided by Facebook can increase depressive symptoms and decrease quality of life [112]. It may depend in part on the characteristics of the user. For example, SNS users' perceived communication competence—an overall evaluation of communication skills and behaviors—plays a role in determining the level of satisfaction they feel is generated from their SNS social support. Wright et al [79] demonstrated that better perceived communication competence predicted higher ratings of both face-to-face social support and Facebook social support satisfaction, which in turn were significantly negatively related to depression.

Social Connectedness

Facebook social connectedness encompasses subjective feelings of belonging and closeness to an individual's social network [113]. Grieve et al [113] demonstrated that higher levels of Facebook social connectedness were related to lower levels of depression and anxiety and higher levels of subjective well-being (life satisfaction). Feelings of social connectedness may mediate the impact an increase in posting behavior has on decreasing loneliness [102].

Social Comparison

Social comparison on SNSs, where individuals compare themselves as having more positive (downward comparison) or negative (upward comparison) qualities than others, is a significant risk factor for depression and anxiety [68,77,114,115]. Several studies found that Facebook envy, a hostile evaluation of others from their social information on SNSs, is associated with higher ratings of depressive symptoms [78,116]. Lee [114] found that depression and anxiety were positively related to the frequency of social comparison on Facebook. Feinstein et al [115] extended these findings by revealing rumination as a mediator in the relationship between negative (upward) social comparison on Facebook and depressive symptoms. This relationship changed over time; at a 3-week follow up, more frequent negative social comparison on Facebook was associated with increases in rumination and a subsequent increase of depressive symptoms.

Appel et al [116] examined how depression may influence an SNS user's interpretation of the profile information of other users. Individuals with depression were more likely to rate themselves as being unhappier (or inferior) in comparison with profiles of any type (attractive or unattractive) than those without depression. Individuals with depression also experienced greater envy than those without depression in response to viewing the unattractive profile, with this difference being greater after viewing the attractive profile.

Social comparison of any direction (upward, nondirectional, or downward) may also indirectly mediate the association between the time spent on Facebook and depression. Across 2 studies, as individuals spend more time on Facebook they engage in more frequent negative (upward) and nondirectional social comparison and less positive (downward) social comparison, which in turn relates to more depressive symptoms [77].

Envy potentially plays a destructive role in passive Facebook use (eg, viewing or browsing profiles; see Table 1). Where Facebook envy is high, greater frequency of passive Facebook use is associated with greater depressive symptoms, and where Facebook envy is low (or not present), passive Facebook use is associated with reduced depressive symptoms [78]. Indeed, research into Instagram (a photo-sharing SNS) [68] has shown that more positive (downward) social comparisons are associated with decreased depressive symptoms. Social network composition, additionally, may moderate the relationship between frequent Instagram use and increases in depressive symptoms via social comparison [68].

Addictive or Problematic Social Networking Site Use

"SNS addiction" and "problematic SNS use" are linked with depression and anxiety [58,60,62,65,104,106,117-121], although associations most likely are bidirectional in nature. It has been suggested that such maladaptive SNS use is only present for a small subset of users [62,106], although one study suggested that 41.9% of adolescents had a Facebook addiction [119]. While depression and social anxiety explain much of the variance in problematic SNS use or SNS addiction, other variables (younger age, male, and more frequent SNS or general Internet use) have also emerged as significant predictors [58,62,118]. Through

cluster analysis, Moreau et al [120] showed that problematic Facebook use is most prevalent in individuals high in borderline personality traits and depressive and social anxiety symptoms compared with groups low in those symptoms or high in sensation seeking (but low in psychopathology). Their findings may indicate considerable comorbidity between psychopathological symptoms and SNS addiction.

Wegmann et al [121] suggested that depressive symptoms and social anxiety have both a significant direct relationship with SNS-specific addiction and a partially mediated pathway to SNS-specific addiction via 2 cognitive styles: self-regulation and Internet use expectancies. In these pathways, higher levels of depression and anxiety are related to lower levels of self-regulation, which are in turn related to higher SNS-specific addiction scores. Internet use expectancies, the perception that the Internet can aid in increasing pleasure and decreasing negativity, were greater for those with higher depression or anxiety symptoms, which again lead to greater vulnerability for SNS-specific addiction. They suggest that depression and social anxiety may predispose SNS users to these cognitive styles.

In contrast, Andreassen et al [117] found that while social anxiety was positively related to addictive SNS use, depression was negatively related to addictive SNS use. This was interpreted as reflecting social withdrawal characteristics of depression and CMC's social compensation for individuals with social anxiety [117]. Indeed, addiction and the compensatory uses of SNSs have been demonstrated to be related to higher levels of social anxiety [106]. Some evidence suggests that the addictive use of SNSs arises from the need to compensate for the social functions affected by social anxiety symptoms. Casale and Fioravanti [104], for example, show that addressing unmet face-to-face social needs, such as the need to belong, to be perceived as socially competent, and to be assertive in communication, may drive problematic SNS use. However, associations may depend on gender. For males and females, a direct association between social anxiety and problematic SNS use has been demonstrated; however, a significant mediator (motivations for competent self-presentation) in this relationship only emerged for males [104]. Lee-Won et al [65] suggested that when the need for social reassurance (ie, motivations to seek social interactions and feelings of belonging) is high or moderate, the relationship between social anxiety and problematic SNS use is strengthened. Thus, social anxiety may only be a risk factor for problematic use of SNSs where the need for social connection is also high.

Physiology and Facebook

Finally, one study examined the impact of Facebook or face-to-face exposure as a primer for physiological arousal [122]. Arousal was greater for individuals when observing someone face-to-face after browsing their Facebook profile than for individuals exposed to a face-to-face encounter followed by the Facebook condition. Social anxiety was a significant moderator, with a more pronounced increase in arousal for those high in social anxiety, particularly in the Facebook than face-to-face exposure. The authors suggested that for the high social anxiety group, the initial exposure to Facebook may prime social comparison and self-presentation concerns for the

http://mental.jmir.org/2016/4/e50/

subsequent face-to-face meeting. However, as emotional valence was not measured, it is unclear if the arousal experienced by participants was perceived as a positive or negative event.

Discussion

Principal Findings

This systematic review examined associations between SNS use and anxiety and depression. Across 70 studies reviewed, a number of positive and negative correlates have been suggested, as well as moderators and mechanisms of these associations. On the basis of this review, it is likely that there are differing engagement and interactional styles on SNSs for users high in social anxiety and depression. These may be driven or defined by both symptoms and motives to compensate for needs that are not met face-to-face. Negative interactions, frequent social comparison, and SNS addiction or problematic use are related to higher levels of depression and anxiety. Furthermore, cognitive response styles such as rumination or brooding may exacerbate the negative interactions between SNS use, depression, or anxiety for some individuals.

While these potential risks exist for mental health, it is also clear that SNSs can provide considerable benefits to their users. Positive quality interactions, social support, and social connectedness most consistently related to lower levels of depression and anxiety. Social support and connectedness derived from SNS use may be uniquely beneficial to individuals with social anxiety who are unable to access these resources face-to-face. However, especially for those with depression, some evidence suggests that there is a discrepancy between the perceptions of interaction quality and social support and the actual content of their SNS communications, which may attenuate the potential positive impacts of SNS use.

Across a number of studies, observable SNS features such as language use and expressions of identity on user profiles have been demonstrated to provide insight into the depression and anxiety status of the SNS user. With continuing research these characteristics may be a useful tool for monitoring mental health. The content and quality of interactions on SNSs may provide the clearest candidates for monitoring depression and anxiety and may be potential intervention targets for improving mental health and well-being through engaging with SNSs.

Social Aspects of Social Networking Sites

Across studies, social aspects, including feelings of social support, social connectedness, and positive interaction quality, emerged as protective factors for SNS users. The SNS network structure itself may play an important role in supporting mental health, in that some platforms may better provide social resources to individuals with depression. Indeed, more integrated social networks on SNSs were associated with lower levels of depression [89]. Studies suggest that social support and social connectedness derived from SNSs are constructs distinct from general social support or connectedness [111,113]. SNSs may therefore be contributing additional benefit to their users by creating another domain in which individuals can access, or have greater perceived access to, social support, especially with individuals for whom face-to-face interaction is difficult

[123-125]. The broad and visibly articulated social context on SNSs may contribute to the feeling of social connectedness derived from SNSs and its association with better mental health outcomes [126]. As such, SNSs may provide an environment where those already high in social skills and resources are benefiting from their cumulative sources of social support ("rich-get-richer"; see [28]) as well as augmenting social support access for those who have difficulties engaging face-to-face [111,123-125].

Consistent with offline research, the perception of social support appears to be more important than actual support [126-128]. Findings demonstrated that perceived social support was greater in those with lower depression scores and that perceived communication competence may contribute to this relationship [79,93]. Greater perceived positive interaction quality and greater reciprocity in interactions are also indicative of lower depression and anxiety. Similarly, Valkenburg et al [32] demonstrated higher levels of life satisfaction and self-esteem for those who frequently reported positive peer experiences on SNSs. However, aspects of the individual that drive depressive feelings and social anxiety, greater use of negative language, and cognitive aspects such as social comparison and rumination, can prevent the user from perceiving support that is actually there [93], further contributing to depressive or anxious symptoms.

Emotional Aspects of Social Networking Sites

The valence of posts on SNSs may both reflect and impact depression and anxiety. Individuals scoring higher on depression scales in the reviewed studies generally expressed more negative affect on SNSs and were more likely to perceive negative interactions. The way individuals interpret emotional and social content on SNSs may place depression as antecedent to maladaptive SNS use, which may, in turn, maintain depressive symptoms. For individuals who are already depressed, ambiguous interactions are often interpreted as negative [13,129], which may attenuate the potential benefits available through SNS use.

Evidence suggests that frequent positive expressions are associated with better mental health, and frequent negative expressions are associated with depression and poorer life satisfaction [67,91,96]. While therapeutic writing can provide some benefits in reducing distress and improving well-being [30,31], online writing may serve a different function, with Web-based expressions reflecting the lived experience of the individual (eg, [91,130-132]), rather than providing a therapeutic outlet. Indeed, relative increases in posting frequency were shown to be associated with greater depressive symptoms [84]. For others, the presence of social anxiety may hinder the use of posting functions for emotional disclosure on SNSs [59], which may decrease access to potential social interaction [98]. As emotional content can be effectively communicated on the Web [133], SNSs represent another space in which positive and negative interactions can be enacted and may provide key behavioral insights into the mental health and well-being of a SNS user. Alternatively, increases in self-expression on SNSs may be more beneficial to well-being domains (such as connectedness, social support, and life satisfaction) but may not

have an impact on depression or anxiety. A direct comparison of these relationships has not been conducted, and might be an area to investigate in the future.

Cognitive Aspects as Mechanisms and Moderators

The prominent risk factors for depression and anxiety that emerged from this review included frequent SNS social comparison, negative perceived interaction quality, addictive or problematic SNS use, and rumination (or brooding). These factors represent cognitive and interactional styles that have well-established associations with depression and anxiety but may be enhanced by the enduring nature of social content on SNSs. Although the total frequency of SNS use does not appear to be directly related to either depression or anxiety, there are different moderating and mediating factors [68,73,77,78] and patterns in the functions of SNS use by individuals with higher depression or anxiety that may contribute to or exacerbate symptoms [69,74-76,78].

One of the risk factors for depression and an individual's interaction with SNSs was rumination. Greater rumination is frequently associated with higher ratings of depression and also impacts well-being by maintaining a focus on negative affect [134,135]. Rumination is a likely mechanism for the relationship between negative interactions with SNSs and depression based on its role in SNS negative emotional expression [67] and social comparison [115]. There is considerable potential for SNSs to amplify and assist ruminative processes by exposing SNS users to a constant stream of rich social information that can be selectively reflected on as permanent content on a user's profile [54,115].

Similar to depression, the cognitive risk factors for social anxiety include social comparison (via brooding) and the perception of frequent negative interactions. However, the pathway to and importance of these risk factors may differ from depression. In contrast to those with depression, those high in social anxiety mainly use SNSs for passive browsing and private communication, not for content production [75]. The passive uses of SNSs may place individuals at greater risk of more frequent social comparison, which may have negative mental health effects [114]. This differs from the relative benefit of content production on SNSs for an individual with social anxiety, as posts are often rated as being more appreciated by friends in the network [98], which may have a flow-on effect to the perception of SNS-derived social support [111] and may even reflect more positive interactions with peers [103].

The reduced social cues on SNSs may be attractive to individuals with social anxiety, as has previously been suggested in the general Internet literature [124]. However, the need to compensate for a lack of belonging and social reassurance in face-to-face interactions, in conjunction with lower self-regulation, may drive problematic SNS use for individuals with social anxiety [65,104,106,117]. Similarly, these motives may also contribute to individuals with social anxiety generating more content on their profile pages than others [57], and for those highest in social anxiety it may contribute to a higher frequency of SNS use [69]. On the whole, there appear to be a number of well-being benefits to using SNSs for individuals high in social anxiety that cannot be gained in face-to-face

interactions; however, the pattern of SNS use may negatively affect other domains.

Mixed Results and Nonpredictors

The frequency of SNS use as a whole suggested no clear association with depression and anxiety. Longitudinal research suggests that depression and anxiety remain stable in the context of how frequently a user engages with SNSs [54,56,61,63,77] and the function of use holds clearer associations with depression and anxiety [75]. This is consistent with the literature examining general Internet use where total frequency of use is often not a predictor of depression, particularly when examining the social features of the Internet [28,125]. For example, when examining different functions on the Internet, Morgan and Cotten [29] showed that more hours spent using the Internet for social activities (IM'ing, chat rooms) are associated with decreased levels of depression and that informational uses and gaming are associated with increases in depression.

While total SNS use may not affect psychopathology, it may be related to subjective well-being. This was illustrated in the study by Kross et al [63], in which more frequent SNS use was related to experiencing more negative affect and reducing life satisfaction. As frequent experience of negative affect may contribute to the onset and maintenance of depression, it is likely that a pathway to poorer mental health outcomes exists via the impact SNS use has on the frequency of experiencing positive and negative emotions [54,63,67]. Additionally, other SNS features and cognitive processes (eg, network size, structure, and composition, tendency to ruminate, frequent social comparison) may be more informative in describing the impact frequent SNS use has on mental health.

In contrast with the literature examining social network size and structure offline [12,136], SNS friendship network size, on the whole, was not associated with depression or anxiety. However, some evidence has shown distinct network structure differences between individuals with depression and those without in terms of the interconnection between friends within a network [84]. Individuals with depression or anxiety have previously been shown to have more impoverished social networks, and changes in mental health are often associated with changes in an individual's social network [12,137]. Impoverished social networks are often a risk factor for depression and anxiety by reducing access to "buffering" social support and increasing feelings of isolation [138-140]. They may also result from poor-quality social interactions, often typical of depression and anxiety [137].

The absence of a clear association between depression or anxiety and the number of friends on SNSs may be explained by one of the major differences between the offline and online social networks; that is, the way friendships are maintained over time. As SNSs do not necessitate direct social interaction to maintain the status of "friendship," many users may not actively redefine their networks [141]. It is likely that the social pruning and the dissolution of social ties associated with mental illnesses such as depression and anxiety may not be visible on SNSs. Social pruning does occur for many SNS users (eg, 63% of American SNS users endorsed that they had removed friends from the "friends" list; [141]), but how comprehensively this behavior is performed remains unknown. Therefore, change in mental health status for SNS users may not be as accurately detected by a decreased social network size online as it may be when observing offline networks. Other metrics, such as communication output and reciprocity, may be more informative in describing the social network changes associated with depression and anxiety. For instance, De Choudhury et al [91] demonstrated that the volume of tweets and the associated replies were reduced in Twitter users with depression compared with those without.

Strengths and Limitations

As with any study, there are both strengths and limitations of this review. We included a basic criterion for bias that focused on evaluating the methodology of studies, which considered whether papers included (1) the use of psychometrically reliable and valid measures; (2) an external measurement criterion for mental health; and (3) description of sample demographics that included basic SNS user activity statistics. Only 9 studies were excluded for bias, suggesting that there is relative strength in defining the variables of interest in this field. However, a greater focus on defining the SNS characteristics of the sample is required.

The review attempted to characterize the research in terms of the populations and specific SNSs that have been studied. Studies have focused rather narrowly on the young adult population. While these individuals tend to represent the highest membership category of SNSs, recent estimates have suggested that SNS use is becoming more evenly represented across the life span, with more than 50% of older Internet users (65+ years) now also using SNSs [7]. This is an important consideration for future research as the social connection that may be gained through SNSs may provide more benefit for older users as quality of the interactions, particularly through language use, may vary significantly over the life span [142].

Despite the systematic approach to this review, the identified themes are not exhaustive. Other themes such as the differences between SNS users and nonusers and SNS use motives may have been extracted and more explicitly discussed. The discussion of results was limited to the depression or anxiety context and did not discuss findings outside this scope. Well-being, which clearly is becoming a growing area of interest (Figure 1), was only included if there was also a focus on depression or anxiety. Future studies might extend to other aspects of mental illness and wellness.

Finally, although we identified some moderating characteristics, few studies have considered individual differences such as gender and personality and their interaction with SNS variables. Future studies might give greater attention to how characteristics of users impact the identified factors.

Implications and Future Directions

The results of this systematic review have revealed considerable support for the importance of examining the content and quality of the interactions a user has with SNSs. As such, the language used in interactions on SNSs could become a target of interest, particularly as it has been shown to be sensitive in identifying individuals with depression [91,92,94,143]. Further research

should also focus on the interplay between the network structure components and dynamic interactions observable on SNSs. The SNS friend structure could be instrumental in defining the type and efficiency with which social resources may be accessed on SNSs. Examining network structure in concert with the quality of interactions, characteristics such as perceived social support, and mental health could provide rich explanations for why some people benefit from SNS use and others are placed at risk, echoing the detailed social network research that has occurred offline (eg, [12]).

Only a few studies in this review utilized SNS-derived data to answer their research questions. The majority focused on the use of self-report survey and relied on participant estimates of their SNS behaviors, which may have introduced considerable retrospective bias. This bias was addressed to some extent by including ESMs that more accurately sample a participant's lived experience [144]. The studies directly observing SNS behaviors indicate that the mental health status of SNS users may be at least partly derived from their patterns of use, language expression, and profile information. These findings provide more weight to the potential of using computational science techniques within psychological research, particularly in characterizing well-being in large community samples [33-35,145,146], as well as predicting personality [147]; see also [148]. In reference to depression and anxiety, SNS data hold huge potential for early identification and time-sensitive monitoring of symptoms [143]. SNS data should be leveraged in future research as a part of ESMs to provide real-time, unobtrusive accounts of social behavior in a natural setting.

Conclusions

This systematic review examined the recent research on associations between SNSs and depression and anxiety. It examined findings in association with the suggested mediators and moderators and the links made with well-being. With more than 50% of adults using multiple SNSs [7], they permeate many aspects of daily life. For many, SNSs represent a way to socially connect with others. However, for others, SNSs may encourage and perpetuate maladaptive tendencies. SNSs maintain and reflect the complexities of the offline social environment and the risks and benefits it may pose to mental health. SNSs represent a novel, unobtrusive, real-time way to observe and leverage mental health and well-being information in a natural setting, with the ultimate potential to positively influence mental health.

Conflicts of Interest

None declared.

Multimedia Appendix 1

Summary of studies included in the systematic review.

[PDF File (Adobe PDF File), 321KB - mental_v3i4e50_app1.pdf]

Multimedia Appendix 2

Main results table: associations between depression, anxiety, and social networking site outcomes across the 70 reviewed studies.

[PDF File (Adobe PDF File), 106KB - mental_v3i4e50_app2.pdf]

References

- Ellison NB, Boyd DM. Sociality through social network sites. In: Dutton WH, editor. The Oxford Handbook of Internet Studies. Oxford: Oxford University Press; 2013:151-172.
- Best P, Manktelow R, Taylor B. Online communication, social media and adolescent wellbeing: a systematic narrative review. Child Youth Serv Rev 2014 Jun;41:27-36. [doi: <u>10.1016/j.childyouth.2014.03.001</u>]
- Spies Shapiro LA, Margolin G. Growing up wired: social networking sites and adolescent psychosocial development. Clin Child Fam Psychol Rev 2014 Mar;17(1):1-18 [FREE Full text] [doi: 10.1007/s10567-013-0135-1] [Medline: 23645343]
- Boyd DM, Ellison NB. Social network sites: definition, history, and scholarship. J Comput-Mediat Comm 2007;13(1):210-230. [doi: 10.1111/j.1083-6101.2007.00393.x]
- Back MD, Stopfer JM, Vazire S, Gaddis S, Schmukle SC, Egloff B, et al. Facebook profiles reflect actual personality, not self-idealization. Psychol Sci 2010 Mar;21(3):372-374. [doi: 10.1177/0956797609360756] [Medline: 20424071]
- Baek YM, Bae Y, Jang H. Social and parasocial relationships on social network sites and their differential relationships with users' psychological well-being. Cyberpsychol Behav Soc Netw 2013 Jul;16(7):512-517. [doi: <u>10.1089/cyber.2012.0510</u>] [Medline: <u>23697533</u>]
- Duggan M, Ellison NB, Lampe C, Lenhart A, Madden M. Social media update 2014. Pew Research Center. 2015. URL: http://www.pewinternet.org/files/2015/01/PI_SocialMediaUpdate20144.pdf [WebCite Cache ID 6W9EOZpGn]
- Marroquín B. Interpersonal emotion regulation as a mechanism of social support in depression. Clin Psychol Rev 2011 Dec;31(8):1276-1290. [doi: 10.1016/j.cpr.2011.09.005] [Medline: 21983267]
- Baxter AJ, Scott KM, Vos T, Whiteford HA. Global prevalence of anxiety disorders: a systematic review and meta-regression. Psychol Med 2013 May;43(5):897-910. [doi: <u>10.1017/S003329171200147X</u>] [Medline: <u>22781489</u>]

http://mental.jmir.org/2016/4/e50/

JMIR Ment Health 2016 | vol. 3 | iss. 4 | e50 | p.12 (page number not for citation purposes)
- Ferrari AJ, Somerville AJ, Baxter AJ, Norman R, Patten SB, Vos T, et al. Global variation in the prevalence and incidence of major depressive disorder: a systematic review of the epidemiological literature. Psychol Med 2013 Mar;43(3):471-481. [doi: 10.1017/S0033291712001511] [Medline: 22831756]
- Hirschfeld RM. The comorbidity of major depression and anxiety disorders: recognition and management in primary care. Prim Care Companion J Clin Psychiatry 2001 Dec;3(6):244-254 [FREE Full text] [Medline: <u>15014592</u>]
- Rosenquist JN, Fowler JH, Christakis NA. Social network determinants of depression. Mol Psychiatry 2011 Mar;16(3):273-281 [FREE Full text] [doi: 10.1038/mp.2010.13] [Medline: 20231839]
- Steger MF, Kashdan TB. Depression and everyday social activity, belonging, and well-being. J Couns Psychol 2009 Apr;56(2):289-300 [FREE Full text] [doi: 10.1037/a0015416] [Medline: 20428460]
- De Silva MJ, McKenzie K, Harpham T, Huttly SR. Social capital and mental illness: a systematic review. J Epidemiol Commun H 2005 Aug;59(8):619-627 [FREE Full text] [doi: 10.1136/jech.2004.029678] [Medline: 16020636]
- Keyes CL. Mental illness and/or mental health? investigating axioms of the complete state model of health. J Consult Clin Psychol 2005 Jun;73(3):539-548. [doi: <u>10.1037/0022-006X.73.3.539</u>] [Medline: <u>15982151</u>]
- Barnett PA, Gotlib IH. Psychosocial functioning and depression: distinguishing among antecedents, concomitants, and consequences. Psychol Bull 1988 Jul;104(1):97-126. [Medline: <u>3043529</u>]
- Derks D, Fischer AH, Bos AE. The role of emotion in computer-mediated communication: a review. Comput Hum Behav 2008 May;24(3):766-785. [doi: 10.1016/j.chb.2007.04.004]
- Grav S, Hellzèn O, Romild U, Stordal E. Association between social support and depression in the general population: the HUNT study, a cross-sectional survey. J Clin Nurs 2012 Jan;21(1-2):111-120. [doi: <u>10.1111/j.1365-2702.2011.03868.x</u>] [Medline: <u>22017561</u>]
- Wilson RE, Gosling SD, Graham LT. A review of Facebook research in the social sciences. Perspect Psychol Sci 2012 May;7(3):203-220. [doi: 10.1177/1745691612442904] [Medline: 26168459]
- Ellison NB, Steinfield C, Lampe C. The benefits of Facebook "friends": social capital and college students' use of online social network sites. J Compt-Mediat Comm 2007;12:1143-1168. [doi: 10.1111/j.1083-6101.2007.00367.x]
- Guo Y, Li Y, Ito N. Exploring the predicted effect of social networking site use on perceived social capital and psychological well-being of Chinese international students in Japan. Cyberpsychol Behav Soc Netw 2014 Jan;17(1):52-58. [doi: <u>10.1089/cyber.2012.0537</u>] [Medline: <u>23971431</u>]
- Jin B. How lonely people use and perceive Facebook. Comput Human Behav 2013 Nov;29(6):2463-2470. [doi: 10.1016/j.chb.2013.05.034]
- Lee KT, Noh MJ, Koo DM. Lonely people are no longer lonely on social networking sites: the mediating role of self-disclosure and social support. Cyberpsychol Behav Soc Netw 2013 Jun;16(6):413-418. [doi: <u>10.1089/cyber.2012.0553</u>] [Medline: <u>23621716</u>]
- Manago AM, Taylor T, Greenfield PM. Me and my 400 friends: the anatomy of college students' Facebook networks, their communication patterns, and well-being. Dev Psychol 2012 Mar;48(2):369-380. [doi: <u>10.1037/a0026338</u>] [Medline: <u>22288367</u>]
- Nabi RL, Prestin A, So J. Facebook friends with (health) benefits? Exploring social network site use and perceptions of social support, stress, and well-being. Cyberpsychol Behav Soc Netw 2013 Oct;16(10):721-727. [doi: 10.1089/cyber.2012.0521] [Medline: 23790356]
- Oh HJ, Ozkaya E, LaRose R. How does online social networking enhance life satisfaction? The relationships among online supportive interaction, affect, perceived social support, sense of community, and life satisfaction. Comput Hum Behav 2014 Jan;30:69-78. [doi: 10.1016/j.chb.2013.07.053]
- Sowislo JF, Orth U. Does low self-esteem predict depression and anxiety? A meta-analysis of longitudinal studies. Psychol Bull 2013 Jan;139(1):213-240. [doi: 10.1037/a0028931] [Medline: 22730921]
- Kraut R, Kiesler S, Boneva B, Cummings J, Helgeson V, Crawford A. Internet paradox revisited. J Soc Issues 2002 Jan;58(1):49-74. [doi: 10.1111/1540-4560.00248]
- Morgan C, Cotten SR. The relationship between Internet activities and depressive symptoms in a sample of college freshmen. Cyberpsychol Behav 2003 Apr;6(2):133-142. [doi: 10.1089/109493103321640329] [Medline: 12804025]
- Pennebaker JW. Writing about emotional experiences as a therapeutic process. Psychol Sci 1997 May 01;8(3):162-166. [doi: 10.1111/j.1467-9280.1997.tb00403.x]
- Smyth JM. Written emotional expression: effect sizes, outcome types, and moderating variables. J Consult Clin Psychol 1998 Feb;66(1):174-184. [Medline: 9489272]
- Valkenburg PM, Peter J, Schouten AP. Friend networking sites and their relationship to adolescents' well-being and social self-esteem. Cyberpsychol Behav 2006 Oct;9(5):584-590. [doi: 10.1089/cpb.2006.9.584] [Medline: 17034326]
- Coviello L, Sohn Y, Kramer AD, Marlow C, Franceschetti M, Christakis NA, et al. Detecting emotional contagion in massive social networks. PLoS One 2014;9(3):e90315 [FREE Full text] [doi: <u>10.1371/journal.pone.0090315</u>] [Medline: <u>24621792</u>]
- Kramer AD. The spread of emotion via Facebook. 2012 Presented at: Proceedings of the SIGCHI Conference on Human Factors in Computing Systems; 2012; Austin, TX. [doi: <u>10.1145/2207676.2207787</u>]

http://mental.jmir.org/2016/4/e50/

JMIR Ment Health 2016 | vol. 3 | iss. 4 | e50 | p.13 (page number not for citation purposes)

- Kramer AD, Guillory JE, Hancock JT. Experimental evidence of massive-scale emotional contagion through social networks. Proc Natl Acad Sci U S A 2014 Jun 17;111(24):8788-8790 [FREE Full text] [doi: <u>10.1073/pnas.1320040111</u>] [Medline: <u>24889601</u>]
- Cole DA, Turner Jr JE. Models of cognitive mediation and moderation in child depression. J Abnorm Psychol 1993 May;102(2):271-281. [Medline: <u>8315139</u>]
- Phillips WJ, Hine DW, Thorsteinsson EB. Implicit cognition and depression: a meta-analysis. Clin Psychol Rev 2010 Aug;30(6):691-709. [doi: 10.1016/j.cpr.2010.05.002] [Medline: 20538393]
- Prinstein MJ, Aikins JW. Cognitive moderators of the longitudinal association between peer rejection and adolescent depressive symptoms. J Abnorm Child Psychol 2004 Apr;32(2):147-158 [FREE Full text] [Medline: 15164857]
- Bosacki S, Dane A, Marini Z, YLC-CURA. Peer relationships and internalizing problems in adolescents: mediating role of self-esteem. Emotional Behav Difficulties 2007 Dec;12(4):261-282. [doi: 10.1080/13632750701664293]
- Nolen-Hoeksema S, Wisco BE, Lyubomirsky S. Rethinking rumination. Perspect Psychol Sci 2008 Sep;3(5):400-424. [doi: 10.1111/j.1745-6924.2008.00088.x] [Medline: 26158958]
- Michl LC, McLaughlin KA, Shepherd K, Nolen-Hoeksema S. Rumination as a mechanism linking stressful life events to symptoms of depression and anxiety: longitudinal evidence in early adolescents and adults. J Abnorm Psychol 2013 May;122(2):339-352 [FREE Full text] [doi: 10.1037/a0031994] [Medline: 23713497]
- Wang X, Cai L, Qian J, Peng J. Social support moderates stress effects on depression. Int J Ment Health Syst 2014;8(1):41 [FREE Full text] [doi: 10.1186/1752-4458-8-41] [Medline: 25422673]
- Higgins JPT, Altman DG, Sterne JAC. Assessing risk of bias in included studies. In: Higgins JPT, Green S, editors. Cochrane Handbook for Systematic Reviews of Interventions. Version 5.1.0 (updated March 2011): The Cochrane Collaboration and John Wiley & Sons Ltd; 2011:187-235.
- Cavazos-Rehg PA, Krauss MJ, Sowles S, Connolly S, Rosas C, Bharadwaj M, et al. A content analysis of depression-related tweets. Comput Human Behav 2016 Jan 1;54:351-357. [doi: <u>10.1016/j.chb.2015.08.023</u>] [Medline: <u>26392678</u>]
- Larsen ME, Boonstra TW, Batterham PJ, O'Dea B, Paris C, Christensen H. We feel: mapping emotion on Twitter. IEEE J Biomed Health Inform 2015 Jul;19(4):1246-1252. [doi: <u>10.1109/JBHI.2015.2403839</u>] [Medline: <u>25700477</u>]
- Masuda N, Kurahashi I, Onari H. Suicide ideation of individuals in online social networks. PLoS One 2013;8(4):e62262. [doi: 10.1371/journal.pone.0062262.t002]
- Nambisan P, Luo Z, Kapoor A, Patrick TB, Cisler RA. Social media, big data, and public health informatics: ruminating behavior of depression revealed through Twitter. 2015 Presented at: Proceedings of the 48th Hawaii International Conference on System Sciences; 5-8 Jan, 2015; Hawaii, USA p. 2906-2913. [doi: 10.1109/HICSS.2015.351]
- Preito VM, Matos S, Àlvarez M, Cacheda F, Oliveira JL. Twitter: a good place to detect health conditions. PLos One 2014 Jan 29;9(1):e86191. [doi: <u>10.1371/journal.pone.0086191.t001</u>]
- Semenov A, Natekin A, Nikolenko S, Upravitelev P, Trofimov M, Kharchenko M. Discerning depression propensity among participants of suicide and depression-related groups of Vk.com. In: Khachay MY, Konstantinova N, Panchenko A, Ignatov DI, Labunets VG, editors. Analysis of Images, Social Networks and Texts. Russia: Switzerlandpringer International Publishing; 2015:24-35.
- Wilson ML, Ali S, Valstar MF. Finding information about mental health in microblogging platforms: a case study of depression. 2014 Presented at: Proceedings of the 5th Information Interaction in Context Symposium; 26-29 August, 2014; Regensburg, Germany p. 8-17. [doi: 10.1145/2637002.2637006]
- Yang W, Mu L. GIS analysis of depression among Twitter users. Appl Geogr 2015 Jun;60:217-223. [doi: 10.1016/j.apgeog.2014.10.016]
- Yang W, Mu L, Shen Y. Effect of climate and seasonality on depressed mood among twitter users. Appl Geogr 2015 Sep;63:184-191. [doi: 10.1016/j.apgeog.2015.06.017]
- Banjanin N, Banjanin N, Dimitrijevic I, Pantic I. Relationship between internet use and depression: focus on physiological mood oscillations, social networking and online addictive behavior. Comput Human Behav 2015 Feb;43:308-312. [doi: 10.1016/j.chb.2014.11.013]
- Davila J, Hershenberg R, Feinstein BA, Gorman K, Bhatia V, Starr LR. Frequency and quality of social networking among young adults: associations with depressive symptoms, rumination, and corumination. Psychol Pop Media Cult 2012 Apr 1;1(2):72-86 [FREE Full text] [doi: 10.1037/a0027512] [Medline: 24490122]
- Farahani HA, Kazemi Z, Aghamohamadi S, Bakhtiarvand F, Ansari M. Examining mental health indices in students using Facebook in Iran. Procedia Soc Behav Sci 2011;28:811-814. [doi: <u>10.1016/j.sbspro.2011.11.148</u>]
- Feinstein BA, Bhatia V, Hershenberg R, Davila J. Another venue for problematic interpersonal behavior: the effects of depressive and anxious symptoms on social networking experiences. J Soc Clin Psychol 2012 Apr;31(4):356-382. [doi: 10.1521/jscp.2012.31.4.356]
- Fernandez KC, Levinson CA, Rodebaugh TL. Profiling: predicting social anxiety from Facebook profiles. Soc Psychological Pers Sci 2012 Jan 19;3(6):706-713. [doi: 10.1177/1948550611434967]
- Giota KG, Kleftaras G. The role of personality and depression in problematic use of social networking sites in Greece. Cyberpsychology J Psychosoc Res Cyberspace 2013;7(3):Article 1. [doi: <u>10.5817/CP2013-3-6</u>]

http://mental.jmir.org/2016/4/e50/

JMIR Ment Health 2016 | vol. 3 | iss. 4 | e50 | p.14 (page number not for citation purposes)

- Green T, Wilhelmsen T, Wilmots E, Dodd B, Quinn S. Social anxiety, attributes of online communication and self-disclosure across private and public Facebook communication. Comput Human Behav 2016 May;58:206-213. [doi: <u>10.1016/j.chb.2015.12.066</u>]
- Hong F, Huang D, Lin H, Chiu S. Analysis of the psychological traits, Facebook usage, and Facebook addiction model of Taiwanese university students. Telemat Inform 2014 Nov;31(4):597-606. [doi: 10.1016/j.tele.2014.01.001]
- Jelenchick LA, Eickhoff JC, Moreno MA. "Facebook depression?" Social networking site use and depression in older adolescents. J Adolesc Health 2013 Jan;52(1):128-130. [doi: <u>10.1016/j.jadohealth.2012.05.008</u>] [Medline: <u>23260846</u>]
- Koc M, Gulyagci S. Facebook addiction among Turkish college students: the role of psychological health, demographic, and usage characteristics. Cyberpsychol Behav Soc Netw 2013 Apr;16(4):279-284. [doi: <u>10.1089/cyber.2012.0249</u>] [Medline: <u>23286695</u>]
- Kross E, Verduyn P, Demiralp E, Park J, Lee DS, Lin N, et al. Facebook use predicts declines in subjective well-being in young adults. PLoS One 2013;8(8):e69841 [FREE Full text] [doi: 10.1371/journal.pone.0069841] [Medline: 23967061]
- Labrague LJ. Facebook use and adolescents' emotional state of depression, anxiety, and stress. Health Sci J 2014;8(1):80-89. [doi: 10.1145/1316624.1316682]
- Lee-Won RJ, Herzog L, Park SG. Hooked on Facebook: the role of social anxiety and need for social assurance in problematic use of Facebook. Cyberpsychol Behav Soc Netw 2015 Oct;18(10):567-574. [doi: <u>10.1089/cyber.2015.0002</u>] [Medline: <u>26383178</u>]
- Lin LY, Sidani JE, Shensa A, Radovic A, Miller E, Colditz JB, et al. Association between social media use and depression among U.S. young adults. Depress Anxiety 2016 Apr;33(4):323-331. [doi: 10.1002/da.22466] [Medline: 26783723]
- Locatelli SM, Kluwe K, Bryant FB. Facebook use and the tendency to ruminate among college students: testing mediational hypotheses. J Educ Comput Res 2012 Sep 7;46(4):377-394. [doi: <u>10.2190/EC.46.4.d</u>]
- Lup K, Trub L, Rosenthal L. Instagram #instasad?: exploring associations among instagram use, depressive symptoms, negative social comparison, and strangers followed. Cyberpsychol Behav Soc Netw 2015 May;18(5):247-252. [doi: <u>10.1089/cyber.2014.0560</u>] [Medline: <u>25965859</u>]
- McCord B, Rodebaugh TL, Levinson CA. Facebook: social uses and anxiety. Comput Hum Behav 2014 May;34:23-27. [doi: 10.1016/j.chb.2014.01.020]
- Mok WT, Sing R, Jiang X, See SL. Investigation of social media on depression. 2014 Presented at: Proceedings of the 9th International Symposium on Chinese Spoken Language Processing; 12-14 Sept, 2014; Singapore p. 488-491. [doi: 10.1109/ISCSLP.2014.6936690]
- Morin-Major JK, Marin MF, Durand N, Wan N, Juster RP, Lupien SJ. Facebook behaviors associated with diurnal cortisol in adolescents: is befriending stressful? Psychoneuroendocrinology 2016 Jan;63:238-246. [doi: 10.1016/j.psyneuen.2015.10.005] [Medline: 26519778]
- Pantic I, Damjanovic A, Todorovic J, Topalovic D, Bojovic-Jovic D, Ristic S, et al. Association between online social networking and depression in high school students: behavioral physiology viewpoint. Psychiatr Danub 2012 Mar;24(1):90-93 [FREE Full text] [Medline: 22447092]
- Rae JR, Lonborg SD. Do motivations for using Facebook moderate the association between Facebook use and psychological well-being? Front Psychol 2015;6:771 [FREE Full text] [doi: 10.3389/fpsyg.2015.00771] [Medline: 26124733]
- Rosen LD, Whaling K, Rab S, Carrier LM, Cheever NA. Is Facebook creating "iDisorders"? The link between clinical symptoms of psychiatric disorders and technology use, attitudes and anxiety. Comput Hum Behav 2013 May;29(3):1243-1254. [doi: 10.1016/j.chb.2012.11.012]
- Shaw AM, Timpano KR, Tran TB, Joormann J. Correlates of Facebook usage patterns: the relationship between passive Facebook use, social anxiety symptoms, and brooding. Comput Hum Behav 2015 Jul;48:575-580. [doi: 10.1016/j.chb.2015.02.003]
- Simoncic TE, Kuhlman KR, Vargas I, Houchins S, Lopez-Duran NL. Facebook use and depressive symptomatology: investigating the role of neuroticism and extraversion in youth. Comput Hum Behav 2014 Nov;40:1-5. [doi: 10.1016/j.chb.2014.07.039]
- Steers MN, Wickham RE, Acitelli LK. Seeing everyone else's highlight reels: how Facebook usage is linked to depressive symptoms. J Soc Clin Psychol 2014 Oct;33(8):701-731. [doi: <u>10.1521/jscp.2014.33.8.701</u>]
- Tandoc EC, Ferrucci P, Duffy M. Facebook use, envy, and depression among college students: is Facebooking depressing? Comput Hum Behav 2015 Feb;43:139-146. [doi: 10.1016/j.chb.2014.10.053]
- Wright KB, Rosenberg J, Egbert N, Ploeger NA, Bernard DR, King S. Communication competence, social support, and depression among college students: a model of facebook and face-to-face support network influence. J Health Commun 2013;18(1):41-57. [doi: 10.1080/10810730.2012.688250] [Medline: 23030518]
- Baker AE, Jeske D. Assertiveness and anxiety effects in traditional and online interactions. Int J Cyber Behav Psychol Learning 2015;5(3):30-46. [doi: 10.4018/IJCBPL.2015070103]
- Frison E, Subrahmanyam K, Eggermont S. The short-term longitudinal and reciprocal relationships between peer victimization on Facebook and adolescents' well-being. J Youth Adolesc 2016 Sep;45(9):1755-1771. [doi: <u>10.1007/s10964-016-0436-z]</u> [Medline: <u>26880284</u>]

http://mental.jmir.org/2016/4/e50/

- Weidman AC, Levinson CA. I'm still socially anxious online: offline relationship impairment characterizing social anxiety manifests and is accurately perceived in online social networking profiles. Comput Hum Behav 2015 Aug;49:12-19. [doi: <u>10.1016/j.chb.2014.12.045</u>]
- Park S, Lee SW, Kwak J, Cha M, Jeong B. Activities on Facebook reveal the depressive state of users. J Med Internet Res 2013;15(10):e217 [FREE Full text] [doi: <u>10.2196/jmir.2718</u>] [Medline: <u>24084314</u>]
- Park S, Kim I, Lee S, Yoo J, Jeong B, Cha M. Manifestation of depression and loneliness on social networks: a case study of young adults on Facebook. 2015 Presented at: Proceedings of the 18th ACM Conference on Computer-Supported Cooperative Work and Social Computing; 2015; Vancouver, Canada p. 557-570. [doi: 10.1145/2675133.2675139]
- Moreno MA, Jelenchick LA, Egan KG, Cox E, Young H, Gannon KE, et al. Feeling bad on Facebook: depression disclosures by college students on a social networking site. Depress Anxiety 2011 Jun;28(6):447-455 [FREE Full text] [doi: 10.1002/da.20805] [Medline: 21400639]
- Davidson T, Farquhar LK. Correlates of social anxiety, religion, and Facebook. J Media Religion 2014 Nov 18;13(4):208-225. [doi: 10.1080/15348423.2014.971566]
- Tsai C, Shen P, Chiang Y. Meeting ex-partners on Facebook: users' anxiety and severity of depression. Behav Inf Technol 2015;34(7):668-677. [doi: 10.1080/0144929X.2014.981585]
- Mota-Pereira J. Facebook enhances antidepressant pharmacotherapy effects. Scientific World J 2014;2014:1-6 [FREE Full text] [doi: 10.1155/2014/892048] [Medline: 24574930]
- Homan CM, Lu N, Tu X, Lytle MC, Silenzio VM. Social structure and depression in TrevorSpace. 2014 Presented at: Proceedings of the 17th ACM Conference on Computer Supported Cooperative Work and Social Computing; 2014; ACM, Baltimore, Maryland p. 615-624. [doi: 10.1145/2531602.2531704]
- Takahashi Y, Uchida C, Miyaki K, Sakai M, Shimbo T, Nakayama T. Potential benefits and harms of a peer support social network service on the Internet for people with depressive tendencies: qualitative content analysis and social network analysis. J Med Internet Res 2009;11(3):e29 [FREE Full text] [doi: 10.2196/jmir.1142] [Medline: 19632979]
- De Choudhury M, Counts S, Horvitz E. Social media as a measurement tool of depression in populations. 2013 Presented at: Proceedings of the 5th Annual ACM Web Science Conference; 2013; Paris, France p. 47-56. [doi: 10.1145/2464464.2464480]
- De Choudhury M, Counts S, Horvitz EJ, Hoff A. Characterizing and predicting postpartum depression from shared facebook data. 2014 Presented at: Proceedings of the 17th ACM Conference on Computer Supported Cooperative Work and Social Computing; 2014; Baltimore, Maryland p. 626-638. [doi: 10.1145/2531602.2531675]
- Park J, Lee DS, Shablack H, Verduyn P, Deldin P, Ybarra O, et al. When perceptions defy reality: the relationships between depression and actual and perceived Facebook social support. J Affect Disord 2016 Aug;200:37-44. [doi: 10.1016/j.jad.2016.01.048] [Medline: 27126138]
- Moreno MA, Grant A, Kacvinsky L, Moreno P, Fleming M. Older adolescents' views regarding participation in Facebook research. J Adolesc Health 2012 Nov;51(5):439-444 [FREE Full text] [doi: <u>10.1016/j.jadohealth.2012.02.001</u>] [Medline: <u>23084164</u>]
- Settanni M, Marengo D. Sharing feelings online: studying emotional well-being via automated text analysis of Facebook posts. Front Psychol 2015;6:1045 [FREE Full text] [doi: 10.3389/fpsyg.2015.01045] [Medline: 26257692]
- Tsugawa S, Kikuchi Y, Kishino F, Nakajima K, Itoh Y, Ohsaki H. Recognising depression from Twitter activity. 2015 Presented at: Proceedings from the 33rd ACM Conference on Human Factors in Computing Systems; 2015; Seoul, Korea p. 3187-3196. [doi: 10.1145/2702123.2702280]
- Dumitrache SD, Mitrofan L, Petrov Z. Self-image and depressive tendencies among adolescent Facebook users. Revista de Psihologie 2012;58(4):285-295.
- Deters FG, Mehl MR, Eid M. Social responses to Facebook status updates: the role of extraversion and social anxiety. Comput Hum Behav 2016 Aug;61:1-13. [doi: 10.1016/j.chb.2016.02.093]
- 99. Ghosh A, Dasgupta S. Psychological predictors of Facebook use. J Indian Acad Appl Psychol 2015;41(1):101-109.
- Baker JR, Moore SM. Distress, coping, and blogging: comparing new Myspace users by their intention to blog. Cyberpsychol Behav 2008 Feb;11(1):81-85. [doi: 10.1089/cpb.2007.9930] [Medline: 18275317]
- Baker JR, Moore SM. Blogging as a social tool: a psychosocial examination of the effects of blogging. Cyberpsychol Behav 2008 Dec;11(6):747-749. [doi: <u>10.1089/cpb.2008.0053</u>] [Medline: <u>19072151</u>]
- Deters FG, Mehl MR. Does posting Facebook status updates increase or decrease loneliness? An online social networking experiment. Soc Psychol Pers Sci 2013 Sep 1;4(5) [FREE Full text] [doi: 10.1177/1948550612469233] [Medline: 24224070]
- Szwedo DE, Mikami AY, Allen JP. Qualities of peer relations on social networking websites: predictions from negative mother-teen interactions. J Res Adolesc 2011 Sep;21(3):595-607 [FREE Full text] [doi: 10.1111/j.1532-7795.2010.00692.x] [Medline: 21860584]
- Casale S, Fioravanti G. Satisfying needs through social networking sites: a pathway towards problematic Internet use for socially anxious people? Addictive Behav Rep 2015 Jun;1:34-39. [doi: 10.1016/j.abrep.2015.03.008]
- Burke TJ, Ruppel EK. Facebook self-presentational motives: daily effects on social anxiety and interaction success. Commun Stud 2015;66(2):204-217. [doi: 10.1080/10510974.2014.884014]

http://mental.jmir.org/2016/4/e50/

JMIR Ment Health 2016 | vol. 3 | iss. 4 | e50 | p.16 (page number not for citation purposes)

- Bodroža B, Jovanović T. Validation of the new scale for measuring behaviors of Facebook users: psycho-Social Aspects of Facebook Use (PSAFU). Comput Hum Behav 2016 Jan;54:425-435. [doi: <u>10.1016/j.chb.2015.07.032</u>]
- Landoll RR, La Greca AM, Lai BS. Aversive peer experiences on social networking sites: development of the Social Networking-Peer Experiences Questionnaire (SN-PEQ). J Res Adolesc 2013 Dec 1;23(4) [FREE Full text] [doi: 10.1111/jora.12022] [Medline: 24288449]
- Moberg FB, Anestis MD. A preliminary examination of the relationship between social networking interactions, internet use, and thwarted belongingness. Crisis 2015 May;36(3):187-193. [doi: 10.1027/0227-5910/a000311] [Medline: 26088827]
- Hong J, Hwang M, Hsu C, Tai K, Kuo Y. Belief in dangerous virtual communities as a predictor of continuance intention mediated by general and online social anxiety: the Facebook perspective. Comput Hum Behav 2015 Jul;48:663-670. [doi: 10.1016/j.chb.2015.02.019]
- Frison E, Eggermont S. The impact of daily stress on adolescents' depressed mood: the role of social support seeking through Facebook. Comput Hum Behav 2015 Mar;44:315-325. [doi: <u>10.1016/j.chb.2014.11.070</u>]
- Indian M, Grieve R. When Facebook is easier than face-to-face: social support derived from Facebook in socially anxious individuals. Pers Indiv Differ 2014 Mar;59:102-106. [doi: <u>10.1016/j.paid.2013.11.016</u>]
- McCloskey W, Iwanicki S, Lauterbach D, Giammittorio DM, Maxwell K. Are Facebook "Friends" Helpful? Development of a Facebook-based measure of social support and examination of relationships among depression, quality of life, and social support. Cyberpsychol Behav Soc Netw 2015 Sep;18(9):499-505. [doi: <u>10.1089/cyber.2014.0538</u>] [Medline: <u>26348809</u>]
- Grieve R, Indian M, Witteveen K, Tolan GA, Marrington J. Face-to-face or Facebook: can social connectedness be derived online? Comput Hum Behav 2013 May;29(3):604-609. [doi: 10.1016/j.chb.2012.11.017]
- Lee SY. How do people compare themselves with others on social network sites?: the case of Facebook. Comput Hum Behav 2014 Mar;32:253-260. [doi: 10.1016/j.chb.2013.12.009]
- Feinstein BA, Hershenberg R, Bhatia V, Latack JA, Meuwly N, Davila J. Negative social comparison on Facebook and depressive symptoms: rumination as a mechanism. Psychol Pop Med Cult 2013;2(3):161-170. [doi: <u>10.1037/a0033111</u>]
- Appel H, Crusius J, Gerlach AL. Social comparison, envy, and depression on Facebook: a study looking at the effects of high comparison standards on depressed individuals. J Soc Clin Psychol 2015 Apr;34(4):277-289. [doi: 10.1521/jscp.2015.34.4.277]
- Andreassen CS, Billieux J, Griffiths MD, Kuss DJ, Demetrovics Z, Mazzoni E, et al. The relationship between addictive use of social media and video games and symptoms of psychiatric disorders: a large-scale cross-sectional study. Psychol Addict Behav 2016 Mar;30(2):252-262. [doi: 10.1037/adb0000160] [Medline: 26999354]
- Błachnio A, Przepiórka A, Pantic I. Internet use, Facebook intrusion, and depression: results of a cross-sectional study. Eur Psychiatry 2015 Sep;30(6):681-684. [doi: 10.1016/j.eurpsy.2015.04.002] [Medline: 25963476]
- Hanprathet N, Manwong M, Khumsri J, Yingyeun R, Phanasathit M. Facebook addiction and its relationship with mental health among Thai high school students. J Med Assoc Thai 2015 Apr;98(Suppl 3):S81-S90. [Medline: <u>26387393</u>]
- Moreau A, Laconi S, Delfour M, Chabrol H. Psychopathological profiles of adolescent and young adult problematic Facebook users. Comput Hum Behav 2015 Mar;44:64-69. [doi: 10.1016/j.chb.2014.11.045]
- Wegmann E, Stodt B, Brand M. Addictive use of social networking sites can be explained by the interaction of Internet use expectancies, Internet literacy, and psychopathological symptoms. J Behav Addict 2015 Sep;4(3):155-162 [FREE Full text] [doi: 10.1556/2006.4.2015.021] [Medline: 26551905]
- Rauch SM, Strobel C, Bella M, Odachowski Z, Bloom C. Face to face versus Facebook: does exposure to social networking web sites augment or attenuate physiological arousal among the socially anxious? Cyberpsychol Behav Soc Netw 2014 Mar;17(3):187-190. [doi: <u>10.1089/cyber.2012.0498</u>] [Medline: <u>24180223</u>]
- Gross EF, Juvonen J, Gable SL. Internet use and well-being in adolescence. J Soc Issues 2002 Jan;58(1):75-90. [doi: 10.1111/1540-4560.00249]
- Baker LR, Oswald DL. Shyness and online social networking services. J Soc Pers Relat 2010 Sep 10;27(7):873-889. [doi: 10.1177/0265407510375261]
- Morahan-Martin J. Internet use and abuse and psychological problems. In: Joinson AN, McKenna KYA, Postmes T, Reips UD, editors. Oxford Handbook of Internet Psychology. New York: Oxford University Press; 2007.
- Kawachi I, Berkman LF. Social ties and mental health. J Urban Health 2001 Sep;78(3):458-467 [FREE Full text] [doi: 10.1093/jurban/78.3.458] [Medline: 11564849]
- Uchino BN. Understanding the links between social support and physical health: a life-span perspective with emphasis on the separability of perceived and received support. Perspect Psychol Sci 2009 May;4(3):236-255. [doi: 10.1111/j.1745-6924.2009.01122.x] [Medline: 26158961]
- Wills TA, Shinar O. Measuring perceived and received social support. In: Cohen S, Underwood LG, Gottlieb BH, editors. Social Support Measurement and Intervention: A Guide for Health and Social Scientists. New York: Oxford University Press; 2000:86-135.
- Clark DM, McManus F. Information processing in social phobia. Biol Psychiatry 2002 Jan 1;51(1):92-100. [Medline: <u>11801234</u>]

- 130. Preotiuc-Pietro D, Sap M, Schwartz HA, Ungar LH. Mental illness detection at the World Well-Being Project for the CLPsych 2015 shared task. 2015 Presented at: Proceedings of the Workshop on Computational Linguistics and Clinical Psychology: from Linguistic Signal to Clinical Reality, NAACL; June 5, 2015; Denver, Colarado p. 40-45.
- Schwartz HA, Ungar LH. Data-driven content analysis of social media: a systematic overview of automated methods. Ann Am Acad Polit SS 2015 Apr 09;659(1):78-94. [doi: <u>10.1177/0002716215569197</u>]
- Schwartz HA, Sap M, Kern ML, Eichstaedt JC, Kapelner A, Agrawal M, et al. Predicting individual well-being through the language of social media. Pac Symp Biocomput 2016;21:516-527 [FREE Full text] [Medline: 26776214]
- Hancock J, Gee K, Ciaccio K, Lin J. I'm sad you're sad: emotional contagion in CMC. 2008 Presented at: Proceedings of the ACM Conference on Computer Supported Cooperative Work; November 8-12, 2008; San Diego, CA, USA p. 295-298. [doi: 10.1145/1460563.1460611]
- Papageorgiou C, Wells A. An empirical test of a clinical metacognitive model of rumination and depression. Cognitive Ther Res 2003;27(3):261-273. [doi: <u>10.1023/a:1023962332399</u>]
- Flynn M, Kecmanovic J, Alloy LB. An examination of integrated cognitive-interpersonal vulnerability to depression: the role of rumination, perceived social support, and interpersonal stress generation. Cognit Ther Res 2010 Oct;34(5):456-466 [FREE Full text] [doi: 10.1007/s10608-010-9300-8] [Medline: 25429169]
- Santini ZI, Koyanagi A, Tyrovolas S, Mason C, Haro JM. The association between social relationships and depression: a systematic review. J Affect Disord 2015 Apr 1;175:53-65. [doi: <u>10.1016/j.jad.2014.12.049</u>] [Medline: <u>25594512</u>]
- Schaefer DR, Kornienko O, Fox AM. Misery does not love company: network selection mechanisms and depression in homophily. Am Sociol Rev 2011 Sep 28;76(5):764-785. [doi: 10.1177/0003122411420813]
- Cornwell EY, Waite LJ. Social disconnectedness, perceived isolation, and health among older adults. J Health Soc Behav 2009 Mar;50(1):31-48 [FREE Full text] [Medline: <u>19413133</u>]
- House JS, Umberson D, Landis KR. Structures and processes of social support. Ann Rev Sociol 1988 Aug;14(1):293-318. [doi: 10.1146/annurev.so.14.080188.001453]
- Turner RJ, Marino F. Social support and social structure: a descriptive epidemiology. J Health Soc Behav 1994 Sep;35(3):193-212. [Medline: <u>7983334</u>]
- Madden M. Privacy management on social media sites. Pew Research Center. URL: <u>http://www.pewinternet.org/files/ old-media/Files/Reports/2012/PIP_Privacy_management_on_social_media_sites_022412.pdf</u> [accessed 2014-08-04] [WebCite Cache ID 6RZp0D5NC]
- Schwartz HA, Eichstaedt JC, Kern ML, Dziurzynski L, Ramones SM, Agrawal M, et al. Personality, gender, and age in the language of social media: the open-vocabulary approach. PLoS One 2013;8(9):e73791 [FREE Full text] [doi: 10.1371/journal.pone.0073791] [Medline: 24086296]
- 143. Schwartz HA, Eichstaedt J, Kern ML, Park G, Sap M, Stillwell D, et al. Toward assessing changes in degree of depression through Facebook. 2014 Presented at: Proceedings of the Workshop on Computational Linguistics and Clinical Psychology: From Linguistic Signal to Clinical Reality; June 27, 2014; Baltimore, Maryland p. 118-125.
- Csikszentmihalyi M, Larson R. Validity and reliability of the Experience-Sampling Method. J Nerv Ment Dis 1987 Sep;175(9):526-536. [Medline: <u>3655778</u>]
- Bollen J, Gonçalves B, Ruan G, Mao H. Happiness is assortative in online social networks. Artif Life 2011;17(3):237-251. [doi: 10.1162/artl a 00034] [Medline: 21554117]
- Schwartz H, Eichstaedt J, Kern ML, Dziurzynski L, Lucas RE, Agrawal M, et al. Characterising geographic variation in well-being using tweets. 2013 Presented at: Proceedings of the Seventh International AAAI Conference on Weblogs and Social Media (ICWSM); 2013; Boston, MA.
- Park G, Schwartz HA, Eichstaedt JC, Kern ML, Kosinski M, Stillwell DJ, et al. Automatic personality assessment through social media language. J Pers Soc Psychol 2015 Jun;108(6):934-952. [doi: 10.1037/pspp0000020] [Medline: 25365036]
- Kern ML, Park G, Eichstaedt JC, Schwartz HA, Sap M, Smith LK, et al. Gaining insight from social media language: methodologies and challenges. Psychol Methods 2016 Aug 8 (forthcoming). [doi: 10.1037/met0000091] [Medline: 27505683]

Abbreviations

CMC: computer-mediated communication ESM: experience sampling method LIWC: Linguistic Inquiry and Word Count MDD: major depressive disorder SNS: social networking site

http://mental.jmir.org/2016/4/e50/

Edited by B O'Dea; submitted 05.04.16; peer-reviewed by M Cha, S Rice, J Gritton; comments to author 24.05.16; revised version received 12.09.16; accepted 18.09.16; published 23.11.16 <u>Please cite as:</u> Seabrook EM, Kern ML, Rickard NS Social Networking Sites, Depression, and Anxiety: A Systematic Review JMIR Ment Health 2016;3(4):e50 URL block for the bioscience of Color (50)

JMIR Ment Health 2016;3(4):e50 URL: <u>http://mental.jmir.org/2016/4/e50/</u> doi:<u>10.2196/mental.5842</u> PMID:<u>27881357</u>

©Elizabeth M Seabrook, Margaret L Kern, Nikki S Rickard. Originally published in JMIR Mental Health (http://mental.jmir.org), 23.11.2016. This is an open-access article distributed under the terms of the Creative Commons Attribution License (http://creativecommons.org/licenses/by/2.0/), which permits unrestricted use, distribution, and reproduction in any medium, provided the original work, first published in JMIR Mental Health, is properly cited. The complete bibliographic information, a link to the original publication on http://mental.jmir.org/, as well as this copyright and license information must be included.

http://mental.jmir.org/2016/4/e50/

JMIR Ment Health 2016 | vol. 3 | iss. 4 | e50 | p.19 (page number not for citation purposes)

		20.00	Sample	Bias	Well-being	Mediators/
Study	ropulation	CNIC	Size	Rating		moderators
Andreassen et al. [117]	General	General	23533	-		
Appel et al. [116]	General	Facebook	89	1	>	
Baker & Jeske [80]	General	Facebook	184	0	>	
Baker & Moore [100]	General	MySpace	134	1		
Baker & Moore [101]	General	MySpace	58	1		
Banjanin et al. [53]	Adolescents	Facebook	336	0		
Blachnio et al. [118]	General	Facebook	672	1		
Bodroža & Jovanović [106]	General	Facebook	804	0		
Burke & Ruppel [105]	Young adults	Facebook	152	1		>
Casale & Fioravanti [104]	Young adults	Facebook	400	1		>
Davidson & Farguhar [86]	Young adults	Facebook	336	1		
Davila et al. [54]	Young adults	Facebook/MySpace	718	0		>
De Choudhury et al. [91]	General	Twitter	489	1		
De Choudhury et al. [92]	Other	Facebook	165	0		
große Deters & Mehl [102]	Young adults	Facebook	86	0	>	>
große Deters et al. [98]	Young adults	Facebook	362	1		>
Dumitrache et al. [97]	Adolescents	Facebook	123	7		
Farahani et al. [55]	Young adults	Facebook	265	1		
Feinstein et al. [115]	Young adults	Facebook	268	1		>
Feinstein et al. [56]	Young adults	Facebook/MySpace	301	0		
Fernandez et al. [57]	Young adults	Facebook	62	0		
Frison & Eggermont [110]	Adolescents	Facebook	910	1	>	>
Frison et al. [81]	Adolescents	Facebook	1621	0	>	>
Ghosh & Dasgupta [99]	General	Facebook	120	7	>	
Giota & Kleftaras [58]	Young adults	Facebook	143	1		
Green et al. [59]	General	Facebook	306	0		
Grieve et al. [113]	Young adults	Facebook	618	1	>	
Hanprathet et al. [119]	Adolescents	Facebook	832	-1		
Homan et al. [89]	Young adults	Trevorspace	195	1		
Hong et al. [109]	Young adults	Facebook	230	0		>

review

		Table 2.2 co	nt.			
Study	Population	SNS	Sample Size	Bias Rating	Well-being	Mediators/ moderators
Hong et al. [60]	Young adults	Facebook	241	-	>	>
Indian & Grieve [111]	General	Facebook	299	0	>	>
Jelenchick et al. [61]	Young adults	Facebook	190	0		
Koc & Gulyagci [62]	Young adults	Facebook	447	0		
Kross et al. [63]	General	Facebook	82	0	>	>
Labrague [64]	Young adults	Facebook	76	0		
Landoll et al. [107]	Young adults	General	430	-1		
Lee [114]	Young adults	Facebook	191	7	>	
Lee-Won et al. [65]	Young adults	Facebook	243	0		>
Lin et al. [66]	Adults	General	1787	0		
Locatelli et al. [67]	Young adults	Facebook	251	0	>	>
Lup et al. [68]	Young adults	Instagram	117	0		>
McCloskey et al. [112]	Young adults	Facebook	633		>	
McCord et al. [69]	Young adults	Facebook	216	0		
Moberg & Anestis [108]	Young adults	General	305	0		
Mok et al. [70]	General	General	59	2		
Moreau et al. [120]	Adolescents	Facebook	456	1		
Moreno et al. [85]	Young adults	Facebook	200	0		
Moreno et al. [94]	Young adults	Facebook	307	1		
Morin-Major et al. [71]	Adolescents	Facebook	88	0	>	
Mota-Pereira [88]	Clinical (depression)	Facebook	60	6		
Pantic et al. [72]	Adolescents	General	160	1		
Park et al. [83]	Young adults	Facebook	55	0		
Park et al. [84]	General	Facebook	212	0		
Park et al. [93]	Young adults	Facebook	103	0		
Rae & Lonborg [73]	Young adults	Facebook	119	0	>	>
Raunch et al. [122]	Young adults	Facebook	26	1		
Rosen et al. [74]	General	Facebook	1143	1		
Settani & Marengo [95]	Adults	Facebook	201	0		
Shaw et al. [75]	Young adults	Facebook	75	0		>

cont.
2.2
Table

Study	Donulation	SNC	Sample	Bias	Well-being	Mediators/
(nnic)	nomemdo 1	6116	Size	Rating		moderators
Simoncic et al. [76]	Young adults	Facebook	237	0		>
Steers et al. [77]	Young adults	Facebook	332	0		>
Szwedo et al. [103]	Young adults	Facebook/MySpace	138	1		
Takahashi et al. [90]	General	Depression specific SNS (unnamed)	105	1		
Tandoc et al. [78]	Young adults	Facebook	736	0		>
Tsai et al. [87]	General	Facebook	202	0		
Tsugawa et al. [96]	General	Twitter	209	0		
Wegmann et al. [121]	Young adults	General	334	1		>
Weidmann & Levinson [82]	Young adults	Facebook	77	0		>
Wright et al. [79]	Young adults	Facebook	361	0		

I

Note. SNS = social network site. Check indicates that well-being or mediators/ moderators were included. See main article for references.

	Study	Mental Health Focus	Key SNS Variables	Direct	Additional Notes and
				Associations	Complex Relationships
Frequency of SNS					
use					
	Banjanin et al. [53]	Depression	Average time on	NS	
	8	I	SNS (timeframe		
			unclear)		
	Davila et al. [54]	Depression	Daily average time on SNS	NS	
	Farahani et al. [55]	Depression	Daily average time	NS	
			SNIC UO		
	Feinstein et al. [56]	Depression	Daily average time on SNS	NS	
	Frison et al. [81]	Depression	Daily average time on SNS	+	
	Giota & Kleftaras [58]	Depression	Daily average time on SNS	SN	
	Jelenchick et al. [61]	Depression	Daily average time	NS	
			on SNS		
	Labrague [64]	Depression	Daily average time on SNS	+	
				,	
	Lin et al. [66]	Depression	Daily average time on SNS	+	
	Locatelli et al. [67]	Depression	Daily average time	NS	
			on SNS		
	Lup et al. [68]	Depression	Daily average time on SNS	+	Moderated by the proportion of strangers in a network and
					was only significant at the highest level It became NS at
					the lower levels of strangers
					followed.

Study		Mental Health Focus	Key SNS Variables	Direct Associations	Additional Notes and Complex Relationships
Pantic et al. [72]		Depression	Daily average time on SNS	+	
Rae & Lonborg	[73]	Depression	Daily average time on SNS	NS	Significant interaction- relationship is positive for those higher in motivations to use Facebook for connection purposes
Simoncic et al. [[26]	Depression	Daily average time on SNS	NS	
Steers et al. [77]	5	Depression	Daily average time on SNS	+ (cross- sectional)	Social comparison was a mediator
				NS (ESM diaries)	
Tandoc et al. [78	[8]	Depression	Daily average time on SNS	NS	
Hong et al. [60]	_	Depression	Daily average time on SNS (apps, news feed, chat)	NS	
Davila et al. [54]	4	Depression	Daily average time on SNS interacting with others	NS	
Rosen et al. [74]	F	Depression	Frequency of SNS use	NS	
Shaw et al. [75]	_	Depression	Frequency of SNS use	NS	

Study	Mental Health Focus	Key SNS Variables	Direct Associations	Additional Notes and Complex Relationships
Shaw et al. [75]	Depression	Frequency of SNS Interactive	+	
Shaw et al. [75]	Depression	Communication Use Frequency of SNS Content Production	+	
Shaw et al. [75]	Depression	Use Frequency of Descive SNS nee	NS	
Tandoc et al. [78]	Depression	Frequency of Passive SNS use	NS	Significant negative association with depression when mediated by Facebook envy
Rosen et al. [74]	Depression	Frequency of impression/profile	+	
Simoncic et al. [76]	Depression	Frequency of Active SNS use	NS	Active uses of Facebook were associated with lower depression in a three-way interaction also including gender and neuroticism (female, high neuroticism).
Davila et al. [54]	Depression	Number of times checking SNS per	NS	
Lin et al. [66]	Depression	Number of times checking SNS per week	+	

Study	Mental Health Focus	Key SNS Variables	Direct Associations	Additional Notes and Complex Relationships
Koc & Gulyagci [62]	Depression	Weekly average time on SNS	NS	
Mok et al. [70]	Depression	Weekly average time on SNS	×	No inferential statistics are presented
Morin-Major et al. [71	Depression	Weekly average time on SNS	NS	Findings were also NS in reference to the frequency of peer interactions and self- presentation behaviours
Wright et al. [79]	Depression	Weekly average time on SNS	+	
Farahani et al. [55]	Anxiety	Daily average time on SNS	+	
Labrague [64]	Anxiety	Daily average time on SNS	+	
Rae & Lonborg [73]	Anxiety	Daily average time on SNS	NS	Significant interaction- relationship is positive for those higher in motivations to use Facebook for connection purposes
Feinstein et al. [56]	Anxiety	Daily average time on SNS interacting with others	NS	
Koc & Gulyagci [62]	Anxiety	Weekly average time on SNS	NS	
Baker & Jeske [80]	Social Anxiety	Daily average time on SNS	NS	

Study	Mental Health Focus	Key SNS Variables	Direct	Additional Notes and
•			Associations	Complex Relationships
Baker & Jeske [80]	Social Anxiety	Daily average time	•	
	(Facebook-specific)	on SNS		
Green et al. [59]	Social Anxiety	Daily average time	NS	
		on SNS		
Lee-Won et al. [65]	Social Anxiety	Daily average time	NS	
		CNIC IIO		
Feinstein et al. [56]	Social Anxiety	Daily average time	NS	
		on SNS interacting		
		with others		
Fernandez et al. [57]	Social Anxiety	Frequency of SNS	NS	
		use		
Shaw et al. [75]	Social Anxiety	Frequency of SNS	+	When included in regression with other functions of SNS
				use this becomes NS
McCord et al. [69]	Social Anxiety	Frequency of Social	NS	Significant positive
		SNS use		association when moderated
				by the degree of AntArety of Escobool? only for these in
				taceous out to mose methode here the second second
				are men group.
Shaw et al. [75]	Social Anxiety	Frequency of SNS	NS	
		Interactive		
		Communication		
Shaw et al. [75]	Social Anxiety	Frequency of SNS	NS	
		Content Production		

	Additional Notes and s Complex Relationships	Remains significant when controlling for depression and anxiety symptoms and was mediated by brooding.	An alternative mediation model provided support for social anxiety as a mediator of passive Facebook use and brooding.							Depressive symptoms did not moderate these relationships	
	Direct Association	+		NS	ı	NS	NS	NS	NS	,	NS
Table 2.5 Colle.	Key SNS Variables	Frequency of Passive SNS Use		Weekly average time on SNS	Daily average time on SNS (apps, news feed, chat)	Frequency of SNS use	Weekly average time on SNS				
	Mental Health Focus	Social Anxiety		Social Anxiety	Life Satisfaction	Life Satisfaction	Psychological Well- being	Self-Esteem	Self-Esteem	Life Satisfaction Affective well-being	Self-Esteem
	Study	Shaw et al. [75]		Burke & Ruppel [105]	Frison et al. [81]	Locatelli et al. [67]	Rae & Lonborg [73]	Baker & Jeske [80]	Hong et al. [60]	Kross et al. [63]	Morin-Major et al. [71

	Study	Mental Health Focus	Key SNS Variables	Direct Associations	Additional Notes and Complex Relationships
Size and Structure of SNS					
	Banjanin et al. [53]	Depression	Number of Facebook Friends	NS	
	Fernandez et al. [57]	Depression	Number of Facebook Friends	NS	
	Labrague [64]	Depression	Number of Facebook Friends	NS	
	Locatelli et al. [67]	Depression	Number of Facebook Friends	NS	
	Morin-Major et al. [71	Depression	Number of Facebook Friends	NS	
	Park et al. [84]	Depression	Number of Facebook Friends		Group Differences: depressed participants had significantly fewer Facebook friends than non-depressed participants
	Park et al. [83]	Depression (BDI)	Number of Facebook Friends	ı	NS when depression was measured by the CES-D
	Rae & Lonborg [73]	Depression	Number of Facebook Friends		
	Rosen et al. [74]	Depression	Number of Facebook Friends (index)	·	Technology attitudes and anxiety were held constant.
	Tandoc et al. [78]	Depression	Number of Facebook Friends	NS	
	Wright et al. [79]	Depression	Number of Facebook Friends	NS	

S	study	Mental Health Focus	Key SNS Variables	Direct Associations	Additional Notes and Complex Relationships
Homan e	et al. [89]	Depression	Network Structure Characteristics	Differ in high and low depression groups	High depression scores have significantly less integrated online social networks
Takahash	hi et al. [90]	Depression	Network Structure Characteristics		Described network characteristics and connections across different levels of depression severity. Qualitative analysis is also included
Mota-Per	reira [88]	Depression	Facebook use with psychiatrist as a "friend"		"Facebook Use" group and "Facebook Use with a psychiatrist as a friend" group, depressive symptoms decreased significantly over a 3-month period compared to a control group.
Tsai et al	l. [87]	Depression	Accepting friend requests from former partners	+	
Tsai et al	l. [87]	Anxiety	Accepting friend requests from former partners	+	
Labrague	e [64]	Anxiety	Number of Facebook Friends	NS	
Rae & Lo	onborg [73]	Anxiety	Number of Facebook Friends	NS	

	Study	Mental Health Focus	Key SNS Variables	Direct Associations	Additional Notes and Complex Relationships
	Javidson &	Anxiety (Facebook-	Number of unique	NS	
ц	arquhar [86]	specific)	groups on Facebook		
D	Davidson &	Social Anxiety	Number of unique	NS	
Ĥ	arquhar [86]		groups on Facebook		
Ŧ	ernandez et al. [57]	Social Anxiety	Number of Facebook	•	
			Friends		
> .	Veidmann &	Social Anxiety	Number of Facebook	·	
, ב	evinson [82]		Friends		
Г	ocatelli et al. [67]	Life Satisfaction	Number of Facebook Friends	NS	
R	kae & Lonborg [73]	Life Satisfaction	Number of Facebook	NS	Significant interaction –
			Friends		relationship is positive for those high in friendship
					motivations for using
					Facebook
R	kae & Lonborg [73]	Positive Affect	Number of Facebook	+	
20	Aorin-Major et al. 71	Self-Esteem	Number of Facebook Friends	NS	
2					
Language Features and Observable SNS Activity					
	De Choudhury et al.	Depression	Twitter Use Data	Predictive	
<u></u>	(Te			language and SNS use	
				features identified	

Chudu	Mantal Haalth Eacus	Table 2.3 cont. Korr SNS Variables	Direct	Additional Notes and
Study	Mental Health Focus	Ney DIND Variables	Associations	Additional Notes and Complex Relationships
Tsugawa et al. [96]	Depression	Twitter Use Data	Predictive language and SNS use	
			features identified	
De Choudhury et al. [92]	Postpartum Depression	Facebook Use Data	Predictive language and SNS use	
			features identified	
Dumitrache et al. [97]	Depression	Identity items on profile	+	
Fernandez et al. [57]	Depression	Identity items on profile	+	Depression was associated with a greater amount of profile information posted
Moreno et al. [85]	Depression	Status update content	Predictive language	Depression symptoms can be observed and coded based on the DSM-IV criteria for MDD
Moreno et al. [94]	Depression	Status update content	+	Depression displays in status updates (based on the DSM- IV criteria for MDD) were associated with PHQ-9 scores

Study	Mental Health Focus	Key SNS Variables	Direct Associations	Additional Notes and Complex Relationships
Park et al. [84]	Depression	Inbound comments	NS	Group differences- fewer inbound comments to depressed participants compared to non-depressed participants
Park et al. [84]	Depression	Inbound likes	NS	Group differences- fewer inbound likes to depressed participants compared to non- depressed participants
Park et al. [84]	Depression	Outbound comments		Group differences- fewer outbound comments from depressed participants compared to non-depressed participants
Park et al. [84]	Depression	Wall post rate	+	
Park et al. [93]	Depression	Positive disclosure in status updates	NS	Group differences- fewer between those with MDD and those without. MDD participants disclose more negative and less positive content than non-depressed participants.
Park et al. [93]	Depression	Negative disclosure in status updates	NS	

Stu	dy	Mental Health Focus	Key SNS Variables	Direct	Additional Notes and
				Associations	Complex Relationships
Settani & I	Marengo	Depression	Positive emotion in	NS	1
[95]			status updates		
Settani & D	Marengo	Depression	Negative emotion in	+	
[95]			status updates		
Settani & I	Marengo	Anxiety	Positive emotion in	NS	
[95]			status updates		
Settani & D	Marengo	Anxiety	Negative emotion in	+	
[95]			status updates		
Fernandez	et al. [57]	Social Anxiety	Identity items on	+	Social interaction anxiety wa
			profile		associated with a greater
					amount of profile information
					posted
Fernandez	et al. [57]	Social Anxiety	Number of posts by	NS	
Fernandez	et al. [57]	Social Anxiety	Number of status	NS	
	-		updates		
große Dete [98]	ers et al.	Social Anxiety	Number of status updates	SN	
große Dete	ers et al.	Social Anxiety	Number of likes on	NS	
[98]			status updates		

cont.
2.3
Table

			1 anic 2.5 colle.		
	Study	Mental Health Focus	Key SNS Variables	Direct Associations	Additional Notes and Complex Relationships
	Weidmann & Levinson [82]	Social Anxiety	Facebook Profile Inactivity	NS	Group Differences- participants listing a single relationship status compared to a relationship with other people and an absence of status updates compared to profiles with status updates had significantly higher scores of social anxiety.
	Weidmann & Levinson [82]	Social Anxiety	Observer ratings of social anxiety from profile content	+	
SNS for Self- disclosure and expression					
	Baker & Moore [100]	Depression	Intention to blog	+	Group differences- intending bloggers had significantly higher depression scores than non-bloggers
	Baker & Moore [101]	Depression	Blogging on MySpace	NS	Bloggers (and non-bloggers) had no change in depression scores between T0 and T1 (2- month interval)
	große Deters & Mehl [102]	Depression	Increased status posting	NS	

		lable 2.3 cont.		
Study	Mental Health Focus	Key SNS Variables	Direct Associations	Additional Notes and Complex Relationships
Locatelli et al. [67]	Depression	Frequency of negative status updates	+	Rumination indirectly mediated the association between negative posts and depression
Locatelli et al. [67]	Depression	Frequency of positive status updates	NS	Rumination indirectly mediated the association between positive posts and depression (significant negative association)
Baker & Moore [100]	Anxiety	Intention to blog	+	Group differences- intending bloggers had significantly higher anxiety scores than non-bloggers
Baker & Moore [101]	Anxiety	Blogging on MySpace	NS	Bloggers (and non-bloggers) had no change in anxiety scores between T0 and T1 (2- month interval)
Davidson & Farquhar [86] Baker & Jeske [80]	Anxiety (Facebook- specific) Social Anxiety	Self-presentation (role conflict) Assertiveness on SNS	+ '	Group differences – Higher social anxiety has lower assertiveness scores than low social anxiety

Additional Notes and Complex Relationships	Both males and females have a positive association between social anxiety and the feelings of assertiveness in SNS communication compared to face-to-face settings.		Higher social anxiety was associated with greater concern of presenting a negative self-image on Facebook	Both males and females have a positive association between social anxiety and the feeling they have more control over SNS self- presentation compared to face-to-face settings.	
Direct Associations	+	+	+	+	+
Key SNS Variables	Assertiveness on SNS	Self-presentation	Self-presentation	Self-presentation	Self-presentation (role conflict)
Mental Health Focus	Social Anxiety	Social Anxiety	Social Anxiety	Social Anxiety	Social Anxiety
Study	Casale & Fioravanti [104]	Bodroža & Iovanović [106]	Burke & Ruppel [105]	Casale & Fioravanti [104]	Davidson & Farquhar [86]

Study	Mental Health Focus	Key SNS Variables	Direct	Additional Notes and
			Associations	Complex Relationships
große Deters et al.	Social Anxiety	Social Reciprocity	+ (Study 1)	Positive posts elicit a greater
[98]		and Valence of		number of 'likes', particularly
		Status Updates	NS (Study 2)	for those high in social
				anxiety. This was not
				supported in Study 2
Ghosh & Dasgupta	Social Anxiety	SNS membership	,	Group differences –
[66]				Facebook users had
				significantly lower social
				anxiety scores than non-users.
				This difference was more
				pronounced for female non-
				users compared to males.
Green et al. [59]	Social Anxiety	SNS self-disclosure	NS	
		(public)		
Green et al. [59]	Social Anxiety	SNS self-disclosure	NS	Significant mediated pathway
		(private)		from social anxiety to private
				Facebook self-disclosure via
				the characteristics of online
				communcauon and disinhihition
Locatelli et al. [67]	Life Satisfaction	Frequency of		Rumination indirectly
		negative status		mediated the association
		updates		between negative posts and life satisfaction

cont.	
2.3	
Table	

	Study	Mental Health Focus	Key SNS Variables	Direct Associations	Additional Notes and Complex Relationships
	Locatelli et al. [67]	Life Satisfaction	Frequency of positive status updates	NS	Rumination indirectly mediated the association between positive posts and
	große Deters & Mehl	Loneliness	Increased status	+	life satisfaction (significant positive association) Loneliness significantly
	[102]		posting		decreased in the experimental group (increased posting) from T1 to T2 (1-week
					interval) and did not change in the control condition. The decrease in loneliness was mediated by feelings of social
					connectedness
	Ghosh & Dasgupta [99]	Self-Esteem	Facebook membership	+	Group differences – Facebook users had significantly higher self- esteem scores than non-users
	große Deters & Mehl [102]	Subjective Happiness	Increased status posting	NS	
Quality of Interactions	Davila et al. [54]	Depression	Perceived frequency	+	Greater perceived frequency
	n 4	a	of negative interactions		of negative interactions (T1) predicted increases in depressive symptoms (T2)

đ			ļ	
Study	Mental Health Focus	Key SNS Variables	Direct Associations	Additional Notes and Complex Relationships
Davila et al. [54]	Depression	Perceived frequency of positive interactions	NS (Study 1) - (Study 2)	Lesser perceived frequency of positive interactions (T1) predicted increases in depressive symptoms (T2)
Feinstein et al. [56]	Depression	Perceived frequency of negative interactions	+	Depressive symptoms at T1 predicted more negative interactions at T2 with close friends and romantic partners
Feinstein et al. [56]	Depression	Perceived frequency of positive interactions		Depressive symptoms at T1 predicted less positive interactions at T2 with romantic partners
Frison et al. [81]	Depression	Negative peer experiences on SNSs	+	Cross-lagged analyses suggested a unidirectional relationship between depression (T1) and increases in negative Facebook experiences (T2; 6-month interval)
Landoll et al. [107]	Depression	Negative peer experiences on SNSs	+	

Study	Mental Health Focus	Key SNS Variables	Direct Associations	Additional Notes and Complex Relationships
Moberg & Anestis [108]	Depression	SNS interactions (very negative – very positive)	1	When controlling for depression, interaction ratings and thwarted belongingness had a significant negative relationship
Szwedo et al. [103]	Depression	Positive peer relationshin quality	NS	
Szwedo et al. [103]	Depression	Negative peer relationship quality	• +	Symptoms at age 13 (T1) predicted less deviancy talk from peers at age 20 (T2)
				Symptoms at age 20 (T2) predicted more verbally aggressive comments from peers at T2
Feinstein et al. [56]	Anxiety	Perceived frequency of negative interactions	+	Global Anxiety symptoms at T1 did not predict interactions at T2, though cross-sectional associations were observed.
Feinstein et al. [56]	Anxiety	Perceived frequency of positive interactions		

Study	Mental Health Focus	Key SNS Variables	Direct Associations	Additional Notes and Complex Relationships
Feinstein et al. [56]	Social Anxiety	Perceived frequency of negative interactions	+	Social Anxiety symptoms at T1 did not predict interactions at T2, though cross-sectional associations were observed
Feinstein et al. [56]	Social Anxiety	Perceived frequency of positive interactions		Relationship is NS when only observing interactions with close friends
Hong et al. [109]	Social Anxiety (online)	Facebook continuance intention		Belief that virtual communities are dangerous (including the potential to encounter hostile/negative interactions) is positively associated with online and general social anxiety subsequently links to reduced continuance intention for using Facebook
Landoll et al. [107]	Social Anxiety	Negative peer experiences on SNSs	+	
Szwedo et al. [103]	Social Anxiety	Positive Peer Relationship Quality	+	Symptoms at age 20 (T2) predicted more supportive comments from peers at T2

Social Support	Study zwedo et al. [103] rison et al. [81] rison & Eggermont rison & Eggermont rison & Eggermont	Mental Health Focus Social Anxiety Life Satisfaction Depression Depression	Key SNS Variables Negative Peer Relationship Quality Negative peer experiences on SNSs experiences on SNSs Perceived Social Support (via SNS) Social Support Seeking (via SNS)	Direct Associations + +	Additional Notes and Complex Relationships Symptoms at age 13 (T1) predicted fewer verbally aggressive comments from peers at age 20 (T2) Cross-lagged analyses suggested a bidirectional relationship between life satisfaction (T1/T2) and negative Facebook experiences (T1/T2) social support seeking through Facebook and its
<u>'</u>					association with depressed mood was mediated by perceived social support (greater perceived social support decreased depressed mood).

Additional Notes and s Complex Relationships	Moderated by perceived friend support (low, medium, high), the association between negative Facebook experiences (T1) and depressed mood (T2) was significant only for low and medium levels.						Significant positive relationship where the individual was disclosing	negative sentiment and they were higher in depressive symptoms.
Direct Association		NS	+	+	NS	'	NS	
Key SNS Variables	Perceived Friend Support (via SNS)- moderator (Negative Facebook experiences)	Perceived social support (via SNS)	Emotional support (via SNS)	Negative social support (via SNS)	Instrumental support (via SNS)	Perceived Social Support (via SNS)	Actual Social Support (via SNS)	
Mental Health Focus	Depression	Depression	Depression	Depression	Depression	Depression	Depression	
Study	Frison et al. [81]	McCloskey et al. [112]	McCloskey et al. [112]	McCloskey et al. [112]	McCloskey et al. [112]	Park et al. [93]	Park et al. [93]	

Study	Mental Health Focus	Key SNS Variables	Direct Associations	Additional Notes and Complex Relationships
Park et al. [93]	Depression	Discrepancy between actual and perceived support	+	Group differences -discrepancy is larger for individuals with MDD compared to non-depressed participants.
Wright et al. [79]	Depression	SNS Social support satisfaction	1	Path from higher CMC competence predicts more Facebook Social Support Satisfaction and subsequently lower depression scores. Motives to use Facebook for social integration and interpersonal communication also contributed to this model
Indian & Grieve [111]	Social Anxiety (high)	Perceived social support (via SNS)	+ (subjective well-being)	Controlled for offline social support
				Group comparisons also revealed no significant differences in perceived Facebook social support between the low and high social anxiety groups
Indian & Grieve [111]	Social Anxiety (low)	Perceived social support (via SNS)	NS (subjective well-being)	

	Study	Mental Health Focus	Key SNS Variables	Direct Associations	Additional Notes and Complex Relationships
	Frison et al. [81]	Life Satisfaction	Perceived Friend Support (via SNS) - moderator <i>(Negative Facebook experiences)</i>	+	Moderated by perceived friend support (low, medium, high), the association between negative Facebook experiences (T1) and life satisfaction (T2) was significant only for low and medium levels.
	McCloskey et al. [112] McCloskey et al. [112]	Quality of Life Quality of Life	Perceived social support (via SNS) Emotional support (via SNS)	NS -	Direct association only with psychological well-being of the WHOQOL-BREF domains
	McCloskey et al. [112] McCloskey et al. [112]	Quality of Life Quality of Life	Negative social support (via SNS) Instrumental support (via SNS)	- SN	
Social Connectedness	Grieve et al. [113]	Depression	Facebook social connectedness		
	Grieve et al. [113] Grieve et al. [113]	Anxiety Life Satisfaction	Facebook social connectedness Facebook social connectedness	ı +	

	Study	Mental Health Focus	Key SNS Variables	Direct Associations	Additional Notes and Complex Relationships
Social Comparison A	ppel et al. [116]	Depression	Envy	+	Significant interaction – envy was greater in response to viewing an attractive Facebook profile, with it being more so in a depressed group compared to a non- depressed group
Т	andoc et al. [78]	Depression	Envy	+	Facebook envy mediates the association between surveillance uses of Facebook and depression
Г	up et al. [68]	Depression	Downward Social Comparison on SNSs		Social comparison was an indirect mediator of the relationship between the frequency of Instagram use and depression. This pathway was additionally moderated by the proportion of strangers followed in the network.
S	teers et al. [77]	Depression	Downward Social Comparison on SNSs	+	Social comparison mediated the relationship between time spent on Facebook and depressive symptoms

Stud	ly	Mental Health Focus	Key SNS Variables	Direct	Additional Notes and
				Associations	Complex Relationships
Steers et al.	[27]	Depression	Non-directional Social Comparison on SNSs	+	Social comparison mediated the relationship between time spent on Facebook and depressive symptoms
Steers et al.	[77]	Depression	Upward Social Comparison on SNSs	+	Social comparison mediated the relationship between time spent on Facebook and depressive symptoms
Feinstein et	al. [115]	Depression	Social Comparison on SNSs	+	More negative social comparison on Facebook at T1 was associated with increases in depressive symptoms at T2 via increases in rumination (3-week interval).
Lee [114]		Depression	Social Comparison on SNSs	+	
Lee [114]		Anxiety	Social Comparison Frequency on SNSs	+	
Appel et al.	[116]	Self-Esteem	Envy	,	
Lee [114]		Self-Esteem	Social Comparison on SNSs		
			l adle 2.3 cont.		
---------------------------------	---	--------------------------	--	------------------------	--
	Study	Mental Health Focus	Key SNS Variables	Direct Associations	Additional Notes and Complex Relationships
Addictive or Problematic Use	Andreassen et al. [117]	Depression	Addictive SNS use		Controlled for demographic characteristics, OCD, and ADHD
	Hanprathet et al. [119]	Depression	Addictive SNS use	+	
	Hong et al. [60] Koc & Gulyagci [62]	Depression Depression	Addictive SNS use Addictive SNS use	+ +	
	Wegmann et al. [121]	Depression	Addictive SNS Use	+	Internet use expectancies and self-regulation mediate the relationship between SNS addiction and depression/social anxiety
	Blachnio et al. [118]	Depression	Facebook Intrusion (behavioural	+	
	Giota & Kleftaras [58]	Depression	Problematic SNS use	+	NS when accounting for personality and average daily SNS use

Study	Mental Health Focus	Key SNS Variables	Direct	Additional Notes and
			Associations	Complex Relationships
Moreau et al. [120	0] Depression	Problematic SNS use	+	Cluster analysis indicated that
				the 'borderline group' which
				had significantly higher
				social anxiety and depression
				scores than two other
				clusters, also had a higher
				Level of problematic
Andwarena at al	American	Addicting CNC new	4	Facebook use Controlled for domographic
Alluredssell et al. [117]	AUMERY	Addictive Stronger	F	Controlled for demographic characteristics, OCD, and
				ADHD
Hanprathet et al. [119]	Anxiety	Addictive SNS use	+	
Bodroža &	Social Anxiety	Addictive SNS use	+	
Jovanović [106]				
Wegmann et al.	Social Anxiety	Addictive SNS Use	+	
[171] C1- 8 E		Duckless CNIC	-	
Casale & Floravar [10.4]	inti Social Anxiety	Problematic SNS use	+	Males – The relationship bottoon social anvioty and
[+ot]				between social anticity and problematic SNS use is
				indirectly mediated by the
				need for self-presentation

Table 2.3 cont.

			Table 2.3 cont.		
	Study	Mental Health Focus	Key SNS Variables	Direct	Additional Notes and
				Associations	Complex Relationships
	Lee-Won et al. [65]	Social Anxiety	Problematic SNS use	+	NSA moderated this relationship with it being more pronounced for those with high and medium NSA, and becoming NS for those low in NSA
	Moreau et al. [120] Hong et al. [60]	Social Anxiety Self Esteem	Problematic SNS use Facebook Addiction	+ S	
Physiological Arousal and Facebook	Raunch et al. [122]	Social Anxiety	Facebook exposure	+	High social anxiety had greater physiological arousal in a face-to-face encounter after prior exposure to a Facebook profile.
Note: "+" = significan SNS = Social Network Reassurance; CES-D = WHO Quality of Life- 2	t positive relationship r ing Site; CMC = Comp c Center for Epidemiolo BREF; BDI = Beck De	<pre>sported; "-" = significant 1 uter Mediated Communic gic Studies Depression So pression Inventory; ESM</pre>	negative relationship rep :ation; MDD = Major D cale; PHQ-9 = Patient H = Experience Sampling	orted; "NS" = r epressive Disorv ealth Questionn Method; T0 = b	ion-significant findings reported; der; NSA = Need for Social aire – 9; WHOQOL-BREF = aseline; T1 = Time 1; T2 = Time

Frequency of SNS use refers to studies using a Likert type question with no defined time measurement (e.g. never - very frequently).

2.3 Concluding Remarks

This chapter presented a systematic review of the research exploring depression and anxiety in the context of social networking site use. It identified both risk and protective factors for mental health. Beneficial aspects of social media use for mental health included reducing loneliness, providing an avenue for social support, and positive social interactions. Detrimental aspects of social media for depression and anxiety use included social comparisons, problematic and addictive social media use, and negative social interactions. Further, cognitive and individual characteristics were identified as potential moderators and mediators to the relationship between social media use and depression or anxiety.

Another systematic review was published concurrently with the work presented here by Baker and Algorta (2016). This review also addressed social networking site use and depression, but did not include articles addressing anxiety. The authors reviewed 30 papers and identified similar themes to those identified in this chapter. They also suggested a complex relationship between social media use and depression that is likely moderated and/or mediated by a number of psychological and individual factors like rumination and social comparison. Consistent with the systematic review presented in this chapter, among the limitations in the literature Baker and Algorta (2016) identified sources of bias including the reliance of self-report of social media behaviours, population bias (young, university student aged), and platform bias (predominantly Faccebook research) which may have impacted on the reliability, validity, and generalisability of findings in the literature.

Several other reviews have been published addressing social media use and mental health outcomes since the publication of the paper presented in this chapter, broadly reviewing the same literature and with findings consistent with those presented here. Frost and Rickwood (2017) reviewed the literature of mental health and Facebook use, examining findings related to depression, anxiety, body image and disordered eating, drinking and alcohol use, other mental health problems, and Facebook addiction. In terms of depression they similarly identified lower

depressive symptoms being related to perceived social support, and higher depressive symptoms associated with the type of Facebook use (active, or passive), negative content in status updates, and not perceiving social support from online sources. Mixed findings were again found for the time spent on Facebook and depression. To clarify these mixed findings, Huang (2017) conducted a meta-analysis of the research examining the frequency of social networking sites use and its association with well-being (which included depression). Across the studies examining depression (and other negative well-being indices) the correlation with the time spent on social media was weak.

In a practical sense, others have explored the way natural language processing from social media data can be applied to detect mental illness. Guntuku et al. (2017) presented an integrative review of the approaches seeking to detect depression from observable behaviours (including language use) on social media. The features used in these approaches broadly consider demographic variables, lexical features, behavioural features (e.g., time of posting), and social features (e.g., size of a network) (Calvo, Milne, Hussain, & Christensen, 2017; Guntuku et al., 2017). Overall, while approaches to creating prediction models varied, the automated analysis of social media content was revealed to be feasible and the prediction performance of most approaches was considered to outperform unaided clinician assessment (Guntuku et al., 2017).

A limitation of the systematic review presented in this chapter was that it did not include a synthesis of findings using meta-analysis. While it provided a detailed description of the area by narratively synthesising the reviewed papers by theme, it was unable to make clear conclusions about the size or direction of effects. This limitation is most relevant to the discussion of mixed findings in relation to the frequency of SNS use. As above, this has recently been investigated by Huang (2017).

In the next chapter, the general methodology of this thesis is presented, providing rationale for the data collection methods used and additional context for the empirical papers presented in Chapters 4 and 5.

Chapter 3

General Methodology

3.0 Introduction

This chapter provides an overview of the rationale for the research design of the experimental papers (Chapters 4 and 5). It offers an extended methodology (that is, beyond the brief methodologies included in each publication) outlining the development of the data collection method used, an experience sampling smartphone app - *MoodPrism*. This included the development and testing of a word count script capable of reliably collecting language and emoji data for positive and negative emotion categories from Facebook and Twitter. Privacy and ethical considerations are discussed, and the participant selection processes are outlined.

3.1 The Need to Integrate Social Media Data Collection with Other Experience Sampling Methods

The systematic review presented in Chapter 2 highlighted several key themes in social media and mental health research. Across the 70 studies reviewed however, few directly sampled social media data as a part of their study design, while the remainder utilised self-report as a means of estimating the online activity of their participants. The extensive use of self-report methods for accessing social media behaviour may account for some of the mixed findings highlighted in the review, particularly in relation to the frequency of time spent online, and the size of friend networks with depression and anxiety. Recent work has indicated that people are inaccurate when estimating their time spent on sites like Facebook, often overestimating their time spent online by more than 4 times (Araujo, Wonneberger, Neijens, & Vreese, 2017; Junco, 2013). Estimates may also be influenced by cognitive biases, particularly in relation to perceptions of the quality of interactions occurring on social media (e.g., Davila et al., 2012; Szwedo, Mikami, & Allen, 2011). This was highlighted in the study conducted by Park et al. (2016), where individuals with depression self-reported little social support online, but objective examination of social media communications revealed that they had received significantly more social support than they perceived. While self-

report does provide important information about subjective experiences of social media use, direct access to social media data would add time-sensitive and more objective assessment of social media behaviour.

There are several advantages to utilising social media data as a behavioural marker of mental health. With close to 2 billion Facebook accounts (Nowak, 2017), the ability to observe social interactions online and in real-time provides researchers with access to previously hidden behaviours on a greater scale than by any other method (Kosinski, Matz, Gosling, Popov, & Stillwell, 2015). The use of social media data provides insight into participants' lives as an unobtrusive and "ongoing experience sampling method" (Park et al., 2015, p. 935) that has qualitative and quantitative properties. Indeed, natural language use on social media has been shown to provide accurate language markers of mental health and well-being that are also sensitive to temporal patterns (Guntuku et al., 2017; Larsen et al., 2015; Hansen Andrew Schwartz et al., 2013).

However, the use of social media data also has inherent biases and limitations. Despite its reach, social media samples are not representative of the general population (Kosinski et al., 2015; Tufekci, 2014). On average, social media users are younger and more technology literate (Blank, 2017; Kosinski et al., 2015). There are also demographic, social, and, geopolitical influences that impact on the type of social media platform used and how individuals engage with it (Correa, 2016; Haight, Quan-Haase, & Corbett, 2014). An example of the geopolitical variation in Facebook use is provided in Figure 1 where the visualisation of (a) Facebook's global social network created by Paul Butler is contrasted with the (b) "UnFacebook World" by Ian Wojtowicz showing the areas of the world that were, and were not connected by friendship ties on Facebook in 2010. The second panel was created by subtracting Facebook's global social network from NASA's *Earth at Night*, thus showing areas of the world connected to electricity (in yellow), but not represented in "Visualising Friendship".

(a) Visualizing Friendship; *Image:* Paul Butler (2010)



(b) UnFacebook World; Image: Ian Wojtowicz (2011)



Figure 1. Visualisation of the geopolitical discrepancies between the (a) Facebook and (b) UnFacebook world. These connections have implications for the representativeness and generalisability of social media sample to the population. Image sources:

(a) https://www.facebook.com/notes/facebook-engineering/visualizing-friendships/469716398919/

 $(b)\ https://ianwojtowicz.com/UnFacebook\% 20 World/False-Color-Facebook-NASA-Mashup.png$

In addition to the factors influencing social media usage generally, not all social media platforms attract the same users or perform the same functions (Phua, Jin, & Kim, 2017; Tufekci, 2014). Phua et al. (2017) discusses this with reference to the Uses and Gratifications Theory (UGT) where individual's selection of social media – as consumers – is guided by how that social media platform can cater to specific needs and wants. They suggest "people may use Facebook to stay in touch with friends, Twitter to follow news and trending topics, Snapchat to instantly share short videos with selected individuals, and Instagram to easily filter and upload visual images" (Phua et al., 2017, p. 414). They found that Snapchat users (compared to Instagram, Twitter, and Facebook) used the platform more for sharing problems, entertainment and passing time. This highlights how the mechanisms of social interaction on different social media platforms may influence the type of content shared between users. This may have implications for prediction of mental health, with data from different platforms potentially providing insight into different underlying behaviours.

Because social media platforms attract different demographic groups, serve different purposes, and have different social mechanisms guiding engagement and interaction (Greenwood, Perrin, & Duggan, 2016; Tufekci, 2014), there is a need to integrate more detailed information about the composition of social media samples. This will better describe who is using social media and account for complexities in the human-technology social environment. It will also provide information about the individual that is not available via social media platforms. For example, depression or anxiety status may not be directly accessible via a status updates. Despite this there are several studies who have used social media data alone to make inferences about mental health (such studies were excluded from analysis in Chapter 2). Without an external criterion of mental health, the validity of conclusions made in such approaches is poor and further highlights the need to incorporate other variables that are not derived from social media data alone (Tufekci, 2014).

Taken together, the methodological considerations discussed above suggest the need for an integrated experience sampling method (ESM) that accesses time-sensitive social media data and supplements this with access to traditional measures that have sound psychometric properties. Such

an integrated method is congruent with several practical steps suggested by Tufekci (2014) to increase the strength of research examining social media data. These include the need to (1) sample data from sources external to the social media data set to serve as reliable and valid dependent variables, (2) examine behavioural variation by including qualitative pull-outs, (3) conducting multi-platform analyses and using complementary methods.

3.2 The Development of *MoodPrism*

To address the need for an integrated ESM tool which collects both subjective and objective data relevant to depression from social media and from psychological measures, a new smartphone app, *MoodPrism*, was developed. The requirements of *MoodPrism* for this thesis was that the tool be capable of collecting (1) a valid and reliable measure of depression (external criterion), (2) subjective mood on a daily basis, and (3) social media data from at least two platforms on a daily basis.

MoodPrism was developed for iPhone and Android as a part of a larger project in collaboration with digital product developers *TwoBulls* (Melbourne, VIC). *MoodPrism* delivered an engaging experience sampling method to its users, collected and monitored changes in self-reported emotional well-being over time, collected social media data, and provided its users with colourful and intuitive mood, mental health, and well-being feedback. A detailed account of *MoodPrism*'s development, testing, and procedure is provided in Appendix A "*Development of a mobile phone app to support self-monitoring of emotional well-being: A mental health digital innovation*" (Rickard et al., 2016).

3.2.1 Participants

There were 2,081 downloads of the *MoodPrism* app between April 2016 and May 2017. Participants were recruited through a variety of means which included: targeted Web and Facebook advertising, community engagement and promotion, as well as naturally occurring downloads by individuals seeking a mood tracking app on the Australian Apple and Google Play stores. To be

included in the broader *MoodPrism* project participants had to be aged 13 years or older, owned their own smartphone, and were not currently taking psychotropic medication.

Inclusion in the experimental papers (Chapter 4 and 5) of this thesis also required participants:

- to complete the Patient Health Questionnaire- 9 (PHQ-9) from the baseline surveys on MoodPrism;
- opt-in to provide either Facebook or Twitter data and;
- to have posted at least 10 status updates over more than 7 days on either Facebook or Twitter within the 12 months prior to completing the PHQ-9;

For the study presented in Chapter 5 the additional inclusion criteria required were:

- completion of at least 70% (21 days) of daily mood reports;
- to have posted at least 10 status updates during the 30 days of *MoodPrism* use.

Figure 2 shows the flow of participant selection for Chapters 4 and 5. It shows from the total downloads 72.4% completed the consent procedures, and from the total sample (N = 1, 518) the opt-in rate for contributing social media data was 14.7%.



Figure 2. Participant selection flow chart. ^a = To create independent groups for Chapter 4, n = 3 who had dual Facebook and Twitter records were allocated to the Facebook group only due to a greater number of language samples. They were then reincluded when assessing suitability for Chapter 5.

Data were available for participants who had opted-in and opted-out to contributing social media data to *MoodPrism*. Table 1 shows the mean scores, standard deviations and frequency counts for individuals in the opt-in and opt-out groups. Sample sizes on each variable differed and these are also indicated in the table. Interestingly, there was a divergence between the self-reported use of social media and the actual use of social media in the opt-in sample. Sixteen participants (1.1%) indicated that they did not use social media in their self-report but also provided social media language samples to the study.

Table 1

Frequency Counts, Means, and Standard Deviations of Participants Who Opted-In and Opted-Out of Contributing Social Media Data to MoodPrism.

Variable	Ор	t-in	Opt	t-out	Total s	ample
	n	%	n	%	n	%
Sample	223	14.7	1,295	85.3	1,518	100
Gender (total)	197	13.0	1,164	76.7	1,361	89.7
Male	67	4.4	317	20.9	384	25.3
Female	128	8.4	841	55.4	969	63.8
Other	2	0.1	3	0.2	5	0.3
Don't want to	0	0	3	0.2	3	0.2
answer						
Age	154	10.1	918	60.5	1072	70.6
M(SD)	33.30	(12.26)	31.86	(13.61)	32.07 (13.43)
Highest level of	197	13.0	1161	76.5	1,358	89.5
education (total)						
Primary	5	0.3	12	0.8	17	1.1
Secondary	42	2.8	357	23.5	399	26.3
Tertiary	91	6.0	473	31.2	564	37.2
Post-graduate	58	3.8	302	19.9	360	23.7
Don't want to	1	0.1	17	1.1	18	1.2
answer						

Table 1 Continued.

Frequency Counts, Means, and Standard Deviations of Participants Who Opted-In and Opted-Out of Contributing Social Media Data to MoodPrism.

Variable	Ор	t-in	Op	t-out	Total s	ample
	n	%	n	%	n	%
Current study status	196	12.9	1160	76.4	1,356	89.3
Not at all	102	6.7	547	36.0	649	42.8
Part-time	30	2.0	136	9.0	166	10.9
Full-time	64	4.2	477	31.4	541	35.6
Current work status	196	12.9	1157	76.2	1,353	89.1
Not at all, and not	20	1.3	163	10.7	183	12.1
seeking work						
Not at all, but	18	1.2	123	8.1	141	9.3
actively seeking						
work						
Not working due	5	0.3	20	1.3	25	1.6
to sick leave						
Part-time	79	5.2	445	29.3	524	34.5
Full-time	74	4.9	406	26.7	480	13.6
Social Media Use						
Do you use social	159	10.5	975	64.2	1134	74.7
media?						
Yes	143	9.4	853	56.2	996	65.6
No	16	1.1	122	8.0	138	9.1
Mental Health						
PHQ-9	196	12.9	1151	75.8	1347	88.7
M (SD)	10.49	(6.64)	10.52	(6.64)	10.52 ((6.63)
GAD-7	196	12.9	1151	78.5	1347	88.7
M(SD)	8.13	(5.42)	8.37	(5.55)	8.33 (5.53)
WEMWBS	196	12.9	1151	78.5	1347	88.7
M (SD)	41.31	(10.77)	41.13	(9.35)	41.15 ((9.56)

Table 1 Continued.

Frequency Counts, Means, and Standard Deviations of Participants Who Opted-In and Opted-Out

of Contributing Social Media Data to MoodPrism.

Variable	Op	ot-in	Opt	t-out	Total s	ample
	n	%	п	%	п	%
Personality (mini-IPIP)						
Extraversion	161	10.6	1006	66.3	1167	76.9
<i>M</i> (SD)	9.97	(4.24)	9.70	(4.26)	9.74 (4.26)
Agreeableness	165	10.9	1038	68.4	1203	79.2
<i>M</i> (SD)	15.18	(3.95)	15.07	(3.96)	15.08	(3.96)
Conscientiousness	166	10.9	1038	68.4	1204	79.3
<i>M</i> (SD)	12.43	(3.95)	12.30	(4.03)	12.32	(4.01)
Neuroticism	166	10.9	1030	67.9	1196	78.8
<i>M</i> (SD)	13.14	(4.25)	12.65	(4.04)	12.72	(4.07)
Openness to Experience	166	10.9	1038	68.4	1204	79.3
<i>M</i> (SD)	12.99	(3.10)	12.38	(3.33)	12.46	(3.30)
MSPSS	165	10.9	1025	67.5	1190	78.4
Significant Other,	5.09	(1.57)	5.28	(1.50)	5.25 (1.51)
<i>M</i> (SD)						
Family, <i>M</i> (SD)	4.72	(1.49)	4.74	(1.54)	4.73 (1.54)
Friends, M (SD)	4.85	(1.36)	4.91	(1.43)	4.90 (1.42)
Total, M (SD)	4.89	(1.21)	4.97	(1.26)	4.96 (1.25)
RSES	160	10.5	1001	65.9	1161	76.5
<i>M</i> (SD)	15.24	(2.25)	15.14	(2.05)	15.16	(2.08)
SDS	189	12.5	1081	71.2	1270	83.7
<i>M</i> (SD)	6.34	(2.78)	6.37	(2.85)	6.36 (2.84)

Note: PHQ-9 = Patient Health Questionnaire -9; GAD-7 = General Anxiety Disorder -7; WEMWBS = Warwick Edinburgh Mental Well-Being Scale; mini-IPIP = mini International Personality Item Pool; MSPSS = Multidimensional Scale of Perceived Social Support; RSES = Rosenberg Self-Esteem Scale; SDS: Marlowe-Crowne Social Desirability Scale Form-C. As all distributions were non-normal, Mann-Whitney U tests were used to compare mean rank differences between the opt-in and opt-out groups in age, and on the PHQ-9, General Anxiety Disorder-7 (GAD-7), Warwick Edinburgh Mental Well-Being Scale (WEMWBS), factors on the mini- International Personality Item Pool (mini-IPIP), Multidimensional Scale of Perceived Social Support (MSPSS), Rosenberg Self-Esteem Scale (RSES), and the Marlowe-Crowne Social Desirability Scale Form-C (SDS). This revealed that opt-in participants were older (Mdn = 31.00) than the opt-out participants (Mdn = 27.00), U = 62690, p = .024. It also showed that openness to experience was greater in the opt-in (Mdn = 14.00) than for the opt-out group (Mdn = 13.00), U =77304, p = .033).

Relations for the opt-in and opt-out groups on all other variables (gender, highest level of education, current study, and current work status) were explored with Chi-square tests for independence. These revealed that there was a significant association between the opt-in status and the highest level of education achieved (X^2 (4, n = 1,358) = 11.26, p = .024). Standardised residuals showed there was a greater proportion of participants who had completed secondary education in the opt-out compared to the opt-in group.

3.2.2 Privacy and Ethical Considerations.

Ethical approval for the *MoodPrism* project was provided by the Monash University Human Research Ethics Committee (reference number: CF14/968 – 2014000398; Appendix B). Further, permission to promote the *MoodPrism* project to high-school students as a part of the recruitment method was sought and granted by the Victorian Government Department of Education and Training (reference number: 2015_002812; Appendix C). Participant information sheets and the inapp consent screens are provided in Appendix D.

There are currently no universal ethical standards guiding the collection of social media data (Golder, Ahmed, Norman, & Booth, 2017). However, the choice to collect social media data as word counts (aggregate) only and the opt-in process selected for data collection in this thesis was driven by considering the participant's ability to provide informed consent and how to best maintain

their privacy. These considerations have been highlighted in recent systematic review exploring the opinions of both researchers and social media users (Golder et al., 2017). While social media research was broadly perceived to be beneficial, the studies reviewed (n = 17) illustrated concerns around the use of social media users' data and the implications that had for participant privacy and when/if informed consent should be sought (Golder et al., 2017). Social media users and researchers expressed concerns around the use of verbatim quotes and the risk this posed to anonymity by limiting how effectively participants could be deidentified. There were preferences expressed for the use of aggregate data and transparent research disclosure statements and processes that outline the scope of data access and collection. Seeking informed consent also emerged as a contentious issue that was guided by beliefs around the public or private nature of social media activity. For some researchers and social media users, explicitly gaining the informed consent of public social media users was not seen as necessary, particularly where data was anonymised; while for others informed consent was viewed as a necessary part of ethical practice (Golder et al., 2017).

In the context of the *MoodPrism* app, social media data were collected from both public and private social media sources, from potentially vulnerable groups (e.g. minors), and in combination with extensive demographic, mental health and well-being data. It was therefore important to aggregate and anonymise participant social media data into word counts to reduce the risk of identification and maintain the privacy preferences that may have been implicit in each participant's use of social media.

As we also had direct access to our participants, we elected to include an opt-in process for contributing social media data. From the Beta testing and focus groups conducted prior to the public release of *MoodPrism* however, it became evident that the purpose and scope of the social media data collection was unclear to participants and that trust was an important precursor to opting-in to contribute data (see Appendix A). Clarifying the language and detail leading into the social media opt-in permissions was a crucial improvement to the informed consent processes.

In summary, the following steps were taken as a part of the ethical management of our social media data collection:

(1) Informed consent procedures:

- Providing clear explanation of the purpose and scope of the data collected from social media in several formats (within *MoodPrism* leading to the opt-in page; in the explanatory statements; on the website).
- b. Providing participants with an opt-in process for contributing social media data that could be accessed at any time and that was not linked to providing consent to participate in the broader *MoodPrism* study.
- (2) Privacy:
 - Automatically anonymising the participant's social media record upon collection through assigning an alphanumeric code linked to a device ID and not to a user name.
 - b. Storing the social media data as aggregate word counts only, thereby reducing the risk of identification.

3.2.3 Measures

3.2.3.1 Social media data collection by MoodPrism.

While the development of *MoodPrism* was a collaborative effort between Abdullah Arjmand, David Baker, Nikki Rickard, myself, and *TwoBulls*, the development of the social media data collection scripts and testing was a major component of this thesis and to which I was the primary investigator responsible. This utilised the application programming interfaces (APIs) for Twitter (REST API) and Facebook (Graph API). Documentation is available at https://developer.twitter.com/ and https://developers.facebook.com/. The collection of social media data was broadly completed in three steps:

1) Extraction of raw social media data through API commands;

- Summary processing of raw data as word counts using custom-developed word count scripts; and,
- Upload of summary word count data to *MoodPrism* database and deletion of raw social media data from local memory.

Social Media Data Extraction. An API allows software such as *MoodPrism* to communicate with a provider (Facebook or Twitter) using a common programming language to read and retrieve data. The data able to be accessed by *MoodPrism* (as third-party software) is restricted to the functions and permissions granted by both Facebook and Twitter. As outlined above, participants first provided permission by opting-in to have *MoodPrism* extract social media data from their Facebook and/or Twitter accounts. This required participants to log-in with their Facebook or Twitter credentials and consent to providing *MoodPrism* with access to the specific data described in Table 2. This provided *MoodPrism* with a *User Access Token* to query and read data for each participant from the Facebook and Twitter servers. The last 50 status updates on Facebook and/or Twitter for each participant were extracted and any new status update posted over the 30 days of using *MoodPrism* was also called via the API every 24 hours. At this level, the content from status updates was collected; however, it should be noted that this information was never made available to the researchers.

Raw Social Media Data Collected by MoodPrism via the Facebook Graph API and Twitter REST API.

Application Programming Interface	Data Called/Retrieved	Description
	Status update timestamp	Date and time the status update was posted
	Status update ID	Unique status update identifier
Facebook Graph API	Status update content	Content of the status update (string data)
	Likes ^a	Number of likes on the status update
	Comments ^a	Number of comments on the status update
	Location ^a	Location status update was posted (if available)
	Status update timestamp	Date and time the status update was posted
Twitter REST API	Status update ID	Unique status update identifier
	Status update content	Content of the status update (string data)
	Number of retweets	The number of times a status update was retweeted. Indicates content has been reposted and was not generated by the <i>MoodPrism</i> participant
	Number of favourites	Number of likes on the status update

Note: a = errors in calling this information were not resolved during data collection and this data

was unavailable for analysis.

Word count scripts and summary processing. After retrieving the data outlined in Table 2, the content of status updates was run through word count scripts developed for the Android and iPhone implementations of *MoodPrism*. The word count scripts processed the raw social media data locally on the participant's smartphone ensuring the content of status updates continued to reside on the user's smartphone only and was never available to the researchers.

The word count scripts for this research were primarily derived from the *Linguistic Inquiry and Word Count 2007 (LIWC 2007*; Pennebaker, Chung, Ireland, Gonzales, & Booth, 2007). The *LIWC 2007* is a widely used corpus of dictionaries for automated text-analysis. The dictionaries represent 80 language categories that have been used to identify psychological and social information in written and spoken language. Numerous studies have utilised the LIWC 2007 in the examination of individual differences, psychological processes, and for elucidating the language features related to mental health (Tausczik & Pennebaker, 2010). The *LIWC* 2007 was updated during the course of this research to the *LIWC* 2015 (Pennebaker, Boyd, Jordan, & Blackburn, 2015); however, the development costs (time, financial) of updating the dictionaries restricted their inclusion. Despite this, comparison of the 2007 and 2015 versions show very similar performance (Pennebaker Conglomerates, 2015).

A subset of *LIWC* 2007 dictionaries including positive emotion, negative emotion, personal pronoun (first person - singular), and other pronoun (first person plural, second person, third person singular/plural) dictionaries were selected to be included in *MoodPrism*'s word count scripts as they had been shown to be related to mental health outcomes in social media and offline research (De Choudhury et al., 2013; Tausczik & Pennebaker, 2010). Using a subset of target categories (rather than all *LIWC 2007* dictionaries) also reduced the smartphone processing burden. In this research, only the positive and negative emotion dictionaries were supplemented by emojis and internet slang for the emotion categories to better reflect the online environment (Kern et al., 2016). These were generated by conducting an online search to compile a list of emojis and internet slang related to

positive and negative emotion. Ambiguous items (unclear emotion or uncommon) and other problematic items (e.g., those that were impractical to implement due to word count script issues) were filtered out to yield the final list (see Appendix E).

The raw word counts from the supplemented *LIWC 2007* dictionaries were operationalised by summing the number of target words in each respective dictionary category used within a language sample (status update) and calculating their proportion relative to the total word count in that language sample. This meant that the relative proportion of all positive, negative, first person personal pronouns, and other pronouns from all words used in each individual Tweet or status update was calculated. This is seen in Equation 1:

$$p(category) = \sum_{word \in category} p(word) = \frac{\sum_{word \in category} count(word)}{N_words},$$

(1)

where count(word) refers to the total number of *LIWC* 2007 dictionary category words contained in a status update or Tweet, and *N_words* is the total number of words in that status update or Tweet.

Practical implementation and testing of word count scripts in Objective C and Java. As

MoodPrism was developed for both iOS and Android smartphones, the programming language implementations of the word count scripts differed between iOS (Objective C) and Android (Java). To ensure the word count scripts were extracting target words accurately, and consistently across both operating systems, a test set of simulated and real status updates was created that aimed to present simple and complex sentences in order to determine where and if the word count scripts failed at identifying words in the supplemented *LIWC 2007* dictionaries. These test sentences were manually coded for the target words and then automatically processed through *MoodPrism*'s word count script. Broadly, the steps followed by the word count scripts in both OSs were:

- 1) *MoodPrism* extracted the content of a status update in a string data format (a series of characters which can consist of alphanumeric, punctuation, and symbols);
- 2) The string was split into substrings that represented words or emojis;
- 3) Each substring was run through each of the dictionaries and matches were counted.; and,
- 4) The word count of matching words within each dictionary was summed and the summary data was uploaded to the *MoodPrism* server.

Testing the first iteration of the word count script through *MoodPrism* revealed both syntax and logic errors. Syntax refers to the rules of language in software and errors occur when symbols are used incorrectly or there are errors in applying the language rules (Youngs, 1974). Logic errors refer to the errors in the performance of the program solution, where the desired outcome of the program is not achieved (Youngs, 1974).

Syntax errors. Firstly, as the word dictionaries had been directly cut and paste from a Microsoft Word (MS Word) document into Objective C and Java, any words containing an apostrophe were not identified as belonging in a dictionary, impacting on contractions, emojis and internet slang. Characters are encoded by a set of universal numeric values called Unicode, providing a common language across programs (Unicode Inc., 2017). In Microsoft Word, apostrophes are automatically converted into curved 'smart quotes' which indicate open and closed apostrophes and quotation marks (Unicode: U+2018 and U+2019), whereas the status update strings retrieved by *MoodPrism* represented apostrophes in their straight format (Unicode: ASCII U+0027). This syntax discrepancy resulted in the word count script incorrectly rejecting all words with straight apostrophes. The error was resolved by replacing all curved MS Word apostrophes with straight apostrophes in the dictionaries.

Secondly, escape commands had not been included on symbols that represent functions (operators) in Objective C and Java, primarily impacting on the ability of the word count script to identify emojis. Part of the syntax in Objective C and Java includes the use of symbols to perform specific functions. For example, brackets "()" are often used to group a set of statements and symbols such as < operate on other symbols in a statement. These symbols are common characters in emojis, for example. :) and <3. Escape functions (often a backslash; \) specify that components in a string are to be read literally (as a character) and not as a function or operator. The syntax error where emojis were not being read literally as substrings was resolved by preceding all Objective C and Java operators present in the dictionaries with an escape function.

Logic errors. Testing also revealed logic errors between the Objective C and Java versions of the word count script. The logic error occurred in Step 2 (above) in how the stings were split into substrings (words). The Objective C implementation defined substrings as consecutive characters that were separated by whitespace (any alphanumeric or symbol characters separated by a 'space'). The Java implementation in comparison defined substrings as consecutive letters or numbers that were split by non A-Z and 0-9 characters. This had different implications for how the iOS and Android versions of *MoodPrism* identified and counted dictionary words within a status update. Neither solution performed in a way that consistently identified both words (in a context that included punctuation) and emojis:

- 1) In Objective C, splitting by whitespace resulted in dictionary words attached to punctuation script (e.g. excited!) not to be identified by the word count.
- 2) In Java, splitting by non-A-Z, 0-9 broke all emojis into their individual characters and none were identified by the word count script.

Final implementation. Beyond making the splitting logic consistent in both Objective C and Java, the inclusion of emojis presented a significant practical challenge and required an amended search and matching strategy for the word count scripts to follow. Punctuation symbols were the consistent problematic factor in both versions of the splitting logic. To address this, the final implementation separated the positive and negative emotion word dictionaries of the *LIWC* 2007 from the positive and negative emojis and internet slang (which had previously been combined in

master positive and negative emotion dictionaries). The word count script then followed a two-step search and matching strategy:

- From the full-text input of a status update, emojis and internet slang were matched and counted with their respective dictionaries without splitting the string into parts. This results in positive and negative emoji and internet slang word counts; and,
- 2) Then, splitting the string on non A-Z (or apostrophe) substrings representing words are created (letters only). These substrings are then matched and counted for the corresponding positive or negative emotion dictionary of the *LIWC 2007*.

Following the two-step search and matching strategy the word counts from each step are summed to create overall positive and negative emotion word counts.

While this solution allowed for consistency across both OS and resolved the major logic error introduced by punctuation, there are specific cases where a false positive word count occurs. For example in the sentence "Here is a list of colours:Purple, green, and blue", the positive emoji ":P" would be identified in step 1 of the word count where a typographical error had occurred. While we could not assess how often these false positive occurred in the data set as we did not have access to the content of status updates, the rare occurrence of false-positives was considered preferable to the frequent occurrence of false-negatives in the original iteration of the code.

3.2.3.2 Psychological Measures and Daily Mood Report Items from MoodPrism.

MoodPrism delivered psychological questionnaires and surveys at baseline, baseline and follow-up (after 30 days of *MoodPrism* use), and at follow-up only. Further, daily mood reports consisting of 16-items were delivered every-day for 30 days after the baseline measures were completed. Tables 3 and 4 provide a summary of the psychological questionnaires delivered by *MoodPrism* at baseline and follow-up. Table 5 describes the items in the daily mood report. These tables have been adapted from Multimedia Appendix 1 in Rickard et al. (2016).

Measure/Domain	Brief Description	Source
Demographics ^a	Consisted of custom items asking participants	
	gender, age, highest education, current work	
	and study status.	
mini-International Personality	Measure of personality based on the Five-	Donnellan
Item Pool ^a	Factor Model: extraversion, neuroticism,	et al. (2006)
	conscientiousness, agreeableness, openness to	
	experience. Consists of 20 items rated from	
	"1- Very inaccurate" to "5- Very accurate"	
	and are summed within personality factors	
	(range 4-20).	
Marlow-Crowne Social	Measures the tendency to present a desirable	Reynolds
Desirability Scale Form-C ^a	social image to other. Consists of 13	(2006)
	dichotomous items rated as "True" or "False".	
	After reverse scoring of relevant items, scores	
	are summed with higher scores indicating	
	higher social desirability.	
Rosenberg Self-Esteem Scale ^a	Measure of self-esteem. Consists of 10 items	Rosenberg
	rated from "0- Strongly disagree" to "3-	(1965)
	Strongly agree". These are summed after	
	reverse scoring negatively worded items	
	(range 0-30). Higher scores indicate greater	
	self-esteem.	
Multidimensional Scale of	Measure of perceived social support with	Zimet et al.
Perceived Social Support	subscales for significant others, family, and	(1988)
	friends. Consists of 12 items rated from "1-	
	Very strongly disagree" to "7- Very strongly	
	agree". Scores are summed within sub-scales	
	and an average over items obtained. Higher	
	averages indicate stronger perceptions of	
	social support.	
Barcelona Musical Reward	Measure of music rewards across five facets:	Mas-Herrero
Questionnaire	musical seeking, emotional evocation, mood	et al. (2013)
	regulation, sensory-motor, and social reward.	
	Consists of 20 items rated from "1-	
	Completely disagree" to "5- Completely	
	agree". Sub-scale score and a total score of	
	musical reward can be derived.	
Technology Use Survey	A custom developed survey addressing	
	patterns of technology use.	

Measures Delivered at Baseline Only by MoodPrism.

Note: ^a = measure included in analyses presented in this thesis (Chapters 4 and 5).

Measure/Domain	Brief Description	Source
Patient Health Questionnaire-9 ^a	Measure of depression symptom severity over	Kroenke
	previous 2-weeks. Consists of 9 items rated	et al. (2001)
	from "0- Not at all" to "3- Nearly every day"	
	which are summed (range 0-27). Higher	
	scores indicate greater severity.	
General Anxiety Disorder-7	Measure of anxiety symptom severity over	Spitzer et al.
	previous 2-weeks. Consists of 7 items rated	(2006)
	from "0- Not at all" to "3- Nearly every day"	
	which are summed (range 0-21). Higher	
	scores indicate greater severity.	
Warwick Edinburgh Mental	Measure of mental well-being addressing	Tennant et al.
Well-Being Scale	positive functioning, positive affect, and	(2007)
	interpersonal relationships. Consists of 14	
	items rated from "1- None of the time" to "5-	
	All of the time" which are summed (range 14-	
	70). Higher scores indicate greater well-being.	
Emotional Self-Awareness Scale	Measures emotional self-awareness with sub-	Kauer et al.
	scales: recognition, identification,	(2012)
	communication, contextualisation, and	
	decision-making. Consists of 33-items rated	
	from "0- Never, to "4- A lot". Higher scores	
	indicate greater levels of emotional self-	
	awareness.	
Coping Self-Efficacy Scale	Measure of perceived self-efficacy to cope	Chesney
	with challenges. Consists of 26 items rated	et al. (2006)
	from "0- Cannot do at all" to "10- Certain can	
	do". Items are summed, and higher scores	
	indicate greater coping self-efficacy.	
Brief Resilience Scales	Measure of resiliency. Consists of 6 items	Smith et al.
	rated from "1- Strongly disagree" to "5-	(2008)
	Strongly agree". Relevant items are reverse	
	scored, and the items summed. Higher scores	
	indicate greater resiliency to stressors.	
Mental Health Literacy	Measure of mental health literacy through	Adapted
Questionnaire	responses to vignettes.	from Reavley
		et al. (2014)

Measures Delivered at Baseline and Follow-Up by MoodPrism.

Note: a = measure included in analyses presented in this thesis (Chapters 4 and 5).

Measure/Domain	Brief Description	Source
Patient Health Questionnaire-2	 Adapted to measure daily depression symptoms. Consists of: 1. "Little interest or pleasure in doing things", 2. "Feeling down, depressed or hopeless". Rated from "0- Not at all" to "4- extremely". 	Kroenke et al. (2009)
General Anxiety Disorder-2	 Adapted to measure daily anxiety symptoms. Consists of: 1. "Feeling nervous, anxious, or on edge", 2. "Not being able to stop or control worrying". Rated from "0- Not at all" to "4- extremely". 	Kroenke et al. (2009)
Mood ^a	Consist of 3 items based on the circumplex model of emotion. 1. "Active or alert" 2. "Positive or pleasant" ^a 3. "Negative or unpleasant. ^a Rated from "0- Not at all" to "4- extremely".	Adapted from Russell (1980)
Well-being	Custom developed 5 items addressing daily feelings of control, social connection, motivation, life meaning, and self-esteem. 1. "In control of what I'm doing" 2. "Socially connected and supported" 3. "Motivated, engaged, and interested" 4. "Life is meaningful and with purpose" 5. "Feeling good about myself" Rated from "0- Not at all" to "4- extremely".	Adapted from Bech et al. (2003) and Keyes (2005)
Daily Events	Two items addressed positive and negative daily events. If a positive or negative event was endorsed, a further rating items was triggered where participants rated how positive or negative it was from "0- Not at all" to "4- extremely".	Categories are available in Rickard et al. (2016).
Context	Two items addressed context at the time of completing a daily mood report. 1. Where are you? 2. Who are you with?	Categories are available in Rickard et al. (2016).

Items in the Daily Mood Reports Delivered by MoodPrism.

Note: a = measure included in analyses presented in this thesis (Chapters 4 and 5).

In addition to the measures presented in Tables 3, 4, and 5, a feedback questionnaire was presented at follow-up. This was custom developed and was based on the Mobile Application Rating Scale (Stoyanov et al., 2015). Further detail can be found in Appendix A.

Only the daily mood report items addressing mood ("positive or pleasant", "negative or unpleasant") and the PHQ-9 were included in the main analyses of this research. The mini-IPIP, Marlowe-Crowne Social Desirability Scale Form-C, and Rosenberg Self-Esteem Scale were included in supplementary analyses presented in Chapter 4. To reduce potential repetition, detail on the content and psychometric properties of these measures are provided within Chapters 4 and 5.

3.2.4 General procedure.

Figure 3 shows chronological screenshot examples of the *MoodPrism* user experience and general research procedure. Participants of the *MoodPrism* study downloaded the app onto their smartphone from the Apple or Google Play stores. Upon opening the app participants landed on a welcome page (Figure 3a) and were introduced to the purpose of the study (Figure 3b). Access to a downloadable explanatory statement (pdf. format) was presented on the app before digital consent to participate in the study was provided by the participants (Figure 3c). Permissions were also sought to access social media data from Facebook and Twitter as a part of an opt-in process (Figure 3d). Social media permissions could be provided or revoked at any time by accessing the *Permissions* page from the drop-down menu in *MoodPrism* (Figure 3e).

Participants then completed a battery of baseline surveys which included measures addressing mental health, well-being, personality, self-esteem, and technology use previously outlined in Measures. These were organised into blocks on a dashboard (Figure 3f) and each of these blocks needed to be completed before unlocking access to feedback from the experience sampling component of *MoodPrism* (daily mood reports). While each block of questionnaires needed to be accessed and each questionnaire submitted, completion of items within each questionnaire were optional. After completing the baseline surveys participants selected the timeframe within which they wished to be notified by *MoodPrism* to complete a daily mood report

(Figure 3g). Participants were randomly notified by push-notification on their smartphone of an available daily mood report within this self-defined timeframe every day for 30 days.

The daily mood reports were accessed from the mood feedback overview page and began with the prompt screen provided in Figure 3h. The prompt was followed by 16-items that addressed mood (positive and negative affect, arousal), mental health, eudaimonic well-being, significant daily life events (positive and negative), environmental and social context. At the end of 30 days, participants were also asked to complete a battery of follow-up surveys which consisted of a subset of surveys presented at onboarding. Feedback was provided to participants following each daily mood report showing the number of surveys required to unlock more app content (Figure 3i). It was also presented in a colourful calendar format (Figure 3j), in a weekly view (Figure 3k), and in a detailed daily view (Figure 3l).



Figure 3. MoodPrism screenshots showing the participant's experience through the general research procedure. Captions describe the content of the example screenshots.

3.3 Concluding Remarks

This chapter outlined the need for directly collecting social media data in conjunction with other ESM methods to better sample the individual and contextual factors surrounding social media use. It briefly described the development of the smartphone app *MoodPrism* and introduced the participant selection method and sample characteristics for the social media users included in Chapters 4 and 5. It is clear from the social media opt-in rates and final sample sizes that barriers to contributing social media data may have existed for the majority of participants. As social media data was to be collected within the comprehensive *MoodPrism* ESM, it is likely that concerns around privacy and the breadth of personal information being sampled inhibited the choice to opt-in. To improve opt-in rates in future research, researchers should create opportunities for the user to develop trust in the app before requesting social media data and, as implemented in this research, create detailed social media specific informed consent processes to aid participant understanding of the nature of data they are contributing. The final part of this chapter described the social media data collection method utilised by *MoodPrism*, the development and testing of the word count scripts, and the general research procedure.

Chapter 4

Predicting Depression from Language-Based Emotion Dynamics: Longitudinal Analysis of Facebook and Twitter Status Updates 4.1 Preamble to Empirical Paper 1

This chapter presents the first empirical paper of this thesis titled "Predicting Depression from Language-Based Emotion Dynamics: Longitudinal Analysis of Facebook and Twitter Status Updates". From the introduction of this thesis and the discussion presented in Chapter 2, it is evident that there has been limited focus on the temporal specificity of observations obtained from social media data and the way these unfold overtime to reveal processes that may be relevant to depression. In this chapter, behavioural (time of posting) and lexical (emotion words) features are combined to tap into the emotion patterns expressed across status updates as a means of providing insight into the depression status of Facebook and Twitter users. Notably, this chapter presents the first application of emotion dynamic indices to social media language data. It also addresses some of the limitations of the research approaches previously taken in the literature identified by Tufekci (2014) and in Chapter 2 by drawing a sample from the general population, using a complementary data collection method for psychological data, providing a cross-platform comparison, and directly sampling social media data rather than relying on self-report alone.

This paper has been published in the *Journal of Medical Internet Research*. As such, the paper is formatted in accordance with the journal requirements. References are provided in the style of the *American Medical Association* (10th edition)

4.2 Predicting Depression from Language-Based Emotion Dynamics: Longitudinal Analysis of Facebook and Twitter Status Updates

Abstract

Background: Frequent expression of negative emotion words on social media has been linked to depression. However, metrics have relied on average values, not dynamic measures of emotional volatility.

Objective: This study reports on the associations between depression severity and the variability (time-unstructured) and instability (time-structured) in emotion word expression on Facebook and Twitter across status updates.

Method: Status updates and depression severity ratings of 29 Facebook users and 49 Twitter users were collected through the app *MoodPrism*. The average proportion of positive and negative emotion words used, within-person variability, and instability were computed.

Results: Negative emotion word instability was a significant predictor of greater depression severity on Facebook ($r_s(29) = .44$, p = .017, 95% CI [.09, .69]), even after controlling for the average proportion of negative emotion words used (partial $r_s(26) = .51$, p = .006)

and within-person variability (partial $r_s(26) = .49$, p = .009). A different pattern emerged on Twitter where greater negative emotion word variability indicated lower depression severity ($r_s(49) = -.34$, p = .011, 95% CI [-.58, .09]). Differences between Facebook and Twitter users in their emotion word patterns and psychological characteristics were also explored.

Conclusions: The findings suggest that negative emotion word instability may be a simple yet sensitive measure of time-structured variability useful when screening for depression through social media, though its usefulness may depend on the social media platform.

Keywords: Automated text analysis, depression, Facebook, Twitter, emotion, variability, instability.

Introduction

"With as much as we have learned about emotions, it is as if we have been taking still photos of a dance." [1].

Social media is used in different ways by different people, but for many individuals, status updates provide snapshots of their lived experience. Studies to date have primarily considered how the relative frequency of words indicating positive and negative emotion relate to other characteristics such as mental health status, or which words (or set of words) best predict different outcomes. Such studies indicate that the frequent expression of negative emotion words in status updates can accurately identify individuals experiencing symptoms of depression [2–6]. However, an individual's mental health is reflected by more than just the average frequency or the type of words used; variability in emotional expression over time might also provide significant insights. In the current research, fluctuations in emotional expression over time is explored as another window of insight into the psychological health of social media users.

Depression in Status Updates on Social Media

Depression, including major depressive disorder (MDD) and dysphoria, are characterized by persistent low mood (including sadness or emptiness) or anhedonia (inability to experience pleasure from activities that are usually enjoyable) [7]. At a broad level, the frequent expression of negative affect within social media status updates has been associated with higher levels of depression symptoms [2,3,5,8–11]. Frequently expressing positive affect, on the other hand, tends to be associated with lower levels of depression and greater levels of well-being [9,12,13]. The link between expressed emotion in status updates and mental health is unsurprising considering that expressing current emotion and venting frustration have been reported to be a primary purpose for many users posting on Facebook [14]. Indeed, negative and positive emotional language has been observed to occur in approximately 34% and 55% of status updates on Facebook, respectively [15]. Adding to this, depressed individuals have also been shown to post content more frequently than non-depressed persons [16], and changes in depression severity may be signalled by increases in

posting behaviour on social media [17]. Combined, the time-structured features and emotional features of status updates may provide insights into the depression status of social media users.

Several studies have sought to code the content of social media posts for depression disclosures [6]. For instance, Moreno et al. [3] demonstrated that status updates on Facebook with references to depression symptoms such as hopelessness were positively correlated with selfreported depression symptoms. Others extended this by describing the linguistic characteristics of depression in posts and developing coding-schemes to identify depression-indicative Tweets or status updates [2,4,8,18]. While specific topics, keywords, and linguistic features (especially negative emotions) are able to identify depression-indicative posts with high accuracy, many of these features may also be present in posts that are non-indicative of depression (low specificity). For example, Mowery et al. [18] found considerable signal discrepancies - over 70% of tweets identified in their sample containing words related to depression were not actually indicative of depression. Thus, although negative emotion words correlate with the presence of depressive symptoms, it is a noisy and imprecise metric.

This highlights the need to move beyond the frequency of emotional language alone towards other online behavioural indices that may better differentiate depressed and non-depressed individuals. Due to the time-sensitive nature of social media data, examining the dynamic movement of emotion across status updates may provide an additional avenue to tap into the nuanced cognitive-emotional processes underlying depression and may provide a more specific index of maladaptive emotional functioning.

The Emotion Dynamics of Depression

A major change in functioning associated with the onset of depression is the ability to effectively regulate emotion. While the capacity for emotion to vary over time is adaptive and may contribute to psychological well-being, higher levels of emotion variability, especially of negative emotion, have been linked to depression [1,19]. For example, individuals who experience intense negative affect reactivity in response to daily stressors are at greater risk of developing depression
[20–22]. This experience is supported by young people's qualitative accounts of depression, where depression is reported to "[take] over during times of vulnerability such as stress or fatigue" (p. 386) [23]. Negative cognitive biases also contribute to emotion variability in depression. Excessive focus on personal distress (rumination) may lead to persistent experiences of severe negative affect and difficultly regulating mood away from negative rumination [24, 25]. The combination of cognitive-emotional processes result in emergent emotion patterns that can manifest at inappropriate times and in inappropriate ways in response to internal and external events. Maladaptive patterns of emotion build over time to place the individual at an increased risk for depression onset and maintenance [19,26–28].

The emotion variability in depression described above has predominantly been operationalised in two ways. Firstly, variability may be operationalised as within-individual variability as *iSD*, an individual's standard deviation of emotion expression. Like the mean, variability may best be viewed as a "trait-like" measure of emotion expression, as it provides a single number that summarizes the overall variability in affect for an individual across their recording period, but ignores time-structured information [29].

A second operationalisation of variability describes emotional instability and uses the mean squared successive difference statistic, MSSD [30], which quantifies differences between consecutive observations of emotion [1]. This time-structured measure of variability uses the temporal ordering of measurements to quantify the magnitude of incremental changes in emotion [30–32]. Crucially, unlike *iSD* where the same result would be obtained if the same set of emotion expression observations are shuffled through time, the MSSD is sensitive to the time-ordering of observations. For example, for the same distribution of negative emotion values (and thus the same *iSD*), negative emotion increasing in small incremental steps from mild to severe would result in a small MSSD value, whereas negative emotion alternating (or swinging) between mild to severe would result in a large MSSD value. In this way, MSSD captures the temporal instability of positive or negative affect.

DEPRESSION AND LANGUAGE-BASED EMOTION DYNAMICS ON SOCIAL MEDIA

Negative affect instability, as measured by the MSSD, has been linked to more severe depression symptoms across several studies and has been identified as a concomitant and early indicator of depression [25,31,33–36]. It has been shown to be a significant risk factor for more frequent and severe suicidal ideation [35] and may be a unique underlying emotion pattern in depression. Negative affect instability has been shown to continue to predict depression when average negative affect and the frequency of negative event exposure are held constant [25,36]. In addition, reductions in negative affect have been shown to be greater for depressed individuals in response to positive events when compared to those who are not depressed, further contributing to potential moment-to-moment variability [37].

Bowen et al. [33] recently aggregated four studies to examine the differences in mood instability between depressed and non-depressed individuals. Participants completed daily mood diary ratings of negative and positive mood upon awakening and before bedtime for one week. Depressed individuals experienced greater negative mood instability over the course of the week compared to non-depressed individuals. Depressed individuals also reported greater severity in negative mood than the non-depressed group, suggesting depression is characterised by both persistent low mood and more extreme daily variation in its severity.

While depression has also been associated with a blunted emotional response to stimuli and smoother emotional experiences from day to day (inertia) [32,38,39]; variability and instability span major categories of emotion dynamics as they relate to depression and are the focus of the current study. Table 1 outlines the definitions of variability, instability and inertia and describes their conceptual overlap. To best examine the unique associations that emotion dynamic patterns have with mental health it has been recommended that the conceptual overlap between these measures be taken into account and controlled for in analyses [32], as is done in the current study.

Table 1

Name	Definition	Operationalisation	Conceptual Overlap
Variability	The amplitude of an individual's emotion. This is time- unstructured, referring to the "general dispersion" of scores.	Within-person standard deviation (iSD)	Variance
Instability	The amplitude of moment-to- moment changes in emotion. This is time-structured, where higher scores indicate greater variance and less positively correlated between observations.	Mean squared successive differences (MSSD)	Variance, time- dependency
Inertia	How well a previous emotional state predicts the next emotional state. This is time-structured, where greater correlation coefficient indicates increased temporal dependency between observations.	Autocorrelation coefficient (ACF)	Time-dependency

Definitions and Conceptual Overlap of Variability, Instability, and Inertia.

Social Media and Emotion Dynamics

Emotion dynamics may provide important insights into the "building blocks" of depression [28] but it is also challenging and time-intensive to collect adequate longitudinal emotion data. Current approaches rely on experience sampling methods (ESMs), where an individual inputs emotion information throughout a day [1,28,40]. While the potential burden and invasiveness of real-time data collection has been significantly reduced by incorporating new and familiar technologies into ESM design (e.g. smartphones) [41,42], the need to respond to automated prompts creates a context that is different than normal daily activities. Further, these methodologies may not be practical for large-scale monitoring of public mental health.

Social media may be a powerful complementary tool. Considering the frequent use of emotion language in status updates that relate to current experiences [14,15], for a large proportion of the population social media can provide unobtrusive access to time-sensitive and ecologically valid samples of expressed emotion [2,43–45]. Diurnal and seasonal variation in depression severity

have been observed at a population [2] and individual level [4] on social media. In these studies, an increase in the linguistic features predictive of depression risk were observed from day to night, and from summer to winter months. Using the social media platform Reddit, De Choudhury et al. [46] considered transitions from mental health subreddits only to also using a suicide support subreddit. Findings suggested that a shift from commonly expressed sentiment (i.e., the average) may represent a change toward better or poorer mental health, particularly where the magnitude of the change is more pronounced. While observations of emotion variability and instability are yet to be applied to social media as a means of automatically screening for individuals at risk of depression, it is likely that in addition to the ability to track macro-level changes in depression on social media, micro-level changes in emotion (emotion variability) relevant to mental health may also be observable.

The Current Study

Evidence is mounting to suggest that emotion patterns, including variability and instability are early indicators for depression risk [19], and there is a need to utilise scalable and unobtrusive means of collecting emotion data to effectively apply these insights to monitoring public mental health. Targeting emotion variability and instability as indicators of maladaptive emotional functioning in depression is a clear area in need of further research on social media. To date, most studies examining emotion language on social media and depression have provided a static view of emotion by compressing the variation of social media language over time into an overall average, stripping the data of what could be meaningful patterns in temporal variation of emotion expression. While the average emotion that individuals express on platforms like Facebook and Twitter can provide accurate and sensitive insights into the presence of depression, the variability in emotion across posts has yet to be examined as a legitimate individual difference (rather than measurement error) that may be indicative of depression severity.

Taking advantage of the time-sensitive and naturally occurring data available from status updates on Facebook and Twitter, the major aim of this study was to demonstrate the feasibility of

DEPRESSION AND LANGUAGE-BASED EMOTION DYNAMICS ON SOCIAL MEDIA

using status update emotion variability and instability as an indicator of depression severity (measured by the *Patient Health Questionnaire-9, PHQ-9*) [47]. It also aimed to examine if emotion instability was related to depression when controlling for its conceptual overlap with variability.

It was hypothesised that: (1) Self-reported depression severity would be positively related to negative emotion word variability and instability across status updates. (2) Self-reported depression severity would be positively related to the average proportion of negative emotion words used, and negatively related to the average proportion of positive emotion words used in status updates on Facebook and Twitter. (3) Negative emotion word instability would remain positively associated with depression severity when controlling for negative emotion word variability. (4) The emotion word patterns and their association with depression would be consistent across Facebook and Twitter. (5) Depression severity would be positively associated with the average number of status updates per day and negatively associated with the time-interval between consecutive status updates (i.e., shorter periods of time between posts).

Method

Participants

The current study used a subset of users from the *MoodPrism* project. *MoodPrism* is a mood-tracking application (app) that collects data and provides engaging feedback to its users on their mood, mental health, and well-being [42]. *MoodPrism* is available for download on the iOS and Android stores for smartphone. All procedures were approved by the Monash University Human Research Ethics Committee.

Participants were recruited by convenience sampling, community engagement, and targeted online advertising (smartphone owner, interested in mental health, lives in Australia). To be included in this study, participants had to download the *MoodPrism* smartphone app, complete the depression severity index available in the app, and opt-in to contribute their Facebook or Twitter data, which were automatically collected by the *MoodPrism* app. A minimum of 10 status updates over a minimum period of 7 days was required for the inclusion of a participant, to allow robust

calculation of emotion word variability and instability over time. Additionally, status updates were only included if they occurred within the 12 months prior to the administration of the *PHQ-9*. For the Twitter data, only original tweets (not retweets) were used. Although retweets may reflect values or interests of a user and include topics similar to self-authored Tweets [48], they also introduce ambiguity about the author's sentiments [49,50]. Further, the Facebook data did not have a similar repost function, such that self-authored Tweets provide a more direct behavioural comparison.

Of the 1,518 users who downloaded the MoodPrism app between April 2016 and May 2017, 223 (14.7%) provided permission to access their social media data. After applying the inclusion criteria outlined above, three participants were found to have contributed both Facebook and Twitter data. These participants had a greater number of language samples on Facebook than on Twitter, and thus were allocated to the Facebook group. A final sample of 29 Facebook users (11 males, 17 females, 1 missing) with a mean age of 32.77 years (SD = 8.40, range = 19-45, n = 22) and 49 Twitter users (16 males, 32 females, 1 missing) with a mean age of 35.03 years (SD = 12.33, range = 16-57, n = 39) was obtained. Participants were well educated, with 34.5% (Facebook) and 40.8% (Twitter) of participants having completed tertiary education. Chi-square tests revealed no significant differences gender or education between the included samples and those who had opted-in to contribute social media data but did not meet the inclusion criteria. Independent samples t-tests revealed no significant differences between groups in age. There were also no significant differences between the included Facebook (n = 29) and Twitter (n = 49) samples in age, gender or education.

Procedure

After downloading and opening *MoodPrism*, participants read an explanatory statement and provided their consent to participate. They then provided an additional opt-in consent to share their Facebook or Twitter data. If consent was provided, the *MoodPrism* app then automatically extracted the participant's previous status updates on Facebook or Twitter and repeated this extraction for all

new status updates posted while *MoodPrism* was installed on the participant's smartphone. Status updates were processed locally on the participant's smartphone through the app, pulling out the time, total word count, and number of positive and negative emotion words, and then these summaries were uploaded to a secure server every 24-hours, at which point the status update content was permanently deleted from *MoodPrism*'s memory. Thus, the app provided summaries of how often emotion words were expressed, but the actual status updates were unavailable for analysis.

Participants additionally completed several blocks of questionnaires on *MoodPrism*. These blocks included demographic items collecting gender and age information, and measures assessing mental health, personality, and other psychological characteristics (see [40] for the full list of measures). Blocks could be completed in any order at a time of the participants' convenience and collectively took an average of 37m 14s (SD = 11m 33s) to complete.

Measures

All data for the current study was collected via the *MoodPrism* app. Depression symptom severity was measured by *the Patient Health Questionnaire -9* (*PHQ-9*) [47], a nine item self-report measure for depression that indicates the severity of symptoms experienced over the previous twoweeks. Each item on the *PHQ-9* (e.g. "Feeling down, depressed, or hopeless.") is rated from 0 – "Not at all", to 3 – "Nearly every day". These ratings are summed to create a total score ranging from 0-27, where higher scores indicate greater severity of depression symptoms. The *PHQ-9* has been validated for use in the general population (Cronbach $\alpha = .87$) [51] and in primary care settings ($\alpha = .89$) [47]. The internal reliability of the *PHQ-9* was good for both the Facebook (Cronbach alpha=.87) and Twitter (Cronbach alpha=.90) samples.

Language samples from Facebook and Twitter were obtained by *MoodPrism* via the Facebook and Twitter application programming interfaces (API), as detailed in Rickard et al. [40]. The period of posts sampled per participant between their first status update and the administration of the *PHQ-9* ranged from 9 to 365 days (Facebook *M* = 170.69, *SD* = 116.05; Twitter *M* = 145.61, *SD* = 124.97).

MoodPrism's automated scripts identified the number of "words" and positive and negative emotion words in the status updates. Words on social media include both normal words and variants (e.g., misspellings, emoticons, abbreviations) that are common on social media [43]. The scripts incorporated the positive- and negative emotion dictionaries of the *Linguistic Inquiry and Word Count 2007 (LIWC 2007)* [52], a widely-used corpus of dictionaries commonly used for language analysis. The *LIWC 2007* dictionaries were supplemented by common emoji's and internet slang that indicated positive or negative emotion (see Multimedia Appendix 1). While not definitive, these inclusions were made to better reflect the language used on social media (for further discussion see [43]).

MoodPrism also collected data on the psychological characteristics of participants which included personality, self-esteem, and social desirability. Multimedia Appendix B presents additional analyses, complementary to the findings presented here, exploring Facebook and Twitter user differences across these characteristics that may inform the patterns of emotion expressed over time.

Data Analysis

Within person variability, instability, and the average proportion of positive and negative emotion words in status updates on Facebook and Twitter were calculated for each participant (defined below), and then correlations with PHQ-9 scores were calculated. The distributions of all Facebook and Twitter variables were non-normal, consequently Spearman's rho was selected for computing correlations. Exploratory post-hoc comparison between the Twitter and Facebook samples on their psychological characteristics were also performed using Mann-Whitney U tests due to non-normal distributions. All analyses were performed in IBM SPSS Statistics, Version 24 [53].

Average proportion.

The relative proportion of positive and negative emotion words was calculated for each status update collected to adjust for the total number of words expressed, as described in Kern et al. [43] and defined in Equation 1 below. An average of these proportions was taken for each participant, resulting in the *average proportion of positive emotion words* and *average proportion of negative emotion words* across all status updates (range: 0 to 1):

$$p(category) = \sum_{word \in category} p(word) = \frac{\sum_{word \in category} count(word)}{N_words},$$
(1)

where *count(word)* refers to the total number of positive emotion words (or negative emotion words; the *LIWC 2007* category) contained in a status update, and N_words is the total number of words in that status update.

Variability.

The within person variability (*iSD*) was computed for each participant across their status updates as:

$$iSD = \sqrt{\frac{\sum_{i} s_{i}^{2}}{n-1}},$$

(2)

where the sum is taken over posts, *i*, *s_i* indicates deviations from the mean in an individual's proportion of positive (or negative) words used in status updates, and *n* refers to the number of status updates for that individual. This resulted in the *positive emotion word variability* and *negative emotion word variability* across status updates for each participant.

Instability.

The MSSD is defined for an individual as:

$$MSSD = \frac{\sum_{i=1}^{n-1} (x_{i+1} - x_i)^2}{n-1},$$

(3)

(4)

where x_i indicates the observation at index *i*, x_{i+1} refers to the next consecutive observation, and *n* refers to the total number of observations for that individual. A major challenge in applying measures of time-structured variability to social media data is managing the irregularly spaced time-intervals between posts. As observations on Facebook and Twitter occur in a natural setting, they often occur at irregular intervals spanning, for example, between hours and months. Thus, in addition to considering time-order, the time elapsed between successive observations also needs to be considered. Emotion instability, when operationalized as MSSD, assumes even sampling of observations to be computed meaningfully [30,31]. Where this is not possible, adjustments can be applied to the data to provide a weighted estimate of time-structured variability [31]. As in Jahng et al. [31] the following time-adjusted MSSD, which accounts for an uneven sampling of observations through time was applied:

time-adjusted
$$MSSD = \frac{median(\Delta t)}{(n-1)} \sum_{i=1}^{n-1} \frac{(x_{i+1} - x_i)^2}{(t_{i+1} - t_i)},$$

where *median*(Δt) is the median of incremental time differences across the whole recording period. This effectively makes observations closer together in time *more important*, and those further apart *less important* to the reweighted MSSD statistic, relative to a participant's median time increment between posts. Importantly, the time-adjusted MSSD, Eq. (4), reduces to the standard MSSD, Eq. (3), in the case that samples are spaced equally through time. The benefit of this adjustment in relation to social media data is the ability to utilise every observation without imposing strict inclusion criteria on the data (e.g., a status update each day). As illustrated in Figure 1, this means that the overall variability contributed by all points can be included and the potential contribution



from points that may appear temporally correlated, if assumed to have occurred near in time (points C and D), is reduced when observations are in fact distant.

Figure 1. A simulated time-series showing the proportion of negative emotion words used in status updates over 14 days. This irregularity of status updates (i.e., missing observations on days 4-8 above) can be accounted for by reweighting pairs of observations by the time elapsed between them, resulting in a lower weight for the pair of points (points C and D). The observations within the box show similar levels of negative emotion word expression, but occur 6 days apart and may appear to be temporally correlated if their relative temporal distance is not accounted for. The red points show the hypothetical unobserved fluctuations in negative affect that may have occurred during the intermediate 6 days.

Applied to the Facebook and Twitter data for each participant, incremental time differences $(t_{i+1} - t_i)$ between each status update were computed. The median of these time differences was then taken for each participant and applied to each incremental time difference and successive difference $(x_{i+1} - x_i)$ in the proportion of positive or negative emotion words in a status update as in Equation 4. The average of the squared reweighted successive differences was then computed, resulting in the time-adjusted MSSD, or the *positive emotion word instability* or *negative emotion word instability*

across status updates. Here, greater values of time-adjusted MSSD indicate a greater magnitude of change in the proportion of emotion words expressed between all consecutive pairs of status updates relative to their median temporal separation.

Results

In total, 1,856 status updates were collected (Facebook = 538; Twitter = 1,318) with 29,809 words expressed (Facebook = 10,373; Twitter = 19,436). In the Facebook sample participants posted an average 18.55 (SD = 10.01) status updates across the recording period; 55.8% of the collected status updates contained positive emotion words and 29.2% contained negative emotion words. In the Twitter sample participants posted an average 26.90 (SD = 11.71) status updates across the recording period; 63.6% of the collected Tweets contained positive emotion words and 56.2% contained negative emotion words.

Table 2 report the means, standard deviations, median and interquartile range of the *PHQ-9* scores and all Facebook and Twitter variables. It also presents descriptive statistics for the temporal aspects of posting status updates in the sample. Mann-Whitney U tests revealed no significant differences between Facebook and Twitter groups in the length of recording period sampled (U = 590.50, p = .215), though there were differences in the median time difference (U = 344.00, p < .001) and average number of status updates per day (U = 509, p = .038) where Twitter users posted status updates more frequently and, based on their individual median, had smaller intervals in minutes between status updates.

Table 2

Descriptive Statistics of the PHQ-9, Status Update Frequency, and the Emotion Features Expressed

in	Status l	Updates on	Facebook	(n = 29)) and Twitter	(n = 49)	י).
		- p			/	1	

Variable		Facebook			Twitter			
	Range	Mean (SD)	Mdn (IQR)	Range	Mean (SD)	Mdn (IQR)		
Depression severity (<i>PHQ-9</i>) ^a	1 - 22	11.48 (6.38)	10 (5.5–17)	0 - 26	9.80 (6.81)	9 (4–14)		
Status Update Frequency								
Recording period (days) ^b	22-356	170.69 (116.05)	134 (54 – 290)	9 - 365	145.61 (124.97)	74.00 (33.50 – 272.00)		
Status updates per day	.03-1.72	.03 (.36)	.16 (.07 - .51)	.03-4.56	.79 (1.09)	.40 (.0990)		
Interval difference (minutes) between status updates ^c	661 – 34827	8446.65 (8724.25)	3818.00 (1877.75– 13522.75)	4.0 – 28428.5	3939.79 (6616.84)	1037 (206.25 – 4571.25)		
Positive Emotion Words								
Average proportion	.0257	.10 (.10)	.08 (.05– .11)	.0114	.07 (.03)	.08 (.05 –.09)		
Variability (<i>iSD</i>) ^d	.0447	.13 (.09)	.10 (.07– .16)	.0317	.07 (.03)	.08 (.05– .09)		
Instability (time- adjusted <i>MSSD</i>) ^e	0.003 – 11.54	1.14 (2.94)	.11 (.02– .47)	0.0002 - 26.80	1.49 (4.40)	.12 (.02–.83)		
Negative Emotion Words								
Average proportion	.0017	.04 (.04)	.02 (.01– .05)	.0126	.09 (.06)	.09 (.04–.12)		
Variability (<i>iSD</i>) ^d	.0031	.07 (.08)	.03 (.02– .09)	.0214	.08 (.03)	.08 (.06–.11)		
Instability (time- adjusted <i>MSSD</i>) ^e	0.00 - 1.23	0.11 (0.24)	.01 (.002– .14)	0.0006 – 37.99	1.31 (5.43)	.15 (.03–.49)		

^a*PHQ-9* = Patient Health Questionnaire– 9

^b Recording period refers to the range of days between the first status update collected and the administration of the PHQ-9

^c The median interval differences between status updates

^d iSD = within-person variability

^e time-adjusted *MSSD* = mean squared successive differences.

Tables 3 and 4 respectively report the two-tailed Spearman correlations (alpha level = .05) between the PHQ-9 and the positive emotion variables and the negative emotion variables from Facebook (above the diagonal) and Twitter (below the diagonal).

Table 3

Spearman's rho Correlation Analyses between Depression Severity (as Rated by the PHQ-9) and the Positive Emotion Features Expressed in Status Updates on Facebook (n = 29) and Twitter (n = 49).^{a, b}

Variable	1.	2.	3.	4.
1. PHQ-9°		.04 [33, .40]	.17 [21, .51]	04 [40, .33]
2. Average proportion	.02		.79**	.48*
	[26, .30]		[.60, .90]	[.14, .72]
3. Variability (<i>iSD</i>) ^d	09	.49**		.61**
	[36, .20]	[.24, .68]		[.31, .80]
4. Instability (time- adjusted <i>MSSD</i>) ^e	20 [46, .09]	.31* [.03, .54]	.48** [.23, .67]	

^a Twitter correlations are shown *below* the diagonal; Facebook correlations are shown *above* the

diagonal

^b Confidence intervals are reported at 95% and are presented in brackets.

^c *PHQ-9* = Patient Health Questionnaire– 9

^d *iSD* = within-person variability

^e time-adjusted *MSSD* = mean squared successive differences.

p = p < .05; p < .001.

Table 4

Spearman's rho Correlation Analyses between Depression Severity (as Rated by the PHQ-9) and the Negative Emotion Features Expressed in Status Updates on Facebook (n = 29) and Twitter (n = 49).^{a, b}

Variable	1.	2.	3.	4.
1. <i>PHQ-9</i> ^b		.12	.20	.44*
2. Average proportion	14	[26, .46]	[18, .53] .95**	[.09, .69] .72*
	[41, .15]		[.90, .98]	[.48, .86]
3. Variability (<i>iSD</i>) ^c	36*	.57**		.82**
	[58, .09]	[.34, .73]		[.65, .91]
4. Instability (time- adjusted MSSD) ^d	20 [46, .09]	.28 [001, .52]	.49** [.24, .68]	

^a Twitter correlations are shown *below* the diagonal; Facebook correlations are shown *above* the

diagonal

^b Confidence intervals are reported at 95% and are presented in brackets.

^c *PHQ-9* = Patient Health Questionnaire– 9

^d *iSD* = within-person variability

^e time-adjusted *MSSD* = mean squared successive differences.

* = p < .05; ** = p < .001.

Facebook Emotion Variability and Depression

Facebook users reported an average depression rating of 11.48 (*SD* = 6.38) on the *PHQ-9* and expressed 9.5% positive emotion words and 3.5% negative emotion words on average across their status updates. Depression severity was not significantly related to the average proportion of positive or negative emotion words expressed, positive or negative emotion word variability (*iSD*), or positive emotion word instability (time-adjusted MSSD). Negative emotion word instability did, however, show a significant positive association with depression severity ratings, sharing 19% of the variability. This indicates that successive status updates differed more in their proportion of negative emotion words used for individuals with higher self-reported depression symptoms.

When controlling for the average proportion of negative emotion words expressed in status updates, negative emotion word instability remained strongly associated with depression severity (partial Spearman correlation: $r_s(26) = .51$, p = .006). Similarly, when controlling for negative emotion word variability, negative emotion word instability remained moderately associated with depression severity ($r_s(26) = .49$, p = .009). To illustrate this effect, Figure 2 shows samples of the pattern of negative emotion word instability from two participants; one with low (2a) and one with high (2b) self-reported depression symptoms. As can be seen in Figure 2a, the individual with low depression severity shows small magnitude changes in their use of negative emotion words across status updates. In contrast, the individual in Figure 2b with high depression severity exhibits greater magnitude spikes in negative emotion word expression.



Figure 2. Graphs showing the proportion of negative emotion words used in individual status updates on Facebook across 35 days. (a) Shows an individual with low self-reported depression severity (PHQ-9 score = 9) who demonstrated little post-to-post variation in the proportion of negative emotion words used, with the maximum difference of .03. The horizontal trend line shows the median proportion of negative emotion words used (.022) and interpolation lines link consecutive status updates. (b) Shows an individual with high self-reported depression severity (PHQ-9 score = 22), who demonstrates large post-to-post changes in the proportion of negative emotion words used in status updates with the largest difference being .21. The horizontal trend line shows the median proportion words used (.01) and interpolation lines link consecutive status updates. Here, instability is independent of variability. See Multimedia Appendix 3 for consideration of instability under fixed variability conditions.

Twitter Emotion Variability and Depression

Twitter users reported an average depression rating of 9.80 (SD = 6.81) on the *PHQ-9* and expressed 7.4% positive emotion words and 9.2% negative emotion words on average across their status updates. Depression severity was not significantly related to the average proportion of positive or negative emotion words expressed, positive emotion word variability (*iSD*), or positive or negative emotion word instability (time-adjusted MSSD). Negative emotion word variability, however, was significantly negatively associated with depression severity ratings, sharing 13% of the variability. That is, a greater general dispersion of negative emotion across status updates on Twitter was associated with lower depression severity. When controlling for the average proportion of negative emotion words expressed in status updates, negative emotion word variability retained its association with depression severity in a partial Spearman correlation $r_s(46) = -.35$, p = .014.

To illustrate this effect, Figure 3 shows samples of the pattern of negative emotion word variability from two participants; one with low (3a) and one with high (3b) self-reported depression symptoms. As can be seen in Figure 3a, the individual with low depression severity shows larger overall variability in their use of negative emotion words across status updates. In contrast, the individual in Figure 3b with high depression severity exhibits more restricted variability in negative emotion word expression. It is important to note that in Figure 3 Tweets often occurred on the same day which accounts for the clustering in the figure.



Figure 3. Graphs showing the proportion of negative emotion words used in individual status updates across (a) 160 and (b) 182 days. (a) Shows an individual with low self-reported depression severity (PHQ-9 score = 8) and high variability in the proportion of negative emotion words used across their recording period. The horizontal trend line shows the median proportion of negative emotion words (.17) and interpolation line links status updates. (b) Shows an individual with high self-reported depression severity (PHQ-9 score = 16) and low variability in the proportion of negative emotion words used across their recording period. The median proportion of negative emotion words used across their recording period. The status updates. (b) Shows an individual with high self-reported depression severity (PHQ-9 score = 16) and low variability in the proportion of negative emotion words used across their recording period. The median proportion of negative words used was .00 and is therefore not shown. The interpolation line links status updates.

Facebook and Twitter Status Update Frequency and Depression

Descriptive statistics for the average number of status updates per day and the median time interval between status updates are presented in Table 2. Spearman correlations revealed a significant positive association between the average number of status updates per day and depression severity for Facebook users, $r_s(29) = .48$, p = .008. There was also a significant negative association between the median time interval between status updates and depression severity for Facebook users, $r_s(29) = .61$, p < .001. Depression severity was not significantly related to the average number of status updates per day or the median interval between status updates for Twitter users.

Differences in Emotion Language Patterns

As indicated in Tables 3 and 4, the pattern of relationships between depression and emotion language use varied between Facebook and Twitter users. To explore this further, comparisons of the social media emotion language variables between the samples were conducted. As all variables were non-normally distributed Mann-Whitney U tests were used to compare the mean rank differences between Facebook and Twitter users in their emotion language patterns. The Twitter sample expressed more negative language (U = 256.00, p < .001) that was more variable (U =400.00, p = .001) and unstable (U = 379.00, p = .001) than did the Facebook group across the recording period. Twitter users also expressed greater variability in their positive emotion compared to Facebook users (U = 413.00, p = .002).

Discussion

This study aimed to determine whether emotion variability and instability across status updates on Facebook and Twitter are useful indicators of depression. Differences between the social media platforms were also explored. The findings suggest that instability in the negative emotion content across Facebook status updates may indeed be a useful indicator for depression, and that the time-adjusted MSSD is an effective index of instability that accounts for the uneven temporal sampling of social media posts. As hypothesised, negative emotion word instability retained its association with depression severity when the average proportion of negative emotion word use and negative emotion word variability were controlled. This index may provide additional sensitivity over basic frequency indices that are typically used in social media and depression studies. However, negative emotion word instability did not emerge as a predictor of depression on Twitter. Rather, in contrast to expectations, negative emotion word variability was negatively associated with depression severity. Further, the average proportions of negative and positive emotion word use were not significantly associated with depression severity on either Facebook or Twitter. Other temporal features, the average number of status updates per day and the median time-interval between status updates were also associated with depression severity, but only for Facebook users.

Negative Affect Instability on Facebook

Greater negative emotion word instability on Facebook was associated with individuals experiencing greater depression severity. The time-adjusted MSSD scores were driven by the pattern of frequent, high magnitude changes in negative emotion word use between status updates, not variability alone. This finding is consistent with previous studies measuring negative affect through self-report over time that have demonstrated negative affect instability to be predictive of depression [25,31,33–36].

Many users post on Facebook to broadcast emotion [14], and emotion words are often present in posts [15]. Individuals with depression are more likely to produce more content on social media when experiencing more severe symptoms [16,17] and this often relates to the disclosure of symptoms, negative experiences, or posting to seek social support [3,8,10,11]. Indeed, this was reflected in the current sample where Facebook users with greater depression severity ratings posted more status updates per day, more frequently (i.e. there was a smaller median time-interval between consecutive status updates). Large changes in the proportion of negative emotion words used between consecutive status updates could reflect patterns of Facebook use that mirror the inherent

variation in the severity of depression symptoms over time. In this way, the negative emotion word instability in the status updates on Facebook may reveal the ebb and flow of depression symptoms and emotion dysregulation in daily life [36].

Negative emotion word instability on Facebook may also be tied to specific events, capturing momentary responses to internal and external stressors. Individuals exposed to an emotional event generally post status updates in a mood congruent way (e.g., happy or sad) [54]. The proportion of negative emotion words used in a status update may reflect the extremity with which an event is perceived as negative or positive. In this light, status updates could provide insight into emotional reactivity to events. A depression-specific pattern of instability in status update expression on Facebook may exist that reflects the amplification of negative emotion in response to ambiguous or negative events [55] and a mood brightening effect in response to positive stimuli where there is a large reduction in expressed negative affect [36,56].

The unique fluctuating pattern of negative emotion expression in individuals with more severe depression symptoms was further supported by negative emotion word instability, which remained associated with greater depression symptoms when controlling for the average proportion of negative emotion words used in status updates. This suggests that the time-structured patterns of emotion expressed on Facebook may provide better differentiation between individuals with and without depression where they express similar levels of negative emotion words. The current study suggests that the poor hit rate in some keyword approaches to classifying depression in status updates, as described by Mowery et al. [18] may be enhanced by including measures of moment-tomoment variability in emotion word use.

Within Person Variability in Emotional Expression on Twitter

Contrary to expectations, Twitter users who had lower variability in their use of negative emotion words across the recording period were more likely to have greater self-reported depression severity. This sits in contrast with a recent meta-analysis that showed negative emotion variability shares a positive association with depression [1].

It could be that the greater variability in emotion expressed by individuals lower in depression on Twitter reflects adaptive emotional functioning. In addition to personal disclosures and using Twitter to talk about daily events [57], people turn to Twitter to post content about politics, world events, and to share information [58]. Expressing a wide range of negative emotion in response to these diverse personal and community related events may be appropriate to the context or be a part of effective emotion regulation strategies. Indeed, expressive emotional writing has been linked to better psychological and physical outcomes in offline and online settings [59–62].

On the other hand, Twitter users with higher levels of depression expressed a more clustered spread of negative emotion. Emotion appraisals of internal and external events and their subsequent expression in status updates may be more restricted or blunted for Twitter users with higher levels of depression. This is consistent with studies indicating that MDD is associated with reduced emotion reactivity [63].

Variability Differences between Facebook and Twitter

Two divergent emotion patterns relating to depression emerged from the Facebook and Twitter samples. This highlights the importance of collecting data from multiple social media platforms, as differences in the communication mechanisms and population demographics across social media sites greatly impacts on the generalisability of findings [49]. In terms of emotion expression, Twitter users expressed more negative emotion that was more variable across the recording period than Facebook users. This may be due to the 140-character restriction placed on Tweets (recently increased to 280-characters) [64] compared to the 63,206-character limit on Facebook [65] which may impact on the total proportion of emotion words expressed and the magnitude of change observed between posts. On Twitter, when an emotion word is used it is likely to occur in the context of fewer total words and will result in a greater proportion emotion expressed per Tweet. In contrast, when a Facebook user expresses emotion it may occur in the context of more total words, potentially reducing the overall proportion of emotion words expressed.

Other confounding variables may also create differences between negative emotion expression on Facebook and Twitter. For example, Twitter allows users to generate anonymous accounts, whereas Facebook accounts are likely to be linked to a real name. The anonymity may release the user from social norms and increase expression of negative emotion [66,67]. Twitter also is less symmetrical, with weaker relational ties, and less dense network structures, which impacts on the emotion people express to their networks [68,69]. These different social contexts and related norms are a fruitful area for future research.

Averages of Negative and Positive Emotion Word Use Are Not Associated with Depression

Inconsistent with many previous findings [2,3,5,9], the average proportion of positive and negative emotion words used across status updates on both Facebook and Twitter were not significantly associated with depression. Other approaches using the *LIWC 2007* positive and negative dictionaries have found that as negative emotion expression increases, so does the ratings of self-reported depression severity (e.g., [5]). This could be due in part to the small sample used here; language is noisy [43], and with only 29 and 49 participants in the Facebook and Twitter samples respectively, the signal may not be enough to counteract that noise (see Kern et al. [43] for further consideration of language and sample size considerations). Amongst this noise, it is notable with the small number of participants that a robust association between negative affect instability and depression on Facebook was found, suggesting a strong relationship between these two quantities. While this result needs to be replicated in other samples, this suggests that when a smaller number of participants are available, instability may be a more sensitive measure than frequency in detecting depression severity.

The null findings between depression and the average proportion of words in status updates may also reflect the lack of precision that frequency measures provide. As shown by Mowery et al. [18], using a keyword approach to identifying depression in social media posts results in a large proportion of false-positives, reducing the specificity with which depression can be identified through the average emotion expressed over time. Context matters [43], such that the use of a word may not directly link to an experienced emotion (e.g., "I went to visit Happy Valley" does not indicate positive emotion). It is important to acknowledge also that negative emotion expression is not the exclusive domain of individuals with higher levels of depressive symptoms. It is also possible that in this study, the amended negative emotion word dictionary of the *LIWC 2007* alone was not sufficient in identifying the words most indicative of depression. Indeed, the dictionaries were recently updated [70], and future studies should examine whether the updated *LIWC 2015* dictionaries offer a better indication of depression. Personality, gender, and age have all previously been shown to impact on the number of negative emotion words people use online (cf. [71]) and this complexity might also be considered in future research.

Limitations and Future Directions

There are several limitations to the current study. Firstly, while emotion scores were calculated, the actual posts were not available (due to privacy considerations), such that the context of their content could not be considered. It is therefore possible that some posts may have obtained a negative emotion word count where a positive message was conveyed. Future research should seek to apply more sophisticated open vocabulary approaches or postprocessing of status updates [18,43] to provide greater detail and accuracy of the language use context.

Secondly, the sample analysed was small and this may have impacted on the power to detect significant associations between variables. This may have obscured potential associations between the expression of negative emotion words and depression severity. It is also likely that, due to

sample size, the findings obtained here may not be generalisable to the Facebook and Twitter populations. Replication is required in larger samples.

Third, only original posts were used, with retweets or shared posts excluded. While retweets may be an indirect indication of a person's emotions, beliefs, values, and behaviours, the intentions underlying reposting are unclear. Further, at the time of data collection, reposting updates was less common in Facebook, so excluding retweets provided a clearer comparison. Future studies might explore the extent to which reposts (retweets and the sharing of posts) reflect a user's values and emotions and indicate depression status.

It is also important to note that it was unknown if the emotion words expressed on Facebook or Twitter accurately reflected same-day subjective changes in mood. Further research should seek to link consecutively measured mood ratings with social media data to strengthen the assumption that interpretation of social media content reflects real-world emotion experience.

Finally, studies should seek to explicitly consider inertia in the emotion expressed in status updates as a predictor of depression and consider how the sensitivity and accuracy of frequency and instability metrics changes across different sample sizes. Such analyses will, however, require adjustments be made to calculations to account the sparseness and irregularity of social media data.

Conclusion

The current study suggests that instability in the negative emotion expressed on Facebook provides insight into the presence of depression symptoms for social media users, and greater variability of negative emotion expression on Twitter may be protective for mental health. These findings provide proof-of-concept that temporally measures of emotion language in social media posts provide a sensitive and specific index of depression. If replicated in other samples, emotion dynamics might be applied to big data approaches for depression screening at a population level, providing insight into the emotion processes underlying depression and improving the specificity of depression identification above using language averages alone. The time-adjusted MSSD

DEPRESSION AND LANGUAGE-BASED EMOTION DYNAMICS ON SOCIAL MEDIA

appropriately accounts for the uneven temporal sampling of real-world social media data, providing a sensitive measure of emotion instability that may be used as an early indicator of (or identified as a risk factor for) depression. Variability is often seen as a nuisance factor that creates noise and obscures other associations. Treating emotion variability as a legitimate individual difference may be an important step in better describing the micro-processes that lead to psychopathology. The findings also point to possible differences across the online culture created by a particular social media platform, such that different platforms may provide different insights into mental health.

The widespread and frequent use of social media has generated considerable concern around its impact on mental health. Yet social media is also revealing itself to be a valuable avenue for the ongoing monitoring of depression. This study contributes to understanding the best approaches for using the technology to help users suffering from depression.

Acknowledgements

We thank Mehmet Özmen for his statistical advice in this study. Elizabeth M. Seabrook is a recipient of an Australian Government Research Training Program Scholarship.

Conflicts of Interest

None declared.

Abbreviations

MDD: Major Depressive Disorder
MSSD: Mean squared successive differences
iSD: within-person standard deviation
PHQ-9: Patient Health Questionnaire 9
LIWC 2007: Linguistic Inquiry and Word Count 2007
RSES: Rosenberg Self-Esteem Scale
M-C Form C: Marlowe-Crowne Social Desirability Scale Form C
Multimedia Appendix 1 [Emoji and internet slang supplements to the LIWC 2007 dictionaries]
Multimedia Appendix 2 [Supplementary Analyses: Personality, Self-esteem, and Social Desirability]
Multimedia Appendix 3 [Illustrative participant data showing instability at fixed levels of

variability]

References

- Houben M, Van Den Noortgate W, Kuppens P. The relation between short-term emotion dynamics and psychological well-being: A meta-analysis. *Psychol Bull*. 2015;141(4):901-930.
 PMID: 25822133.
- De Choudhury M, Counts S, Horvitz E. Social media as a measurement tool of depression in populations. In: *Proceedings of the 5th Annual ACM Web Science Conference*. ACM; 2013:47–56. doi: 10.1145/2464464.2464480.
- Moreno MA, Christakis DA, Egan KG, et al. A pilot evaluation of associations between displayed depression references on Facebook and self-reported depression using a clinical scale. J Behav Health Serv Res. 2012;39(3):295–304. PMID: 21863354.
- Schwartz HA, Eichstaedt J, Kern ML, et al. Towards assessing changes in degree of depression through Facebook. In: *Workshop on Computational Linguistics and Clinical Psychology: From Linguistic Signal to Clinical Reality*. Baltimore, Maryland, USA: Association for Computational Linguistics; 2014:118-125.
- 5. Settanni M, Marengo D. Sharing feelings online: studying emotional well-being via automated text analysis of Facebook posts. *Front Psychol*. 2015;6. PMID: 26257692.
- Guntuku SC, Yaden DB, Kern ML, Ungar LH, Eichstaedt JC. Detecting depression and mental illness on social media: an integrative review. *Curr Opin Behav Sci*. 2017;18(Supplement C):43-49. doi:10.1016/j.cobeha.2017.07.005
- American Psychiatric Association. Depressive Disorders. In: *Diagnostic and Statistical Manual of Mental Disorders*. DSM Library. American Psychiatric Association; 2013. doi: 10.1176/appi.books.9780890425596.dsm04.

- 8. Cavazos-Rehg PA, Krauss MJ, Sowles S, et al. A content analysis of depression-related tweets. *Comput Hum Behav.* 2016;54:351-357. PMID: 26392678.
- Locatelli SM, Kluwe K, Bryant FB. Facebook use and the tendency to ruminate among college students: Testing mediational hypotheses. *J Educ Comput Res*. 2012;46(4):377-394. doi:10.2190/EC.46.4.d.
- Moreno MA, Jelenchick LA, Egan KG, et al. Feeling bad on Facebook: Depression disclosures by college students on a social networking site. *Depress Anxiety*. 2011;28(6):447-455. PMID: 21400639.
- Park J, Lee DS, Shablack H, et al. When perceptions defy reality: The relationships between depression and actual and perceived Facebook social support. *J Affect Disord*. 2016;200:37-44. PMID: 27126138.
- Bollen J, Gonçalves B, Ruan G, Mao H. Happiness is assortative in online social networks. *Artif Life*. 2011;17(3):237–251. PMID: 21554117.
- Schwartz HA, Eichstaedt JC, Kern ML, et al. Characterizing geographic variation in wellbeing using Tweets. 2013 Presented at: *Proceedings of the Seventh International AAAI Conference on Weblogs and Social Media (ICWSM)*; 2013; Boston, MA.
- Manago AM, Taylor T, Greenfield PM. Me and my 400 friends: The anatomy of college students' Facebook networks, their communication patterns, and well-being. *Dev Psychol*. 2012;48(2):369-380. PMID: 22288367.
- Kramer AD. The spread of emotion via Facebook. In: *Proceedings of the SIGCHI Conference* on Human Factors in Computing Systems. ACM; 2012:767–770. doi: 10.1145/2207676.2207787.

- Shaw AM, Timpano KR, Tran TB, Joormann J. Correlates of Facebook usage patterns: The relationship between passive Facebook use, social anxiety symptoms, and brooding. *Comput Hum Behav.* 2015;48:575-580. doi:10.1016/j.chb.2015.02.003
- Park S, Kim I, Lee SW, Yoo J, Jeong B, Cha M. Manifestation of depression and loneliness on social networks: a case study of young adults on Facebook. In: *Proceedings of the 18th ACM Conference on Computer Supported Cooperative Work & Social Computing*. ACM; 2015; Vancouver, Canada p. 557–570. doi: 10.1145/2675133.2675139
- Mowery D, Smith H, Cheney T, et al. Understanding depressive symptoms and psychosocial stressors on Twitter: A Corpus-Based Study. *J Med Internet Res*. 2017;19(2):e48. PMID: 28246066.
- 19. Kuppens P, Verduyn P. Emotion dynamics. Curr Opin Psychol. in press.
- 20. Shapero BG, Black SK, Liu RT, et al. Stressful life events and depression symptoms: the effect of childhood emotional abuse on stress reactivity. *J Clin Psychol*. 2014;70(3):209-223. PMID: 23800893.
- Wichers M, Geschwind N, Jacobs N, et al. Transition from stress sensitivity to a depressive state: longitudinal twin study. *Br J Psychiatry*. 2009;195(6):498-503. doi:10.1192/bjp.bp.108.056853.
- Wichers M, Jacobs N, Derom C, Thiery E, van Os J. Depression: Too much negative affect or too little positive affect? *Twin Res Hum Genet*. 2007;10(S1):19-20. doi:10.1375/twin.10.supp.19.
- Dundon E. Adolescent depression: A metasynthesis. *J Pediatr Health Care*. 2006;20(6):384-392. doi:10.1016/j.pedhc.2006.02.010.

- Nolen-Hoeksema S, Wisco BE, Lyubomirsky S. Rethinking rumination. *Perspect Psychol Sci.* 2008;3(5):400–424. PMID: 26158958.
- Thompson RJ, Berenbaum H, Bredemeier K. Cross-sectional and longitudinal relations between affective instability and depression. *J Affect Disord*. 2011;130(1-2):53-59. PMID: 20951438.
- Bradley B, DeFife JA, Guarnaccia C, et al. Emotion dysregulation and negative affect: association with psychiatric symptoms. *J Clin Psychiatry*. 2011;72(5):685-691. PMID: 21658350.
- 27. Gross JJ, Muñoz RF. Emotion regulation and mental health. *Clin Psychol Sci Pract*.
 1995;2(2):151-164. doi:10.1111/j.1468-2850.1995.tb00036.x.
- Wichers M. The dynamic nature of depression: a new micro-level perspective of mental disorder that meets current challenges. *Psychol Med.* 2014;44(7):1349-1360. PMID: 23942140.
- Ram N, Gerstorf D. Time-structured and net intraindividual variability: tools for examining the development of dynamic characteristics and processes. *Psychol Aging*. 2009;24(4):778-791. PMID: 20025395.
- Von Neumann J, Kent RH, Bellinson HR, Hart BI. The mean square successive difference. Ann Math Stat. 1941;12(2):153-162. doi:10.1214/aoms/1177731746.
- Jahng S, Wood PK, Trull TJ. Analysis of affective instability in ecological momentary assessment: Indices using successive difference and group comparison via multilevel modeling. *Psychol Methods*. 2008;13(4):354-375. PMID: 19071999.

- 32. Koval P, Pe ML, Meers K, Kuppens P. Affect dynamics in relation to depressive symptoms: Variable, unstable or inert? *Emotion*. 2013;13(6):1132-41. PMID: 23914765.
- Bowen R, Peters E, Marwaha S, Baetz M, Balbuena L. Moods in clinical depression are more unstable than severe normal sadness. *Front Psychiatry*. 2017;8. PMID: 28446884.
- Gershon A, Eidelman P. Affective intensity and instability: predictors of depression and functional impairment in bipolar disorder. *J Behav Ther Exp Psychiatry*. 2015;46:14-18. doi:10.1016/j.jbtep.2014.07.005.
- 35. Palmier-Claus JE, Taylor PJ, Gooding P, Dunn G, Lewis SW. Affective variability predicts suicidal ideation in individuals at ultra-high risk of developing psychosis: An experience sampling study. *Br J Clin Psychol*. 2012;51(1):72-83. PMID: 22268542.
- 36. Thompson RJ, Mata J, Jaeggi SM, Buschkuehl M, Jonides J, Gotlib IH. The everyday emotional experience of adults with major depressive disorder: examining emotional instability, inertia, and reactivity. *J Abnorm Psychol*. 2012;121(4):819-829. PMID: 22708886.
- 37. Peeters F, Berkhof J, Rottenberg J, Nicolson NA. Ambulatory emotional reactivity to negative daily life events predicts remission from major depressive disorder. *Behav Res Ther*. 2010;48(8):754-760. PMID: 20537317.
- Koval P, Sutterlin S, Kuppens P. Emotional inertia is associated with lower well-being when controlling for differences in emotional context. Frontiers in Psychology. 2016:8(6). PMID: 26779099.
- Kuppens P, Sheeber LB, Yap MBH, Whittle S, Simmons JG, Allen NB. Emotional inertia prospectively predicts the onset of depressive disorder in adolescence. *Emotion*. 2012;12(2):283-289. PMID: 21988744.

- Larson R, Csikszentmihalyi M. The experience sampling method. In: *Flow and the Foundations of Positive Psychology*. Dordrecht: Springer Netherlands; 2014:21-34. doi: 10.1007/978-94-017-9088-8_2.
- 41. Miller G. The smartphone psychology manifesto. *Perspect Psychol Sci.* 2012;7(3):221-237.
 PMID: 26168460.
- Rickard N, Arjmand H-A, Bakker D, Seabrook E. Development of a mobile phone app to support self-monitoring of emotional well-being: a mental health digital innovation. *JMIR Ment Health*. 2016;3(4):e49. PMID: 27881358.
- 43. Kern ML, Park G, Eichstaedt JC, et al. Gaining insights from social media language: Methodologies and challenges. *Psychol Methods*. 2016;21(4):507-525. PMID: 27505683.
- 44. Schwartz HA, Ungar LH. Data-driven content analysis of social media: a systematic overview of automated methods. *Ann Am Acad Pol Soc Sci.* 2015;659(1):78-94. doi:10.1177/0002716215569197.
- 45. Larsen ME, Boonstra TW, Batterham PJ, O'Dea B, Paris C, Christensen H. We Feel: mapping emotion on Twitter. *IEEE J Biomed Health Inform*. 2015;19(4):1246-1252. doi:10.1109/JBHI.2015.2403839.
- De Choudhury M, Kiciman E, Dredze M, Coppersmith G, Kumar M. Discovering shifts to suicidal ideation from mental health content in social media. In: ACM Press; 2016:2098-2110. doi:10.1145/2858036.2858207.
- 47. Kroenke K, Spitzer RL, Williams JBW. The PHQ-9. J Gen Intern Med. 2001;16(9):606-613.
 PMID: 11556941.

- Geva H, Oestreicher-Singer G, Saar-Tsechansky M. Using retweets to shape our online persona: A topic modeling approach. Rochester, NY: Social Science Research Network; 2016. doi: 10.2139/ssrn.2759811
- 49. Tufekci Z. Big questions for social media big data: Representativeness, validity and other methodological pitfalls. In *Proceedings of the 8th International Conference on Weblogs and Social Media, 1CWSM.* 2014:505-514. The AAAI Press.
- boyd d, Golder S, Lotan G. Tweet, tweet, retweet: Conversational aspects of retweeting on twitter. In: *System Sciences (HICSS), 2010 43rd Hawaii International Conference on*. IEEE; 2010:1–10. doi: 10.1109/HICSS.2010.412
- Kocalevent R-D, Hinz A, Brähler E. Standardization of the depression screener patient health questionnaire (PHQ-9) in the general population. *Gen Hosp Psychiatry*. 2013;35(5):551-555.
 PMID: 23664569.
- Pennebaker JW, Chung CK, Ireland M, Gonzales A, Booth RJ. The LIWC 2007 Application.
 2007. http://www.liwc.net.
- 53. IBM Corp. IBM SPSS Statistics for Windows. Armonk, NY: IBM Corp.; 2016.
- 54. Kreitler CM, Stenmark CK, Serrate J, Winn N. The role of individual differences and emotion in Facebook activity. *J Psychol Behav Sci.* 2016;4(1). doi:10.15640/jpbs.v4n1a1.
- Gotlib IH, Joormann J. Cognition and depression: Current status and future directions. *Annu Rev Clin Psychol*. 2010;6:285-312. PMID: 20192795.
- Bylsma LM, Taylor-Clift A, Rottenberg J. Emotional reactivity to daily events in major and minor depression. *J Abnorm Psychol*. 2011;120(1):155-167. PMID: 21319928.

- 57. Java A, Song X, Finin T, Tseng B. Why We Twitter: An analysis of a microblogging community. In: *Advances in Web Mining and Web Usage Analysis*. Lecture Notes in Computer Science. Springer, Berlin, Heidelberg; 2009:118-138. doi: 10.1145/1348549.1348556.
- Park CS. Does Twitter motivate involvement in politics? Tweeting, opinion leadership, and political engagement. *Comput Hum Behav*. 2013;29:1641–1648. doi:10.1016/j.chb.2013.01.044.
- 59. Baikie KA, Wilhelm K. Emotional and physical health benefits of expressive writing. *Adv Psychiatr Treat*. 2005;11(5):338-346. doi:10.1192/apt.11.5.338.
- 60. Smyth JM. Written emotional expression: effect sizes, outcome types, and moderating variables. *J Consult Clin Psychol*. 1998;66(1):174-184. PMID: 9489272.
- 61. Pennebaker JW. Theories, therapies, and taxpayers: On the complexities of the expressive writing paradigm. *Clin Psychol Sci Pract*. 2004;11(2):138-142. doi:10.1093/clipsy.bph063.
- 62. Baker JR, Moore SM. Blogging as a social tool: A psychosocial examination of the effects of blogging. *Cyberpsychol Behav*. 2008;11(6):747-749. PMID: 19072151.
- 63. Bylsma LM, Morris BH, Rottenberg J. A meta-analysis of emotional reactivity in major depressive disorder. *Clin Psychol Rev.* 2008;28(4):676-691. doi:10.1016/j.cpr.2007.10.001.
- 64. Rosen A, Ihara I. Twitter. *Giving you more characters to express yourself*. September 2017. https://blog.twitter.com/official/en_us/topics/product/2017/Giving-you-more-characters-to-express-yourself.html. Archived at: http://www.webcitation.org/6tm7Re3Bg
- 65. Taylor C. Mashable. Your new Facebook status: 63,206 characters or less. January 2011. http://mashable.com/2011/11/30/facebook-status-63206-characters/#oOlxKAXmjSqu. Archived at: //www.webcitation.org/6uPxF2hcd
- 66. Peddinti ST, Ross KW, Cappos J. "On the Internet, nobody knows you're a dog": A Twitter case study of anonymity in social networks. In: *Proceedings of the Second ACM Conference on Online Social Networks*.2014. ACM; 2014:83–94. doi:10.1145/2660460.2660467
- 67. Correa D, Silva LA, Mondal M, Benevenuto F, Gummadi KP. The many shades of anonymity: characterizing anonymous social media content. In: *Proceeding of the Ninth International Conference on Web and Social Media ,ICWSM*. 2015:71–80. Retrieved from https://www.aaai.org/ocs/index.php/ICWSM/ICWSM15/paper/view/10596
- Lerman K, Arora M, Gallegos L, Kumaraguru P, Garcia D. Emotions, demographics and sociability in twitter interactions. *ArXiv151007090 Phys.* October 2015. http://arxiv.org/abs/1510.07090.
- 69. Myers SA, Sharma A, Gupta P, Lin J. Information network or social network?: The structure of the twitter follow graph. In: ACM Press; 2014:493-498. doi:10.1145/2567948.2576939
- Pennebaker JW, Boyd RL, Jordan K, Blackburn K. *The Development and Psychometric Properties of LIWC2015*.; 2015. Austin, TX: University of Texas at Austin.
- Schwartz HA, Eichstaedt JC, Kern ML, et al. Personality, gender, and age in the language of social media: The open-vocabulary approach. *PLoS ONE*. 2013;8(9):e73791. PMID: 24086296.

Positive	Negative	e Emoji's	Positi	ve Emotio	n Negative Emotion Inter			
Emoji s	. 1	. /	Aisht	rnet Slang	1	Sla	forme	
=D>	1	-/	Aight	J/J :1-	Туке	Abit		
=D	.1 V(:/ T T	Aight	JK :/1-	marvy	Acgai	galik amah 1	
			Aita	J/K :1-0	njoy ~1	Anc	grindi 1.0	
:-D	X-(1/1	Alte	JKS	qi	Arsea	n8 1-90	
(:)		QQ	Airt	J2I :46	qool	AWK	180r	
(:-)	в (=0::::	alryt	J41	qt	Bm	h83r	
>:D<	>:-(= (Alol	J4g	qte	Boom	h8ed	
:))	>:-)	&.(Ami	J41	rofl*	Bord	h8r	
:-X	(:-&	(;_;)	Aprece	JIg	*S*	Bovered	hored	
:X	:@	:(Apreci8	Jfl	S´ok	Catwot	h8s h8t	
:)	:(÷	Awes	jfk	Snog	Cba	h8t0r	
:-)	:-t	:'-(Awsm	jja	Sok	Cbb	h8t3r	
~~	:-11	:*(Awsome	JKZ	Sof	Cbf	h8te	
:,')	:11	:*-(Bahaha	k	Stm	Cbfa	h8tr	
:-")	:t	:_(Besos	kay	Ub3r	Cbfed	hait	
:")	:-@	:[Bilu	kewel	W00t	Cmeo	heyt	
;)	:{	;_;	Btwilu	kewl	Woot	Cmw	htr	
;-)	:-Z	=\	Btwitiailwu	kk	Xellent	Cof	id10d	
O/	:Z	:1	Byak	kl	Xlnt	Cotf	idjit	
;-(??	:(&	Chillin	koo	xtc	Dc	idot	
;(@_@	:-E	Ctm	kool		Dbm	idyat	
:')	(*)	:-e	Cut3	kss		Dfc	invu	
:'-)	(0_0)	DX	Eil kssd		Dgac	irhy		
>-)	b-(/:(Fab	ab kul		Dgaf	isb	
:p	%+l	:-[Fah	kute		Dgara	kmn	
:P	:S	:[Fi9	Fi9 101z		Dgas	kmp	
:-P	:-S	:-}	Funee laff		Eejit	lsr		
:-p	(:-	:}	Funy	lal		Eedyat	lzr	
:>	:(#-)	*g*	lol		Ef	nv	
:->	:-(#)	Gtm	lols		Ef-ing	ofcol	
11	1-o	:-6	Gud	lel		Effed	oh noes	
^.^	\-o	:6	Gudd	liek*		Effin	oh noez	
^?^	:-0	:-@	Heh	lik		Effing	p/oed	
^_^	=(((:(Нрру	lk		F#cking	poed	
%)	+-((:-<	Нру	lke		F'n	ph33r	
%-)	3</td <td>><</td> <td>Haha</td> <td>llc</td> <td></td> <td>f-ing</td> <td>phayl</td>	><	Haha	llc		f-ing	phayl	
<3	(U)	://	I8	llf		fml	phail	
<u3< td=""><td>:-c</td><td></td><td>Iatb</td><td>lmao</td><td></td><td>f.m.l</td><td>r8p</td></u3<>	:-c		Iatb	lmao		f.m.l	r8p	
	:c		light	lmfao		farg	r8pist	
	O.0		ligh	rofl		fck	stupd	
	%)		Ijdl	lml		fcked	suk	
	%-)		Ili	lov		fckin	sukz	
	(@_@)		Ilml	luv		fcking	suxx	
	:s		Ilms	luff		feck	wrdo	
	:-Q		Ilu*	luvv		fk	wtf	
	:Q		Ilshipmp	lurve		fkd		
	:-\$		Irly	lv		fker		
	:\$		Ite	lve		fkin		
			Jj	lyk		fking		

Multimedia Appendix 1 Emoji & internet slang additions to the *LIWC 2007* emotion dictionaries

Multimedia Appendix 2

Supplementary Analyses: Personality, Self-esteem, and Social Desirability

Supplementary analyses were conducted to explore the individual characteristics that may be involved in the different emotion expression patterns between Facebook and Twitter.

Measures

MoodPrism collected data on the psychological characteristics of participants which included personality, self-esteem, and social desirability.

Personality, based on the Five Factor Model (extraversion, neuroticism, conscientiousness, agreeableness, and openness to experience), was measured by the Mini-IPIP scales [1]. The Mini-IPIP has 20-items, with four items addressing each personality factor. Example items include: Extraversion (e.g. "I am the life of the party"); Neuroticism (e.g. "I have frequent mood swings"); Conscientiousness (e.g. "I get chores done straight away"); Agreeableness (e.g. "I sympathize with others' feelings"); and Openness (or Intellect/Imagination) (e.g. "I have a vivid imagination"). Statements are rated on a 5-point Likert scale ranging from "1- Very Inaccurate" to "5 - Very Accurate", with a mid-point of "3 – Neither agree nor disagree). The five-factor structure of the Mini-IPIP has been supported across several studies and has adequate internal reliability for all sub-scales (α 's > .60) [1,2].

Self-esteem was measured with the Rosenberg Self-Esteem Scale (RSES) [3]. Participants rate 10 statements about their feelings toward themselves on a 4-point Likert scale from "0 – Strongly Disagree" to "3 – Strongly Agree". Five items relate to positive feelings (e.g. "I feel that I have a number of good qualities"), and five are negative and are reverse scored (e.g. "I certainly feel useless at times"). Ratings are summed to generate a total score ranging between 0-30 where higher scores indicate greater self-esteem. The RSES demonstrates good reliability across a range of demographic characteristics ($\alpha = .91$) [4].

DEPRESSION AND LANGUAGE-BASED EMOTION DYNAMICS ON SOCIAL MEDIA

Social desirability, the tendency with which an individual presents a desirable impression of themselves, was measured by the short form Marlowe-Crowne Social Desirability Scale Form C (M- C Form C) [5]. Participants respond to 13 dichotomous items describing socially desirable (but rare) or undesirable (but common) behaviours (e.g. "No matter who I'm talking to, I'm always a good listener"). These are rated as being "True" or "False" (items are reverse-scored to the socially-desirable direction) and are summed to create a score ranging from 0-13, where higher scores indicate a greater tendency to present a socially desirable image of oneself to others. The M-C Form C has demonstrated good reliability (Kuder-Richardson r = .76) [5].

Results

Table 1 presents the means and standard deviations of the psychological characteristics for participants with available data in the Facebook and Twitter samples.

Table 1

Means and Standard Deviations of Psychological Characteristics in the Facebook and Twitter Samples.

Variable ^a	Scale Range	Platform				
		Facebook M (SD)	Twitter M (SD)			
Depression Severity	0-27	11.48 (6.38)	9.80 (6.81)			
Extraversion	1 - 20	8.95 (3.02)	10.34 (4.50)			
Agreeableness	1 – 20	14.35 (3.69)	15.34 (4.09)			
Conscientiousness	1 – 20	11.48 (3.57)	13.59 (3.69)			
Neuroticism	1 - 20	14.39 (3.27)	13.02 (4.61)			
Openness to Experience	1 - 20	12.87 (2.67)	13.10 (3.27)			
Self-Esteem	0 - 30	14.05 (2.09)	15.30 (1.91)			
Social Desirability	0 - 13	5.13 (2.69)	6.49 (2.50)			

^a Sample sizes differ across variables. For Facebook *n* ranges between 21 -29; for Twitter *n* ranges between 40 - 49.

DEPRESSION AND LANGUAGE-BASED EMOTION DYNAMICS ON SOCIAL MEDIA

As all distributions were non-normal, Mann-Whitney U tests were selected to compare mean ranks on psychological characteristics between the Facebook and Twitter users. No significant differences were observed in depression severity, extraversion, agreeableness, neuroticism, or openness to experience. Twitter users did have significantly higher mean ranks in conscientiousness (U = 288.50, p = .034), self-esteem (U = 250.00, p = .017), and social desirability (U = 419.00, p = .046) compared to Facebook users.

Discussion

Compared to the Twitter sample, Facebook users had lower, conscientiousness, self-esteem, and lower scores of social desirability. Lower levels of conscientiousness have associated with less impulse control and can result in greater reactivity in negative emotion [6]. Combined with a reduced tendency to present a positive self-image to others, this may account for the negative emotion instability observed on Facebook but not Twitter; particularly as low conscientiousness also plays as role in less cautious online behaviour [7, 8]. This may elicit status updates on Facebook directly tied to emotional experiences with less reflective construction. Twitter users in contrast may be more inclined to monitor their online emotion expression and present more favourable representations of themselves. Though we lacked the power to control for these characteristics in regression analyses they may be informative targets for future research.

References

1. Donnellan MB, Oswald FL, Baird BM, Lucas RE. The Mini-IPIP Scales: Tiny-yet-effective measures of the Big Five Factors of Personality. *Psychol Assess*. 2006;18(2):192-203. doi:10.1037/1040-3590.18.2.192.

2. Baldasaro RE, Shanahan MJ, Bauer DJ. Psychometric Properties of the Mini-IPIP in a large, nationally representative sample of young adults. *J Pers Assess*. 2013;95(1):74-84. doi:10.1080/00223891.2012.700466.

Rosenberg M. Society and the Adolescent Self-Image. Princeton, NJ: Princeton University Press;
 1965. doi: 10.1126/science.148.3671.804

4. Sinclair SJ, Blais MA, Gansler DA, Sandberg E, Bistis K, LoCicero A. Psychometric properties of the Rosenberg Self-Esteem Scale: Overall and across demographic groups living within the united states. *Eval Health Prof.* 2010;33(1):56-80. PMID: 20164106.

5. Reynolds WM. Development of reliable and valid short forms of the Marlowe-Crowne social desirability scale. *J Clin Psychol*. 1982;38(1):119-125. doi:10.1002/1097-4679(198201)38:1<119::AID-JCLP2270380118>3.0.CO;2-I.

6. Fayard JV, Roberts BW, Robins RW, Watson D. Uncovering the affective core of conscientiousness: The role of self-conscious emotions. *J Pers*. 2012;80(1):1-32. PMID: 21241309.

7. Hollenbaugh EE, Ferris AL. Facebook self-disclosure: Examining the role of traits, social cohesion, and motives. *Comput Hum Behav.* 2014;30:50-58. doi:10.1016/j.chb.2013.07.055.

8. Seidman G. Self-presentation and belonging on Facebook: How personality influences social media use and motivations. *Personal Individ Differ*. 2013;54(3):402-407.

doi:10.1016/j.paid.2012.10.009.



Illustrative participant data showing instability at fixed levels of variability.

Multimedia Appendix 3

Figure 1. Example participant data showing instability under fixed variability (iSD) conditions. Panel (a) shows a participant with a low instability value (.015) and variability of .104. Panel (b) shows a participant with a similar variability value (.108) and a high instability value (.290). Brackets show the iSD around the mean (horizontal trend line). Time-scale is not shown here to compress Facebook record into a visible format for both participants.

4.3 Concluding Remarks

In this chapter a paper was presented that applied emotion dynamics indices to the emotion language expressed in status updates on Facebook and Twitter. It found that greater instability in the use of negative words across status updates on Facebook was associated with depression severity. In contrast, lower variability in the use of negative emotion words across the Tweets on Twitter was association with lesser depression severity. For Facebook users only, behavioural features indicating more frequent posting of status updates also predicted depression severity. Supplementary analyses were also presented suggesting individual characteristics, such as personality, may be involved in platform-specific user differences that potential impact on the emotion language patterns expressed over time. While this paper demonstrated the feasibility and potential utility of using emotion dynamics features as a part of identifying depression from social media user's status updates, it is still unclear how well the patterns observed accurately reflect lived experience. This is addressed in the next chapter in a case-series analysis seeking to critically explore the daily use of emotion words on Twitter with concurrent ratings of self-reported mood.

Chapter 5

Do people feel what they Tweet? A case-series exploration of daily mood ratings and the emotions expressed on Twitter

5.1 Preamble to Empirical Paper 2

This chapter presents the second empirical paper of this thesis, titled "Do people feel what they Tweet? A case-series exploration of daily mood ratings and the emotions expressed on Twitter". Measures of self-reported mood and the emotion language used in Tweets were taken concurrently across a 30-day period and the resulting mood profiles are contrasted. This paper provides a critical examination of the reliability and validity issues in using the emotion language on social media as an indicator of experienced mood and the implications this has for depression prediction from social media data. A descriptive case-series approach was taken in this paper due to the sample size obtained. As such, it provides an in-depth description of the mood profiles obtained from *MoodPrism's* daily self-report and from the emotion expressed on Twitter by applying emotion dynamics features as descriptive summaries of the time-series plots. This chapter highlights points of convergence and divergence between the mood profiles and discusses the impact depression may have on the observations obtained.

This paper was submitted to *Computers in Human Behavior*. As such, the paper is formatted in accordance with the journal requirements. References are provided in the style of the *American Psychological Association* (6th edition).

5.2 Do people feel what they Tweet? A case-series exploration of daily mood ratings and the emotions expressed on Twitter

Abstract

Background: Status updates on social media have been used to identify users' mental health status, with emotion words emerging as a primary indicator of assumed depression status. However, it remains unclear how accurately expressed emotions reflect a user's mood.

Objective: This case-series compares the mood profiles of Twitter users with their concurrent selfreported mood to examine how consistently same-day mood co-occurs across data types and collection methods.

Method: Self-reported mood and Tweets were collected from five users over a 30-day period using a mobile app. Tweets were automatically counted for emotion words. Daily levels, individual variability, instability, and acute changes in self-reported mood and expressed emotion were compared.

Results: Across the five participants, Spearman correlations revealed little similarity in terms of mood levels within the same day, but visual inspection suggested variability profiles across 30-days appeared similar. Twitter and self-report mood profiles aligned as depression severity increased. **Conclusion:** Although the growing amount of data available through social media makes it possible to detect mental health difficulties at collective levels, greater understanding of how this unfolds over time at the individual level is needed. This study provides a starting point, demonstrating the feasibility of emotion dynamic data analytic methods for language.

Keywords: social media, linguistic analysis, emotion, dynamics, experience sampling methodologies.

1. Introduction

Over the past decade, online social media, such as Facebook, Twitter, Weibo, and Instagram, has become a dominant part of many people's lives. People frequently turn to social media to obtain and broadcast information, express thoughts and feelings, discuss current events, feel connected, and even to discuss mental health issues (e.g., Berry et al., 2017; Deters & Mehl, 2013; Grieve, Indian, Witteveen, Anne Tolan, & Marrington, 2013; Manago, Taylor, & Greenfield, 2012; Park, 2013). The language expressed online has been proposed to be useful for identifying individuals and communities experiencing poor mental health (e.g., Bollen, Gonçalves, Ruan, & Mao, 2011; De Choudhury, Gamon, Counts, & Horvitz, 2013; De Choudhury, Kiciman, Dredze, Coppersmith, & Kumar, 2016; Mowery et al., 2017; Schwartz et al., 2014). Numerous models and algorithms with varying degrees of sophistication have been developed, with adequate ability to predict depression and other mental illnesses (cf. Guntuku, Yaden, Kern, Ungar, & Eichstaedt, 2017).

Emotion is typically included in such models, with studies finding that depressed individuals express more negative emotion words than non-depressed persons (De Choudhury et al., 2013; Schwartz et al., 2014; Settanni & Marengo, 2015). Because language is intricately linked to cognitive and affective processes (Lindquist, Gendron, & Satpute, in press), tracking emotion words over time may be a valuable tool for observing the internal processes that signal change in depression status. However, it is yet to be clearly demonstrated how accurately emotional language used on social media aligns with a user's concurrent mood. Using a case study approach, the current study explores the connections and differences between Twitter language and self-reported mood, providing insights into the reliability and sensitivity with which social media content may be useful at the individual user level for screening or monitoring depression.

1.1 The Language of Depression on Social Media

Evidence is mounting to suggest that social media behaviours can signal the presence of depression (Seabrook, Kern, & Rickard, 2016). Depressed persons post content more frequently on social media, and increases in posting frequency may signal increases in depression severity. On Twitter, people Tweet about mental health (and depression) as a way of accessing social support, developing community, and finding a safe place for expression (Berry et al., 2017). Individuals with greater depression symptoms also self-report that they write negative content in their status updates more frequently than those without depression (Locatelli, Kluwe, & Bryant, 2012). Direct examination of status updates has revealed that young people with a greater number of self-reported depressive symptoms express indicators reflecting symptoms of major depressive disorder (MDD) across their Facebook status updates (Moreno et al., 2011; Moreno et al., 2012).

Language has emerged as one of the core features of social media behaviour that can successfully identify depression (Guntuku et al., 2017). Users with a diagnosis of depression are more likely to use negative emotion words than those without a depression diagnosis (De Choudhury et el., 2013). Depression severity can be observed through social media language content (Schwartz et al., 2014), and there are unique linguistic characteristics that signal suicide risk, such as the use of language related to death (De Choudhury et al., 2016; Larsen et al., 2015). Depression has been shown to influence the linguistic style of an individual in several offline studies (Bernard, Baddeley, Rodriguez, & Burke, 2016; Pennebaker, Mehl, & Niederhoffer, 2003; Rude, Gortner, & Pennebaker, 2004). Aligned with Beck's (1974) cognitive model of depression, negative cognitive biases contributing to depressive thoughts and feelings may colour the negative emotional tone in the language communicated by depressed persons, resulting in the more frequent expression of negative emotion seen online.

1.2 The False-Positives

There is a great deal of variation in the methodological approaches and prediction performance of automated screening methods for depression from social media data (Guntuku et al.,

2017). While depression identification from social media has considerable promise, approaches targeting words also return many false-positive identifications of possible depression (Cavazos-Rehg et al., 2016; Mowery et al., 2017).

First, linguistic features, including negative emotion words, are not used solely by those experiencing depression. For example, content analysis of a sample of depression-related Tweets has revealed that while around one-third of the Tweets were directly related to personal experiences of depression, the remainder either made trivial reference to depression or expressed a message of support (Cavazos-Rehg et al., 2016). Recent work has also indicated that the induction of negative mood, regardless of depression status, is the most significant factor linked to later use of negative emotion words in expressive writing tasks (Bernard et al., 2016).

Second, confounding variables such as neuroticism, self-esteem, and age have also been linked to the use of negative emotion words on social media and may be involved in the way people communicate online, potentially obscuring how accurately depression status can be detected (Forest & Wood, 2012; Park et al., 2015; Preotiuc-Pietro et al., 2015).

Third, current approaches to examining linguistic features associated with depression have had limited focus on temporal variation. As an adequate sample of words per user is needed (see Kern et al., 2016), user data is often combined across posts. While this provides greater overall predictive ability, it eliminates the rich temporal structure of responses, information that may help differentiate depressed from non-depressed individuals. If cognitive and affective schemas are visible in language, it is also likely that the dynamic features of emotion might be visible over multiple samples of language. Emotion variability and instability (i.e., time-structured variability) have been consistently linked to depression and even predict depression onset (Houben, Van Den Noortgate, & Kuppens, 2015; Koval, Pe, Meers, & Kuppens, 2013; Wichers, 2014). *Variability* provides information about the general dispersion of emotion observations over time and can be measured by statistics such as the within-person standard deviation (iSD) (Houben et al., 2015). While variability measures are insensitive to the time-ordering of the emotion variation (only their

overall spread), measures of *instability* capture patterns in this temporal structure, such as the magnitude of change between consecutive pairs of emotion observations over time, the mean squared successive differences (MSSD) (Houben et al., 2015; Von Neumann, Kent, Bellinson, & Hart, 1941). Instability has been shown to be important to depression prediction in a small sample of Facebook users, where greater post-to-post fluctuations in the expression of negative words (i.e., negative affect instability) was positively associated with depression severity (Seabrook, Kern, Fulcher, & Rickard, 2018).

As individuals frequently post about life events on Twitter, it is possible that patterns of emotion dysregulation better differentiate between depressed and non-depressed individuals than average language use alone. High emotion variability or instability, as seen on social media, is likely to be driven by reactions to positive and negative events. Indeed, specific changes in negative affect may also be observed by computing acute changes in mood between days, and serve as a complementary measure to instability by describing the probability of clinically relevant changes (probability of acute change; PAC) and when they occur (acute change; AC) (Jahng, Wood, & Trull, 2008).

1.3 The Current Study

Social media platforms like Twitter provide a unique opportunity for passive data collection as a means of predicting and monitoring depression. By directly sampling the naturally occurring, real-time expressions of social media users, status updates provide momentary data that offer potential as a practical means of monitoring depressive symptoms (at the individual level), and public mental health status (at the collective level) (Kern et al., 2016; Larsen et al., 2015). The prevalence of negative emotion has been successfully used to distinguish depressed and nondepressed individuals, but less is known about how such symptoms unfold within individual users.

Using a case-series approach, we examine the extent to which self-reported emotions reflect expressed emotion online across five Twitter users. Following other studies that have considered emotion frequency as a predictor of depression (e.g., De Choudhury et al., 2013; Gkotsis et al.,

2016; Settanni & Marengo, 2015), we classified emotion using the positive and negative emotion dictionaries from the Linguistic Inquiry and Word Count (LIWC) program (Pennebaker, Chung, Ireland, Gonzales, & Booth, 2007). Ecological momentary assessments (EMAs) of mood have been demonstrated to reveal emotion patterns that predict depression and have high ecological validity (Armey, Schatten, Haradhvala, & Miller, 2015; Ebner-Priemer & Trull, 2009; Houben et al., 2015). Enabling EMAs, *MoodPrism* (Rickard, Arjmand, Bakker, & Seabrook, 2016) is a smartphone application (app) which provides daily self-reports of positive and negative emotion. We had one methodological and two substantive aims.

Methodologically, we used EMAs accessed via smartphones to compare mood profile data across Twitter and self-report (collected by the app) across a 30-day period, addressing a need to provide complementary evaluation of social media data by also sampling non-social media variables (Tufekci, 2014). We computed four simple *time series features* for summarizing each 30day emotion time series: (i) average mood, (ii) variability, (iii) instability, and (iv) acute changes in mood.

Substantively, we considered how similar (or dissimilar) the time-structured mood profiles obtained from Twitter and *MoodPrism's* self-report were over a 30-day period. We asked:

1. Is there a same-day relationship between the measures of mood obtained through selfreport and through Twitter?

2. From visual inspection, are there any time-series features that are similar in magnitude across the self-report and Twitter time-series?

In addition, we considered whether there were any systematic variations in the four time-series features from self-report and Twitter that may be linked to depression severity.

2. Method

2.1 Procedure Overview

Participants for the current study were selected from the larger *MoodPrism* project assessing a mood-tracking app (cf. Rickard, Arjmand, Bakker, & Seabrook, 2016). *MoodPrism* allows users

to track their mood once every day for 30 days and presents engaging mood and mental health feedback.

For the broader project, participants were recruited using strategies that included targeted Facebook advertising recruiting individuals who were smartphone users and lived in Australia, and convenience sampling. After downloading *MoodPrism* (available on both the Android and iOS Australian app stores), participants opened the app and read explanatory statements. They provided consent to participate in the research electronically within the app. Participants were then invited to allow access to their Twitter data. It was explained that only word counts and timestamps would be collected and that the content of Tweets would not be stored. They could complete the measures and study without allowing access to their Twitter data.

Participants then completed baseline measures. These were organised into blocks that could be completed at a time of the participant's convenience before unlocking the mood-tracking function of *MoodPrism*. As reported in Arjmand and Rickard (2018) these surveys took on average 37m 14s (SD = 11m 33s) to complete. After completing the baseline measures, participants were prompted at a random time within a preferred timeframe every day for 30-days to complete daily mood reports. Participants elected timeframes could range between 3:00 am, and 3:00 am the following day). Completing the daily mood reports took on average 1m 34s (SD = 44s) to complete (Arjmand & Rickard, 2018). After 30 days of using *MoodPrism*, participants completed follow-up surveys through the app, which primarily addressed mental health change (including the *PHQ-9*). All procedures were approved by the Monash University Human Research Ethics Committee (Approval # CF14/968 – 2014000398).

2.2 Participants

Five participants were identified from the broader project who: (1) completed the baseline surveys on *MoodPrism*, (2) completed at least 70% of the daily mood reports over 30 days (defined below), (3) provided access to their Twitter account, and (4) Tweeted during their period of *MoodPrism* use. Participants were assigned a pseudonym for data analysis and reporting. They

contributed data for a maximum of 30 consecutive days within the data collection period occurring between April and December 2016. Table 1 provides the participants' demographic characteristics. All five participants were female and were aged between 18 and 35 years old. At the time of the study they were employed or undergoing study, and all had achieved at least secondary school qualifications.

Table 1

	Participant									
Variable	1	2	3	4	5					
Age	33	33	18	35	29					
Education	Tertiary	Secondary	Secondary	Tertiary	Post-					
					graduate					
Occupational	Part-time	Full-time	Full-time	Full-time	Not					
status					employed					
Study status	Not studying	Full-time	Part-time	Not studying	Part-time					

Demographic Characteristics of Participants/Cases

2.3 Measures

All measures for the current study were delivered and/or collected by *MoodPrism* (Rickard et al., 2016). *MoodPrism* delivered push-notifications to prompt participants to complete daily mood self-report for 30 days and accessed the emotion language expressed in Tweets automatically from the standard Twitter application programming interface (API) during the same period (d1 – d30). Depression severity ratings were also provided to *MoodPrism* at baseline (T1) and at the 30-day follow-up (T2). Age, gender, current occupational status, and highest completed level of education were collected at baseline.

2.3.1 Daily mood reports (d1-d30). *MoodPrism* delivered push-notification prompts asking: "How are you feeling today?" at quasi-random times each day for 30 consecutive days while *MoodPrism* was installed on a participant's smartphone. The daily mood reports included items assessing mental health, eudaimonic well-being, self-esteem, significant daily positive and negative events, and social- environmental context (Rickard et al., 2016); the current study focused on current mood.

Based on the Circumplex Model of Emotion (Russell, 1980) participants responded to three items rated on a 5-point Likert scale ("1 – Not at all", "5 – Extremely") rating their current mood: (1) Positive or pleasant; (2) Negative or unpleasant; (3) Active or alert. These ratings represent the core affect space of an individual at the time of the *MoodPrism* prompt. Mood ratings (positive, negative) were considered as distinct dimensions within each day. The 30-day time series of selfreported positive and negative mood obtained for each participant will be referred to as SR-Positive and SR-Negative, respectively.

2.3.2 Automated Tweet word count (tw1-tw30). To protect the privacy and confidentiality of users, *MoodPrism* pre-processed Tweets, providing information about the contents of the Tweets, but not the Tweets themselves. After consenting to contribute Twitter data, the user's Twitter handle was automatically accessed by *MoodPrism* via the Twitter API, collecting all Tweets posted during the period *MoodPrism* was installed on the participant's smartphone. *MoodPrism* parsed the collected Tweets through a word count script that extracted the number of positive and negative words, total words, and time-stamp for each Tweet (process depicted in Figure 1). This script included the LIWC2007 positive and negative emotion dictionaries (Pennebaker et al., 2007) which were supplemented with common emojis and internet slang (see Appendix A for the full list). Retweets were not included in our analysis.



Figure 1. MoodPrism procedure for Twitter word count extraction.

The relative proportion of emotion words used within each Tweet was calculated and is defined in Equation 1 (c.f. Kern et al., 2016).

$$p(category) = \sum_{word \in category} p(word) = \frac{\sum_{word \in category} count(word)}{N_words},$$
(1)

where *count(word)* refers to the total number of target words (for each of two categories: 'positive' or 'negative' emotion words) within an individual Tweet, and N_words refers to the total number of words in a Tweet. This results in a proportion of emotion language ranging from 0-1. A daily average of these proportions was then obtained as summary of the mood expressed on Twitter. The resulting 30-day time series for daily aggregate positive and negative mood Tweets will be referred to as T-Positive and T-Negative, respectively.

2.3.3 Depression severity. Self-reported depression severity was assessed with the nine item *Patient Health Questionnaire -9 (PHQ-9*; Kroenke, Spitzer, & Williams, 2001) at baseline (T1, immediately after downloading *MoodPrism*) and at follow-up (T2) after 30 days of *MoodPrism* use. Items from the PHQ-9 are rated from "0 – Not at all" to "3- Nearly every day", describing the frequency of depression symptoms such as "Feeling down, depressed or hopeless" over the previous two weeks. Ratings are summed to generate a total score ranging between 0-27. More severe

depression symptoms are indicated by higher total scores and cut-off values are provided to aid interpretation (0-4 = minimal; 5-9 = mild; 10-14 = moderate; 15-19 = moderately severe; 20-27 = severe). The PHQ-9 has demonstrated good reliability in the general population (Cronbach α = .87; Kocalevent, Hinz, & Brähler, 2013) and in primary care settings (α = .89; Kroenke et al., 2001).

2.4 Data Analysis

The *MoodPrism* and Twitter mood data sets were collated in chronological order and timematched for each participant. Spearman cross-correlations were computed to examine the relationship the SR- and T-Positive, and SR- and T-Negative mood profiles on same-day measurements across the 30-day recording period. Feature comparisons (average, variability, instability, and PAC) were conducted by visual inspection.

2.4.1 Within-person time-series features. Four mood profile features were selected for examination, identified from the emotion dynamics literature as targets that may aid depression identification (Houben et al., 2015). Only instability and PAC (defined below) incorporate temporal structure into their measurement. It should be noted that these features are not exhaustive and other indices may also be informative (e.g., spin, autocorrelation), but are beyond the scope of this study.

(1) The **average** positive and negative mood across all observations, defined by:

$$M = \frac{1}{n} \sum_{i=1}^{n} x_i = \frac{1}{n} (x_1 + x_2 + \Lambda x_n),$$
(2)

where M is the average of emotion values (x) taken over observations, i, and divided by n, the total number of observations for a participant within the 30-day recording period.

(2) Within-person variability of positive or negative mood was computed for each participant as:

$$iSD = \sqrt{\frac{\sum_{i} {s_i}^2}{n-1}} ,$$

(3)

where the sum is taken over observations, i, s_i indicates deviations from the mean of the time series, and n is the number of observations.

(3) Instability of positive or negative mood was computed for each participant as:

$$MSSD = \frac{\sum_{i=1}^{n-1} (x_{i+1} - x_i)^2}{n-1},$$

(4)

where x_i indicates the observation at index *i*, x_{i+1} refers to the next consecutive observation, and *n* refers to the total number of observations for that individual. Instability was calculated from the *z*-transformed time-series to allow comparison between self-report and Twitter, and thus represents the relative patterns of fluctuation through time.

Due to missing data the following adjustment was made to the MSSD to weight successive differences based on the time-interval between observations (Jahng et al., 2008).

time-adjusted
$$MSSD = \frac{median(\Delta t)}{(n-1)} \sum_{i=1}^{n-1} \frac{(x_{i+1} - x_i)^2}{(t_{i+1} - t_i)},$$
(5)

where $median(\Delta t)$ is the median of incremental time differences across the recording period of the study (Seabrook et al., submitted).

(5) **Proportion of Acute Change** (PAC; Jahng et al., 2008) was computed for each participant as:

$$PAC = \frac{1}{N-1} \sum_{i=1}^{N-1} AC_{i+1},$$

(6)

where successive differences $(x_{i+1}-x_i)$ were computed between each consecutive pair of observations, time-adjusted as in Eq. (5), and *z*-transformed. As there are no theoretically established cut-offs for defining the values of acute change (AC) in this data, a statistical cut-off was selected following the recommendations of Jahng et al. (2008). ACs were defined by a *z*-score

cut point of 1.282 which represents values in the top 10% of the normal distribution. Therefore, $AC_{i+1} = 1$ when $x_{i+1} - x_i \ge 1.282$. The calculation of successive differences disregards the sign, or direction of change, AC falling above the 1.282 cut-off indicate a large magnitude change in either an increasing or decreasing direction. Visual inspection of the data was conducted to elucidate the direction of AC.

2.4.2 Data transformations. To allow qualitative visual comparisons of values between the two data types (self-report and Twitter), data were transformed in two ways. First, before computing the average and iSD, raw scores were rescaled as a proportion of the total range of scores (minimum to maximum values) within self-report and Twitter as defined in Eq. (7).

$$x' = \frac{x - \min(x)}{\max(x) - \min(x)}$$

(7)

where x is the raw value. Minimums and maximums were defined the by range of observed values in the entire sample (n = 5) for the SR- and T- positive and negative series. This resulted in values from both self-report and Twitter ranging from 0-1, where 1 consistently represents the maximum observed value on each scale . The average and iSD was computed from these rescaled values to aid in visual comparison between self-report and Twitter. Second, as noted above, instability was computed from *z*-transformed time series and represents the pattern of fluctuations relative to the mean and standard deviation of a participant's self-report or Twitter emotion profile. Raw-score values and dual-axis plots of the mood profiles are presented in Appendix B.

3. Results

A total of 386 Tweets were extracted from the five participants. Table 2 presents depression status at baseline (T1) and follow-up (T2), and the Tweet and daily mood report activity across the 30-day study period. Participants 1 and 4 showed no signs of depression at baseline, whereas participants 2, 3, and 5 showed signs of moderate to severe depression. The users typically posted

multiple times per day, resulting in a combined total of 5,805 words, 209 (3.6%) of which were positive emotion words and 282 (4.9%) were negative emotion words.

Table 2

Depression Scores (T1, T2), and Tweet and Daily Mood Report Activity across the 30-Day Study Period (d1-d30).

Variable	Participant									
	1	2	3	4	5					
					16					
T1 Depression	9 (mild)	26 (severe)	4 (none)	5 (mild)	(moderately					
					severe)					
T2 Depression	5 (mild)	NT A	10	7 (mild)	10					
12 Depression	S (IIIIId)	NA	(moderate)	7 (mma)	(moderate)					
Daily Mood										
Reports	27	22	29	26	30					
Completed										
		Twitter A	Activity							
Total Tweets	46	53	104	95	88					
Total Words	074	072	1228	1403	1228					
	774)12	1220	1405	1220					
Tweets per day	1.53 (3.66)	1.77 (3.50)	3.47 (2.97)	3.17 (2.51)	2.93 (4.55)					
M (SD)	1100 (0100)	1117 (0.00)	0.117 (21977)	0.117 (2.01)	2198 (1188)					
Words per day	32.47	32,40 (83,07)	40 93 (29 63)	46 76 (39 93)	40 93 (62 11)					
<i>M</i> (SD)	(117.61)	32.10 (03.07)	10.95 (29.05)	10.70 (37.75)	10.95 (02.11)					
Words per										
Tweet	21.17 (4.16)	18.34 (6.78)	11.80 (5.34)	14.77 (6.07)	13.95 (7.48)					
<i>M</i> (SD)										

Note: For depression, clinical cut-off labels are present in parentheses. T1 = baseline, T2 = follow up.

3.1 Profile Comparison

Figure 2 provides z-transformed time series plots of positive (left) and negative (right) emotion for self-report and Twitter data for each participant. A reference line is provided at 0.00. Missing observations, which was common in Twitter time series, appear as a blank. Visual inspection suggests that for Participants 1, 3, and 4, SR-Positive and T-Positive are dissimilar over time, whereas for Participants 2 and 5 there is a similar overall shape. SR-Negative and T-Negative series appear similar in shape across all participants.



Figure 2. Panel plots of z-transformed positive and negative mood per day from MoodPrism self-report and Twitter emotion word use.



Fig 2 Continued. Panel plots of z-transformed positive and negative mood per day from MoodPrism self-report and Twitter emotion word use.



Fig 2 Continued. Panel plots of z-transformed positive and negative mood per day from MoodPrism self-report and Twitter emotion word use. A

horizontal trend line is provided for reference at 0.00. Daily mood ratings are denoted by circles and the emotion words in Tweets are shown with squares.

Same-day cross-correlations (Spearman's rho) between the SR and T time series for both positive and negative mood, and Lin's Concordance Correlation Coefficient (CCC) are presented in Table 3. The number of observation pairs differ by participant.

Table 3

Spearman Ranked Correlations between SR- and T- Positive (and Negative) Mood Profiles by Participant.

	Participant	Participant	Participant	Participant	Participant			
	1	2	3	4	5			
Observations	<i>n</i> = 9	<i>n</i> = 10	<i>n</i> = 27	<i>n</i> = 23	<i>n</i> = 14			
Positive Mood								
Correlation		.11	03	08	03			
(rho)								
		Negativ	ve Mood					
Correlation	.00	.41	05	.19	.09			
(rho)								

Note. Correlations could not be computed for Participant 1 positive mood due to constant SR-Positive ratings.

Visual inspection of Table 3 suggests that positive mood expressed on Twitter does not share a consistent relationship with same-day self-reported positive mood over time for these five participants across their 30-day recording period. Participant 2 returned the highest *rho* coefficient, though this was weak; and Participant 4 revealed a very weak negative relationship between their SR-Positive and T-Positive mood profiles. These correlation values are informed by inspection of Figure 2 where Participant 2 appears to have a similar (though not identical) pattern of positive mood across 30-days, whereas, for Participant 4 the mood profiles appear to be at an inverse for most of the 30-days, though does become more similar after d20. For the remaining participants (3

and 5), there is no clear co-varying pattern visible in Figure 2. For example, Participant 3 appears to have an inverse relationship between her SR and T-positive series between d1-25, with it becoming similar from d26-d30.In contrast to positive mood, we observe stronger relationships between negative mood expressed on Twitter and same-day self-reported negative mood for Participants 2, 4, and 5. These relationships are moderate (Participant 2) and weak (Participants 4 and 5). Visual inspection of Figure 2 highlights the close relationship between SR and T negative mood profiles for Participant 2 on days with concurrent observations. Similarly, Participants 4 and 5 appear to have similar SR and T-negative mood profiles over the 30 days, though for Participant 4 between d1-d7 there is a great deal more variation in the T-negative series that does not mirror the SR-negative series.

3.2 Time-Series Feature Comparison

Our methodological aim was to compare four time-series features (average mood, individual variability, instability, and PAC) for the self-report and Twitter mood profiles within individuals. Table 4 presents each feature by participant.

Feature	Participant 1		Participant 2		Participant 3		Participant 4		Participant 5	
	SR	Twitter	SR	Twitter	SR	Twitter	SR	Twitter	SR	Twitter
Observations	<i>n</i> = 27	<i>n</i> = 9	<i>n</i> = 22	<i>n</i> = 12	<i>n</i> = 29	<i>n</i> = 28	<i>n</i> = 26	<i>n</i> = 26	<i>n</i> = 30	<i>n</i> = 14
Positive Mood										
Average ^a	.47	.11	.24	.20	.55	.33	.64	.15	.48	.06
iSD ^a	.08	.09	.24	.16	.18	.23	.14	.15	.23	.09
Instability ^{b, c,}	2.06	2.75	1.53	1.69	0.95	1.84	1.47	1.59	1.13	1.47
PAC ^{b, c}	.19	.00	.28	.09	.04	.11	.04	.12	.10	.15
Negative Mood										
Average ^a	.09	.31	.60	.22	.07	.14	.02	.43	.22	.24
iSD ^a	.19	.34	.30	.21	.13	.14	.07	.23	.22	.24
Instability ^{b, c,}	1.40	2.13	1.80	0.58	2.05	1.94	1.80	2.14	1.68	1.75
PAC ^{b, c}	.08	.13	.17	.09	.07	.11	.12	.08	.14	.15

Descriptive Mood Profile Features from Self-Report and Twitter Positive and Negative Mood Data.

Note: ^a = Features were computed from rescaled raw scores and represent a proportion of the total range of values of positive and negative mood from self-report and Twitter; ^b = Features were computed from z-transformed scores. ^c = observations = n - 1 for Instability and PAC as they are calculated from the successive differences between observations. *iSD* = within-person standard deviation; PAC = Probability of Acute Change.

Average positive mood was higher in the SR-Positive time series than in the T-Positive time series for participants 1, 3, 4, and 5, but they were similar for Participant 2. For negative mood, Participants 1, 3, and 4 had higher mean levels in the T-Negative series. Participant 2 had higher levels in the SR series, whereas levels of negative mood on the SR and T series were comparable for Participant 5.

For variance, Participant 5 demonstrated greater variability in the SR-Positive series than in the T series, whereas positive mood variability was comparable across SR-Positive and T-Positive series for the other participants. For negative mood, Participants 1 and 4 had greater variability in the T-Negative series than SR-Negative series, Participant 2 had greater variability in the SR-Negative series, and Participants 3 and 5 demonstrated comparable variability between their SR and T-Negative series.

In terms of instability, Participants 2, 4, and 5 demonstrated comparable instability for positive mood between their SR and T-Positive series, whereas Participants 1 and 3 demonstrated greater instability in the T-Positive series. For negative mood, Participants 3, and 5 demonstrated comparable instability between the SR and T-Negative series. For Participants 1 and 4, there was greater instability in the T-Negative series, and for Participant 2 there was greater instability in the SR-Negative series.

For the proportion of acute change (PAC), Participants 1 and 2 demonstrated a higher PAC value in the SR-Positive series than in the T-Positive series, whereas Participants 3, 4, and 5 had a higher PAC value in the T-Positive series. For negative mood, the PAC was higher in the SR-Negative series than in the T-Negative series for Participants 2 and 4, whereas PAC was higher in the T-Negative series for Participants 1 and 3. PAC was equivalent for Participant 5.

As a complementary approach to examining acute change, Figure 3 illustrates AC by successive difference across the study period. The ACs represented indicate large magnitude increases or decreases in mood between days. Light grey cells indicate AC in either positive or negative T time series, dark grey cells indicate AC in either positive or negative SR time series, and

black cells show where there is a AC on both the SR and T time series in the same successive difference.



Figure 3. Acute change between mood observations on self-report and Twitter time-series. Light grey cells indicate AC in either positive or negative T time series, dark grey cells indicate AC in either positive or negative SR time series, and black cells show where there is a AC on both the SR and T time series in the same successive difference.

Co-occurrence of AC on both the SR and T time series was only evident for Participant 5 as indicated by the black cells in Figure 3. At successive difference 4 (between days 4 and 5) there was a marked decrease in SR negative mood and an increase in the T positive mood. At successive difference 21, (between days 21 and 22) there was a marked decrease in SR positive mood and an increase in T negative mood. Finally, at successive difference 22 (between days 22 and 23) there was a marked decrease in both SR and T negative mood. Overall, the patterns of AC for Participant 5 suggest similarity between the signals of SR and T in indicating sudden and extreme shifts in positive or negative mood from day to day. The other participants did not demonstrate such shifts.

3.3 The Impact of Depression Severity

Finally, we considered whether the time-series features varied according to depression status. Table 2 indicates baseline and 30-day follow up scores. Participants 1 and 4 demonstrated no or mild signs of depression across both assessments, Participant 3 increased in depression, Participant 5 decreased. As Participant 2 did not complete the T2 administration of the PHQ-9, we examined her responses to the PHQ-2 within the *MoodPrism* daily mood reports. Ratings were

summed within each day and averaged across the last 14 days of the study. The resulting average (M = 2.25; range: 0 - 6) suggests Participant 2 would screen positive for the presence of depression (Arroll et al., 2010), with their daily ratings describing depression symptoms that were moderately severe.

Table 5 shows the average, within-person variability, instability, and PAC for positive and negative mood for each participant in terms of their salient differences between the SR and T series. It also presents participants' T1 and T2 depression status.

Table 5

Salient Differences by Feature and by Participant between the Positive and Negative Mood Profile Features of Self-Report and Twitter.

	Depr	ession	Positive Mood Profile Comparisons				Negative Mood Profile Comparisons				
	Sta	atus									
Participant	T1	T2	Average	Variability	Instability	PAC	Average	Variability	Instability	PAC	Total by
											Participant
1	mild	mild	SR		Т	SR	Т	Т	Т	Т	7
2	severe	mod*				SR	SR	SR		SR	4
3	none	mod	SR		Т	Т	Т			Т	5
4	mild	mild	SR			Т	Т	Т	Т	SR	6
5	mod-	mod	SR	SR		Т					3
	severe										
Total by Feature			4	1	2	5	4	3	2	4	

Note: SR indicates the self-report mood profile had higher values on a feature than the Twitter mood profile. T indicates the Twitter mood profile had higher values on a feature than the self-report mood profile. Blank cells indicate comparable feature values on the SR and T mood profile. mod = moderate depression severity; mod-severe = moderately-severe depression severity.

* = depression status at T2 was calculated from the PHQ-2.

Salient differences between the magnitude of values for mood profile features (SR and T) were identified through visual inspection of the values in Table 4 and were discussed in the Feature Comparisons, above. These are summarised in the context of depression severity within-participants and within features in Table 5 which suggests that there are fewer salient differences between the SR and T series for the participants with higher depression severity at T1 and T2 (Participants 2 and 5). Where salient differences do exist, they are in relation to higher values from the SR series. In contrast, those with lower depression severity at T1 and T2 (Participants 1 and 4) have a greater number of salient differences overall, where the T-series (specifically T-negative) have higher values than the SR-series.

3.3.1 Self-Report

Participants with no-to-mild depression severity at T1 demonstrated higher mean values of positive mood on the SR-Positive series and lower mean values of negative mood on the SR-Negative series (Participants 1, 3, and 4) relative to those with higher severe depression severity (Participants 2 and 5). An exception to this was Participant 5 (moderately-severe depression severity; T1) who improved between assessments (moderate depression severity; T2) who reported equivalent levels of positive mood to those with mild depression severity.

For variability, Participants 1 and 4 with no-to-mild depression severity at both assessments had low variability in the SR-Positive and SR-Negative series. In contrast, variability was higher for Participants 2 and 5 (moderate-to-severe depression severity) on both positive and negative series. Participant 3, who declined between assessments (T1-none; T2- moderate) demonstrated equivalent levels of variability to Participants 2 and 5.

Instability on the SR-Positive series had no clear variations between participants that may be linked depression severity, however, the SR-Negative instability appeared higher for those with moderate-to-severe depression severity (Participants 2 and 5) and for those whose depression severity increased between assessments (Participant 3). Participant 4 also demonstrated equivalent levels of SR-Negative instability to those with more severe depression severity, though visual
inspection of Figure 2 suggests that these fluctuations occur in the context of longer periods of stable negative mood which contrasts with Participants 2, 3, and 5 who show frequent and large fluctuations. Similarly, the PAC values do not appear to systematically vary with depression severity. Notably, Participant 2 (severe depression) demonstrated the highest PAC values on both the SR-Positive and SR-Negative series.

3.3.2 Twitter

For T-Positive, those with lower depression severity (Participants 1 and 4), or who improved between assessments (Participant 5), had lower mean levels in positive mood, whereas those with higher depression severity (Participant 2) or had increased depression severity between assessments (Participant 3) demonstrated higher mean positive mood. For T-Negative, mean negative mood was higher for those with no-mild depression severity (Participants 1 and 4) and was lower for those with moderate-to-severe depression severity (Participants 2 and 5) and for those who had increased depression severity between assessments (Participant 3).

In terms of variability, there did not appear to be any systematic variations related to depression severity on either the T-Positive or T-Negative series. Similarly, for instability there did not appear to be any systematic variation based in depression severity in the T-Positive or T-Negative series and this was reflected in the PAC values. Notably, Participant 2 recorded the lowest T-Negative instability and visual inspection of Figure 2 indicates a relatively smooth increase in negative mood over time.

4. Discussion

The growing amount of data available through online social media platforms such as Facebook and Twitter have raised the possibility of using language and other features to detect and monitor mental health difficulties (Guntuku et al., 2017). Yet while there is some predictive success at collective levels, less is known about how social media language intersects with felt daily emotions. Using a case-series approach, this study examined the convergence between mood profiles obtained through the emotion language used on Twitter, and through a self-report mood

tracking app for five different users. We examined four different time-series features (average, variability, instability, and PAC), comparing profiles across positive and negative emotion. In addition, we considered how profiles varied based on an individual's depression severity.

Considering the profiles alone, there was little agreement between the same day mood expressed through SR and Twitter. However, the time series features suggested a more complex picture. While there were often salient differences between the SR and T series in terms of average mood, variability and instability appeared more similar. Examination of AC however, suggested that large daily fluctuations in positive or negative mood in this sample rarely co-occurred on both the SR and T series. When considered in the context of depression severity, there was greater similarity between the SR and T series at higher depression severity levels. The average positive and negative mood in the SR and T series appeared to systematically vary by depression severity when comparing participants in this sample. Positive and negative variability and instability also suggested variations based on depression severity in the SR series, though for the T series, no depression-related systematic variations were evident.

4.1 Inconsistencies between Self-Reported Mood and Twitter Language

Overall, concurrent observations of mood from Twitter and self-report were inconsistent. Although the face validity of negative emotion word expression on Twitter may be high, particularly when considered in the context of depression prediction, its concurrent validity as a measure from which to infer mood (i.e., revealing persistent low mood over time) may be low. Several features of posting a Tweet differ from expressive writing in a private or laboratory setting. Posting a Tweet is a brief form of public disclosure and may be intended for a specific audience or crafted to encourage reciprocal communication (Marwick & boyd, 2011). Language choice may be influenced by self-presentation motives (i.e., motivations to present a positive self-image to others in the social network) (Gil-Or, Levi-Belz, & Turel, 2015; Seidman, 2013; Selim, Long, & Vignoles, 2014) and online social norms (that positive disclosures are more favourable) may influence emotional expression (Waterloo, Baumgartner, Peter, & Valkenburg, 2017). These confounding

variables may impact greatly on how reliably Twitter emotion language can sample mood from one moment to the next.

4.2 Feature Comparison

Along with comparing profiles, we had a methodological aim to consider various time series features as approaches to considering convergence and inconsistencies across the two types of data. On the whole, SR-Positive mood profiles revealed larger mean values than in T-Positive mood profiles, whereas T-Negative mood profiles tended to have higher mean values than SR-Negative mood profiles. Similarly, the variability features (*iSD*, instability, and PAC) of the positive and negative mood profiles did not clearly converge. These discrepancies may be a result of the specific environments from which the mood data is sampled. The mood data collected in the daily mood reports was signal-contingent and were temporally-bound to the moment a participant responded to a prompt. The mood collected following these prompts was likely to be tapping into current or recent emotional experiences.

A major advantage of using smartphone assisted experience sampling method is that retrospective biases are reduced, and the ecological validity of the data is high (Miller, 2012). In contrast, while Tweets have high ecological validity in the sense they are sampled from a naturally occurring behaviour in real time, it is likely that the mood referred to in Tweets may reflect a broader, less temporally-bound timeframe. Tweets may be past, present or future oriented (Park et al., 2017). This has the potential to introduce retrospective biases into the estimation of mood from Twitter, particularly relevant in the context of findings indicating that retrospective accounts of mood often overestimate negative emotion (Sato & Kawahara, 2011).

These findings highlight prominent differences in emotional responding when sampling experiential emotion and behavioural emotion (i.e. Tweets). Mauss et al. (2005) discuss consistency between measures of emotion as response coherence, where there is a presumed theoretical coordination between the expressive and experienced components of emotion. In practice, however,

the relationship between these components is not as clear, particularly where the time-frame for sampling emotion experience and behaviour is broad or varied as it is in this study.

4.3 The Impact of Depression Severity

The five participants varied in terms of their baseline and follow up depression statuses, ranging from mild to moderately severe. The patterns suggested that the signals between mood profiles converged as depression severity increased, a finding consistent with research suggesting coherence between experiential and behavioural emotion increases as emotion intensity increases (Mauss et al., 2005). Those with greater depression severity may have their language use on Twitter more clearly driven by experiences of negative mood resulting in the high sensitivity of identification for these individuals. In contrast, variables other than depression severity may be involved in defining the language used for those with no-to mild symptoms which may contribute to the likelihood of obtaining false positives when using language to predict depression on Twitter.

The pattern of convergence was particularly apparent in Participant 2, who differed from other participants in that they reported severe depression severity (26 on the PHQ-9) at T1 compared to the other participants (ranging 4-16 on the PHQ-9). They also self-reported the most intense negative mood across the period of the study compared to other participants. The moderate association between negative language and negative mood is consistent with recent findings indicating that negative affect drives more negative language use (Bernard et al., 2016) and cognitive theories of depression that suggest negative language use manifests from increased levels of negative thinking, or negative cognitive biases in interpreting information from the environment (Beck, 1974). Depression has also been linked to a greater tendency to post present-focused Tweets (Park et al., 2017). It is likely that at the more severe levels of depression severity, Tweets may act as a more direct measure of experienced mood as there is a present focus (consistent with the methodology of the daily mood reports) and the influence of confounding variables (discussed above) that alter the language used on Twitter may be less impactful in this context.

Notably, different features appeared relevant for SR versus Twitter mood profiles. Features from the SR-Positive and SR-Negative mood profile differed between mild and moderate depression severity in relation to the average, variability and instability. In terms of the average, as might be expected, participants with lower depression severity self-reported higher positive and lower negative mood than those with higher depression severity. Those with moderate depression severity had greater variability in their SR-Positive and SR-Negative mood over 30-days than those with mild depression severity. This is consistent with previous research which suggests that more variable positive and negative emotion is associated with poorer mental health outcomes (Gruber, Kogan, Quoidbach, & Mauss, 2013; Houben et al., 2015).

While there were no clear patterns associated with depression severity and SR-Positive instability, SR-Negative instability was higher for the participants with moderate to severe depression severity. Greater instability of negative mood (moment to moment fluctuations) has been associated with poorer psychological outcomes (Houben et al., 2015). Visual inspection of the SR-Positive and SR-Negative mood profiles suggested that while those with mild depression severity had more pronounced acute changes in positive mood than those with moderate depression severity, they also had longer period of reporting stable positive mood, a characteristic that has been linked to higher levels of well-being and lower levels of psychological distress (Cummins, 2010; Gruber et al., 2013).

Contrary to much of the existing literature, in Twitter across the 30 days, participants with mild depression severity used more negative emotion words than those with moderate depression severity. An example of this pattern can be seen in Participant 4, who at both T1 and T2 reported mild depression severity, but expressed the highest proportion of negative emotion words on Twitter over the 30-day study. This also contrasted with their SR-Negative mood profile, which revealed a significant lower mean negative mood over time than was expressed on Twitter. Negative emotion words are not only used by individuals with high depression severity. Personality is likely to be implicated in the characteristic mood tone people use online. For example, Park and

colleagues (2015) found that the use of negative emotion words on Facebook were related to lower levels of agreeableness, lower levels of conscientiousness, and lower levels of emotional stability (neuroticism).

Similarly, the purpose for using Twitter may influence language use. In relation to talking about mental health on Twitter, some users go online to vent how they are feeling and to seek social support (Berry et al., 2017). Talking about emotion on Twitter could be an adaptive combination of self- and interpersonal- emotion regulation strategies for some, where frequently expressing negative experience online may contribute to better mental health. Expressive writing on Twitter may have positive impacts on well-being overtime as it does in offline settings (Baikie & Wilhelm, 2005; Smyth, 1998) and beneficial online interpersonal processes (e.g. social support) also have an opportunity to occur. While our data do not provide insights into the most likely reason those with mild depression severity tended to express more negative language, the findings highlight how important confounding variable explanations may be for interpreting the psychological meaning/context/relevance of emotion language used on social media.

4.4 Strengths, Limitations and Future Directions

Our study provides a detailed exploration of mood data over time from two methods of experience sampling. Importantly, we highlight individual variation in within and between Twitter and self-report measures of mood, and demonstrate potential validity and reliability issues in sampling Twitter data as a measure inferring mood over time.

However, the study is clearly limited by the small sample size. We provide an in depth look at these participants, but all patterns need to be replicated in larger samples. Notably, the study considers four different time series features that might be considered in the future to compare self-report and Twitter profiles over time. The range of depression severity for our participants at T2 was also small (PHQ-9 scores of between 5 and 10).

Secondly, the specificity with which Tweets and self-reported mood was linked was hampered by technical issues. *MoodPrism* was built as a native app for iOS and Android and had

varying data collection accuracy and success in recording timestamp data for the daily mood reports. As a result, the capacity to link Tweets (with a discrete/specific timestamp) to daily mood reports was limited to the daily level and the frame within which a Tweet and daily mood report may have co-occurred is unknown. It is likely that there may have been greater similarity between the mood profiles presented if measurements of self-reported mood were linked to Tweets within a more restricted timeframe (e.g. 6hr intervals).

Third, to protect the user privacy, the app automatically pre-processed the Twitter data, such that we only had frequencies of the LIWC categories, and not the tweets themselves. While the LIWC dictionaries have been used across numerous studies to classify sentiment, they are also limited. The use single words, and are prone to signal discrepancies, such as lexical ambiguities and signal negation (Kern et al., 2016).

Finally, it should be emphasised that the interpretation of finding presented here should be read in the context of this sample only and not as findings generalizable to the population. The cases demonstrate just some of the complexity in mood and depression detection on Twitter and place this in the context of one other EMA method (smartphone self-report).

Our findings highlight the individual-level heterogeneity in mood expression and reporting on Twitter and in self-report. However, more research is required to explore third variables explanations for (1) why the mood signals from self-report and Twitter differ; (2) which individual characteristics influence language use in the context of depression and may disguise or augment psychological distress. This has significant implications for the way we use Twitter language as a passive data collection method for predicting depression and may aid in improving the overall sensitivity and specificity of prediction models.

5. Conclusions

Within-person or user-level observations are critical to the ongoing development of tailored and dynamic mental health monitoring. The current study examined the mood profiles of five participants obtained through smartphone self-report and via the emotion language used on Twitter

over a 30-day period. Participants mostly did not Tweet in a way that was consistent with their selfreported mood from day-to-day on average, but some patterns in the way people express emotion online, like how variable or unstable their emotion is, may be similar to their self-reported mood experiences over time.

There is a great deal of heterogeneity in the way people express emotion on Twitter and in how they self-report mood over time. The mood features that are informative of depression on social media may not be the same as those informative of depression through self-report. Exploring the individual characteristics that make these two mood signals similar or different is a crucial next step to improving the way information about mood and emotion dysregulation overtime can be used to predict depression, both from Twitter and in self-report.

References

- Arjmand, H.-A., & Rickard, N. S. (2018). Exploring the utility of a smartphone experience sampling applications (ESA) for exploring resilience to daily stressors. *Manuscript in Preparation*.
- Armey, M. F., Schatten, H. T., Haradhvala, N., & Miller, I. W. (2015). Ecological momentary assessment (EMA) of depression-related phenomena. *Current Opinion in Psychology*, 4, 21–25. https://doi.org/10.1016/j.copsyc.2015.01.002
- Arroll, B., Goodyear-Smith, F., Crengle, S., Gunn, J., Kerse, N., Fishman, T., ... Hatcher, S. (2010). Validation of PHQ-2 and PHQ-9 to screen for major depression in the primary care population. *Annals of Family Medicine*, 8(4), 348–353. https://doi.org/10.1370/afm.1139
- Baikie, K. A., & Wilhelm, K. (2005). Emotional and physical health benefits of expressive writing. *Advances in Psychiatric Treatment*, *11*(5), 338–346. https://doi.org/10.1192/apt.11.5.338
- Beck, A. T. (1974). The development of depression: A cognitive model. In *The psychology of depression: Contemporary theory and research*. (p. xvii, 318-xvii, 318). Oxford, England: John Wiley & Sons.
- Bernard, J. D., Baddeley, J. L., Rodriguez, B. F., & Burke, P. A. (2016). Depression, language, and affect: An examination of the influence of baseline depression and affect induction on language. *Journal of Language and Social Psychology*, 35(3), 317–326. https://doi.org/10.1177/0261927X15589186
- Berry, N., Lobban, F., Belousov, M., Emsley, R., Nenadic, G., & Bucci, S. (2017).
 #WhyWeTweetMH: Understanding why people use twitter to discuss mental health problems. *Journal of Medical Internet Research*, *19*(4), e107. https://doi.org/10.2196/jmir.6173

- Bollen, J., Gonçalves, B., Ruan, G., & Mao, H. (2011). Happiness is assortative in online social networks. Artificial Life, 17(3), 237–251. https://doi.org/10.1162/artl_a_00034
- Cavazos-Rehg, P. A., Krauss, M. J., Sowles, S., Connolly, S., Rosas, C., Bharadwaj, M., & Bierut,
 L. J. (2016). A content analysis of depression-related tweets. *Computers in Human Behavior*, 54, 351–357. https://doi.org/10.1016/j.chb.2015.08.023
- Cummins, R. A. (2010). Subjective wellbeing, homeostatically protected mood and depression: a synthesis. *Journal of Happiness Studies*, *11*(1), 1–17. https://doi.org/10.1007/s10902-009-9167-0
- De Choudhury, M., Counts, S., & Horvitz, E. (2013). Social media as a measurement tool of depression in populations. In *Proceedings of the 5th Annual ACM Web Science Conference* (pp. 47–56). ACM, New York, NY, USA. http://doi.org/10.1145/2464464.2464480
- De Choudhury, M., Kiciman, E., Dredze, M., Coppersmith, G., & Kumar, M. (2016). Discovering shifts to suicidal ideation from mental health content in social media (pp. 2098–2110). ACM Press. https://doi.org/10.1145/2858036.2858207
- Deters, F. G., & Mehl, M. R. (2013). Does posting facebook status updates increase or decrease loneliness? an online social networking experiment. *Social Psychological and Personality Science*, 4(5), 579–586. https://doi.org/10.1177/1948550612469233
- Ebner-Priemer, U. W., & Trull, T. J. (2009). Ecological momentary assessment of mood disorders and mood dysregulation. *Psychological Assessment*, 21(4), 463–475. https://doi.org/10.1037/a0017075
- Forest, A. L., & Wood, J. V. (2012). When social networking is not working: Individuals with low self-esteem recognize but do not reap the benefits of self-disclosure on Facebook. *Psychological Science*, 23(3), 295–302. https://doi.org/10.1177/0956797611429709

- Gil-Or, O., Levi-Belz, Y., & Turel, O. (2015). The "Facebook-self": Characteristics and psychological predictors of false self-presentation on Facebook. *Frontiers in Psychology*, 6. https://doi.org/10.3389/fpsyg.2015.00099
- Gkotsis, G., Oellrich, A., Hubbard, T. J., Dobson, R. J., Liakata, M., Velupillai, S., & Dutta, R.
 (2016). The language of mental health problems in social media. In *Third Computational Linguistics and Clinical Psychology Workshop (NAACL)* (pp. 63–73). Retrieved from http://www.anthology.aclweb.org/W/W16/W16-0307.pdf
- Grieve, R., Indian, M., Witteveen, K., Anne Tolan, G., & Marrington, J. (2013). Face-to-face or Facebook: Can social connectedness be derived online? *Computers in Human Behavior*, 29(3), 604–609. https://doi.org/10.1016/j.chb.2012.11.017
- Gruber, J., Kogan, A., Quoidbach, J., & Mauss, I. B. (2013). Happiness is best kept stable: Positive emotion variability is associated with poorer psychological health. *Emotion*, 13(1), 1–6. https://doi.org/10.1037/a0030262
- Guntuku, S. C., Yaden, D. B., Kern, M. L., Ungar, L. H., & Eichstaedt, J. C. (2017). Detecting depression and mental illness on social media: An integrative review. *Current Opinion in Behavioral Sciences*, 18(Supplement C), 43–49. https://doi.org/10.1016/j.cobeha.2017.07.005
- Houben, M., Van Den Noortgate, W., & Kuppens, P. (2015). The relation between short-term emotion dynamics and psychological well-being: A meta-analysis. *Psychological Bulletin*, 141(4), 901–930. https://doi.org/10.1037/a0038822
- Jahng, S., Wood, P. K., & Trull, T. J. (2008). Analysis of affective instability in ecological momentary assessment: Indices using successive difference and group comparison via multilevel modeling. *Psychological Methods*, 13(4), 354–375. https://doi.org/10.1037/a0014173

- Kern, M. L., Park, G., Eichstaedt, J. C., Schwartz, H. A., Sap, M., Smith, L. K., & Ungar, L. H.
 (2016). Gaining insights from social media language: Methodologies and challenges. *Psychological Methods*, 21(4), 507–525. https://doi.org/10.1037/met0000091
- Kocalevent, R.-D., Hinz, A., & Brähler, E. (2013). Standardization of the depression screener patient health questionnaire (PHQ-9) in the general population. *General Hospital Psychiatry*, 35(5), 551–555. https://doi.org/10.1016/j.genhosppsych.2013.04.006
- Koval, P., Pe, M. L., Meers, K., & Kuppens, P. (2013). Affect dynamics in relation to depressive symptoms: Variable, unstable or inert? *Emotion*, 13(6), 1132-41. http://doi.org/ 10.1037/a0033579
- Kroenke, K., Spitzer, R. L., & Williams, J. B. W. (2001). The PHQ-9. *Journal of General Internal Medicine*, *16*(9), 606–613. https://doi.org/10.1046/j.1525-1497.2001.016009606.x
- Larsen, M. E., Boonstra, T. W., Batterham, P. J., O'Dea, B., Paris, C., & Christensen, H. (2015).
 We Feel: Mapping emotion on Twitter. *IEEE Journal of Biomedical and Health Informatics*, *19*(4), 1246–1252. https://doi.org/10.1109/JBHI.2015.2403839
- Lindquist, K., Gendron, M., & Satpute, A. B. (in press). Language and emotion: Putting words into feelings and feelings into words. In *Handbook of Emotions* (4th ed.). New York, NY: The Guilford Press. Retrieved from https://www.unc.edu/~kal29/docs/Lindquistetal Handbook inpress.pdf
- Locatelli, S. M., Kluwe, K., & Bryant, F. B. (2012). Facebook use and the tendency to ruminate among college students: Testing mediational hypotheses. *Journal of Educational Computing Research*, *46*(4), 377–394. https://doi.org/10.2190/EC.46.4.d
- Marwick, A. E., & boyd, d. (2011). I tweet honestly, I tweet passionately: Twitter users, context collapse, and the imagined audience. *New Media & Society*, *13*(1), 114–133. https://doi.org/10.1177/1461444810365313

- Mauss, I. B., Levenson, R. W., McCarter, L., Wilhelm, F. H., & Gross, J. J. (2005). The tie that binds? Coherence among emotion experience, behaviour, and physiology. *Emotion*, 5(2), 175-190. http://doi.org/10.1037/1528-3542.5.2.175
- Miller, G. (2012). The smartphone psychology manifesto. *Perspectives on Psychological Science*, 7(3), 221–237. https://doi.org/10.1177/1745691612441215
- Moreno, M. A., Christakis, D. A., Egan, K. G., Jelenchick, L. A., Cox, E., Young, H., ... Becker, T. (2012). A pilot evaluation of associations between displayed depression references on Facebook and self-reported depression using a clinical scale. *The Journal of Behavioral Health Services & Research*, 39(3), 295–304.
- Moreno, M. A., Jelenchick, L. A., Egan, K. G., Cox, E., Young, H., Gannon, K. E., & Becker, T. (2011). Feeling bad on Facebook: depression disclosures by college students on a social networking site. *Depression and Anxiety*, 28(6), 447–455. https://doi.org/10.1002/da.20805
- Mowery, D., Smith, H., Cheney, T., Stoddard, G., Coppersmith, G., Bryan, C., & Conway, M. (2017). Understanding depressive symptoms and psychosocial stressors on twitter: A corpus-based study. *Journal of Medical Internet Research*, 19(2), e48. https://doi.org/10.2196/jmir.6895
- Park, G., Schwartz, H. A., Eichstaedt, J. C., Kern, M. L., Kosinski, M., Stillwell, D. J., ... Seligman, M. E. P. (2015). Automatic personality assessment through social media language. *Journal of Personality and Social Psychology*, *108*(6), 934–952. https://doi.org/10.1037/pspp0000020
- Park, G., Schwartz, H. A., Sap, M., Kern, M. L., Weingarten, E., Eichstaedt, J. C., ... Seligman, M.
 E. P. (2017). Living in the past, present, and future: Measuring temporal orientation with language. *Journal of Personality*, 85(2), 270–280. https://doi.org/10.1111/jopy.12239

- Pennebaker, J., Mehl, M., & Niederhoffer, K. (2003). Psychological aspects of natural language use: our words, our selves. *Annual Review of Psychology*, 54, 547-77. http://doi.org/ 10.1146/annurev.psych.54.101601.145041
- Pennebaker, J. W., Chung, C. K., Ireland, M., Gonzales, A., & Booth, R. J. (2007). The LIWC 2007 Application. Retrieved from http://www.liwc.net
- Preotiuc-Pietro, D., Eichstaedt, J. C., Park, G., Sap, M., Smith, L., Tobolsky, V., ... Ungar, L. H.
 (2015). The role of personality, age and gender in tweeting about mental illness (pp. 21–30).
 Presented at the *NAACL: Workshop on Computational Linguistics and Clinical Psychology: From Linguisitc Signal to Clinical Reality*, Denver, CO, USA. Retrieved from
 www.aclweb.org/anthology/W15-1203
- Rickard, N., Arjmand, H.-A., Bakker, D., & Seabrook, E. (2016). Development of a mobile phone app to support self-monitoring of emotional well-being: A mental health digital innovation. *JMIR Mental Health*, 3(4), e49. https://doi.org/10.2196/mental.6202
- Rude, S., Gortner, E.-M., & Pennebaker, J. (2004). Language use of depressed and depressionvulnerable college students. *Cognition and Emotion*, 18(8), 1121–1133. https://doi.org/10.1080/02699930441000030
- Russell, J. A. (1980). A circumplex model of affect. *Journal of Personality and Social Psychology*, *39*(6), 1161–1178. https://doi.org/10.1037/h0077714
- Sato, H., & Kawahara, J. (2011). Selective bias in retrospective self-reports of negative mood states. *Anxiety, Stress, and Coping*, 24(4), 359–367. https://doi.org/10.1080/10615806.2010.543132
- Schwartz, H. A., Eichstawdt, J., Kern, M. L., Park, G., Sap, M., Stillwell, D., ... Ungar, L. (2014).
 Towards assessing changes in degree of depression through Facebook. In Workshop on
 Computational Linguistics and Clinical Psychology: From Linguistic Signal to Clinical

Reality (pp. 118–125). Baltimore, Maryland, USA: NAACL Retrieved from http://www.acl2014.org/acl2014/W14-32/pdf/W14-3214.pdf.

- Seabrook, E. M., Kern, M. L., Fulcher, B. D., & Rickard, N. S. (2018). Depression is predicted by emotional instability on Facebook, but by reduced emotion variability on Twitter. Submitted manuscript.
- Seabrook, E. M., Kern, M. L., & Rickard, N. S. (2016). Social networking sites, depression, and anxiety: A systematic review. *JMIR Mental Health*, 3(4), e50. https://doi.org/10.2196/mental.5842
- Seidman, G. (2013). Self-presentation and belonging on Facebook: How personality influences social media use and motivations. *Personality and Individual Differences*, 54(3), 402–407. https://doi.org/10.1016/j.paid.2012.10.009
- Selim, H. A., Long, K. M., & Vignoles, V. L. (2014). Exploring identity motives in Twitter usage in Saudi Arabia and the UK. *Studies in Health Technology and Informatics*, 199, 128–132.
- Settanni, M., & Marengo, D. (2015). Sharing feelings online: studying emotional well-being via automated text analysis of Facebook posts. *Frontiers in Psychology*, 6. https://doi.org/10.3389/fpsyg.2015.01045
- Smyth, J. M. (1998). Written emotional expression: effect sizes, outcome types, and moderating variables. *Journal of Consulting and Clinical Psychology*, 66(1), 174–184.
- Tufekci, Z. (2014). Big questions for social media big data: Representativeness, validity and other methodological pitfalls (pp. 505-514). In *Proceedings of the 8th International Conference on Weblogs and Social Media*, ICWSM, Ann Arbor, USA. Retrieved from http://arxiv.org/abs/1403.7400

- Von Neumann, J., Kent, R. H., Bellinson, H. R., & Hart, B. I. (1941). The mean square successive difference. *The Annals of Mathematical Statistics*, 12(2), 153–162. https://doi.org/10.1214/aoms/1177731746
- Waterloo, S. F., Baumgartner, S. E., Peter, J., & Valkenburg, P. M. (2017). Norms of online expressions of emotion: Comparing Facebook, Twitter, Instagram, and WhatsApp. *New Media & Society*, 1461444817707349. https://doi.org/10.1177/1461444817707349
- Wichers, M. (2014). The dynamic nature of depression: A new micro-level perspective of mental disorder that meets current challenges. *Psychological Medicine*, 44(7), 1349–1360. https://doi.org/10.1017/S0033291713001979

Appendix A

Emoji & internet slang additions to the LIWC 2007 emotion dictionaries

Positive	Negative		Posit	ive Emotior	Negative Emotion Internet				
Emoji's	Emoji's		Internet Slang			Slang			
=D>	:-1	:s	Aight	Ilms	lmao	Abft	fkd	r8pist	
=D	:1	:-Q	Aightz	Ilu*	lmfao	Acgaf	fker	stupd	
:D	X(:Q	Aiight	Ilshipmp	rofl	Aiic	fkin	suk	
:-D	X-(:-\$	Aite	Irly	lml	Arsed	fking	sukz	
(:)	0	:\$	Alrt	Ite	lov	Awk	fomo	suxx	
(:-)	b (:-/	alryt	Jj	luv	Bm	gank	wrdo	
>:D<	>:-(:/	Alol	j/j	luff	Boom	grmbl	wtf	
:))	>:-)	T_T	Aml	jk	luvv	Bord	h8		
:-X	(:-&	T^T	Aprece8	j/k	lurve	Bovered	h80r		
:X	:@	QQ	Apreci8	jks	lv	Catwot	h83r		
:)	:(=O::::	Awes	j2f	lve	Cba	h8ed		
:-)	:-t	='(Awsm	j4f	lyk	Cbb	h8r		
~~	:-11	&.(Awsome	j4g	lyke	Cbf	h8red		
:,')	:11	(;_;)	Bahaha	j41	marvy	Cbfa	h8s		
:-")	:t	:'(Besos	jfg	njoy	Cbfed	h8t		
:")	:-@	:'	Bilu	jfl	ql	Cmeo	h8t0r		
;)	:{	:'-(Btwilu	jfk	qool	Cmw	h8t3r		
;-)	:-Z	:*(Btwitiailwu	jja	qt	Cof	h8te		
\0∕	:Z	:*-(Byak	jkz	qte	Cotf	h8tr		
;-(??	:_(Chillin	k	rofl*	Dc	hait		
;(@_@	:[Ctm	kay	*s*	Dbm	heyt		
:')	(*)	;_;	Cut3	kewel	S'ok	Dfc	htr		
:'-)	(0_0)	=\	Eil	kewl	Snog	Dgac	1d10d		
>-)	b-(:1	Fab	KK	Sok	Dgaf	ıdjit		
:p	%+l	:(& E	Fah	kl 1	Sof	Dgara	1dot		
:P	:5	:-E	F19	KOO	Stm	Dgas	idyat		
:-P	:-5	:-e	Funee	KOOI	Ub3r	Eejit	invu		
:-p	(:-		Funy	KSS	WOUT	Eedyat	1rny		
:>	:(/:(*g	KSSO	WOOU	EI	1SD		
:->	:-(:-[.r	Gtm		Xellent Vlnt	EI-ing	kmn lrmn		
	1-0	:[.)	Gud		AIII	Effect	KIIIP 1an		
^ <u>.</u> ^	\-0	;-} .)	Guua	loff	XIC	Ellin	18f 1ar		
		;} #)	Hell	101		Elling E#alsing	IZI		
0(4)		#-) #)	Пру			F'n	IIV ofcol		
<i>%)</i>	+-((#) • 6	Пру			Г II fing	olcol		
%-) ~2	$\langle J \rangle$	0 :6	Папа 19			1-mg	noos		
	(0)	.0 :_@	Io	liek*		fm1	n/oed		
\\u_J		@	light	lik		1.111.1 farg	proced		
		(.('-<	ligh	11K 11k		fck	pbeu nh33r		
	0.0 %)	>	Iidl	lke		fcked	phool		
	%_)	•//	I)u IIi			fckin	phayi		
	(@ @)	•/ /	Ilml	llf		fcking	r8n		
				111		feck	roh		
						fk			

Appendix B Raw Data Panel Plots and Features Comparison Table

Table B.1

Raw score values for descriptive mood profile features from Self-Report and Twitter positive and negative mood data.

Feature	Participant 1		Participant 2		Participant 3		Participant 4		Participant 5		
	SR	Twitter	SR	Twitter	SR	Twitter	SR	Twitter	SR	Twitter	
Observations	<i>n</i> = 27	<i>n</i> = 9	<i>n</i> = 22	<i>n</i> = 12	<i>n</i> = 29	<i>n</i> = 28	<i>n</i> = 26	<i>n</i> = 26	<i>n</i> = 30	<i>n</i> = 14	
Positive Mood											
Average	1.89	.02	0.95	.04	2.21	.07	2.58	.03	1.93	.01	
iSD	0.32	.02	0.95	.04	0.73	.05	0.58	.03	0.91	.02	
Instability	0.21	.001	1.38	.002	0.50	.004	0.49	.002	0.93	< .001	
PAC	.19	.00	.28	.09	.04	.11	.04	.12	.10	.15	
Negative Mood											
Average	.37	.07	2.41	.04	0.28	.03	0.08	.09	0.87	.05	
iSD	.74	.07	1.18	.04	0.53	.03	0.27	.05	0.86	.06	
Instability	.77	.01	2.51	.001	0.57	.002	0.13	.01	1.24	.004	
PAC	.08	.13	.17	.09	.07	.11	.12	.08	.14	.15	

Note: Self-report mood absolute range from 0 to 4; Twitter emotion word proportions absolute range from 0 to 1.



Negative Mood



Figure B.1. Raw score dual axis plots of self-reported daily mood and daily emotion words in Tweets across the 30-day study.

Positive Mood

Negative Mood



B.1. continued. Raw score dual axis plots of self-reported daily mood and daily emotion words in Tweets across the 30-day study.



B.1. continued. Raw score dual axis plots of self-reported daily mood and daily emotion words in Tweets across the 30-day study.

5.3 Concluding Remarks

This chapter presented the self-report and Twitter data from five participants over a 30-day period of using *MoodPrism*. It demonstrated that for these participants, there was considerable inconsistency between their self-reported daily mood and the emotion they expressed on Twitter. Greater depression severity may influence the extent to which self-report and Twitter mood profiles converge, with the cognitive and emotional aspects of depression becoming more salient in Tweets than other confounding factors. In the following chapter, the findings of this thesis are summarised, and an integrated discussion provided.

Chapter 6

General Discussion

6.0 Introduction

This final chapter provides an integrated discussion of the findings presented in the previous chapters. It begins with an overview of the broad research aims, followed by a summary of the key findings and contributions from each of the three studies. The key contributions from each paper are then discussed together and integrated with theory and previous research. Strengths, limitations and implications of the research are presented, including practical recommendations for social media users and providers, clinicians, and researchers. Directions for future research are identified, followed by concluding remarks.

6.1 Summary of the Main Findings

This thesis set out to explore the use of emotion language on social networking sites and its association with depression. To achieve this objective, three studies were conducted. Investigation began with a broad lens encompassing depression, anxiety, and well-being in a systematic review seeking to identify the social media behaviours that were linked to mental health in previous research. Following this review, the scope of the remaining papers was narrowed to focus on depression and the content and tone of social media disclosures to address gaps identified in this literature. Critically, both experimental papers introduced the use of time-sensitive observations of emotion variability identified in social media user's language.

6.1.1 Systematic review of the literature.

The first paper presented in this thesis was a systematic review of the literature. The aim of the systematic review was to integrate the findings related to depression and anxiety across social media platforms. As social networking sites are inherently complex environments, the review also aimed to examine potential mediators and moderators between social networking site use and depression and anxiety. It also explored how well-being has been integrated into the existing research as a part of presenting a balanced view of the impact that social networking site use may have across the dual continuum of mental health. This systematic review also held the purpose of contextualising the development of research questions for the remaining studies presented in this thesis.

Due to the broad focus of the systematic review presented here, the understanding of social networking sites, and their intersection with mental health, has been significantly extended from that of previous reviews that had predominantly focused on adolescents and young people (e.g., Best, Manktelow, & Taylor, 2014; Spies Shapiro & Margolin, 2014) or on a single social networking site platform (e.g. Wilson, Gosling, & Graham, 2012). There was also a growing need for a systematic review dedicated to the examination of high prevalence disorders, such as depression and anxiety. This need is highlighted by the simultaneous publication of another systematic literature review examining the association between social networking sites and depression by Baker and Algorta (2016).

The systematic review presented in Chapter 2 returned 70 studies meeting the selection criteria and addressed the following themes: frequency of social networking site use, size and structure of social networking sites, language features and observable social networking site activities, self-disclosure and expression, quality of interactions, social support, social connectivity, social comparison, addictive and problematic behaviours, and physiological associations. Overall the findings from studies within these themes revealed both protective and risk factors for depression and anxiety. The inclusion of both depression and anxiety also highlighted potential differences in the behavioural patterns of social networking site users; predominantly, differences in the way individuals used social networking sites (passively or actively) and in the interactional style with which individuals communicated with peers. Some of the most consistent associations between depression and social networking site use were in relation to the content and tone of status updates; although, there was little evidence to suggest status update content was also linked to

detecting anxiety. Other features, such as the size of an online social network and the overall time spent on social networking sites, were less informative of depression and anxiety. Importantly, cognitive aspects, personality, gender, and age were frequently revealed to have moderating or mediating influences on the relationships between social networking site use and depression or anxiety. Therefore, the consideration of individual differences is an important area for future research about social networking site use.

Several gaps in previous research were identified in the systematic review and guided the empirical investigation of Chapters 4 and 5. First, there was a clear sampling bias in the literature where most samples consisted of young people (adolescents to university aged). In the context of an ever-evolving social networking site landscape, both in terms of platform popularity and the types of individuals that that use them (Greenwood, Perrin, & Duggan, 2016), it becomes increasingly important to sample individuals from a wide range of backgrounds to better represent the population of SNS users and the likely changes in usership over time in the online environment. Further, the majority of studies only considered Facebook as the SNS platform for investigation. This limits the generalisability of findings, specifically in relation to understanding the social networking site specific mechanisms relevant to the mental health. Tufekci (2014) discussed this limitation in the context of big data research on social media, with Twitter as a model organism. There are SNS platform-specific interactions and characteristics (e.g., retweets, friending versus following) which attract particular users as well as enable specific patterns of behaviour that may not transfer across different SNS services. Therefore, multi-platform comparisons are needed to reduce *mechanisms bias* and inform how results from different SNS platforms are interpreted (Tufekci, 2014).

Additionally, across the literature there has been an overreliance on the use of self-reported estimates of social media behaviour. As with most self-report methods, sampling retrospective accounts of behaviour may provide inaccurate measures thus threatens the validity of the conclusions that can be made (Trull & Ebner-Priemer, 2009). The potential for retrospective

sampling to bias results was demonstrated by Steers, Wickham, and Acitelli (2014) where the association between the frequency of SNS use and depression was assessed by retrospective survey (Study 1) and by daily ESM diaries (Study 2) and findings differed by method (Study 1: positive association; Study 2: no association). While ESM approaches to estimating SNS behaviour increase the precision of recall (Steers et al., 2014; Trull & Ebner-Priemer, 2009), directly sampling SNS data further addresses retrospective biases by taking advantage of observing the digital record of a participants SNS use. From a practical standpoint, building understanding of mental health directly from social networking site data also has the major advantage of providing both (1) an unobtrusive method of data collection from participants and (2) a potential method of intervention or service delivery to social networking site users.

Finally, while not explicitly considered within the systematic review paper there are other unexplored factors that could be important. It became apparent that among the studies directly sampling social networking site data, few studies utilised the time-structured nature of observations to explore behavioural patterns across time at the individual level. While some studies have included cross-sectional follow ups (T1 and T2) (e.g., Davila et al., 2012; Feinstein et al., 2013; Kross et al., 2013; Szwedo, Mikami, & Allen, 2011), fewer have incorporated experience sampling methods or direct observation that included discrete time variables (De Choudhury, Gamon, Counts, & Horvitz, 2013; Jelenchick, Eickhoff, & Moreno, 2013; Kross et al., 2013). Similarly, few studies have sought to critically examine the features extracted from social networking site data that were associated with depression for their content validity. For example, the use of negative emotion words in status updates was frequently associated with depression, though the accuracy with which these expressions sampled negative mood was unknown.

Taken together, these research gaps were addressed by building social media data collection into the *MoodPrism* app. A paper detailing the development and testing of this app was presented in Appendix A of this thesis and a description of the social media data collection development and testing was presented in Chapter 3. This method of data collection indirectly addressed some of the sampling bias in the literature by drawing from the general population, and addressed the mechanisms bias by including a cross-platform comparison capability. Also, *MoodPrism* allowed the explicit introduction of time variables and the collection of non-social networking site variables (e.g., daily mood reports, demographic characteristics) to explore patterns of social media emotion expression over time and in the context of a complimentary smartphone experience sampling method.

6.1.2 Examination of emotion dynamics is social media posts in relation to a static depression severity index.

Building on the findings of the systematic literature review, Chapter 4 aimed to further explore the content and tone of status updates on Facebook and Twitter and their relationship with depression. This was achieved by incorporating emotion variability and instability as potential predictors of depression severity. The time-ordering of emotion word observations was addressed by exploring patterns of instability over the study recording period. It was hypothesised that negative emotion instability would be positively associated with depression severity on both Facebook and Twitter and that variability features would be more sensitive to depression severity than the average proportion of emotion words alone.

The results of this study revealed two different patterns of association on Facebook and Twitter between the emotion language variables sampled and depression severity. For Facebook users, depression severity was significantly and positively associated with negative emotion word instability. This relationship remained even when controlling for the average proportion and variability of negative words used. In contrast, for Twitter users, there was a negative association between negative emotion word variability and depression severity that was retained when controlling for the average proportion of negative words. The average proportion of positive and negative emotion words used was not associated with depression severity on either Facebook or

Twitter, contrary to hypotheses. Further, posting more frequently (greater average number of posts per day, smaller median time interval between posts) was positively associated with depression severity, but only on Facebook.

Exploratory analyses were also performed to explore potential reasons for the different patterns between Facebook and Twitter in the emotion variability features that were associated with depression. Firstly, there were differences in the overall language patterns used between platforms. Compared to Facebook users, Twitter users expressed a greater average proportion of negative emotion words that was more variable and more unstable. Supplementary analyses also revealed several psychological differences between samples, primarily that participants using Twitter had significantly higher levels of conscientiousness, self-esteem, and social desirability than users of Facebook.

6.1.3 Emotion language expressed on social media and concurrent self-reported emotion: The impact of depression severity.

The final study sought to critically explore the use of emotion language as an indicator of depression severity. This was achieved by looking at the link between the mood expressed on Twitter and the same-day self-reported mood to *MoodPrism*. It compared users' mood data from both sources by considering same-day associations as well as looking at broader patterns of mood over time to probe the validity of using social media language as an indicator of daily mood.

Through the exploratory investigation of individual cases, this study revealed that there was little similarity between self-reported mood and the mood expressed on Twitter at a same-day level and at a broader mood profile level (i.e. average, variability, instability, and acute changes). Data suggested that as depression severity increases, there is greater similarity between self-report and Twitter mood profiles. It could be that for individuals with more severe depression severity, Twitter emotion language may be a more valid and reliable indicator of depression risk by accurately tapping into patterns of negative mood. However, this may also highlight emotion language may

not be driven by mood experiences or depression symptoms alone for cases with mild depression. Instead, it may be driven by other third variables, thereby reducing emotion language's reliability in sampling mood and contributing to false-positive depression identification in language-driven depression prediction models.

6.2 Contributions of this Thesis

Bringing together the findings from the systematic review and the two empirical papers, this thesis extends the understanding of the links between the emotion expressed on social media and depression by incorporating an investigation of mood profiles over time and considering this in the context of a complementary experience sampling method. The research presented here specifically sought to explore:

(1) What emotion dynamic features in the mood profiles generated across status updates are associated with depression severity? and;

(2) Do the emotions expressed on social media through language accurately reflect subjective daily mood?

6.2.1 Mood profile features in status updates predict depression severity.

Examining emotion dynamic features through social media language data is a novel approach developed in this work made much more accessible by smartphone technology. While there are caveats to the use of social media language for inferring mood (discussed below), the work reported in this thesis demonstrated that several indices for examining the mood profile features of language across status updates on social media were useful and illustrated how they might predict depression severity. Beyond looking at the average proportion of emotion language use alone, this research introduced variability, instability, and the proportion of acute change as metrics that may provide additional insights for depression by tapping into emotion processes on social media.

By focusing on the time-sensitive nature of the data collected from Facebook and Twitter, the findings in Chapters 4 and 5 revealed that the way emotion is expressed on social media is not necessarily stable over time (as might be described when taking an average) and, for some individuals, there was a great deal of variation in the way they expressed emotion online. Negative emotion word instability was associated with greater depression severity for Facebook users, consistent with the studies examining emotional instability using self-report and experience sampling methods (Houben, Van Den Noortgate, & Kuppens, 2015). Negative emotion variability was associated with lower depression severity for Twitter users. This pattern contrasted with that found for Facebook users and suggested a more restricted range of negative emotion expression for depressed users of Twitter and highlights the importance of considering cross-platform comparisons due to differences in the communication mechanisms and user demographics (Tufekci, 2014).

Exploring the emotion dynamics of language on social media is challenging due to the heterogeneity in the way people post content overtime. Compared to signal-based experiencing sampling methods which seek to collect data from participants following a signal at specific time-points of within certain time-frames, social media data is irregular as it is event-based (i.e., the posting of a status update) and derived from a natural context (Park et al., 2015; Wheeler & Reis, 1991). Measures such as the MSSD and PAC require even time-intervals between observations to be calculated meaningfully. To account for the unique pattern of posting for each individual and to adjust for the uneven spacing between status updates, a median time-adjustment to the MSSD and PAC (introduced by Jahng, Wood, and Trull, 2008) was applied in the current research. In this context, fluctuations in emotion between language samples were considered in the context of an individual's median posting habit. This is an elegant solution to the irregular nature of social media data to implement and is recommended for consideration in samples where the formal assumptions of time-series analysis cannot be met. The research presented in this thesis demonstrated the

application of these temporally-sensitive metrics, adding depth to the use of emotion language from social media as an indicator of depression.

6.2.2 The language-mood gap: Implications for depression prediction from social media content.

A major contribution of this thesis was linking subjective daily reports of mood to the daily aggregate mood expressed on social media, specifically on Twitter. This was primarily addressed in Chapter 5 and revealed a 'language-mood gap' for most participants between the 30-day mood profiles obtained in self-report and from Tweets.

Psychological constructivist theories argue emotion experiences are partially constructed and understood through language and that language may also play a role in regulating emotion through reappraisal (Brooks et al., 2017; Lindquist, Gendron, & Satpute, in press; Wood, Lupyan, & Niedenthal, 2016). As indicated in the systematic review presented in Chapter 2, the cognitive processes involved in depression (e.g., negative cognitive biases, rumination) manifest on social media, particularly in the content and tone of status updates, providing the opportunity to observe processes of emotion (dys)regulation over time.

This thesis demonstrated that measures of emotion language variability and instability hold important information about depression by potentially sampling emotion processes over time, like heightened reactivity to events or alternatively a restricted range of emotional experiences (discussed in Chapter 4). However, the patterns of emotion expression on Twitter did not clearly reflect the patterns of self-reported mood in Chapter 5. This could indicate that the emotion language expressed on social media is not sensitive to the fluctuations of daily mood, where expressions of emotion on social media are amplified or dulled in relation to lived experiences by other third variables such as personality (i.e. its impact on expressive online behaviours and as factors that may constitute emotional traits [extraversion, neuroticism]; Park et al., 2015). This is important in the context of emotion response coherence, in that behavioural and experiential emotion are impacted upon contextual and personal factors that may distort the similarities between signals (Mauss et al., 2005).

The different patterns in mood profile features between self-report and Twitter may also reflect a key difference in the nature of emotion information that is sampled from the two methods. In the daily mood reports, participants were asked to respond to how they were currently feeling in the moment which is designed to capture emotion-episodes and reduce recall bias (Robinson & Clore, 2002). In contrast to the brief responses provided to *MoodPrism*, expressing emotion on social media is not necessarily temporally bound to the time of the post; it may refer to an event earlier in the day, providing time for social media users to reflect and reconstruct their experiences over time (Park et al., 2017). Writing and editing a status update may perform an emotion regulatory role similar to expressive writing interventions where writing about an experience aids cognitive processing (Baikie & Wilhelm, 2005; Baker & Moore, 2008; Pennebaker, 1997). Rather than sampling emotion in-the-moment, the emotion language on social media may sample emotion 'post-regulation', more closely reflecting the additional cognitive processes tied to written emotion expression. These may include rumination or reframing, and in the context of social media may also reflect conscious choices related to constructing status updates as a part of presenting an online identity or catering to specific audiences (Baikie & Wilhelm, 2005; Derks, Fischer, & Bos, 2008; Lin, Tov, & Qiu, 2014; Lindquist et al., in press; Locatelli, Kluwe, & Bryant, 2012; Qiu, Lin, Leung, & Tov, 2012).

As such, there are several differences between posting a status update on social media and communicating emotion to others face to face or in private online messaging (IM). In contrast to IM which are private messages to a small audience, status updates are a one-to-many, may be broadcast (or intended for) close friends, acquaintances, and/or strangers, and may be posted as a part of presenting an ideal online persona (boyd & Ellison, 2007; Ellison & boyd, 2013; Gil-Or, Levi-Belz,

& Turel, 2015; Lin et al., 2014; Seidman, 2013). Positive self-presentation may be enhanced on social media due to the enduring nature of the content posted and potential audience, impacting on the valence of emotion expressed (Gil-Or et al., 2015; Seidman, 2013). Similarly, emotional traits (extraversion and neuroticism) may also influence the average valence expressed on social media, obscuring the emotion expression related to mental health (Park et al., 2015; Winter et al., 2014). Other personality factors like conscientiousness also impact on social media behaviours by predisposing individuals to impulsive online behaviour (low conscientiousness) or highly controlled online expression (high conscientiousness; Baiocco et al., 2017; Hughes, Rowe, Batey, & Lee, 2012). As demonstrated in the Chapter 4 supplementary materials, these individual characteristics differed between our Facebook and Twitter samples and may partially account for the differences in emotion patterns linked to depression. Similarly, the divergent mood signals between Twitter and the *MoodPrism* self-report may be explained through the influence of other characteristics that impact on the use of emotion language in addition to the presence of depressive symptoms.

This highlights two parallel conversations in the literature. On the one hand, research has consistently demonstrated that personality, social media use motivations, gender and age are implicated in the way people use social media and in the relationships social media use have with mental health (Baker & Algorta, 2016; Frost & Rickwood, 2017; Preotiuc-Pietro et al., 2015; Schwartz et al., 2013; Seidman, 2013; Settanni & Marengo, 2015). In the systematic review presented in Chapter 2, these variables emerged as moderators (or mediators) to the relationship between social media use and depression or anxiety. On the other hand, the practical application of predictive techniques to detect mental illness such as depression have predominantly utilised social-media only variables or features (language, time of use, reciprocity, social network structure) (Calvo, Milne, Hussain, & Christensen, 2017; Guntuku, Yaden, Kern, Ungar, & Eichstaedt, 2017). For instance, in a recent integrative review of the research using social media to detect mental illness, only two of the 12 studies reviewed utilised non-social media predictors like age, gender

and personality (Guntuku et al., 2017). These studies showed that personality impacts on who might disclose about mental health and that age improves the detection performance for post-traumatic stress disorder and anxiety from among other comorbid conditions health (Benton, Mitchell, & Hovy, 2017; Preotiuc-Pietro et al., 2015). It is clear the integration of non-social media variables is a necessary next step to improve the predictive accuracy of depression status, particularly in reducing the potential for false-positive.

6.3 Practical Implications

The findings of this thesis suggest that the observation of emotion language on social media can be used to predict depression, though further attention should be paid to the time-dependent nature of observations and the additional insights for depression that we can gain through looking at (maladaptive) patterns in emotion language. Individual variability in emotion expression may be a useful discriminator of depression severity for individuals who express similar levels of negative emotion on Facebook. As a tool for detecting depression, social media is a scalable and unobtrusive means of accessing information relevant to monitoring public mental health. At the same time, there are caveats to the use of social media emotion language alone as a means of inferring mental health and addressing the gap between online expression and self-reported experiences of mood may improve the sensitivity and specificity of language models predicting depression.

6.3.1 For Users of Social Media

In Chapter 2, the use of social media was revealed to potentially have both positive and negative impacts on a social media user's mental health. Rather than clearly being linked to the time spent online, it is likely that the content and tone of interactions on social media play a greater role in depression and anxiety. Similarly, feelings of social media addiction were linked to poorer mental health. Despite these potential negative impacts, much research revealed positive influences of social media use, particularly in reducing loneliness, increasing social connectedness, and providing a forum to receive social support (Berry et al., 2017; Deters & Mehl, 2013; Lee, Noh, &

Koo, 2013; Radovic, Gmelin, Stein, & Miller, 2017). Social media is a place where people seek mental health support through their personal networks and through the formation of online groups (Berry et al., 2017; Naslund et al., 2017). Social support is likely to be readily accessed by those with high social anxiety, but may less clearly perceived by those with depression (Indian & Grieve, 2014; Park et al., 2016). Self-reflection may be a useful tool for social media user who feel it is having a negative impact on their daily functioning and spending some time away from social media may be a good strategy.

6.3.2 For Researchers

It is clear there is complexity in using language as a means of inferring mood and mental health. This thesis has suggested the presence of a language-mood gap that presents a significant challenge for improving the predictive sensitivity and specificity of automatic language-based screening of depression from social media data. Language is not a perfect data source for inferring internal events and there appears to be a need to conduct research establishing the content validity and reliability of the features extracted from social media records as accurate indicators of the psychological constructs they are argued to reflect. This thesis presented a method for approaching this task – an integrated experience sampling method that collects social media data and concurrent self-reported mood.

There are some notable advantages to developing smartphone apps as a means of collecting data for research. They allow researchers to leverage data from existing databases (e.g., Twitter and Facebook) and can be customised to meet specific research goals (Miller, 2012; Rickard, Arjmand, Bakker, & Seabrook, 2016). The advantages for social media research are more pronounced as data collection via a smartphone app can provide *ground truth* data for a range of psychological and demographic characteristics. There are, however, challenges to using a smartphone app for research. Apps are a consumer choice and require researchers to consider engaging and user-friendly interfaces to ensure participant retention, particularly where an app is required to be on the

participants smartphone for an extended period of time to collect data. In the context of *MoodPrism* this was achieved by providing engaging daily feedback to users on their mood experiences. Bakker, Kazantzis, Rickwood, and Rickard (2016) provide further discussion on the integration of gamification, simple user interface, and notification in designing smartphone apps for interventions. Further discussion on of the advantages and challenges to designing and implementing a smartphone app for research are presented in Appendix A.

This thesis also suggested that looking at emotion dynamic features expressed through language, in addition to the relative frequency of emotion word use, may improve the ability to discriminate between depressed and non-depressed individuals. Indeed, the cognitive processes involved in emotion dysregulation may be better revealed through social media language as it is likely to reflect emotion post-regulation. In this way, exploring emotion dynamics on social media may be a useful way to tap into daily experiences that may build toward psychopathology. Across Chapters 4 and 5, this thesis demonstrated the use of several metrics for examining emotion dynamic features and how they may be applied to social media language. Further exploration of these metrics may yield useful reference values or pattern identification for social media language for the detection of depression.

6.3.3 For Social Media Providers and Mental Health Services.

The use of social media undoubtedly has an impact on mental health and, by the same token, has the opportunity to improve access to care for those who need it. Social media providers should be wary of the potential negative impacts of their platforms (reviewed in Chapter 2) and work through design to encourage behaviour that may be beneficial for mental health. For example, Facebook's research team have recently indicated that they are working on ways to increase prosocial interactions that may have positive impacts for user's well-being and social connectedness (Morris, 2017).
Increased collaboration between social media providers, researchers, and local mental health services are required to develop comprehensive approaches to depression detection (or triage) and the delivery of interventions or resources to those who may benefit from them. A recent example of such a collaboration is between Twitter and ReachOut an – Australian youth e-mental health support service – who developed online resources to support young people in navigating distressing social media discussions (ReachOut, 2017). Recommendations included reflecting on the impact online discussions were having and encouraging youth to learn to take a break from social media when it was having a persistent negative impact (ReachOut, 2017).

The research in this thesis highlighted several behaviour patterns that are likely to be uniquely associated with depression. Researchers developing predictive models for depression may wish to integrate these tools with local community resources or online interventions (as above) and collaborate with social media providers to deliver this content through push notifications, targeted advertising, or alerts. Indeed, recent findings have indicated that individuals who self-identify with mental illness would be open to the delivery of these approaches via social media (Naslund et al., 2017). Such an approach may be even more impactful for those who are low in help-seeking behaviour or alternatively are little insight into their distress. While key discussions are required around the ethical considerations of using social media data to infer mental health status and how to best obtain social media users informed consent, such collaborations provide the opportunity to apply research and community resources in detecting previously undetected cases of depression, and deliver interventions at an early stage (Guntuku et al., 2017).

6.4 Strengths and Limitations

In this section the research strengths and limitations of the research are discussed. This section also reflects on the technical considerations of integrating social networking site data with daily experience sampling of self-reported mood.

A major strength of the method presented in this thesis was the ability to integrate multiplatform comparisons and complimentary experience sampling data collection methods with the use of the *MoodPrism* app. As discussed earlier, comparing social media platforms is critical in ensuring the mechanisms specific to each SNS are accounted for in the interpretation of findings (Tufekci, 2014). As shown in Chapter 4, the patterns of emotion expression across Twitter and Facebook differed significantly, as did their associations with depression, and this may have been due to the type of people using the platform and/or the differing communication mechanisms. A similar multi-platform comparison was not performed in Chapter 5 as there were no participants within the total *MoodPrism* sample who had posted on Facebook concurrently with the completion of the daily mood reports. In our sample, on average Facebook users posted less frequently than Twitter users (see Chapter 4). It suggests the need for an alternate methodology that may better capture Facebook users. This may involve observing Facebook users over a longer time-period (e.g. 12 months) and linking self-report prompt notifications directly to posting activity on Facebook to capture a concurrent mood rating.

The data collection method in this thesis also integrated two methods of ecological momentary assessment of mood, one from signal-dependent self-report (that is, the daily mood reports) and the other from event-contingent sampling (that is, Tweets). These complementary methods collected data in real-time in a minimally intrusive way and allowed the incorporation of temporal features in analysis. This methodology also revealed potential reliability and validity issues in inferring mood from social media content by providing a comparative concurrent external criterion or "ground truth", which had not yet been addressed in other research. While both EMA methods contain their own limitations, Tufecki (2014) suggests that such multi-method approaches generate "richer answers" to research questions on social media.

Despite the benefits provided by the methodology used in this thesis, as highlighted in Chapters 4 and 5, a major limitation of this research was sample size. An original research plan had

the intention to incorporate and control for third variables consistently across the analyses (personality, gender, age, social desirability, significant life events), however, the sample size obtained meeting the research inclusion criteria did not make this feasible. There are several potential reasons a larger sample size was not achieved, which are important for researchers embarking on work in this area are advised to be aware:

1) *Trust and Privacy*. While incorporating social media data collection into *MoodPrism* had several strengths, it may have also impacted on participation. Firstly, due to the potentially detailed nature of the information collected from Facebook and Twitter, as well as from *MoodPrism*, it is likely participants may have had concerns around privacy. This was highlighted in focus groups testing the beta version of *MoodPrism* where "trust" in the app was identified as a barrier to contributing social media data (see Appendix A). While changes to the opt-in procedure were made to improve the information around contributing social media data and to provide additional opportunities to opt-in after developing trust in the app, time constraints prohibited the evaluation of if these changes had addressed participant's "trust" concerns and increased their likelihood to opt-in.

The overall sample size in this research may have been increased by making the contribution of social media data compulsory to the study, however, the opt-in procedure was deemed important to increase participant autonomy, provide opportunities to better understand the purpose of each component of the broader *MoodPrism* project (informed consent), and reduce the likelihood of a sampling bias to the broader project by potentially restricting the sample to only those with social media accounts who were also willing to share that data.

2) *Attrition*. While *MoodPrism* was a minimally obtrusive method of data collection, there were approximately 45 minutes of surveys to complete after download and this was followed by the 30-day daily mood report procedure. This may have been burdensome for participants and resulted in a steep attrition rate at the outset of the study impacting on the social media and individual

characteristics data collection for Chapters 4 and 5. This time-cost for participants was unavoidable in the context of the broader *MoodPrism* project, however, future research may overcome this limitation by presenting a limited quantity of brief baseline surveys. Figure 1 shows the compliance rates of social media opt-in and social media opt-out participants across 30 days of *MoodPrism* use from download to the completion of the follow-up surveys. The participants who opted-in to contribute social media data are presented in dark grey. It is important to note here that the social media opt-in participants may have been excluded from analyses due to other criteria in Chapters 4 and 5 and do not necessarily represent final sample sizes.



MoodPrism User Compliance

Figure 1. The *MoodPrism* compliance curve across the 30-day study. Social media opt-in users are shown in dark grey and those who opted-out are shown in light grey.

3) *Technical Issues*. During the *MoodPrism* study participants contacted the research team reporting technical issues such as app crashes and not receiving push notifications to complete the daily mood reports (this was an iOS only error). In the backend database timestamp recording errors were also evident for the daily mood reports. While these issues were resolved promptly, data could

not be recovered in cases where a participant had uninstalled *MoodPrism*. Researchers or clinicians seeking to implement mobile ESMs should be wary and responsive to such technical issues. In many cases, this may require an ongoing relationship with an app developer with the technical expertise to resolve bugs in a timely manner. Some of these issues are discussed in more detail in Appendix A.

More broadly, there were several limitations in only looking at the emotion language expressed on social media. First, as the content of social media status updates were not collected we were unable to manually code status updates to ensure the emotion word counts reflected the sentiment in the text. For example, negation and sarcasm often reverse the emotional meaning of a sentence, despite the presence of the opposite emotion word (Hogenboom, Van Iterson, Heerschop, Frasincar, & Kaymak, 2011). This likely would have contributed to false-positive emotion word counts within either the positive or negative emotion categories.

Second, the proportion of emotion words in a status update was used to infer emotion intensity, in that greater values reflected more negative or positive emotion compared to smaller values. As highlighted in Chapter 4, this may be particularly problematic in relation to crossplatform comparisons as the upper-bound on the total number of words expressed may be restricted (as in the case of Twitter), or expansive (as in the case of Facebook), thereby impacting of the proportion of emotion words obtained from each platform. Further, some emotion words may indicate greater intensity than others, for example "bad" may be low intensity and "horrible" may be high intensity. In this research, we used a polarity-based approach (positive or negative) where all emotion words were weighted equally. Valence-based approaches though weighting words to better reflect the intensity of the emotion may have added nuance to the size of emotion fluctuations between status updates (Hutto & Gilbert, 2014).

Third, this research had a narrow scope by only focusing on the emotion language defined by the *LIWC 2007*. Other word categories, such as the use of personal pronouns are also informative

to depression as indicators of cognitive processes indicating excessive self-focus (Brockmeyer et al., 2015; Rude, Gortner, & Pennebaker, 2004). Discrete emotions categories, such as sadness and anger, may also increase the granularity in observing emotion processes over time and has previously been shown to differentiate between depressed, anxious, and stressed individuals of different age groups (Settanni & Marengo, 2015). The research presented here only selected those with more than 10 status updates and was therefore not representative of all social media users. Many people use social media in a passive way (i.e. scrolling through the newsfeed) and periods of online social withdrawal may also be informative of depression (Shaw, Timpano, Tran, & Joormann, 2015). This limitation is also true of other language only approaches to examining depression status and highlights the importance of incorporating other observable social media features in prediction models.

Finally, social media is inherently a social environment that encompasses many complex interpersonal interactions that may impact on emotion and more broadly on mental health. Some of these risk and protective factors were outlined in Chapter 2 and it is a limitation of this thesis that these factors were not explored in the experimental chapters. They are, however, important targets for further research, particularly in building a more comprehensive understanding of how personal characteristics, social interaction, and mental health all combine to define the emotion language used on social media and how, in turn, that data may be utilised as a tool for depression detection.

6.5 Future Research Directions

Considering the importance of time-sensitive observations highlighted in this thesis, a clear avenue for future research will be in the formal time series modelling of emotion expression on social media over time. Both variability and instability may be useful predictors of depression and be more sensitive to depression status than taking averages over time. Future research may wish to establish meaningful cut-off values for variability and instability on both Facebook and Twitter that indicate increased depression risk. Time series analysis will be particularly valuable if considered in

the context of a complementary experience sampling method (as presented in Chapter 5). While time series modelling of social media data is complex due to its irregular nature, this thesis presented some simple solutions to analysing the time features in social media data (i.e., median timeframe adjustments suggested by Jahng et al., 2008). More sophisticated methods may be applicable in larger datasets (e.g. Gaussian kernels; Rehfeld, Marwan, Heitzig, & Kurths, 2011, autoregressive models; see de Haan-Rietdijk et al., 2017 for a review of methods) that will increase analytic depth and power with which significant and meaningful associations between social media data and external emotion criterions can be detected.

Cross-lagged analyses were not performed here due to data quality (i.e., missing time stamps) and quantity, though including such analyses in future research would provide valuable insights into the potential drivers of emotion language use on social media. While the findings presented in Chapter 5 did not indicate significant associations between same-day self-reported mood and the mood expressed on Twitter, it is possible that lagged relationships may exist. For example, for some the use of emotion language on social media may be in response to significant external events, where subjective mood predicts the use of emotion language at a later point. For others, the emotion expressed on social media may predict subjective mood at a later point, potentially through processes of rumination, or contagion (Coviello et al., 2014; Davila et al., 2012; Kramer, 2012). Improving the time-specificity of observations (exact timestamps, rather than at a daily level) may also better reveal connections between subjective mood and the mood expressed across language samples on social media.

A further important inclusion for future research is the integration of individual characteristics as moderators of language, communication style, and broader social media use. As discussed above, it is likely that predictive language models for depression are working well for those with severe depression severity as fewer third variables obscure the expression on mood or depression symptoms in language. Research has previously inferred age, gender and personality

from social media text and controlled for these characteristics when predicting self-identified (that is, disclosed in a social media post) depression or post-traumatic stress disorder (Preotiuc-Pietro et al., 2015). This revealed a substantial overlap between the language associated with demographic characteristics and the language that was predictive of depression or PTSD, highlighting the importance of considering these characteristics as contributors to the emotion patterns visible in the language on social media. Including characteristics known to be associated with emotion expression like personality, gender and age (Deng, Chang, Yang, Huo, & Zhou, 2016; Löckenhoff, Costa, & Lane, 2008; Park et al., 2015) will contribute a more nuanced understanding of language used by individuals without depression symptoms, particularly in describing individuals who express emotion language in a way that is similar to that associated with depression status (i.e., high levels of negative emotion word use). This was highlighted in the case studies in Chapter 5 where participants without depression were expressing negative emotion words in a way that might indicate experiences of persistent low mood (see Participant 4). From a preventative perspective, accounting for individual characteristics may assist in revealing patterns of language use that signal depression onset or risk in a manner tailored to the individual, potentially decreasing false positive or negative identifications of depression from social media content.

6.6 Concluding Remarks

The primary aim of this research was to explore the association between depression levels and emotion expressed on social media, and to apply metrics that may be informative for depression identification and prediction. It sought to strengthen the understanding of the language features predictive of depression by incorporating time-sensitive observations of emotion word variability to better describe patterns of mood change over time. Importantly, this thesis linked social media data with self-report experience sampling via a novel smartphone methodology to critically examine how well the emotion language expressed on social media reflects experienced mood. The studies presented here suggest that examining the dynamics of emotion word expression on social media

may provide additional and sensitive insights into the presence of depression for social media users. While these metrics are useful for describing emotion patterns over time, it is also clear that social media records do not clearly mirror lived experiences. This has implications for depression detection from social media language, primarily for finding practical ways to reduce the number of false-positive identifications in prediction models.

The link social media use has with mental health is complex and varied. There are both beneficial and detrimental aspects of using social media for depression, and the quality of social media use can provide greater insight into determining mental health outcomes than quantity of social media use alone. Utilising social media derived data to learn about the mental health of social media users is a key step towards developing automated methods of detecting significant depression risk. Taken together, the findings of this thesis have suggested extensions to the language-based models predicting depression by focusing on the patterns of emotion word change over time. The findings also counsel the need for further research to determine the reliability and validity of using social media language as an indicator of the emotion experiences underlying depression.

Thesis References

- American Psychiatric Association. (2013). Depressive Disorders. In *Diagnostic and Statistical Manual of Mental Disorders* (Vols. 1–0). American Psychiatric Association. https://doi.org/10.1176/appi.books.9780890425596.dsm04
- Andreassen, C. S., Billieux, J., Griffiths, M. D., Kuss, D. J., Demetrovics, Z., Mazzoni, E., & Pallesen, S. (2016). The relationship between addictive use of social media and video games and symptoms of psychiatric disorders: A large-scale cross-sectional study. *Psychology of Addictive Behaviors*, 30(2), 252–262. https://doi.org/10.1037/adb0000160
- Appel, H., Crusius, J., & Gerlach, A. L. (2015). Social comparison, envy, and depression on Facebook: A study looking at the effects of high comparison standards on depressed individuals. *Journal of Social and Clinical Psychology*, *34*(4), 277–289. https://doi.org/10.1521/jscp.2015.34.4.277
- Araujo, T., Wonneberger, A., Neijens, P., & de Vreese, C. (2017). How much time do you spend online? Understanding and improving the accuracy of self-reported measures of internet use. *Communication Methods and Measures*, *11*(3), 173–190. https://doi.org/10.1080/19312458.2017.1317337
- Arjmand, H.-A., & Rickard, N. S. (2018). *Exploring the utility of a smartphone experience sampling applications (ESA) for exploring resilience to daily stressors*. Manuscript in preparation.
- Armey, M. F., Schatten, H. T., Haradhvala, N., & Miller, I. W. (2015). Ecological momentary assessment (EMA) of depression-related phenomena. *Current Opinion in Psychology*, 4, 21–25. https://doi.org/10.1016/j.copsyc.2015.01.002

- Back, M. D., Stopfer, J. M., Vazire, S., Gaddis, S., Schmukle, S. C., Egloff, B., & Gosling, S. D.
 (2010). Facebook profiles reflect actual personality, not self-idealization. *Psychological Science*, *21*(3), 372–374. https://doi.org/10.1177/0956797609360756
- Baek, Y. M., Bae, Y., & Jang, H. (2013). Social and parasocial relationships on social network sites and their differential relationships with users' psychological well-being. *Cyberpsychology, Behavior and Social Networking*, 16(7), 512–517. https://doi.org/10.1089/cyber.2012.0510
- Baikie, K. A., & Wilhelm, K. (2005). Emotional and physical health benefits of expressive writing. *Advances in Psychiatric Treatment*, *11*(5), 338–346. https://doi.org/10.1192/apt.11.5.338
- Baiocco, R., Chirumbolo, A., Bianchi, D., Ioverno, S., Morelli, M., & Nappa, M. R. (2017). How
 HEXACO personality traits predict different selfie-posting behaviors among adolescents
 and young adults. *Frontiers in Psychology*, 7. https://doi.org/10.3389/fpsyg.2016.02080
- Baker, A. E., & Jeske, D. (2015). Assertiveness and anxiety effects in traditional and online interactions. *International Journal of Cyber Behavior, Psychology and Learning*, 5(3), 30–46. https://doi.org/10.4018/IJCBPL.2015070103
- Baker, D. A., & Algorta, G. P. (2016). The relationship between online social networking and depression: A systematic review of quantitative studies. *Cyberpsychology, Behavior and Social Networking*, 19(11), 638–648. https://doi.org/10.1089/cyber.2016.0206
- Baker, J. R., & Moore, S. M. (2008a). Blogging as a social tool: A psychosocial examination of the effects of blogging. *CyberPsychology & Behavior*, 11(6), 747–749. https://doi.org/10.1089/cpb.2008.0053
- Baker, J. R., & Moore, S. M. (2008b). Distress, coping, and blogging: comparing new myspace users by their intention to blog. *CyberPsychology & Behavior*, 11(1), 81–85. https://doi.org/10.1089/cpb.2007.9930

- Baker, L. R., & Oswald, D. L. (2010). Shyness and online social networking services. *Journal of Social and Personal Relationships*, 27(7), 873–889.
 https://doi.org/10.1177/0265407510375261
- Bakker, D., Kazantzis, N., Rickwood, D., & Rickard, N. (2016). Mental health smartphone apps:
 Review and evidence-based recommendations for future developments. *JMIR Mental Health*, *3*(1), e7. https://doi.org/10.2196/mental.4984
- Baldasaro, R. E., Shanahan, M. J., & Bauer, D. J. (2013). Psychometric properties of the mini-IPIP in a large, nationally representative sample of young adults. *Journal of Personality Assessment*, 95(1), 74–84. https://doi.org/10.1080/00223891.2012.700466
- Banjanin, N., Banjanin, N., Dimitrijevic, I., & Pantic, I. (2015). Relationship between internet use and depression: Focus on physiological mood oscillations, social networking and online addictive behavior. *Computers in Human Behavior*, 43, 308–312. https://doi.org/10.1016/j.chb.2014.11.013
- Barnett, P. A., & Gotlib, I. H. (1988). Psychosocial functioning and depression: distinguishing among antecedents, concomitants, and consequences. *Psychological Bulletin*, 104(1), 97– 126. https://doi.org/10.1037/0033-2909.104.1.97
- Baxter, A. J., Scott, K. M., Vos, T., & Whiteford, H. A. (2013). Global prevalence of anxiety disorders: A systematic review and meta-regression. *Psychological Medicine*, 43(5), 897–910. https://doi.org/10.1017/S003329171200147X
- Bazarova, N. N., Choi, Y. H., Whitlock, J., Cosley, D., & Sosik, V. (2017). Psychological Distress and Emotional Expression on Facebook. *Cyberpsychology, Behavior, and Social Networking*, 20(3), 157–163. https://doi.org/10.1089/cyber.2016.0335
- Bech, P., Olsen, L. R., Kjoller, M., & Rasmussen, N. K. (2003). Measuring well-being rather than the absence of distress symptoms: a comparison of the SF-36 Mental Health subscale and

the WHO-Five Well-Being Scale. *International Journal of Methods in Psychiatric Research*, *12*(2), 85–91. https://doi.org/10.1002/mpr.145

- Beck, A. T. (1974). The development of depression: A cognitive model. In *The psychology of depression: Contemporary theory and research*. (p. xvii, 318-xvii, 318). Oxford, England: John Wiley & Sons.
- Benton, A., Mitchell, M., & Hovy, D. (2017). Multi-task learning for mental health using social media text. Proceedings of the European Chapter of the Association for Computational Linguistics. Retrieved from http://arxiv.org/abs/1712.03538
- Bernard, J. D., Baddeley, J. L., Rodriguez, B. F., & Burke, P. A. (2016). Depression, language, and affect: An examination of the influence of baseline depression and affect induction on language. *Journal of Language and Social Psychology*, 35(3), 317–326. https://doi.org/10.1177/0261927X15589186
- Berry, N., Lobban, F., Belousov, M., Emsley, R., Nenadic, G., & Bucci, S. (2017).
 #WhyWeTweetMH: Understanding why people use Twitter to discuss mental health problems. *Journal of Medical Internet Research*, *19*(4), e107.
 https://doi.org/10.2196/jmir.6173
- Best, P., Manktelow, R., & Taylor, B. (2014). Online communication, social media and adolescent wellbeing: A systematic narrative review. *Children and Youth Services Review*, 41, 27–36. https://doi.org/10.1016/j.childyouth.2014.03.001
- Błachnio, A., Przepiórka, A., & Pantic, I. (2015). Internet use, Facebook intrusion, and depression: Results of a cross-sectional study. *European Psychiatry*, 30(6), 681–684. https://doi.org/10.1016/j.eurpsy.2015.04.002

- Blank, G. (2017). The digital divide among Twitter users and its implications for social research. Social Science Computer Review, 35(6), 679–697. https://doi.org/10.1177/0894439316671698
- Bodroža, B., & Jovanović, T. (2016). Validation of the new scale for measuring behaviors of Facebook users: Psycho-Social Aspects of Facebook Use (PSAFU). *Computers in Human Behavior*, 54, 425–435. https://doi.org/10.1016/j.chb.2015.07.032
- Bollen, J., Gonçalves, B., Ruan, G., & Mao, H. (2011). Happiness is assortative in online social networks. *Artificial Life*, *17*(3), 237–251. https://doi.org/10.1162/artl_a_00034
- Bosacki, S., Dane, A., Marini, Z., & YLC-CURA. (2007). Peer relationships and internalizing problems in adolescents: Mediating role of self-esteem. *Emotional and Behavioural Difficulties*, 12(4), 261–282. https://doi.org/10.1080/13632750701664293
- Bowen, R., Peters, E., Marwaha, S., Baetz, M., & Balbuena, L. (2017). Moods in clinical depression are more unstable than severe normal sadness. *Frontiers in Psychiatry*, 8. https://doi.org/10.3389/fpsyt.2017.00056
- boyd, d. m., & Ellison, N. B. (2007). Social network sites: Definition, history, and scholarship. *Journal of Computer-Mediated Communication*, *13*(1), 210–230. https://doi.org/10.1111/j.1083-6101.2007.00393.x
- boyd, d. m., Golder, S., & Lotan, G. (2010). Tweet, tweet, retweet: Conversational aspects of retweeting on Twitter. In 43rd Hawaii International Conference on Systems Sciences (HICSS) (pp. 1-10). IEEE Xplore. http://doi.org/ 10.1109/HICSS.2010.412
- Bradley, B., DeFife, J. A., Guarnaccia, C., Phifer, J., Fani, N., Ressler, K. J., & Westen, D. (2011). Emotion dysregulation and negative affect: Association with psychiatric symptoms. *The Journal of Clinical Psychiatry*, 72(5), 685–691. https://doi.org/10.4088/JCP.10m06409blu

- Brockmeyer, T., Zimmermann, J., Kulessa, D., Hautzinger, M., Bents, H., Friederich, H.-C., ... Backenstrass, M. (2015). Me, myself, and I: self-referent word use as an indicator of selffocused attention in relation to depression and anxiety. *Frontiers in Psychology*, 6. https://doi.org/10.3389/fpsyg.2015.01564
- Brooks, J. A., Shablack, H., Gendron, M., Satpute, A. B., Parrish, M. H., & Lindquist, K. A. (2017).
 The role of language in the experience and perception of emotion: a neuroimaging metaanalysis. *Social Cognitive and Affective Neuroscience*, *12*(2), 169–183. https://doi.org/10.1093/scan/nsw121
- Burke, T. J., & Ruppel, E. K. (2015). Facebook self-presentational motives: Daily effects on social anxiety and interaction success. *Communication Studies*, 66(2), 204–217. https://doi.org/10.1080/10510974.2014.884014
- Butler, P. (2010). Visualizing Friendships [image]. Retrieved from https://www.facebook.com/notes/facebook-engineering/visualizingfriendships/469716398919
- Bylsma, L. M., Morris, B. H., & Rottenberg, J. (2008). A meta-analysis of emotional reactivity in major depressive disorder. *Clinical Psychology Review*, 28(4), 676–691. https://doi.org/10.1016/j.cpr.2007.10.001
- Bylsma, L. M., Taylor-Clift, A., & Rottenberg, J. (2011). Emotional reactivity to daily events in major and minor depression. *Journal of Abnormal Psychology*, *120*(1), 155–167. https://doi.org/10.1037/a0021662
- Calvo, R. A., Milne, D. N., Hussain, M. S., & Christensen, H. (2017). Natural language processing in mental health applications using non-clinical texts. *Natural Language Engineering*, 23(5), 649–685. https://doi.org/10.1017/S1351324916000383

- Casale, S., & Fioravanti, G. (2015). Satisfying needs through Social Networking Sites: A pathway towards problematic Internet use for socially anxious people? *Addictive Behaviors Reports*, *1*, 34–39. https://doi.org/10.1016/j.abrep.2015.03.008
- Cavazos-Rehg, P. A., Krauss, M. J., Sowles, S., Connolly, S., Rosas, C., Bharadwaj, M., & Bierut,
 L. J. (2016). A content analysis of depression-related tweets. *Computers in Human Behavior*, 54, 351–357. https://doi.org/10.1016/j.chb.2015.08.023
- Cepoiu, M., McCusker, J., Cole, M. G., Sewitch, M., Belzile, E., & Ciampi, A. (2008). Recognition of depression by non-psychiatric physicians: A systematic literature review and metaanalysis. *Journal of General Internal Medicine*, 23(1), 25–36. https://doi.org/10.1007/s11606-007-0428-5
- Chesney, M. A., Neilands, T. B., Chambers, D. B., Taylor, J. M., & Folkman, S. (2006). A validity and reliability study of the coping self-efficacy scale. *British Journal of Health Psychology*, *11*(Pt 3), 421–437. https://doi.org/10.1348/135910705X53155
- Clark, D. M., & McManus, F. (2002). Information processing in social phobia. *Biological Psychiatry*, *51*(1), 92–100. https://doi.org/10.1016/S0006-3223(01)01296-3
- Cole, D. A., & Turner, J. E. (1993). Models of cognitive mediation and moderation in child depression. *Journal of Abnormal Psychology*, 102(2), 271–281. http://dx.doi.org/10.1037/0021-843X.102.2.271
- Collins, K. A., Westra, H. A., Dozois, D. J. A., & Burns, D. D. (2004). Gaps in accessing treatment for anxiety and depression: Challenges for the delivery of care. *Clinical Psychology Review*, 24(5), 583–616. https://doi.org/10.1016/j.cpr.2004.06.001
- Connolly, N. P., Eberhart, N. K., Hammen, C. L., & Brennan, P. A. (2010). Specificity of stress generation: A comparison of adolescents with depressive, anxiety, and comorbid diagnoses.

International Journal of Cognitive Therapy, 3(4), 368–379. https://doi.org/10.1521/ijct.2010.3.4.368

- Cornwell, E. Y., & Waite, L. J. (2009). Social disconnectedness, perceived isolation, and health among older adults. *Journal of Health and Social Behavior*, *50*(1), 31–48. https://doi.org/10.1177/002214650905000103
- Correa, D., Silva, L. A., Mondal, M., Benevenuto, F., & Gummadi, K. P. (2015). The many shades of anonymity: Characterizing anonymous social media content. In *Proceedings of the Ninth International Conference on Web and Social Media (ICWSM)* (pp. 71–80). Retrieved from https://www.aaai.org/ocs/index.php/ICWSM/ICWSM15/paper/viewFile/10596/10490
- Correa, T. (2016). Digital skills and social media use: How Internet skills are related to different types of Facebook use among "digital natives." *Information, Communication & Society*, 19(8), 1095–1107. https://doi.org/10.1080/1369118X.2015.1084023
- Coviello, L., Sohn, Y., Kramer, A. D. I., Marlow, C., Franceschetti, M., Christakis, N. A., &
 Fowler, J. H. (2014). Detecting Emotional Contagion in Massive Social Networks. *PLoS ONE*, 9(3), e90315. https://doi.org/10.1371/journal.pone.0090315
- Csikszentmihalyi, M., & Larson, R. (1987). Validity and reliability of the Experience-Sampling Method. *The Journal of Nervous and Mental Disease*, *175*(9), 526–536. http://dx.doi.org/10.1097/00005053-198709000-00004
- Cummins, R. A. (2010). Subjective wellbeing, homeostatically protected mood and depression: A synthesis. *Journal of Happiness Studies*, *11*(1), 1–17. https://doi.org/10.1007/s10902-009-9167-0
- Davidson, T., & Farquhar, L. K. (2014). Correlates of social anxiety, religion, and Facebook. Journal of Media and Religion, 13(4), 208–225. https://doi.org/10.1080/15348423.2014.971566

- Davila, J., Hershenberg, R., Feinstein, B. A., Gorman, K., Bhatia, V., & Starr, L. R. (2012).
 Frequency and quality of social networking among young adults: Associations with depressive symptoms, rumination, and corumination. *Psychology of Popular Media Culture*, *1*(2), 72–86. https://doi.org/10.1037/a0027512
- De Choudhury, M., Counts, S., & Horvitz, E. (2013). Social media as a measurement tool of depression in populations. In *Proceedings of the 5th Annual ACM Web Science Conference* (pp. 47–56). ACM, New York, NY, USA. http://doi.org/10.1145/2464464.2464480
- De Choudhury, M., Counts, S., Horvitz, E. J., & Hoff, A. (2014). Characterizing and predicting postpartum depression from shared Facebook data. In *Proceedings of the 17th ACM Conference on Computer Supported Cooperative Work and Social Computing* (pp. 626–638). ACM, New York, NY, USA. https://doi.org/10.1145/2531602.2531675
- De Choudhury, M., Gamon, M., Counts, S., & Horvitz, E. (2013). Predicting depression via social media. In *ICWSM* (p. 2). Retrieved from http://course.duruofei.com/wpcontent/uploads/2015/05/Choudhury_Predicting-Depression-via-Social-Media_ICWSM13.pdf
- De Choudhury, M., Kiciman, E., Dredze, M., Coppersmith, G., & Kumar, M. (2016). Discovering shifts to suicidal ideation from mental health content in social media (pp. 2098–2110). ACM, New York, NY, USA. https://doi.org/10.1145/2858036.2858207
- De Haan-Rietdijk, S., Voelkle, M. C., Keijsers, L., & Hamaker, E. L. (2017). Discrete- vs. Continuous-Time Modeling of Unequally Spaced Experience Sampling Method Data. Frontiers in Psychology, 8, 1849. http://doi.org/10.3389/fpsyg.2017.01849
- De Silva, M. J. (2005). Social capital and mental illness: A systematic review. Journal of Epidemiology & Community Health, 59(8), 619–627. https://doi.org/10.1136/jech.2004.029678

- Deng, Y., Chang, L., Yang, M., Huo, M., & Zhou, R. (2016). Gender differences in emotional response: Inconsistency between experience and expressivity. *PLOS ONE*, 11(6), e0158666. https://doi.org/10.1371/journal.pone.0158666
- Derks, D., Fischer, A. H., & Bos, A. E. R. (2008). The role of emotion in computer-mediated communication: A review. *Computers in Human Behavior*, 24(3), 766–785. https://doi.org/10.1016/j.chb.2007.04.004
- Deters, F. G., & Mehl, M. R. (2013). Does posting Facebook status updates increase or decrease loneliness? An online social networking experiment. *Social Psychological and Personality Science*, 4(5), 579–586. https://doi.org/10.1177/1948550612469233
- Deters, F. G., Mehl, M. R., & Eid, M. (2016). Social responses to Facebook status updates: The role of extraversion and social anxiety. *Computers in Human Behavior*, 61, 1–13. https://doi.org/10.1016/j.chb.2016.02.093
- Dittmar, N. (1996). Explorations in 'Idiolects'. Amsterdam Studies in the Theory and History of Linguistic Science, 109–128. http://doi.org/10.1075/cilt.138.10dit
- Donnellan, M. B., Oswald, F. L., Baird, B. M., & Lucas, R. E. (2006). The mini-IPIP Scales: Tinyyet-effective measures of the Big Five Factors of Personality. *Psychological Assessment*, 18(2), 192–203. https://doi.org/10.1037/1040-3590.18.2.192
- Duggan, M., Ellison, N. B., Lampe, C., Lenhart, A., & Madden, M. (2015). Social Media Update 2014. Pew Research Center. Retrieved from http://www.pewinternet.org/files/2015/01/PI_SocialMediaUpdate20144.pdf
- Dumitrache, S. D., Mitrofan, L., & Petrov, Z. (2012). Self-image and depressive tendencies among adolescent Facebook users. *Revista de Psihologie*, *58*(4), 285–295.

- Dundon, E. (2006). Adolescent depression: A metasynthesis. *Journal of Pediatric Health Care*, 20(6), 384–392. https://doi.org/10.1016/j.pedhc.2006.02.010
- Ebner-Priemer, U. W., & Trull, T. J. (2009). Ecological momentary assessment of mood disorders and mood dysregulation. *Psychological Assessment*, 21(4), 463–475. https://doi.org/10.1037/a0017075
- Ellison, N. B., & boyd, d. (2013). Sociality through social network sites. In *The Oxford Handbook* of Internet Studies (pp. 151–172). Oxford: Oxford University Press. http://doi.org/10.1093/oxfordhb/9780199589074.001.0001/oxfordhb-9780199589074-e-8
- Ellison, N. B., Steinfield, C., & Lampe, C. (2007). The benefits of Facebook "Friends:" social capital and college students' use of online social network sites. *Journal of Computer-Mediated Communication*, 12(4), 1143–1168. https://doi.org/10.1111/j.1083-6101.2007.00367.x
- Ellison, N. B., Vitak, J., Gray, R., & Lampe, C. (2014). Cultivating social resources on social network sites: Facebook relationship maintenance behaviors and their role in social capital processes. *Journal of Computer-Mediated Communication*, 19(4), 855–870. https://doi.org/10.1111/jcc4.12078
- Farahani, H. A., Kazemi, Z., Aghamohamadi, S., Bakhtiarvand, F., & Ansari, M. (2011).
 Examining mental health indices in students using Facebook in Iran. *Procedia Social and Behavioral Sciences*, 28, 811–814. https://doi.org/10.1016/j.sbspro.2011.11.148
- Fayard, J. V., Roberts, B. W., Robins, R. W., & Watson, D. (2012). Uncovering the affective core of conscientiousness: The role of self-conscious emotions. *Journal of Personality*, 80(1), 1–32. https://doi.org/10.1111/j.1467-6494.2011.00720.x
- Feinstein, B. A., Bhatia, V., Hershenberg, R., & Davila, J. (2012). Another venue for problematic interpersonal behavior: The effects of depressive and anxious symptoms on social

networking experiences. *Journal of Social and Clinical Psychology*, *31*(4), 356–382. http://doi.org/10.1521/jscp.2012.31.4.356

- Feinstein, B. A., Hershenberg, R., Bhatia, V., Latack, J. A., Meuwly, N., & Davila, J. (2013). Negative social comparison on Facebook and depressive symptoms: Rumination as a mechanism. *Psychology of Popular Media Culture*, 2(3), 161–170. https://doi.org/10.1037/a0033111
- Fernandez, K. C., Levinson, C. A., & Rodebaugh, T. L. (2012). Profiling: predicting social anxiety from facebook profiles. *Social Psychological and Personality Science*, 3(6), 706–713. https://doi.org/10.1177/1948550611434967
- Ferrari, A. J., Somerville, A. J., Baxter, A. J., Norman, R., Patten, S. B., Vos, T., & Whiteford, H.
 A. (2013). Global variation in the prevalence and incidence of major depressive disorder: a systematic review of the epidemiological literature. *Psychological Medicine*, 43(3), 471–481. https://doi.org/10.1017/S0033291712001511
- Flynn, M., Kecmanovic, J., & Alloy, L. B. (2010). An examination of integrated cognitiveinterpersonal vulnerability to depression: The role of rumination, perceived social support, and interpersonal stress generation. *Cognitive Therapy and Research*, 34(5), 456–466. https://doi.org/10.1007/s10608-010-9300-8
- Forest, A. L., & Wood, J. V. (2012). When social networking is not working: Individuals with low self-esteem recognize but do not reap the benefits of self-disclosure on Facebook. *Psychological Science*, 23(3), 295–302. https://doi.org/10.1177/0956797611429709
- Frison, E., & Eggermont, S. (2015). The impact of daily stress on adolescents' depressed mood: The role of social support seeking through Facebook. *Computers in Human Behavior*, 44, 315–325. https://doi.org/10.1016/j.chb.2014.11.070

- Frison, E., Subrahmanyam, K., & Eggermont, S. (2016). The short-term longitudinal and reciprocal relations between peer victimization on Facebook and adolescents' well-being. *Journal of Youth and Adolescence*, 45(9), 1755–1771. https://doi.org/10.1007/s10964-016-0436-z
- Frost, R. L., & Rickwood, D. J. (2017). A systematic review of the mental health outcomes associated with Facebook use. *Computers in Human Behavior*, 76, 576–600. https://doi.org/10.1016/j.chb.2017.08.001
- Gershon, A., & Eidelman, P. (2015). Inter-episode affective intensity and instability: Predictors of depression and functional impairment in bipolar disorder. *Journal of Behavior Therapy and Experimental Psychiatry*, 46, 14-18. http://doi.org/10.1016/j.jbtep.2014.07.005
- Geva, H., Oestreicher-Singer, G., & Saar-Tsechansky, M. (2016). Using retweets to shape our online persona: A topic modeling approach. Rochester, NY: Social Science Research Network. Retrieved from https://papers.ssrn.com/abstract=2759811
- Ghosh, A., & Dasgupta, S. (2015). Psychological predictors of Facebook use. Journal of the Indian Academy of Applied Psychology, 41(1), 101. Retrieved from http:// jiaap.org/Listing_Detail/Logo/e9e7e8e8-614b-4d83-9409-50ed9450dd33.pdf
- Gil-Or, O., Levi-Belz, Y., & Turel, O. (2015). The "Facebook-self": Characteristics and psychological predictors of false self-presentation on Facebook. *Frontiers in Psychology*, 6. https://doi.org/10.3389/fpsyg.2015.00099
- Giota, K. G., & Kleftaras, G. (2013). The role of personality and depression in problematic use of social networking sites in Greece. *Cyberpsychology: Journal of Psychosocial Research on Cyberspace*, 7(3), Article 1. https://doi.org/10.5817/CP2013-3-6
- Gkotsis, G., Oellrich, A., Hubbard, T. J., Dobson, R. J., Liakata, M., Velupillai, S., & Dutta, R.(2016). The language of mental health problems in social media. In *Third Computational*

Linguistics and Clinical Psychology Workshop (NAACL) (pp. 63–73). Retrieved from http://www.anthology.aclweb.org/W/W16/W16-0307.pdf

- Golder, S., Ahmed, S., Norman, G., & Booth, A. (2017). Attitudes toward the ethics of research using social media: A systematic review. *Journal of Medical Internet Research*, 19(6), e195. https://doi.org/10.2196/jmir.7082
- Gosling, S. D., Augustine, A. A., Vazire, S., Holtzman, N., & Gaddis, S. (2011). Manifestations of personality in online social networks: Self-reported Facebook-related behaviors and observable profile information. *Cyberpsychology, Behavior, and Social Networking*, 14(9), 483–488. https://doi.org/10.1089/cyber.2010.0087
- Gotlib, I. H., & Joormann, J. (2010). Cognition and depression: Current status and future directions.
 Annual Review of Clinical Psychology, 6, 285–312.
 https://doi.org/10.1146/annurev.clinpsy.121208.131305
- Grav, S., Hellzèn, O., Romild, U., & Stordal, E. (2012). Association between social support and depression in the general population: the HUNT study, a cross-sectional survey. *Journal of Clinical Nursing*, 21(1–2), 111–120. https://doi.org/10.1111/j.1365-2702.2011.03868.x
- Green, T., Wilhelmsen, T., Wilmots, E., Dodd, B., & Quinn, S. (2016). Social anxiety, attributes of online communication and self-disclosure across private and public Facebook communication. *Computers in Human Behavior*, 58, 206–213. https://doi.org/10.1016/j.chb.2015.12.066
- Greenwood, S., Perrin, R., & Duggan, M. (2016). *Social Media Update 2016*. Pew Research Center. Retrieved from http://www.pewinternet.org/2016/11/11/social-media-update-2016/
- Grieve, R., Indian, M., Witteveen, K., Anne Tolan, G., & Marrington, J. (2013). Face-to-face or Facebook: Can social connectedness be derived online? *Computers in Human Behavior*, 29(3), 604–609. https://doi.org/10.1016/j.chb.2012.11.017

- Gross, E. F., Juvonen, J., & Gable, S. L. (2002). Internet use and well-being in adolescence. *Journal of Social Issues*, 58(1), 75–90. https://doi.org/10.1111/1540-4560.00249
- Gross, J. J., & Muñoz, R. F. (1995). Emotion regulation and mental health. *Clinical Psychology: Science and Practice*, 2(2), 151–164. https://doi.org/10.1111/j.1468-2850.1995.tb00036.x
- Gruber, J., Kogan, A., Quoidbach, J., & Mauss, I. B. (2013). Happiness is best kept stable: Positive emotion variability is associated with poorer psychological health. *Emotion*, 13(1), 1–6. https://doi.org/10.1037/a0030262
- Guntuku, S. C., Yaden, D. B., Kern, M. L., Ungar, L. H., & Eichstaedt, J. C. (2017). Detecting depression and mental illness on social media: An integrative review. *Current Opinion in Behavioral Sciences*, 18(Supplement C), 43–49. https://doi.org/10.1016/j.cobeha.2017.07.005
- Guo, Y., Li, Y., & Ito, N. (2014). exploring the predicted effect of social networking site use on perceived social capital and psychological well-being of Chinese international students in Japan. *Cyberpsychology, Behavior, and Social Networking*, *17*(1), 52–58. https://doi.org/10.1089/cyber.2012.0537
- Haight, M., Quan-Haase, A., & Corbett, B. A. (2014). Revisiting the digital divide in Canada: the impact of demographic factors on access to the internet, level of online activity, and social networking site usage. *Information, Communication & Society*, *17*(4), 503–519. https://doi.org/10.1080/1369118X.2014.891633
- Hancock, J. T., Gee, K., Ciaccio, K., & Lin, J. M.-H. (2008). I'm sad you're sad: emotional contagion in CMC. In *Proceedings of the 2008 ACM conference on Computer supported cooperative work* (pp. 295–298). ACM, New York, NY, USA. Retrieved from http://dl.acm.org/citation.cfm?id=1460611

- Hanprathet, N., Manwong, M., Khumsri, J., Yingyeun, R., & Phanasathit, M. (2015). Facebook addiction and its relationship with mental health among Thai high school students. *Journal* of the Medical Association of Thailand, 98(Suppl 3), S81-90. Retrieved from http://www.thaiscience.info/journals/Article/JMAT/10971134.pdf
- Higgins, J. T. ., Altman, D. G., & Sterne, J. A. (2011). Assessing risk of bias in included studies. In
 J. T. Higgins & S. Green (Eds.), *Cochrane Handbook for Systematic Reviews of Interventions. Version 5.1.0* (pp. 187–235). The Cochrane Collaboration and John Wiley & Sons Ltd.
- Hirschfeld, R. M. A. (2001). The comorbidity of major depression and anxiety disorders: recognition and management in primary care. *Primary Care Companion to the Journal of Clinical Psychiatry*, 3(6), 244–254. Retrieved from https://www.ncbi.nlm.nih.gov/pmc/articles/PMC181193/
- Hogenboom, A., Van Iterson, P., Heerschop, B., Frasincar, F., & Kaymak, U. (2011). Determining negation scope and strength in sentiment analysis. In *Systems, Man, and Cybernetics (SMC),* 2011 IEEE International Conference on (pp. 2589–2594). IEEE. http://doi.org/10.1109/ICSMC.2011.6084066
- Hollenbaugh, E. E., & Ferris, A. L. (2014). Facebook self-disclosure: Examining the role of traits, social cohesion, and motives. *Computers in Human Behavior*, 30, 50–58. https://doi.org/10.1016/j.chb.2013.07.055
- Homan, C. M., Lu, N., Tu, X., Lytle, M. C., & Silenzio, V. (2014). Social structure and depression in TrevorSpace. In *Proceedings of the 17th ACM conference on Computer supported cooperative work & social computing* (pp. 615–625). ACM, New York, NY, USA. Retrieved from http://dl.acm.org/citation.cfm?id=2531704

- Hong, F.-Y., Huang, D.-H., Lin, H.-Y., & Chiu, S.-L. (2014). Analysis of the psychological traits,
 Facebook usage, and Facebook addiction model of Taiwanese university students.
 Telematics and Informatics, *31*(4), 597–606. https://doi.org/10.1016/j.tele.2014.01.001
- Hong, J.-C., Hwang, M.-Y., Hsu, C.-H., Tai, K.-H., & Kuo, Y.-C. (2015). Belief in dangerous virtual communities as a predictor of continuance intention mediated by general and online social anxiety: The Facebook perspective. *Computers in Human Behavior*, 48, 663–670. https://doi.org/10.1016/j.chb.2015.02.019
- Houben, M., Van Den Noortgate, W., & Kuppens, P. (2015). The relation between short-term emotion dynamics and psychological well-being: A meta-analysis. *Psychological Bulletin*, 141(4), 901–930. https://doi.org/10.1037/a0038822
- House, J. S., Umberson, D., & Landis, K. R. (1988). Structures and processes of social support. *Annual Review of Sociology*, 14(1), 293–318. https://doi.org/10.1146/annurev.so.14.080188.001453
- Huang, C. (2017). Time spent on social network sites and psychological well-being: a metaanalysis. *Cyberpsychology, Behavior and Social Networking*, 20(6), 346–354. https://doi.org/10.1089/cyber.2016.0758
- Hughes, D. J., Rowe, M., Batey, M., & Lee, A. (2012). A tale of two sites: Twitter vs. Facebook and the personality predictors of social media usage. *Computers in Human Behavior*, 28(2), 561–569. https://doi.org/10.1016/j.chb.2011.11.001
- Hutto, C., & Gilbert, E. (2014). Vader: A parsimonious rule-based model for sentiment analysis of social media text. In *Eighth International AAAI Conference on Weblogs and Social Media*. Retrieved from http://www.comp.social.gatech.edu/papers/icwsm14.vader.hutto.pdf

IBM Corp. (2016). IBM SPSS Statistics for Windows (Version 24). Armonk, NY: IBM Corp.

- Indian, M., & Grieve, R. (2014). When Facebook is easier than face-to-face: Social support derived from Facebook in socially anxious individuals. *Personality and Individual Differences*, 59, 102–106. https://doi.org/10.1016/j.paid.2013.11.016
- Jahng, S., Wood, P. K., & Trull, T. J. (2008). Analysis of affective instability in ecological momentary assessment: Indices using successive difference and group comparison via multilevel modelling. *Psychological Methods*, 13(4), 354–375. https://doi.org/10.1037/a0014173
- Java, A., Song, X., Finin, T., & Tseng, B. (2009). Why we Twitter: an analysis of a microblogging community. In *Advances in Web Mining and Web Usage Analysis* (pp. 118–138). Springer, Berlin, Heidelberg. Retrieved from https://link.springer.com/chapter/10.1007/978-3-642-00528-2_7
- Jelenchick, L. A., Eickhoff, J. C., & Moreno, M. A. (2013). "Facebook Depression?" Social networking site use and depression in older adolescents. *Journal of Adolescent Health*, 52(1), 128–130. https://doi.org/10.1016/j.jadohealth.2012.05.008
- Jin, B. (2013). How lonely people use and perceive Facebook. *Computers in Human Behavior*, 29(6), 2463–2470. https://doi.org/10.1016/j.chb.2013.05.034
- Joormann, J., & Gotlib, I. H. (2010). Emotion regulation in depression: Relation to cognitive inhibition. *Cognition & Emotion*, 24(2), 281–298. https://doi.org/10.1080/02699930903407948
- Junco, R. (2013). Comparing actual and self-reported measures of Facebook use. *Computers in Human Behavior*, 29(3), 626–631. https://doi.org/10.1016/j.chb.2012.11.007
- Kauer, S. D., Reid, S. C., Crooke, A. H. D., Khor, A., Hearps, S. J. C., Jorm, A. F., ... Patton, G.(2012). Self-monitoring using mobile phones in the early stages of adolescent depression:

randomized controlled trial. *Journal of Medical Internet Research*, *14*(3), e67. https://doi.org/10.2196/jmir.1858

- Kawachi, I., & Berkman, L. F. (2001). Social ties and mental health. Journal of Urban Health: Bulletin of the New York Academy of Medicine, 78(3), 458–467. https://doi.org/10.1093/jurban/78.3.458
- Kern, M. L., Park, G., Eichstaedt, J. C., Schwartz, H. A., Sap, M., Smith, L. K., & Ungar, L. H.
 (2016). Gaining insights from social media language: Methodologies and challenges. *Psychological Methods*, 21(4), 507–525. https://doi.org/10.1037/met0000091
- Keyes, C. L. M. (2005). Mental illness and/or mental health? Investigating axioms of the complete state model of health. *Journal of Consulting and Clinical Psychology*, 73(3), 539–548. https://doi.org/10.1037/0022-006X.73.3.539
- Koc, M., & Gulyagci, S. (2013). Facebook addiction among Turkish college students: The role of psychological health, demographic, and usage characteristics. *Cyberpsychology, Behavior,* and Social Networking, 16(4), 279–284. https://doi.org/10.1089/cyber.2012.0249
- Kocalevent, R.-D., Hinz, A., & Brähler, E. (2013). Standardization of the depression screener patient health questionnaire (PHQ-9) in the general population. *General Hospital Psychiatry*, 35(5), 551–555. https://doi.org/10.1016/j.genhosppsych.2013.04.006
- Kosinski, M., Matz, S. C., Gosling, S. D., Popov, V., & Stillwell, D. (2015). Facebook as a research tool for the social sciences: Opportunities, challenges, ethical considerations, and practical guidelines. *American Psychologist*, 70(6), 543–556. https://doi.org/10.1037/a0039210
- Koval, P., Pe, M. L., Meers, K., & Kuppens, P. (2013). Affect dynamics in relation to depressive symptoms: Variable, unstable or inert? *Emotion*, 13(6), 1132-41. http://doi.org/ 10.1037/a0033579

- Koval, P., Sutterlin, S., & Kuppens, P. (2016). Emotional inertia is associated with lower well-being when controlling for differences in emotional context. *Frontiers in Psychology*, 6. https://doi.org/10.3389/fpsyg.2015.01997
- Kramer, A. D. I. (2012). The spread of emotion via Facebook. In *Proceedings of the SIGCHI* Conference on Human Factors in Computing Systems (pp. 767–770). ACM, New York, NY, USA. Retrieved from http://dl.acm.org/citation.cfm?id=2207787
- Kramer, A. D. I., Guillory, J., & Hancock, J. (2014). Experimental evidence of massive-scale emotional contagion through social networks. *Proceedings of the National Academy of Sciences of the United States of America*, 111(24), 8788-8790. https://doi.org/10.1073/pnas.1320040111
- Kraut, R., Kiesler, S., Boneva, B., Cummings, J., Helgeson, V., & Crawford, A. (2002). Internet paradox revisited. *Journal of Social Issues*, 58(1), 49–74. http://doi.org/ doi/10.1111/1540-4560.00248
- Kreitler, C. M., Stenmark, C. K., Serrate, J., & Winn, N. (2016). The role of individual differences and emotion in Facebook activity. *Journal of Psychology and Behavioral Science*, 4(1). https://doi.org/10.15640/jpbs.v4n1a1
- Kroenke, K., Spitzer, R. L., & Williams, J. B. W. (2001). The PHQ-9. *Journal of General Internal Medicine*, *16*(9), 606–613. https://doi.org/10.1046/j.1525-1497.2001.016009606.x
- Kroenke, K., Spitzer, R. L., Williams, J. B. W., & Löwe, B. (2009). An ultra-brief screening scale for anxiety and depression: the PHQ-4. *Psychosomatics*, 50(6), 613–621. https://doi.org/10.1176/appi.psy.50.6.613

- Kross, E., Verduyn, P., Demiralp, E., Park, J., Lee, D. S., Lin, N., ... Ybarra, O. (2013). Facebook use predicts declines in subjective well-being in young adults. *PLoS ONE*, 8(8), e69841. https://doi.org/10.1371/journal.pone.0069841
- Kuppens, P., Sheeber, L. B., Yap, M. B. H., Whittle, S., Simmons, J. G., & Allen, N. B. (2012).
 Emotional inertia prospectively predicts the onset of depressive disorder in adolescence. *Emotion*, 12(2), 283–289. https://doi.org/10.1037/a0025046
- Kuppens, P., Van Mechelen, I., Nezlek, J. B., Dossche, D., & Timmermans, T. (2007). Individual differences in core affect variability and their relationship to personality and psychological adjustment. *Emotion*, 7(2), 262–274. https://doi.org/10.1037/1528-3542.7.2.262
- Kuppens, P., & Verduyn, P. (2017). Emotion dynamics. *Current Opinion in Psychology*, 17, 22–26. https://doi.org/10.1016/j.copsyc.2017.06.004
- Labrague, L. J. (2014). Facebook use and adolescents' emotional states of depression, anxiety, and stress. *Health Science Journal*, 8(9), 80-89. Retrieved from http://hypatia.teiath.gr/xmlui/handle/11400/1481
- Landoll, R. R., La Greca, A. M., & Lai, B. S. (2013). Aversive peer experiences on social networking sites: Development of the Social Networking-Peer Experiences Questionnaire (SN-PEQ). *Journal of Research on Adolescence*, 23(4), 695–705. https://doi.org/10.1111/jora.12022
- Landoll, R. R., La Greca, A. M., Lai, B. S., Chan, S. F., & Herge, W. M. (2015). Cyber victimization by peers: Prospective associations with adolescent social anxiety and depressive symptoms. *Journal of Adolescence*, 42, 77–86. https://doi.org/10.1016/j.adolescence.2015.04.002

- Larsen, M. E., Boonstra, T. W., Batterham, P. J., O'Dea, B., Paris, C., & Christensen, H. (2015).
 We Feel: Mapping emotion on Twitter. *IEEE Journal of Biomedical and Health Informatics*, *19*(4), 1246–1252. https://doi.org/10.1109/JBHI.2015.2403839
- Larson, R., & Csikszentmihalyi, M. (2014). The Experience Sampling Method. In M. Csikszentmihalyi, *Flow and the Foundations of Positive Psychology* (pp. 21–34). Dordrecht: Springer Netherlands. Retrieved from http://doi.org/ 10.1007/978-94-017-9088-8_2
- Lee, K.-T., Noh, M.-J., & Koo, D.-M. (2013). Lonely people are no longer lonely on social networking sites: The mediating role of self-disclosure and social support. *Cyberpsychology, Behavior and Social Networking*, 16(6), 413–418. https://doi.org/10.1089/cyber.2012.0553
- Lee, S. Y. (2014). How do people compare themselves with others on social network sites?: The case of Facebook. *Computers in Human Behavior*, 32, 253–260. https://doi.org/10.1016/j.chb.2013.12.009
- Lee-Won, R. J., Herzog, L., & Park, S. G. (2015). Hooked on Facebook: The role of social anxiety and need for social assurance in problematic use of Facebook. *Cyberpsychology, Behavior, and Social Networking*, 18(10), 567–574. https://doi.org/10.1089/cyber.2015.0002
- Lerman, K., Arora, M., Gallegos, L., Kumaraguru, P., & Garcia, D. (2015). Emotions, demographics and sociability in Twitter interactions. In *Proceedings of the 10th International Conference on Web and Social Media (ICWSM)* (pp. 201-210). Retrieved from http://arxiv.org/abs/1510.07090
- Lin, H., Tov, W., & Qiu, L. (2014). Emotional disclosure on social networking sites: The role of network structure and psychological needs. *Computers in Human Behavior*, 41, 342–350. https://doi.org/10.1016/j.chb.2014.09.045
- Lin, L. I.-K. (1989). A Concordance Correlation Coefficient to Evaluate Reproducibility. *Biometrics*, 45(1), 255–268. https://doi.org/10.2307/2532051

- Lin, L. Y, Sidani, J. E., Shensa, A., Radovic, A., Miller, E., Colditz, J. B., ... Primack, B. A. (2016). Association between social media use and depression among U.S. young adults. *Depression and Anxiety*, 33(4), 323–331. https://doi.org/10.1002/da.22466
- Lindquist, K., Gendron, M., & Satpute, A. B. (in press). Language and emotion: Putting words into feelings and feelings into words. In *Handbook of Emotions* (4th ed.). New York, NY: The Guilford Press. Retrieved from https://www.unc.edu/~kal29/docs/Lindquistetal Handbook inpress.pdf
- Locatelli, S. M., Kluwe, K., & Bryant, F. B. (2012). Facebook use and the tendency to ruminate among college students: Testing mediational hypotheses. *Journal of Educational Computing Research*, *46*(4), 377–394. https://doi.org/10.2190/EC.46.4.d
- Löckenhoff, C. E., Costa, P. T., & Lane, R. D. (2008). Age differences in descriptions of emotional experiences in oneself and others. *The Journals of Gerontology: Series B*, 63(2), P92–P99. https://doi.org/10.1093/geronb/63.2.P92
- Lup, K., Trub, L., & Rosenthal, L. (2015). Instagram #Instasad?: Exploring associations among instagram use, depressive symptoms, negative social comparison, and strangers followed. *Cyberpsychology, Behavior, and Social Networking*, *18*(5), 247–252. https://doi.org/10.1089/cyber.2014.0560
- Madden, M. (2012). Privacy management on social media sites. *Pew Internet Report*, 1–20. Retrieved from www.pewinternet.org/2012/02/24/privacy-management-on-social-media-sites/
- Magaard, J. L., Seeralan, T., Schulz, H., & Brütt, A. L. (2017). Factors associated with help-seeking behaviour among individuals with major depression: A systematic review. *PLoS ONE*, *12*(5). https://doi.org/10.1371/journal.pone.0176730

- Manago, A. M., Taylor, T., & Greenfield, P. M. (2012). Me and my 400 friends: The anatomy of college students' Facebook networks, their communication patterns, and well-being.
 Developmental Psychology, 48(2), 369–380. https://doi.org/10.1037/a0026338
- Marroquín, B. (2011). Interpersonal emotion regulation as a mechanism of social support in depression. *Clinical Psychology Review*, 31(8), 1276–1290. https://doi.org/10.1016/j.cpr.2011.09.005
- Marwick, A. E., & boyd, d. (2011). I tweet honestly, I tweet passionately: Twitter users, context collapse, and the imagined audience. *New Media & Society*, *13*(1), 114–133. https://doi.org/10.1177/1461444810365313
- Mas-Herrero, E., Marco-Pallarés, J., Lorenzo-Seva, U., Zatorre, R., & Rodriguez-Fornells, A.
 (2013). Individual Differences in Music Reward Experiences, *Music Perception*, *31*(2), 118-138. http://doi.org/10.1525/mp.2013.31.2.118
- Masuda, N., Kurahashi, I., & Onari, H. (2013). Suicide ideation of individuals in online social networks. *PloS One*, 8(4), e62262. https://doi.org/10.1371/journal.pone.0062262
- Mauss, I. B., Levenson, R. W., McCarter, L., Wilhelm, F. H., & Gross, J. J. (2005). The tie that binds? Coherence among emotion experience, behaviour, and physiology. *Emotion*, 5(2), 175-190. http://doi.org/10.1037/1528-3542.5.2.175
- Mauss, I. B., & Robinson, M. D. (2009). Measures of emotion: A review. *Cognition & Emotion*, 23(2), 209–237. https://doi.org/10.1080/02699930802204677
- McBride, G. B. (2005). A proposal for strength-of-agreement criteria for Lin's concordance correlation coefficient. NIWA Client Report: HAM2005-062. Retrieved from https://www.medcalc.org/download/pdf/McBride2005.pdf
- McCloskey, W., Iwanicki, S., Lauterbach, D., Giammittorio, D. M., & Maxwell, K. (2015). Are Facebook "Friends" helpful? development of a Facebook-based measure of social support

and examination of relationships among depression, quality of life, and social support. *Cyberpsychology, Behavior, and Social Networking*, *18*(9), 499–505. https://doi.org/10.1089/cyber.2014.0538

- McCord, B., Rodebaugh, T. L., & Levinson, C. A. (2014). Facebook: Social uses and anxiety. *Computers in Human Behavior*, *34*, 23–27. https://doi.org/10.1016/j.chb.2014.01.020
- Michl, L. C., McLaughlin, K. A., Shepherd, K., & Nolen-Hoeksema, S. (2013). Rumination as a mechanism linking stressful life events to symptoms of depression and anxiety: longitudinal evidence in early adolescents and adults. *Journal of Abnormal Psychology*, *122*(2), 339– 352. https://doi.org/10.1037/a0031994
- Miller, G. (2012). The Smartphone Psychology Manifesto. *Perspectives on Psychological Science*, 7(3), 221–237. https://doi.org/10.1177/1745691612441215
- Moberg, F. B., & Anestis, M. D. (2015). A preliminary examination of the relationship between social networking interactions, internet use, and thwarted belongingness. *Crisis*, 36(3), 187– 193. https://doi.org/10.1027/0227-5910/a000311
- Mok, W. T., Sing, R., Jiang, X., & See, S. L. (2014). Investigation of social media on depression. In *The 9th International Symposium on Chinese Spoken Language Processing* (pp. 488–491). https://doi.org/10.1109/ISCSLP.2014.6936690
- Morahan-Martin, J. (2009). Internet use and abuse and psychological problems. In A. N. Joinson, K. Y. A McKenna, T. Postmes, U-D. Reips (Eds.), Oxford Handbook of Internet Psychology. http://doi.org/10.1093/oxfordhb/9780199561803.013.0021
- Moreau, A., Laconi, S., Delfour, M., & Chabrol, H. (2015). Psychopathological profiles of adolescent and young adult problematic Facebook users. *Computers in Human Behavior*, 44, 64–69. https://doi.org/10.1016/j.chb.2014.11.045

- Moreno, M. A., Christakis, D. A., Egan, K. G., Jelenchick, L. A., Cox, E., Young, H., ... Becker, T. (2012). A pilot evaluation of associations between displayed depression references on Facebook and self-reported depression using a clinical scale. *The Journal of Behavioral Health Services & Research*, 39(3), 295–304.
- Moreno, M. A., Grant, A., Kacvinsky, L., Moreno, P., & Fleming, M. (2012). Older adolescents' views regarding participation in Facebook research. *The Journal of Adolescent Health : Official Publication of the Society for Adolescent Medicine*, *51*(5), 439–444. https://doi.org/10.1016/j.jadohealth.2012.02.001
- Moreno, M. A., Jelenchick, L. A., Egan, K. G., Cox, E., Young, H., Gannon, K. E., & Becker, T. (2011). Feeling bad on Facebook: Depression disclosures by college students on a social networking site. *Depression and Anxiety*, 28(6), 447–455. https://doi.org/10.1002/da.20805
- Morgan, C., & Cotten, S. R. (2003). The relationship between internet activities and depressive symptoms in a sample of college freshmen. *Cyberpsychology & Behavior: The Impact of the Internet, Multimedia and Virtual Reality on Behavior and Society*, 6(2), 133–142. https://doi.org/10.1089/109493103321640329
- Morin-Major, J. K., Marin, M.-F., Durand, N., Wan, N., Juster, R.-P., & Lupien, S. J. (2016).
 Facebook behaviors associated with diurnal cortisol in adolescents: Is befriending stressful?
 Psychoneuroendocrinology, 63, 238–246. https://doi.org/10.1016/j.psyneuen.2015.10.005
- Morris, D. (2017, December 16). Facebook admits social media can harm your mental health. Retrieved from http://fortune.com/2017/12/16/facebook-admits-social-media-can-harmyour-mental-health/
- Mota Pereira, J. (2014). Facebook Enhances Antidepressant Pharmacotherapy Effects. *The Scientific World Journal*, 2014, 1–6. https://doi.org/10.1155/2014/892048

- Mowery, D., Smith, H., Cheney, T., Stoddard, G., Coppersmith, G., Bryan, C., & Conway, M. (2017). Understanding depressive symptoms and psychosocial stressors on Twitter: a corpus-based study. *Journal of Medical Internet Research*, 19(2), e48. https://doi.org/10.2196/jmir.6895
- Myers, S. A., Sharma, A., Gupta, P., & Lin, J. (2014). Information network or social network?: The structure of the Twitter follow graph (pp. 493–498). ACM, New York, NY, USA. https://doi.org/10.1145/2567948.2576939
- Nabi, R. L., Prestin, A., & So, J. (2013). Facebook friends with (health) benefits? Exploring social network site use and perceptions of social support, stress, and well-being. *Cyberpsychology, Behavior, and Social Networking*, *16*(10), 721–727.
 https://doi.org/10.1089/cyber.2012.0521
- Nambisan, P., Luo, Z., Kapoor, A., Patrick, T. B., & Cisler, R. A. (2015). Social media, big data, and public health informatics: Ruminating behavior of depression revealed through twitter (pp. 2906–2913). IEEE. https://doi.org/10.1109/HICSS.2015.351
- Naslund, J. A., Aschbrenner, K. A., McHugo, G. J., Unützer, J., Marsch, L. A., & Bartels, S. J. (2017). Exploring opportunities to support mental health care using social media: A survey of social media users with mental illness. *Early Intervention in Psychiatry*. Advance on publication. https://doi.org/10.1111/eip.12496
- Nolen-Hoeksema, S., Wisco, B. E., & Lyubomirsky, S. (2008). Rethinking rumination. *Perspectives on Psychological Science*, *3*(5), 400–424. http://doi.org/ 10.1111/j.1745-6924.2008.00088.x.
- Nowak, M. (2017). Two billion people coming together on Facebook. Retrieved January 17, 2018, from https://newsroom.fb.com/news/2017/06/two-billion-people-coming-together-onfacebook/
- Oh, H. J., Ozkaya, E., & LaRose, R. (2014). How does online social networking enhance life satisfaction? The relationships among online supportive interaction, affect, perceived social support, sense of community, and life satisfaction. *Computers in Human Behavior*, 30, 69– 78. https://doi.org/10.1016/j.chb.2013.07.053
- Palmier-Claus, J. E., Taylor, P. J., Gooding, P., Dunn, G., & Lewis, S. W. (2012). Affective variability predicts suicidal ideation in individuals at ultra-high risk of developing psychosis: An experience sampling study. *British Journal of Clinical Psychology*, *51*(1), 72–83. https://doi.org/10.1111/j.2044-8260.2011.02013.x
- Pantic, I., Damjanovic, A., Todorovic, J., Topalovic, D., Bojovic-Jovic, D., Ristic, S., & Pantic, S. (2012). Association between online social networking and depression in high school students: Behavioral physiology viewpoint. *Psychiatria Danubina*, 24(1), 90–93. Retrieved from

http://www.hdbp.org/psychiatria_danubina/pdf/dnb_vol24_no1/dnb_vol24_no1_90.pdf

- Papageorgiou, C., & Wells, A. (2003). An empirical test of a clinical metacognitive model of rumination and depression. *Cognitive Therapy and Research*, 27(3), 261–273. https://doi.org/10.1023/A:1023962332399
- Park, C. S. (2013). Does Twitter motivate involvement in politics? Tweeting, opinion leadership, and political engagement. *Computers in Human Behavior*, 29, 1641–1648. https://doi.org/10.1016/j.chb.2013.01.044
- Park, G., Schwartz, H. A., Eichstaedt, J. C., Kern, M. L., Kosinski, M., Stillwell, D. J., ... Seligman, M. E. P. (2015). Automatic personality assessment through social media language. *Journal of Personality and Social Psychology*, *108*(6), 934–952. https://doi.org/10.1037/pspp0000020

- Park, G., Schwartz, H. A., Sap, M., Kern, M. L., Weingarten, E., Eichstaedt, J. C., ... Seligman, M.
 E. P. (2017). Living in the Past, Present, and Future: Measuring Temporal Orientation With Language. *Journal of Personality*, 85(2), 270–280. https://doi.org/10.1111/jopy.12239
- Park, J., Lee, D. S., Shablack, H., Verduyn, P., Deldin, P., Ybarra, O., ... Kross, E. (2016). When perceptions defy reality: The relationships between depression and actual and perceived Facebook social support. *Journal of Affective Disorders*, 200, 37–44. https://doi.org/10.1016/j.jad.2016.01.048
- Park, S., Kim, I., Lee, S. W., Yoo, J., Jeong, B., & Cha, M. (2015). Manifestation of depression and loneliness on social networks: a case study of young adults on Facebook. In *Proceedings of the 18th ACM Conference on Computer Supported Cooperative Work & Social Computing* (pp. 557–570). ACM, New York, NY, USA. Retrieved from http://dl.acm.org/citation.cfm?id=2675139
- Park, S., Lee, S. W., Kwak, J., Cha, M., & Jeong, B. (2013). Activities on Facebook reveal the depressive state of users. *Journal of Medical Internet Research*, 15(10), e217. https://doi.org/10.2196/jmir.2718
- Peddinti, S. T., Ross, K. W., & Cappos, J. (2014). "On the Internet, nobody knows you're a dog": A Twitter case study of anonymity in social networks. In *Proceedings of the Second ACM Conference on Online Social Networks* (pp. 83–94). New York, NY, USA: ACM. https://doi.org/10.1145/2660460.2660467
- Peeters, F., Berkhof, J., Rottenberg, J., & Nicolson, N. A. (2010). Ambulatory emotional reactivity to negative daily life events predicts remission from major depressive disorder. *Behaviour Research and Therapy*, 48(8), 754–760. https://doi.org/10.1016/j.brat.2010.04.008

- Pennebaker, J., Mehl, M., & Niederhoffer, K. (2003). Psychological aspects of natural language use: our words, our selves. *Annual Review of Psychology*, 54, 547-77. http://doi.org/ 10.1146/annurev.psych.54.101601.145041
- Pennebaker, J. W. (1997). Writing about emotional experiences as a therapeutic process. *Psychological Science*, 8(3), 162–166. https://doi.org/10.1111/j.1467-9280.1997.tb00403.x
- Pennebaker, J. W. (2004). Theories, therapies, and taxpayers: On the complexities of the expressive writing paradigm. *Clinical Psychology: Science and Practice*, 11(2), 138–142. https://doi.org/10.1093/clipsy.bph063
- Pennebaker, J. W., Boyd, R. L., Jordan, K., & Blackburn, K. (2015). The development and psychometric properties of LIWC2015. Retrieved from https://repositories.lib.utexas.edu/handle/2152/31333
- Pennebaker, J. W., Chung, C. K., Ireland, M., Gonzales, A., & Booth, R. J. (2007). *The LIWC 2007 Application*. Retrieved from http://www.liwc.net
- Pennebaker Conglomerates. (2015). *LIWC 2015: Comparing LIWC 2015 and LIWC 2007*. Retrieved January 17, 2018, from https://liwc.wpengine.com/compare-dictionaries/
- Phillips, W. J., Hine, D. W., & Thorsteinsson, E. B. (2010). Implicit cognition and depression: A meta-analysis. *Clinical Psychology Review*, 30(6), 691–709. https://doi.org/10.1016/j.cpr.2010.05.002
- Phua, J., Jin, S. V., & Kim, J. (2017). Gratifications of using Facebook, Twitter, Instagram, or Snapchat to follow brands: The moderating effect of social comparison, trust, tie strength, and network homophily on brand identification, brand engagement, brand commitment, and membership intention. *Telematics and Informatics*, *34*(1), 412–424. https://doi.org/10.1016/j.tele.2016.06.004

- Preotiuc-Pietro, D., Eichstaedt, J. C., Park, G., Sap, M., Smith, L., Tobolsky, V., ... Ungar, L. H.
 (2015). The role of personality, age and gender in tweeting about mental illness (pp. 21–30).
 Presented at the *NAACL: Workshop on Computational Linguistics and Clinical Psychology: From Linguisitc Signal to Clinical Reality*, Denver, CO, USA. Retrieved from
 www.aclweb.org/anthology/W15-1203
- Preotiuc-Pietro, D., Sap, M., Schwartz, H. A., & Ungar, L. (2015). Mental Illness Detection at the World Well-Being Project for the CLPsych 2015 Shared Task. In *Proceedings of the Workshop on Computational Linguistics and Clinical Psychology: From Linguistic Signal to Clinical Reality.* NAACL. Retrieved from www.aclweb.org/anthology/W15-1205
- Prieto, V. M., Matos, S., Álvarez, M., Cacheda, F., & Oliveira, J. L. (2014). Twitter: A good place to detect health conditions. *PLoS ONE*, *9*(1). https://doi.org/10.1371/journal.pone.0086191
- Prinstein, M. J., & Aikins, J. W. (2004). Cognitive moderators of the longitudinal association between peer rejection and adolescent depressive symptoms. *Journal of Abnormal Child Psychology*, 32(2), 147–158. https://doi.org/10.1023/B:JACP.0000019767.55592.63
- Pulverman, C. S., Lorenz, T. A., & Meston, C. M. (2015). Linguistic changes in expressive writing predict psychological outcomes in women with history of childhood sexual abuse and adult sexual dysfunction. *Psychological Trauma: Theory, Research, Practice and Policy*, 7(1), 50–57. https://doi.org/10.1037/a0036462
- Qiu, L., Lin, H., Leung, A. K., & Tov, W. (2012). Putting their best foot forward: Emotional disclosure on Facebook. *Cyberpsychology, Behavior, and Social Networking*, 15(10), 569–572. https://doi.org/10.1089/cyber.2012.0200
- Radovic, A., Gmelin, T., Stein, B. D., & Miller, E. (2017). Depressed adolescents' positive and negative use of social media. *Journal of Adolescence*, 55, 5–15. https://doi.org/10.1016/j.adolescence.2016.12.002

- Rae, J. R., & Lonborg, S. D. (2015). Do motivations for using Facebook moderate the association between Facebook use and psychological well-being? *Frontiers in Psychology*, 6. https://doi.org/10.3389/fpsyg.2015.00771
- Ram, N., & Gerstorf, D. (2009). Time-structured and net intraindividual variability: Tools for examining the development of dynamic characteristics and processes. *Psychology and Aging*, 24(4), 778–791. https://doi.org/10.1037/a0017915
- Rauch, S. M., Strobel, C., Bella, M., Odachowski, Z., & Bloom, C. (2014). Face to face versus Facebook: Does exposure to social networking web sites augment or attenuate physiological arousal among the socially anxious? *Cyberpsychology, Behavior, and Social Networking*, *17*(3), 187–190. https://doi.org/10.1089/cyber.2012.0498
- ReachOut. (2017). ReachOut and Twitter team up to support young people affected by bad world news. Retrieved January 21, 2018, from https://about.au.reachout.com/reachout-and-twitterteam-up-to-support-young-people-affected-by-bad-world-news/
- Reavley, N. J., Morgan, A. J., & Jorm, A. F. (2014). Development of scales to assess mental health literacy relating to recognition of and interventions for depression, anxiety disorders and schizophrenia/psychosis. *The Australian and New Zealand Journal of Psychiatry*, 48(1), 61– 69. https://doi.org/10.1177/0004867413491157
- Rehfeld, K., Marwan, N., Heitzig, J., & Kurths, J. (2011). Comparison of correlation analysis techniques for irregularly sampled time series. *Nonlinear Processes in Geophysics*, 18(3), 389–404. https://doi.org/ 10.5194/npg-18-389-2011
- Reynolds, W. M. (1982). Development of reliable and valid short forms of the Marlowe-Crowne Social Desirability Scale. *Journal of Clinical Psychology*, 38(1), 119–125. https://doi.org/10.1002/1097-4679

- Rickard, N., Arjmand, H.-A., Bakker, D., & Seabrook, E. (2016). Development of a mobile phone app to support self-monitoring of emotional well-being: A mental health digital innovation. *JMIR Mental Health*, 3(4), e49. https://doi.org/10.2196/mental.6202
- Rosen, A., & Ihara, I. (2017, September 26). Giving you more characters to express yourself. Twitter. Retrieved October 23, 2017, from https://blog.twitter.com/official/en_us/topics/product/2017/Giving-you-more-characters-toexpress-yourself.html
- Rosen, L. D., Whaling, K., Rab, S., Carrier, L. M., & Cheever, N. A. (2013). Is Facebook creating "iDisorders"? The link between clinical symptoms of psychiatric disorders and technology use, attitudes and anxiety. *Computers in Human Behavior*, 29(3), 1243–1254. https://doi.org/10.1016/j.chb.2012.11.012
- Rosenberg, M. (1965). Society and the adolescent self-image. Princeton, NJ: Princeton University Press.
- Rosenquist, J., Fowler, J., & Christakis, N. (2011). Social network determinants of depression. *Molecular Psychiatry*, *16*(3). https://doi.org/10.1038/mp.2010.13
- Rude, S., Gortner, E.-M., & Pennebaker, J. (2004). Language use of depressed and depressionvulnerable college students. *Cognition and Emotion*, 18(8), 1121–1133. https://doi.org/10.1080/02699930441000030
- Russell, J. A. (1980). A circumplex model of affect. *Journal of Personality and Social Psychology*, 39(6), 1161–1178. https://doi.org/10.1037/h0077714
- Santini, Z. I., Koyanagi, A., Tyrovolas, S., Mason, C., & Haro, J. M. (2015). The association between social relationships and depression: A systematic review. *Journal of Affective Disorders*, 175, 53–65. https://doi.org/10.1016/j.jad.2014.12.049

Sato, H., & Kawahara, J. (2011). Selective bias in retrospective self-reports of negative mood states. *Anxiety, Stress, and Coping*, 24(4), 359–367. https://doi.org/10.1080/10615806.2010.543132

- Schaefer, D. R., Kornienko, O., & Fox, A. M. (2011). Misery does not love company: network selection mechanisms and depression homophily. *American Sociological Review*, 76(5), 764–785. https://doi.org/10.1177/0003122411420813
- Schwartz, H. A., Eichstaedt, J. C., Kern, M. L., Dziurzynski, L., Lucas, R. E., Agrawal, M., ... others. (2013). Characterizing geographic variation in well-being using Tweets. In *ICWSM*. Retrieved from http://www.aaai.org/ocs/index.php/ICWSM/ICWSM13/paper/viewPDFInterstitial/6138Han sen/6398
- Schwartz, H. A., Eichstaedt, J. C., Kern, M. L., Dziurzynski, L., Ramones, S. M., Agrawal, M., ...
 Ungar, L. H. (2013). Personality, Gender, and Age in the Language of Social Media: The
 Open-Vocabulary Approach. *PLoS ONE*, 8(9), e73791.
 https://doi.org/10.1371/journal.pone.0073791
- Schwartz, H. A., Eichstawdt, J., Kern, M. L., Park, G., Sap, M., Stillwell, D., ... Ungar, L. (2014). Towards assessing changes in degree of depression through Facebook. In *Workshop on Computational Linguistics and Clinical Psychology: From Linguistic Signal to Clinical Reality* (pp. 118–125). Baltimore, Maryland, USA: NAACL Retrieved from http://www.acl2014.org/acl2014/W14-32/pdf/W14-3214.pdf.
- Schwartz, H. A., Sap, M., Kern, M. L., Eichstaedt, J. C., Kapelner, A., Agrawal, M., ... Ungar, L.
 H. (2016). Predicting individual well-being through the language of social media. *Pacific Symposium on Biocomputing*. *Pacific Symposium on Biocomputing*, 21, 516–527. Retrieved from http://wwbp.org/papers/2016_predicting_wellbeing.pdf

- Schwartz, H. A., & Ungar, L. H. (2015). Data-driven content analysis of social media: A systematic overview of automated methods. *The ANNALS of the American Academy of Political and Social Science*, 659(1), 78–94. https://doi.org/10.1177/0002716215569197
- Seabrook, E. M., Kern, M. L., Fulcher, B. D., & Rickard, N. S. (2018) Depression is predicted by emotional instability on Facebook, but by reduced emotion variability on Twitter.
 Manuscript submitted for publication.
- Seabrook, E. M., Kern, M. L., & Rickard, N. S. (2016). Social networking sites, depression, and anxiety: A systematic review. *JMIR Mental Health*, 3(4), e50. https://doi.org/10.2196/mental.5842
- Segrin, C., & Flora, J. (1998). Depression and verbal behavior in conversations with friends and strangers. *Journal of Language and Social Psychology*, 17(4), 492–503. https://doi.org/10.1177/0261927X980174005
- Seidman, G. (2013). Self-presentation and belonging on Facebook: How personality influences social media use and motivations. *Personality and Individual Differences*, 54(3), 402–407. https://doi.org/10.1016/j.paid.2012.10.009
- Selim, H. A., Long, K. M., & Vignoles, V. L. (2014). Exploring identity motives in Twitter usage in Saudi Arabia and the UK. *Studies in Health Technology and Informatics*, 199, 128–132. http://doi.org/10.3233/978-1-61499-401-5-128
- Semenov, A., Natekin, A., Nikolenko, S., Upravitelev, P., Trofimov, M., & Kharchenko, M. (2015). Discerning depression propensity among participants of suicide and depression-related groups of Vk.com. In *Analysis of Images, Social Networks and Texts* (pp. 24–35). Springer, Cham. https://doi.org/10.1007/978-3-319-26123-2_3

- Settanni, M., & Marengo, D. (2015). Sharing feelings online: studying emotional well-being via automated text analysis of Facebook posts. *Frontiers in Psychology*, 6. https://doi.org/10.3389/fpsyg.2015.01045
- Shapero, B. G., Black, S. K., Liu, R. T., Klugman, J., Bender, R. E., Abramson, L. Y., & Alloy, L.
 B. (2014). Stressful life events and depression symptoms: The effect of childhood emotional abuse on stress reactivity. *Journal of Clinical Psychology*, 70(3), 209–223. https://doi.org/10.1002/jclp.22011
- Shaw, A. M., Timpano, K. R., Tran, T. B., & Joormann, J. (2015). Correlates of Facebook usage patterns: The relationship between passive Facebook use, social anxiety symptoms, and brooding. *Computers in Human Behavior*, 48, 575–580. https://doi.org/10.1016/j.chb.2015.02.003
- Shensa, A., Escobar-Viera, C. G., Sidani, J. E., Bowman, N. D., Marshal, M. P., & Primack, B. A. (2017). Problematic social media use and depressive symptoms among U.S. young adults: A nationally-representative study. *Social Science & Medicine*, 182, 150–157. https://doi.org/10.1016/j.socscimed.2017.03.061
- Simoncic, T. E., Kuhlman, K. R., Vargas, I., Houchins, S., & Lopez-Duran, N. L. (2014). Facebook use and depressive symptomatology: Investigating the role of neuroticism and extraversion in youth. *Computers in Human Behavior*, 40, 1–5. https://doi.org/10.1016/j.chb.2014.07.039
- Sinclair, S. J., Blais, M. A., Gansler, D. A., Sandberg, E., Bistis, K., & LoCicero, A. (2010).
 Psychometric properties of the Rosenberg Self-Esteem Scale: Overall and across demographic groups living within the United States. *Evaluation & the Health Professions*, 33(1), 56–80. https://doi.org/10.1177/0163278709356187

- Smith, B. W., Dalen, J., Wiggins, K., Tooley, E., Christopher, P., & Bernard, J. (2008). The Brief Resilience Scale: Assessing the ability to bounce back. *International Journal of Behavioral Medicine*, 15(3), 194–200. https://doi.org/10.1080/10705500802222972
- Smyth, J. M. (1998). Written emotional expression: effect sizes, outcome types, and moderating variables. *Journal of Consulting and Clinical Psychology*, 66(1), 174–184. http://dx.doi.org/10.1037/0022-006X.66.1.174
- Sowislo, J. F., & Orth, U. (2013). Does low self-esteem predict depression and anxiety? A metaanalysis of longitudinal studies. *Psychological Bulletin*, *139*(1), 213–240. https://doi.org/10.1037/a0028931
- Spies Shapiro, L. A., & Margolin, G. (2014). Growing up wired: Social networking sites and adolescent psychosocial development. *Clinical Child and Family Psychology Review*, 17(1), 1–18. https://doi.org/10.1007/s10567-013-0135-1
- Spitzer, R. L., Kroenke, K., Williams, J. B. W., & Löwe, B. (2006). A brief measure for assessing generalized anxiety disorder: The GAD-7. *Archives of Internal Medicine*, 166(10), 1092– 1097. https://doi.org/10.1001/archinte.166.10.1092
- Steers, M. N., Wickham, R. E., & Acitelli, L. K. (2014). Seeing everyone else's highlight reels: how Facebook usage is linked to depressive symptoms. *Journal of Social and Clinical Psychology*, 33(8), 701–731. https://doi.org/10.1521/jscp.2014.33.8.701
- Steger, M. F., & Kashdan, T. B. (2009). Depression and everyday social activity, belonging, and well-being. *Journal of Counseling Psychology*, 56(2), 289–300. https://doi.org/10.1037/a0015416
- Stoyanov, S. R., Hides, L., Kavanagh, D. J., Zelenko, O., Tjondronegoro, D., & Mani, M. (2015).
 Mobile App Rating Scale: A new tool for assessing the quality of health mobile apps. *JMIR mHealth and uHealth*, *3*(1), e27. https://doi.org/10.2196/mhealth.3422

- Szwedo, D. E., Mikami, A. Y., & Allen, J. P. (2011). Qualities of peer relations on social networking websites: predictions from negative mother-teen interactions. *Journal of Research on Adolescence*, 21(3), 595–607. https://doi.org/10.1111/j.1532-7795.2010.00692.x
- Takahashi, Y., Uchida, C., Miyaki, K., Sakai, M., Shimbo, T., & Nakayama, T. (2009). Potential benefits and harms of a peer support social network service on the internet for people with depressive tendencies: qualitative content analysis and social network analysis. *Journal of Medical Internet Research*, 11(3), e29. https://doi.org/10.2196/jmir.1142
- Tandoc, E. C., Ferrucci, P., & Duffy, M. (2015). Facebook use, envy, and depression among college students: Is Facebooking depressing? *Computers in Human Behavior*, 43, 139–146. https://doi.org/10.1016/j.chb.2014.10.053
- Tausczik, Y. R., & Pennebaker, J. W. (2010). The psychological meaning of words: LIWC and computerized text analysis methods. *Journal of Language and Social Psychology*, 29(1), 24–54. https://doi.org/10.1177/0261927X09351676
- Taylor, C. (2011, January 12). Your New Facebook Status: 63,206 Characters or Less. Mashable. Retrieved October 23, 2017, from http://mashable.com/2011/11/30/facebook-status-63206characters/#oOlxKAXmjSqu
- Tennant, R., Hiller, L., Fishwick, R., Platt, S., Joseph, S., Weich, S., ... Stewart-Brown, S. (2007).
 The Warwick-Edinburgh Mental Well-being Scale (WEMWBS): Development and UK validation. *Health and Quality of Life Outcomes*, *5*, 63. https://doi.org/10.1186/1477-7525-5-63
- Thompson, R. J., Berenbaum, H., & Bredemeier, K. (2011). Cross-Sectional and longitudinal relations between affective instability and depression. *Journal of Affective Disorders*, *130*(1–2), 53–59. https://doi.org/10.1016/j.jad.2010.09.021

- Thompson, R. J., Mata, J., Jaeggi, S. M., Buschkuehl, M., Jonides, J., & Gotlib, I. H. (2012). The everyday emotional experience of adults with major depressive disorder: Examining emotional instability, inertia, and reactivity. *Journal of Abnormal Psychology*, *121*(4), 819– 829. https://doi.org/10.1037/a0027978
- Trull, T. J., & Ebner-Priemer, U. W. (2009). Using Experience Sampling Methods/Ecological Momentary Assessment (ESM/EMA) in clinical assessment and clinical research: Introduction to the special section. *Psychological Assessment*, 21(4), 457–462. https://doi.org/10.1037/a0017653
- Tsai, C.-W., Shen, P.-D., & Chiang, Y.-C. (2015). Meeting ex-partners on Facebook: users' anxiety and severity of depression. *Behaviour & Information Technology*, 34(7), 668–677. https://doi.org/10.1080/0144929X.2014.981585
- Tsugawa, S., Kikuchi, Y., Kishino, F., Nakajima, K., Itoh, Y., & Ohsaki, H. (2015). Recognizing depression from Twitter activity (pp. 3187–3196). ACM, New York, NY, USA. https://doi.org/10.1145/2702123.2702280
- Tufekci, Z. (2014). Big questions for social media big data: Representativeness, validity and other methodological pitfalls (pp. 505-514). In *Proceedings of the 8th International Conference on Weblogs and Social Media,* ICWSM, Ann Arbor, USA. Retrieved from http://arxiv.org/abs/1403.7400
- Turner, R. J., & Marino, F. (1994). Social support and social structure: A descriptive epidemiology. Journal of Health and Social Behavior, 35(3), 193–212. Retrieved from http://www.jstor.org/stable/2137276
- Uchino, B. N. (2009). Understanding the links between social support and physical health: A lifespan perspective with emphasis on the separability of perceived and received support.

Perspectives on Psychological Science: A Journal of the Association for Psychological Science, 4(3), 236–255. https://doi.org/10.1111/j.1745-6924.2009.01122.x

- Unicode Inc. (2017). *What is Unicode?* Retrieved January 18, 2018, from http://unicode.org/standard/WhatIsUnicode.html
- Valkenburg, P. M., Peter, J., & Schouten, A. P. (2006). Friend networking sites and their relationship to adolescents' well-being and social self-esteem. *Cyberpsychology & Behavior: The Impact of the Internet, Multimedia and Virtual Reality on Behavior and Society*, 9(5), 584–590. https://doi.org/10.1089/cpb.2006.9.584
- Verduyn, P., & Brans, K. (2012). The relationship between extraversion, neuroticism and aspects of trait affect. *Personality and Individual Differences*, 52(6), 664–669. https://doi.org/10.1016/j.paid.2011.12.017
- Verduyn, P., Ybarra, O., Résibois, M., Jonides, J., & Kross, E. (2017). Do social network sites enhance or undermine subjective well-being? A Critical Review. *Social Issues and Policy Review*, 11(1), 274–302. http://doi.org/ 10.1111/sipr.12033
- Von Neumann, J., Kent, R. H., Bellinson, H. R., & Hart, B. I. (1941). The mean square successive difference. *The Annals of Mathematical Statistics*, 12(2), 153–162. https://doi.org/10.1214/aoms/1177731746
- Wang, X., Cai, L., Qian, J., & Peng, J. (2014). Social support moderates stress effects on depression. *International Journal of Mental Health Systems*, 8(1), 41. https://doi.org/10.1186/1752-4458-8-41
- Waterloo, S. F., Baumgartner, S. E., Peter, J., & Valkenburg, P. M. (2017). Norms of online expressions of emotion: Comparing Facebook, Twitter, Instagram, and WhatsApp. *New Media & Society*, 1461444817707349. https://doi.org/10.1177/1461444817707349

- Wegmann, E., Stodt, B., & Brand, M. (2015). Addictive use of social networking sites can be explained by the interaction of Internet use expectancies, Internet literacy, and psychopathological symptoms. *Journal of Behavioral Addictions*, 4(3), 155–162. https://doi.org/10.1556/2006.4.2015.021
- Weidman, A. C., & Levinson, C. A. (2015). I'm still socially anxious online: Offline relationship impairment characterizing social anxiety manifests and is accurately perceived in online social networking profiles. *Computers in Human Behavior*, 49, 12–19. https://doi.org/10.1016/j.chb.2014.12.045
- Wheeler, L., & T. Reis, H. (1991). Self-recording of everyday life events: Origins, types, and uses. *Journal of Personality*, 59(3), 339-354. http://doi.org/10.1111/j.1467-6494.1991.tb00252.x
- Wichers, M. (2014). The dynamic nature of depression: A new micro-level perspective of mental disorder that meets current challenges. *Psychological Medicine*, 44(7), 1349–1360. https://doi.org/10.1017/S0033291713001979
- Wichers, M., Geschwind, N., Jacobs, N., Kenis, G., Peeters, F., Derom, C., ... van Os, J. (2009).
 Transition from stress sensitivity to a depressive state: longitudinal twin study. *The British Journal of Psychiatry*, *195*(6), 498–503. https://doi.org/10.1192/bjp.bp.108.056853
- Wichers, M., Jacobs, N., Derom, C., Thiery, E., & van Os, J. (2007). Depression: Too much negative affect or too little positive affect? *Twin Research and Human Genetics*, *10*(S1), 19–20. https://doi.org/10.1375/twin.10.supp.19
- Wills, T. A., & Shinar, O. (2000). Measuring perceived and received social support. In Social support measurement and intervention: A guide for health and social scientists. (pp. 86–135). New York, NY, US: Oxford University Press.

- Wilson, M. L., Ali, S., & Valstar, M. F. (2014). Finding information about mental health in microblogging platforms: a case study of depression (pp. 8–17). ACM, New York, NY, USA. https://doi.org/10.1145/2637002.2637006
- Wilson, R. E., Gosling, S. D., & Graham, L. T. (2012). A review of Facebook research in the social sciences. Perspectives on Psychological Science: A Journal of the Association for Psychological Science, 7(3), 203–220. https://doi.org/10.1177/1745691612442904
- Winter, S., Neubaum, G., Eimler, S. C., Gordon, V., Theil, J., Herrmann, J., ... Krämer, N. C. (2014). Another brick in the Facebook wall – How personality traits relate to the content of status updates. *Computers in Human Behavior*, 34, 194–202. https://doi.org/10.1016/j.chb.2014.01.048
- Wojtowicz, I. (2011). UnFacebook World [image]. Retrieved from https://ianwojtowicz.com/UnFacebook%20World/False-Color-Facebook-NASA-Mashup.png
- Wood, A., Lupyan, G., & Niedenthal, P. (2016). Why do we need emotion words in the first place?
 Commentary on Lakoff (2015). *Emotion Review*, 8(3), 274–275.
 https://doi.org/10.1177/1754073915595103
- World Health Organization. (2017). *Depression and other common mental disorders: global health estimates*. Retrieved from http://apps.who.int/iris/handle/10665/254610
- Wright, K. B., Rosenberg, J., Egbert, N., Ploeger, N. A., Bernard, D. R., & King, S. (2013).
 Communication competence, social support, and depression among college students: A model of Facebook and face-to-face support network influence. *Journal of Health Communication*, 18(1), 41–57. https://doi.org/10.1080/10810730.2012.688250
- Yang, W., & Mu, L. (2015). GIS analysis of depression among Twitter users. Applied Geography, 60, 217–223. https://doi.org/10.1016/j.apgeog.2014.10.016

- Yang, W., Mu, L., & Shen, Y. (2015). Effect of climate and seasonality on depressed mood among twitter users. *Applied Geography*, 63, 184–191. https://doi.org/10.1016/j.apgeog.2015.06.017
- Youngs, E. A. (1974). Human errors in programming. *International Journal of Man-Machine Studies*, 6(3), 361–376. https://doi.org/10.1016/S0020-7373(74)80027-1

Zimet, G. D., Powell, S. S., Farley, G. K., Werkman, S., & Berkoff, K. A. (1990). Psychometric characteristics of the Multidimensional Scale of Perceived Social Support. *Journal of Personality Assessment*, 55(3/4), 610–617. http://dx.doi.org/10.1207/s15327752jpa5503&4_17

Appendices

Appendix A: Development of a Mobile Phone App to Support Self-Monitoring of Emotional Well-Being: A Mental Health Digital Innovation

Development of a Mobile Phone App to Support Self-Monitoring of Emotional Well-Being: A Mental Health Digital Innovation

Nikki Rickard^{1,2*}, BBSc (Hons), PhD; Hussain-Abdulah Arjmand^{1*}, BBNSc (Hons); David Bakker^{1*}, BPsych (Hons); Elizabeth Seabrook^{1*}, BPsych (Hons)

¹Emotion and Well-being Research Unit, School of Psychological Sciences, Monash University, Clayton, Australia

²Centre for Positive Psychology, Melbourne Graduate School of Education, University of Melbourne, Melbourne, Australia *all authors contributed equally

Corresponding Author:

Nikki Rickard, BBSc (Hons), PhD Emotion and Well-being Research Unit School of Psychological Sciences Monash University Building 18 Wellington Rd Clayton, 3800 Australia Phone: 61 400 191 768 Fax: 61 3 9905 3948 Email: nikki.rickard@monash.edu

Abstract

Background: Emotional well-being is a primary component of mental health and well-being. Monitoring changes in emotional state daily over extended periods is, however, difficult using traditional methodologies. Providing mental health support is also challenging when approximately only 1 in 2 people with mental health issues seek professional help. Mobile phone technology offers a sustainable means of enhancing self-management of emotional well-being.

Objective: This paper aims to describe the development of a mobile phone tool designed to monitor emotional changes in a natural everyday context and in real time.

Methods: This evidence-informed mobile phone app monitors emotional mental health and well-being, and it provides links to mental health organization websites and resources. The app obtains data via self-report psychological questionnaires, experience sampling methodology (ESM), and automated behavioral data collection.

Results: Feedback from 11 individuals (age range 16-52 years; 4 males, 7 females), who tested the app over 30 days, confirmed via survey and focus group methods that the app was functional and usable.

Conclusions: Recommendations for future researchers and developers of mental health apps to be used for research are also presented. The methodology described in this paper offers a powerful tool for a range of potential mental health research studies and provides a valuable standard against which development of future mental health apps should be considered.

(JMIR Ment Health 2016;3(4):e49) doi:10.2196/mental.6202

KEYWORDS

eHealth; emotions; mental health; mobile phone; feedback

Introduction

Background

Emotional well-being is broadly defined [1] as, "a positive sense of well-being and an underlying belief in our own and others' dignity and worth" by the Mental Health Foundation (p. 8).

http://mental.jmir.org/2016/4/e49/

Consistent with dual models of well-being, it encompasses both positive functioning (happiness, a sense of control and self-efficacy, and social connectedness) and an absence of stress and depression [2,3]. Monitoring changes in emotional well-being is fundamental to mental health, with increases in emotional well-being associated with resilience, creative thinking, social connectivity, and physical health [4-9]. In

> JMIR Ment Health 2016 | vol. 3 | iss. 4 | e49 | p.1 (page number not for citation purposes)

contrast, significant and sustained decreases in emotional well-being are associated with the development of affective disorders such as depression and anxiety, and reduced physical health [4,5,7].

Monitoring for such changes is crucial for early detection of mental health problems. Rapid response to early risk indicators is one of the key predictors of better health outcomes, enabling preventative health approaches to be initiated early [10]. Regular monitoring of emotional health indices is therefore recommended by various national guidelines [11,12]. In practice, however, it remains difficult for clinicians or professional mental health service providers to obtain frequent monitoring in real time [13,14]. A priority challenge facing the health care system is to achieve practicable and sustainable means of supporting self-management of health and well-being. Self-monitoring is a particularly attractive goal for mental health care, given that many individuals with mental health needs do not seek professional health care support [15-17]. In addition, self-monitoring may develop an individual's insight into their need to seek help. In particular, young people consistently indicate that they prefer nonprofessional or self-managed strategies for addressing mental health issues [18,19]. Obtaining temporally sensitive (eg, daily) information on significant changes in emotional state has the potential to profoundly improve the capacity to promote emotional health [12].

Experience sampling methodologies (ESMs), or ecological momentary assessments, involve the systematic collection of self-report data from individuals at multiple time points throughout their everyday lives [20]. ESMs have been used to monitor changes in affective state, and to predict mental health with success to a certain extent [21,22]. In particular, the variability in emotional state over time provides more substantial information for understanding the causes and nature of psychopathology than do cross-sectional "snapshot" assessments. For example, when sampled multiple times a day for 6 days, negative affect was found to vary more in patients diagnosed with major depressive disorder than that in controls across the day [23]. ESM assessments in individuals diagnosed with panic disorder also revealed that the expectation of a panic attack was a significant precursor for the occurrence of a panic attack [24]. Ben-Zeev et al [25] also found that patients diagnosed with a major depressive disorder retrospectively reported higher levels of symptoms relating to anhedonia, suicidality, and sadness than captured in their ESM reports, highlighting the biases of traditional survey methods. To date, however, it has been methodologically difficult and obtrusive to obtain temporally regular and precise measures of emotional state [21]. The resources required to obtain such information repeatedly over lengthy time frames have made such an intensive monitoring prohibitive. In addition, the use of palm pilots and pagers (which were never as familiar to users as mobile phones have become) to prompt users for this information can be intrusive, and makes it less likely that users will continue to use this form of monitoring for extended periods [26].

Mobile phone technology offers an unprecedented opportunity to unobtrusively track everyday behavior and changes in emotional state, all in real time [27,28]. Mobile phone health tools also offer the potential of immediate response to the

http://mental.jmir.org/2016/4/e49/

XSL•FO

Rickard et al

outcome of this monitoring via delivery of mental health information contingent on changes in real-time emotional state [29]. This technology has not yet been fully leveraged for these purposes, despite mobile phones being one of the few pieces of technology that most people carry on their person every day [30]. This pervasiveness means that mobile phones offer a highly natural and regular means by which information on emotional state could be obtained. Mobile phones now penetrate 77%, 72%, and 68% of the Australian, US, and UK population, respectively [31], and are a cost-effective means of seeking help for mental health issues that may overcome socioeconomic and geographic boundaries [32,33].

Mobile phone health technology holds great potential for facilitating the management of emotional health through its ability to deliver flexible, user-oriented intervention and self-management tools; a feature particularly relevant for young people who often report fear of stigma associated with seeking professional services for sensitive mental health issues [34,35]. In a 2010 study, 76% of an Australian sample reported being interested in using mobile phones to monitor and manage their own mental health [32]. A large number of mobile phone apps are currently available that claim to promote mental health and well-being [36,37] and a subset of these also attempt to track mood or emotional state over time. However, empirical support for the efficacy of these apps is extremely limited [36]. For instance, in a systematic review of 5464 mental health app abstracts, less than 5 apps were found to have experimental evidence [37]. In addition, a few have capitalized on the benefits enabled by the mobile phone technology such as experience sampling and automated data collection in identifying and evaluating potential time-sensitive behavioral indicators of mental health change [36].

Of the mobile phone mental health programs that have utilized ESM to track mood over time, several favorable outcomes have been reported. For example, Reid et al [28,38] found that the majority of their adolescent sample using the mobile phone-based mental health app, mobiletype, completed their self-assessments, and that the use of the app increased the practitioners' understanding of their patients' mental health. Harrison et al [29] reported that the use of the mobile phone accessed Web-based cognitive behavioral therapy (CBT) course MyCompass for 6 weeks significantly reduced symptoms of depression and anxiety and improved self-efficacy. One of the barriers to sustainability of user engagement in such programs, however, is that they require extensive voluntary input from the user. When evaluated, a common theme is initial compliance, followed by high dropout and poor self-reporting rates (eg, less than 10% of the sample trialing MyCompass reported using it every day) [29]. Reasons for discontinued use include problems understanding how to use the program, invasiveness of the questions, the need for repetitive completion of questionnaires, insufficient personalization of the mental health advice, and little motivation to engage with the program [28,29].

An innovative way to meet this challenge is to monitor indices of emotional health using methods that require minimal insight or subjective report from the user. Mobile phones contain a range of embedded sensors and features, including accelerometers and global positioning systems and apps, which

> JMIR Ment Health 2016 | vol. 3 | iss. 4 | e49 | p.2 (page number not for citation purposes)

can automatically record information about a user's behavior [39]. Two recent studies have obtained a combination of data from mobile phones in an attempt to predict participants' self-reported mood. LiKamWa et al [40] found that up to 93% of mood scores were accurately predicted by social activity, physical activity, and general mobile phone use data collected from mobile phones. Asselbergs et al [41] attempted to predict self-reported mood of 27 participants from metadata of 6 mobile phone indices (phone calls, text messages, screen time, app usage, accelerometer, and phone camera events). Although the accuracy of the models was no greater than models obtained without mobile phone data, the methodology was demonstrated to be technically feasible and to hold promise. The authors recommended that inclusion of more meaningful or relevant features from mobile phone data may be the key to improving prediction.

Interestingly, young people use mobile phones for music listening, fitness, and social networking more than any other demographic [42], and these are among the most effective strategies for optimizing emotional health [43-46]. For example, the frequency of app-switching and the content of social network messages were found to predict depression [43] even prior to its onset [47]. Music listening patterns also appear to predict emotional health [48-50] and given that approximately two thirds of music listening by young people is via mobile devices such as mobile phones [31], it is surprising that relatively few apps have attempted to use music for this purpose [27]. Vocal expression too has been found to be a useful index of emotional state [51,52]. Short voice samples have been found to demonstrate 70% accuracy for simple affect recognition [53]. Monitoring a combination of behavioral indices such as physical activity, online social interactions, and music choices therefore offers a promising means of nonintrusive but sensitive assessment of affective state. Advances in statistical methods available through machine learning also enable powerful analysis of this more complex level of individualized multilevel modeling [52,54].

Another limitation of most mental health apps currently available is that they tend to simplify the emotional well-being spectrum, with positive and negative affect anchors on a unidimensional rating scale. Contemporary conceptualizations of well-being however clearly show that optimal "emotional health and well-being" does not emerge from an absence of affective disorder alone, but also requires a state of positive functioning [2,55,56]. Although positive and negative emotional functioning are correlated, there is substantial evidence that they are orthogonal constructs [57]. Mobile phone technology that differentiates the quadrants created by categorizing according to mental illness or languishing and mental health or flourishing [3,55] is therefore encouraged.

Objective

In this paper, we capitalized on the extraordinary role that mobile phones play in people's lives to develop a tool that has the potential to significantly extend the understanding of emotional health and well-being. The aim of this paper was to describe the design of the mobile phone app, *MoodPrism*, which was developed to monitor emotional well-being in context and

http://mental.jmir.org/2016/4/e49/

XSL•FO

Rickard et al

in real time, and provide personalized feedback on the full spectrum of emotional well-being. The paper describes in detail the design and data collection functions of the app, which were incorporated to address major challenges for mental health research and practice, and presents feedback from a small sample of trial users (beta-testers), which tested the functionality and usability of the app.

Methods

Design and Development of the App

MoodPrism was designed and developed in collaboration with a commercial digital creation studio, Two Bulls (Melbourne, Australia). The app was prepared for both the iOS and Android mobile phone platforms and was distributed by the Web-based Apple and Google Play stores, respectively. The term "*MoodPrism*" was selected to reflect its primary purpose of collecting emotional state data across the entire spectrum of emotional health and well-being and converting this into an array of color-coded feedback to the user.

The development of *MoodPrism* involved designing 3 different methods of data collection within the software: (1) automated monitoring of selected online behavior, (2) experience sampling of emotional well-being self-reports, and (3) psychological assessment questionnaires. automated monitoring of selected online behavior, experience sampling of emotional well-being self-reports, and psychological assessment questionnaires. This triangulation of data collection is considered crucial for advancing the measurement of emotional state [58]. As part of the sign-up procedure to the app, permissions for sensitive data had to be obtained. Incentives to continue collecting data over an extended period were also generated.

The development of *MoodPrism* was completed in March 2015. The required forms of data collection were achieved by developing a suite of app components, which were then collated into a cohesive app. The outcomes of this development process are described in the following.

Sign Up

As part of the sign-up procedure for the app, options were offered to users to provide the app with access to social networking and music apps as well as general (postcode) location. These data were then collected continuously and without the need for user input over the month's research period. After sign up and consent procedures, MoodPrism administered the initial surveys that could be completed in multiple sittings and required 30-60 min in total to complete. The participants were then requested to use the app for at least thirty days, during which they would be prompted daily to answer a set of short questions, and weekly to complete a short audio recording. If they were unable to respond to daily prompts, MoodPrism advised they could complete them at a time of their convenience till midnight that day, or alternatively to ignore them. At the end of the 30 days, users were invited to complete a final set of surveys, which in total required 15-30 min to complete.

Users were incentivized to continue using *MoodPrism* through 3 strategies. First, daily mood and mental health feedback was provided to the user, with additional feedback unlocked after

> JMIR Ment Health 2016 | vol. 3 | iss. 4 | e49 | p.3 (page number not for citation purposes)

sustained use (Multimedia Appendix 2). This promoted engagement by rewarding users and encouraging feelings of achievement, adhering to principles of gamification [59], which is recommended in mental health apps [36]. Second, completion of daily reports as well as the final surveys generated entries into a draw for 1 of the 4 \$AU100 (approximately US \$75) gift vouchers. Third, users were informed that their data were contributing to research into the value of mobile phone apps for monitoring mental health and well-being.

Automated Monitoring

MoodPrism acted as a portal for data accessed via several mobile phone sensors and apps. Two validated predictors of emotional state change were targeted: music use and web-based social network site activity. As a part of the sign-up process, users were invited to give permission for the app to access Facebook, Twitter, the user's music library, and location (postcode only).

Facebook, Twitter, and music use data were collected once every 24 h, and the information collected is provided in Multimedia Appendix 1. Data were accessed from Facebook and Twitter through their relevant application programming interfaces (APIs). This allows third-party access to selected data collected by both Facebook and Twitter. Facebook and Twitter content was analyzed automatically and locally on the user's phone using several linguistic dictionaries from the Linguistic Inquiry and Word Count (LIWC) [60]. Summaries were obtained for frequencies of emotion words, which were supplemented with a range of emoticons and Internet slang expressions for emotions. Social words and personal pronoun counts were also obtained. A word count for the target categories in the dictionary was extracted and these counts were uploaded to the server. This was repeated every 24 h to collect the posts that occured across the duration of MoodPrism use. The post content temporarily stored by MoodPrism was then deleted.

Experience Sampling

MoodPrism utilized ESM to deliver a short set of questions to users daily (Figure 1). Prompts were delivered at a quasi-random time between user-defined hours (eg, 9:00 am-9:00 pm) for 30 davs.

Rickard et al

The questions captured a real-time assessment of the user's emotional well-being, event-related experiences, and their context. Emotional state questions comprised 4 questions on psychological illhealth (depression and anxiety), 4 on emotional state (positive and negative affect, arousal, and control), and 4 on positive functioning (social connection, motivation, meaning, and self-esteem). Positive and negative event-related experiences were assessed by the type of event experienced and a rating of the event's affective strength (from "slightly" to "extremely positive or negative"). The type of event was selected from a range of options drawn from stressor event questionnaires [61-65] and modified as a short list of the most common event domains (eg, school or work, physical health, material possessions, or social experience domain). Context was assessed via 2 questions, 1 for social context (who the user was with at the time of the report) and environmental context (where they were at the time of the report). Specific questions are given in Table 1.

In addition, a weekly prompt was delivered that requested a short voice recording to serve as an implicit measure of emotional state [51,53]. Users were prompted to read a standardized piece of text at the start and the end of the recording, and within that window to describe freely how they were feeling at that time.

Psychological Assessment Questionnaires

A number of questionnaires were available for completion at the onset of the app use, providing baseline measures of emotional well-being as well as data on potential moderators or confounding variables (see Figure 2). These questionnaires were categorized into survey "blocks" and displayed on the MoodPrism homescreen until their completion. This served to organize the questionnaires into manageable chunks for users to complete in their own time. A subset of these questionnaires was also delivered at the end of the month-long period to enable assessment of whether the app may have affected the well-being measures. A description of these questionnaires was provided in Table 2.



Figure 1. Screen shots from app showing experience sampling method.

http://mental.jmir.org/2016/4/e49/ XSI+FO

JMIR Ment Health 2016 | vol. 3 | iss. 4 | e49 | p.4 (page number not for citation purpo

Rickard et al



Figure 2. Screenshots showing examples of longer psychological questionnaires.

Table 1.	Qualitative	feedback:	questions	guiding	qualitative	feedback f	orums.
----------	-------------	-----------	-----------	---------	-------------	------------	--------

Broad question	Prompts
Was the app easy to use?	Privacy issues (eg. social networking sites)
	Was it clear to you why you were providing the information that you did?
	Why did you opt-in or opt-out of connecting your social media accounts? What things would be an incentive to opt-in?
	Can you imagine anyone using the app without incentives?
	Who do you think would benefit from using it?
	Was it clear to you that you were earning entries into a draw to win an iPad? Was it clear how the prize entries were being awarded? Did this consciously motivate you to use the app?
	Were the colors or emoticons used in the mood feedback helpful?
How did you find the daily	Did they get in the way at all?
prompts?	Were significant events captured?
	What kind of event did you feel was appropriate to report (major, minor, or both)?
How did you find the feedback?	Mood feedback
	Did you notice yourself paying more attention to the way you feel than usual?
	When you started using the app, was it made clear that reporting your mood could improve your mental health and well-being?
	Surveys
	Mental health info or contacts - did you explore any of these? Were they useful?
	Did you ever find the overview upsetting or negative?

Table 2. Sample feedback provided by beta-testers.

Theme	Sample responses
Positive feedback	
Aesthetically pleasant	It looks nice!
Easy to use	Seamless and smooth to use
Daily reports quick to complete	Simple set of responses takes only a few minutes daily - easy to use daily
Feedback useful and specific	Targeted questions give specific feedback about links between mood and daily activities Colored display of mood was useful representation [sic] Liked unlocking of content - motivated to keep using Feedback was not upsetting
Good to be able to get feedback about how feelings change daily	The ease of the app and being able to check in how exactly I'm feeling at a certain time
Negative feedback	
Wording of some questions confusing	Many questions in the introductory questionnaires are confusing double-negative repeats of previous questions, combined with putting negative responses near the top (where you expect positive ones) is confusing.
	I've never been irked when people expressed ideas very different from my own: "Yes or No". Is it possible to put Agree or Disagree instead?
Some content can make you feel negative	Quite morbid things in the list of "most negative thing to happen to you today" makes me imagine some pretty terrible rare events like "death of a loved one", etc not a great thing to remind someone with depression to think about on a daily basis. / Many questions are quite negative like this you think about how stressed, worried, out of control, etc. you are creates a major disincentive to participating they're not things you want to dwell on when you're depressed.
Feedback clarity	The summary information for tracking well-being across times seems simplistic. For example, if I was in a good but deactivated mood, it said I was "on my way to thriving" - but of course it's not healthy to be highly activated ALL the time.
	The other thing I thought could be made clearer is what the numbers on the main screen mean - they're all different colors for the different days of the month but not sure what those numbers or colors mean
ESM functionality	There are a couple of categories I felt were missing when logging the things that happened today. On the "who are you with" screen, the option of "partner" would be useful. The "won something" category in the positive events screen was less useful. No positive event option for work
Privacy or information issues	Need trust in the app to give permission for social media sharing. So should give permission later on, perhaps after surveys, after built trust in app after some use Location information should be clarified to be postcode, not specific GPS point
Installation issues	Hard to download

Feedback

The final design feature of *MoodPrism* was the provision of a range of feedback to the user on their emotional well-being and mental health. This feedback was organized in consultation with the Australian mental health organizations *beyondblue* [66] and headspace [67], research literature on mental health and well-being, and expert advice on currently available mental health apps.

The feedback was available at several stages (see Multimedia Appendix 2):

- On the completion of a survey block, users were provided a summary of their general score on one of the surveys within that block.
- On completion of each daily report, users were provided with a color-coded brief description and custom emoticon representing their emotional state on that day. Weekly and

monthly overviews were also available when multiple ESMs were completed. On completion of 1 week's worth of ESMs, "positive mental

- health" data provided individualized feedback (based on their positive health responses), which included links to positive health websites and apps.
- On completion of 2 weeks' worth of ESMs, depression and anxiety data were collated to provide individualized feedback on mental illness risk (based on their PHQ-4 responses). Recommendations and supporting links to mental health websites or contacts were also provided, as well as advice suitable to the user's emotional functioning over the past 2 weeks.

Database Security and Storage

With such extensive and potentially identifiable information being collected by *MoodPrism*, data storage and data security became a major priority. The following considerations were

http://mental.jmir.org/2016/4/e49/

XSL-FO

made regarding data storage in adherence with industry and University [68] standards, the *Privacy and Data Protection Act* 2014, and the *Guidelines for Ethical Practice in Psychological Research Online* as outlined by the British Psychological Society [69].

Immediately following the survey collection, data were stored on the user's mobile phone prior to being uploaded encrypted into a secure database every 24 h. All data uploaded from the user's phone was stored on an Amazon Web Services server. This database was protected by a firewall and regularly updated security protocols. The data stored were anonymized at the point of upload. All potentially identifiable information was removed from the data and only the device ID was retained (functioning as a randomly generated participant code). Data were only accessible online by authorized users via Secure Shell (SSH), which authenticates server access with digital certificates and encrypted passwords. All communication between authorized users and the server also occurred through HTTPS. This ensured that all information passed between the server and the researchers was encrypted and cannot be accessed or manipulated by a third party.

With regard to social media data, explicit consent to access Facebook or Twitter accounts ("opt-in") was provided by the user. Their social media credentials were stored locally on the phone but were never uploaded to the server. All Facebook and Twitter posts' content were processed locally in the mobile phone's memory and aggregated word counts were generated. Only the aggregate word count was uploaded to the storage server.

Results

The app was initially tested by both the researchers and the app developers for minor issues and bugs. A small convenience sample of independent, nonclinical users (N=11; age range=16-52 years; 4 males, 7 females) was then recruited to test the app to generate feedback on the functionality and usability of the app to the researchers and app developers. They used *MoodPrism* daily over a 30-day period and kept notes of their user experience. Information about the study was provided to the participants and electronic consent was required before the app could be used.

The test sample was invited to provide more intensive qualitative feedback by either Web-based questionnaire (n=5) or via attendance at a focus group session (n=6). Focus group participants also provided quantitative feedback by completing the Mobile Application Rating Scale (MARS) [70]. The MARS is a multidimensional measure for trialing and rating the quality of mobile phone apps, and has demonstrated interrater reliability and internal consistency. All beta-testers were also invited to discuss or provide emailed notes on their user experience. Broad Rickard et al

questions were posed, and prompts were provided where necessary (see Table 1). (No attempt was made to analyze the emotional well-being data from the beta-testers, as the sample was small, and this aim was beyond the scope of the current paper, the primary aim of which was to provide information on the development of the app.)

Themes extracted from the comments provided via the focus group or Web-based feedback are presented in Table 2.

The testing of the app with this sample was approved by the Monash University Human Research Ethics Committee (Approval # CF14/968 – 2014000398). App development was completed in 2015 and tested over June-July 2015. The app was then revised in response to feedback received and the final version of the app prepared. The app was then released on the Google Play (Android) and Apple (iOS) stores. Future publications will report empirical data from this app, with the scope of the current publication limited to the development process only.

Feedback about the functionality and usability of the app was obtained from 11 beta-testers, who completed a standard survey of app usability, the MARS. The results are presented in Figure 3.

MARS ratings for the *MoodPrism* app exceeded the average rating for 50 apps reviewed by Stoyanov et al [70] for each MARS subscale. Highest satisfaction ratings were obtained for items relating to the app's graphics quality (eg, buttons, icons), gestural design (eg, swipes, scrolls), ease of use (eg, clear menus), credibility of the information sources, the layout aesthetics, and increased awareness of mood. Lowest ratings were obtained for entertainment value (eg, fun to use), customization options, likelihood to change behavior, motivations to address mood and interest, and likelihood to recommend to others.

The results from the focus group sessions and emailed responses from all 11 beta-testers are also summarized in Table 2.

The majority of issues identified by the beta-testers were addressed in the final version of the app. For instance, the order of positively or negatively worded options was made consistent across all questionnaires, additional information on how location and social networking data will be used was provided, with reassurance that information collected was deidentified was added, and an explanatory key was provided for interpreting colors and emoticons. The only issues that were not able to be addressed related to the integrity of psychometrically validated questionnaires (and therefore wording could not be altered), inclusion of negative content (which was important to the primary purpose of the app), or installation difficulties (as they related to the trial version only, and would not be present in the Apple and Android Web-based stores).

http://mental.jmir.org/2016/4/e49/

XSL-FO



Figure 3. Quantitative feedback: beta-tester ratings on the Mobile Application Rating Scale (MARS) subscales (N=11).

Discussion

Principal Findings

In this paper, we demonstrated how mobile phone technology could be harnessed to overcome several challenges in current mental health research and practices. Key needs we aimed to meet by developing this tool included the following: real-time monitoring of emotional functioning, assessing the full spectrum of emotional well-being, confidential access to mental health support and information when required, and to reduce obtrusiveness of regular monitoring.

MoodPrism was developed on both iOS and Android mobile phone platforms as an app to monitor emotional well-being in real time. It achieved this using ESM and collection of behavioral data via mobile phone apps (addressing challenge 1). It included assessment of daily positive psychological functioning (or "flourishing" [55]) in addition to more traditional assessment of negative psychological functioning (depression and anxiety) (addressing challenge 2). MoodPrism offered users a range of resources and links to enhance mental health literacy and access to professional mental health support, which vary depending on their current emotional functioning (addressing challenge 3). MoodPrism also incorporated voice monitoring. social networking site, and music playlist data collection as the first steps toward less obtrusive monitoring of emotional well-being for extended periods (addressing challenge 4)-although extensive algorithmic modeling will be necessary to achieve this goal. In sum, MoodPrism successfully responded to 4 key challenges in the emotional mental health domain. A number of important learnings were also achieved during this project, which may be helpful to outline for future researchers considering developing a mental health app [36].

Considerations When Developing a Research-Based Mental Health App

Development of mental health apps is a relatively young field, and the guidelines to support researchers and app developers are not yet widespread. During the development of *MoodPrism*, a number of key issues were identified that could be helpful to researchers developing apps for mental health research and practice. These issues are briefly outlined in the following and then recommendations for consideration in future research are summarized in Figure 1.

First, it is important to recognize the different priorities of app developers and researchers (and mental health practitioners). For example, the MoodPrism researchers' main goals were database integrity, psychometrically sound questionnaires, and ethical administration of sensitive content. The app developers' main goals were an enjoyable user experience, good design, simple user interface, brief page content, and anonymous data storage. Identifying these goals and coming to an agreement on how they should be prioritized could help design an app that optimizes functionality (and therefore will be used by the participants) with integrity (so that the data are suitable for analysis). With MoodPrism, the researchers' priority to maintain psychometric properties of questionnaires was in conflict with the app developers' priority for good user interface and design. Administration of long questionnaires was overcome by creating brief checkpoints or "blocks" of surveys to complete, each with a portion of feedback provided as a reward to incentivize completion of long surveys. Similarly, the developers' database priorities were guided by industry standards for data collection and storage. At times, this conflicted with the researchers' need to obtain sufficient details; for example, anonymity of social media posts initially prevented the integrity of coding processes from being verified. Coding solutions were eventually achieved,

http://mental.jmir.org/2016/4/e49/



JMIR Ment Health 2016 | vol. 3 | iss. 4 | e49 | p.8 (page number not for citation purposes)

but considerable delays could have been avoided if the database requirements were thoroughly discussed at the project's outset. When these conflicting priorities were identified, a solution was often achieved that produced the unexpected benefit of optimizing outcomes for both stakeholders. For example, the chunking of questionnaires not only improved the user experience, but also was likely to improve the validity of data as participants were less likely to fatigue, or resort to nonserious responding.

Second, sufficient time should be quarantined at the outset for planning, and at the completion for beta testing and revision. App developers' schedules can overlook the details involved in translating research requirements into the app space, and as a result underestimate the time involved. Database APIs for commercial apps also tend to have simpler output requirements than is often essential for advanced statistical analyses. A failure to identify the more complex necessities of the app's function at the outset can result in over simplistic transition of features into the app, and subsequent delays in revision to meet research needs. Time spent presenting the entire app's contents clearly up front to app developers will help avoid significant delays during development. Time should also be sufficient at the outset for complete storyboarding and wireframing of the app to ensure both parties agree on the app's format and presentation. Aesthetics that work well in commercial apps do not always translate well for research content, which may out of a necessity include lengthier content or inflexible formatting or labeling of items (eg, traditional Likert-type scales in psychological questionnaires). Samples of similar app presentations that are known to work effectively with this type of content should if possible be reviewed and the best features identified. Allowing sufficient time for planning should also ensure that clear milestone dates are set, post which no further changes or additional content can be made by researchers or practitioners until trialing. Ongoing modifications can magnify delays for app developers and confuse versions being delivered. Sufficient time when the app is being finalized is also critical. Users should be allowed a sufficient trial period to allow testing of the app in various contexts, and the schedule should also ensure that they are able to report back both individually, and where Rickard et al

possible as a part of group discussion. Focus groups are invaluable for identifying common themes across users, as well as allowing more singular experiences to emerge.

Third, communication among app developers and researchers or practitioners should be managed centrally. A flexible Web-based platform (such as "Basecamp") provides project management tools such as discussion threads, allocation of tasks, a central file repository, and reminders. Progress of tasks should be monitored regularly and updates provided when item check off is delayed. Clear assignment of tasks avoids tasks being overlooked, and ensures accountability.

Fourth, methods to evaluate the app should be included within the app itself. Commercial apps can contain simple "thumbs up" or star ratings, but this is unlikely to be sufficiently informative for research or practitioner needs. Importantly, it is helpful to obtain assessments of the various aspects of the app, including commercial considerations such as aesthetics and functionality as well as those of central interest to researchers, such as ethics or trust and integrity. Published app assessment measures such as the MARS for health apps should be considered if possible. This will allow standardization and comparability across apps in the mental health space, and to build integrity and an evidence base for improvement of mental health apps over time.

Our experiences researching and developing mental health apps have yielded a number of important practical insights of value to researchers in this field. The issues highlighted during the development of *MoodPrism*, taken together with our recommendations documented elsewhere [36], are summarized in Figure 4.

Potential Applications of *MoodPrism* in Psychological Research

The development of a research mobile phone tool such as *MoodPrism* has enormous potential within the mental health field. Several applications of *MoodPrism* currently in progress are summarized in the following to illustrate the power of flexible, real-time monitoring using this platform.

http://mental.jmir.org/2016/4/e49/



JMIR Ment Health 2016 | vol. 3 | iss. 4 | e49 | p.9 (page number not for citation purposes)

Rickard et al

Figure 4. Recommended steps for researchers engaging in the app development process.



Automated Prediction of Mental Health Risk

One of the most exciting promises for data-rich apps like MoodPrism is the development of algorithms which allow automated prediction of emotional health. This modeling could determine the minimum number of constructs required to reliably predict significant changes in emotional well-being, which could be used to inform a more streamlined and userfriendly app. Importantly, it is unlikely that any 1 or 2 variables will provide reliable prediction of such changes; a strength of MoodPrism is that it provides a breadth of variables that can be used to answer diverse and important research questions. Various algorithms may be identified, for instance, which discriminate between periods of stability and decline, and MoodPrism could then unobtrusively monitor for this change, and provide targeted mental health support to the user. This extends previous research that demonstrates feasibility of such modeling [40,41,71] by utilizing predictors already established in previous research to be associated with mental health (such as online social networking) rather than only those mobile phone sensors that are convenient to record (such as app use and activity).

Improving Emotional Self-Awareness, Mental Health Literacy, and Mental Health and Well-Being Outcomes

Bakker et al [36] detail how mental health apps can be categorized as reflection-, education-, or problem-focused. *MoodPrism* is largely a reflection-focused app aimed at improving a user's emotional self-awareness by encouraging the user to report their thoughts, feelings, or behaviors and then reflect upon them. There is also an education component in *MoodPrism* that provides access to mental health information

and resources. Use of this type of mental health app may therefore result in improvements in mental health and well-being. Kauer et al [72] found evidence that using a mobile phone app that promotes self-reflection through mood tracking can increase ESA and decrease depressive symptoms. Furthermore, rigorous study is needed to explore the mental benefits of MoodPrism and other similar health reflection-focused or education-focused apps, as very few randomized controlled trials have been conducted to investigate the efficacy of mental health apps [37]. Importantly, mobile phone technology complements traditional emotion monitoring techniques such as CBT-based recording worksheets [73,74], by increasing recording of subtle changes in behavior in real time. The innovative pairing of changes in emotional well-being with rapid delivery of mental health information has the potential to improve a user's access to relevant resources such as Web-based health portals (eg, cheadspace, eHub), or local GPs when it is needed [75-77].

Leveraging Behavioral Data on Social Media to Gain Insight Into Mental Health and Social Context

Users of social networking sites leave rich digital traces of their social behavior, which includes the structure of their friendship networks and the written interactions between connections [78-80]. The quality of interactions on social network service (SNS) has been shown to hold important relationships with mental health. Positive interactions are associated with better mental health outcomes, and negative interactions may exacerbate mental illness [81-83]. However, how certain individual characteristics might lead a user to gain benefit or detriment from their SNS use is yet to be clearly described [84]. This requires access to both SNS data and the administration

http://mental.jmir.org/2016/4/e49/

XSL-FO

of psychometrically sound surveys to profile the users of SNSs. By profiling SNS users and better tapping into the interindividual variation in SNS use, the accuracy of SNS language models for mental health prediction could be improved [85] and some of the conflicting findings around the use of SNS and its mental health impact could be disentangled [85]. Furthermore, apps like *MoodPrism* enable SNS data to be associated in real time with ESM assessments of mood and psychological surveys. Time-sensitive linking of self-reported mood change and emotional expression in SNS posts may also provide evidence to support the use of SNS data and language analysis as a tool for mood and mental health tracking overtime.

Predicting Resilience Patterns to Everyday Significant Events

Event-based resilience research explores individual capacities to maintain healthy psychological functioning in response to naturally occurring stressor events [86,87]. Previous research methodologies use cross-sectionally designed studies and typically rely on retrospective reports [88-90]. These provide only partial snapshots of an individual's capacity for resilient responding and can be subject to recall biases. The collection of MoodPrism's daily reports of psychological well-being, as well as the presence or absence of stressor events, is therefore pertinent to advancing event-based resilient research methodologies. Such methodological approaches allow for multiple snapshots in mood responding that, when compiled, create more representative, real-time observation of dynamic fluctuations that occur in an individual's mood responses to stressor events. Such data will permit a more accurate exploration and identification of the heterogeneous mood trajectories that individuals display following stressor experiences [85,87,91-93]. Favorable patterns of responding, reflecting the maintenance of psychological functioning, can be identified and profiled to explore important factors that discriminate resilient individuals from other groups that reflect less-resilient patterns of responding.

Conclusions

Development of mental health apps such as MoodPrism maximize health impact by harnessing the opportunities offered by mobile phone technology. Approximately, three quarters of the US and Australian populations own a mobile phone, and around 3 in 4 of those never leave home without their mobile device [31,94]. People check their mobile phones up to 150 times a day [30], demonstrating that mobile devices offer unprecedented access to everyday behavior. Incorporating evidence-based monitoring of emotional health into routine mobile phone apps can provide a powerful and flexible methodology for increasing personal control over one's own emotional health. Capitalizing on inbuilt tools within mobile phones-such as music players, voice recorders, and social network media-to contribute data further enhances the potential of such apps to sensitively monitor emotional health over extended periods of time, while remaining unobtrusive. People (particularly young people) often find mobile phone technologies more engaging, anonymous, and less stigmatizing than other means of accessing help, and therefore are much more likely to use this methodology [16]. The new technologies described in this paper not only complement traditional approaches or educational tools supporting mental health but also have the potential to enhance their reach by overcoming many of the barriers currently challenging the reliable surveillance of emotional well-being.

Acknowledgments

This research was funded by beyondblue.

Conflicts of Interest

None declared.

Multimedia Appendix 1

Details on 3 forms of data (automatic, experience sampling, and psychological surveys) collected from MoodPrism.

[PDF File (Adobe PDF File), 55KB - mental_v3i4e49_app1.pdf]

Multimedia Appendix 2

Feedback generated by the subjects while using MoodPrism.

[PDF File (Adobe PDF File), 594KB - mental_v3i4e49_app2.pdf]

References

- St John T, Leon L, McCulloch A. Childhood and adolescent mental health: understanding the lifetime impacts. The Mental Health Foundation. 2015. [FREE Full text] [doi: 10.1023/A:1007219227883]
- Greenspoon PJ, Saklofske DH. Toward an integration of subjective well-being and psychopathology. Social Indicators
- Research. 2001. (1) p. 81 URL: http://link.springer.com/article/10.1023/A:1007219227883 [WebCite Cache ID 6m18x9BBy]
 Westerhof GJ, Keyes CL. Mental illness and mental health: the two continua model across the lifespan. J Adult Dev 2010 Jun;17(2):110-119 [FREE Full text] [doi: 10.1007/s10804-009-9082-y] [Medline: 20502508]

http://mental.jmir.org/2016/4/e49/

XSL-FO

JMIR Ment Health 2016 | vol. 3 | iss. 4 | e49 | p.11 (page number not for citation purposes)

- Diener E, Chan MY. Happy people live longer: subjective well-being contributes to health and longevity. Appl Psychol Health Well-being 2011;3(1):1-43. [doi: 10.1111/j.1758-0854.2010.01045.x]
- Fredrickson BL. What good are positive emotions? Rev Gen Psychol 1998 Sep;2(3):300-319 [FREE Full text] [doi: 10.1037/1089-2680.2.3.300] [Medline: 21850154]
- Folkman S, Moskowitz JT. Positive affect and the other side of coping. Am Psychol 2000 Jun;55(6):647-654. [Medline: 10892207]
- Garland EL, Fredrickson B, Kring AM, Johnson DP, Meyer PS, Penn DL. Upward spirals of positive emotions counter downward spirals of negativity: insights from the broaden-and-build theory and affective neuroscience on the treatment of emotion dysfunctions and deficits in psychopathology. Clin Psychol Rev 2010 Nov;30(7):849-864 [FREE Full text] [doi: 10.1016/j.cpr.2010.03.002] [Medline: 20363063]
- Isen AM. Positive affect. In: Dalgeish T, Power MJ, editors. Handboook of Cognition and Emotion. New York: Wiley; 1999:521-539.
- Tugade MM, Fredrickson BL, Barrett LF. Psychological resilience and positive emotional granularity; examining the benefits of positive emotions on coping and health. J Pers 2004;72(6):1161-1190. [doi: 10.1111/j.1467-6494.2004.00294.x]
- Goodwin GM. Time in the course of major depressive disorder. Medicographia 2010;32:126-132. [FREE Full text]
 National Institute for Health and Care Excellence. Depression: management of depression in primary and secondary care.
- 2016. URL: <u>https://www.nice.org.uk/guidance/CG23</u> [accessed 2016-06-15] [Web/Cite Cache ID 6ig/MFJ]
- McDermott B, Baigent M, Chanen A. beyondblue Expert Working Committee Clinical Practice Guidelines: Depression in Adolescents and Young Adults. Melbourne: beyondblue; 2010.
- Hetrick SE, Thompson A, Yuen K, Finch S, Parker AG. Is there a gap between recommended and 'real world' practice in the management of depression in young people? A medical file audit of practice. BMC Health Services 2012;12:178. [doi: 10.1186/1472-6963-12-178]
- Morrato EH, Libby AM, Orton HD, Degruy 3rd FV, Brent DA, Allen R, et al. Frequency of provider contact after FDA advisory on risk of pediatric suicidality with SSRIs. Am J Psychiatry 2008 Jan;165(1):42-50. [doi: 10.1176/appi.ajp.2008.165.7.A42] [Medline: 17986680]
- Cooper PJ, Goodyer I. A community study of depression in adolescent girls: I. Estimates of symptom and syndrome prevalence. Br J Psychiatry 1993;163:369-74, 379-80. [Medline: <u>8401968</u>]
- Rickwood D, Deane FP, Wilson CJ, Ciarrochi JV. Young people's help-seeking for mental health problems. Aust eJournal Adv Ment Health 2005;4(3):218-251. URL: <u>http://ro.uow.edu.au/hbspapers/2106/ [WebCite Cache ID 6m19KUttx]</u>
- Rickwood DF, Deane FP, Wilson CJ. When and how do young people seek professional help for mental health problems? Med J Aust 2007;187(7 Suppl):S35-S39. [Medline: <u>17908023</u>]
- Burns JR, Rapee RM. Adolescent mental health literacy: young people's knowledge of depression and help seeking. J Adolesc 2006 Apr;29(2):225-239. [doi: 10.1016/j.adolescence.2005.05.004] [Medline: 15996727]
- Jorm AF, Wright A, Morgan AJ. Where to seek help for a mental disorder? National survey of the beliefs of Australian youth and their parents. Med J Aust 2007 Nov 19;187(10):556-560. [Medline: <u>18021042</u>]
- aan het Rot M, Hogenelst K, Schoevers RA. Mood disorders in everyday life: a systematic review of experience sampling and ecological momentary assessment studies. Clin Psychol Rev 2012;32(6):510-523. [Medline: <u>22721999</u>]
- Myin-Germeys I, Oorschot M, Collip D, Lataster J, Delespaul P, van Os J. Experience sampling research in psychopathology: opening the black box of daily life. Psychol Med 2009 Sep;39(9):1533-1547. [doi: <u>10.1017/S0033291708004947</u>] [Medline: <u>19215626</u>]
- Telford C, McCarthy-Jones S, Corcoran R, Rowse G. Experience Sampling Methodology studies of depression: the state of the art. Psychol Med 2012 Jun;42(6):1119-1129. [doi: 10.1017/S0033291711002200] [Medline: 22008511]
- Peeters F, Berkhof J, Delespaul P, Rottenberg J, Nicolson NA. Diurnal mood variation in major depressive disorder. Emotion 2006;6(3):383-391. [doi: 10.1037/1528-3542.6.3.383] [Medline: 16938080]
- Kenardy J, Fried L, Kraemer H, Taylor CB. Psychological precursors of panic attacks. Br J Psychiatry 1992 May;160:668-673. [Medline: 1591576]
- Ben-Zeev D, Young MA. Accuracy of hospitalized depressed patients' and healthy controls' retrospective symptom reports: an experience sampling study. J Nerv Ment Dis 2010;198(4):280-285. [Medline: 20386257]
- Miller G. The smartphone psychology manifesto. Perspect Psychol Sci 2012 May;7(3):221-237. [doi: 10.1177/1745691612441215] [Medline: 26168460]
- Randall WM, Rickard NS. Development and trial of a mobile Experience Sampling Method (m-ESM) for personal music listening. Music Perception Interdisciplinary J 2012;31(2):157-170. URL: <u>http://mp.ucpress.edu/content/31/2/157 [WebCite Cache ID 6m19NQVP0]</u> [doi: 10.1525/mp.2013.31.2.157]
- Reid SC, Kauer SD, Khor AS, Hearps SJ, Sanci LA, Kennedy AD, et al. Using a mobile phone application in youth mental health - an evaluation study. Aust Fam Physician 2012 Sep;41(9):711-714. [Medline: 22962650]
- Harrison V, Proudfoot J, Wee PP, Parker G, Pavlovic DH, Manicavasagar V. Mobile mental health: review of the emerging field and proof of concept study. J Ment Health 2011 Dec;20(6):509-524. [doi: <u>10.3109/09638237.2011.608746</u>] [Medline: <u>21988230</u>]

http://mental.jmir.org/2016/4/e49/

XSI+FO

JMIR Ment Health 2016 | vol. 3 | iss. 4 | e49 | p.12 (page number not for citation purposes)

- Google. Our mobile planet: United States of America. 2013 May. URL: <u>http://services.google.com/fh/files/misc/omp-2013-us-en.pdf [WebCite Cache ID 6a5EeTpp5]</u>
- Proudfoot J, Parker G, Hadzi PD, Manicavasagar V, Adler E, Whitton A. Community attitudes to the appropriation of mobile phones for monitoring and managing depression, anxiety, and stress. J Med Internet Res 2010;12(5):e64 [FREE Full text] [doi: 10.2196/jmir.1475] [Medline: 21169174]
- Quine S, Bernard D, Booth M, Kang M, Usherwood T, Alperstein G, et al. Health and access issues among Australian adolescents: a rural-urban comparison. Rural Remote Health 2003;3(3):245 [FREE Full text] [Medline: 15882102]
- Poushter J. Smartphone ownership and Internet usage continues to climb in emerging economies. Pew Research Center. 2016 Feb 22. URL: <u>http://www.pewglobal.org/2016/02/22/</u> <u>smartphone-ownership-and-internet-usage-continues-to-climb-in-emerging-economies/</u> [accessed 2016-06-14] [WebCite Cache ID 6iGxlcA5V]
- Burns JM, Durkin LA, Nicholas J. Mental health of young people in the United States: what role can the internet play in reducing stigma and promoting help seeking? J Adolesc Health 2009 Jul;45(1):95-97. [doi: <u>10.1016/j.jadohealth.2008.12.006</u>] [Medline: <u>19541256</u>]
- Rickwood D. Promoting youth mental health through computer-mediated communication. Int J Ment Health Promotion 2010;12(3):32-44.
- Bakker D, Kazantzis N, Rickwood D, Rickard NS. Mental health smartphone apps: review and evidence-based recommendations for future developments. JMIR Ment Health 2016;3(1):e7. [doi: <u>10.2196/mhealth.4984</u>] [Medline: <u>26932350</u>]
- Donker T, Petrie K, Proudfoot J, Clarke J, Birch MR, Christensen H. Smartphones for smarter delivery of mental health programs: a systematic review. J Med Internet Res 2013;15(11):e247. [doi: <u>10.2196/jmir.2791</u>] [Medline: <u>24240579</u>]
- Reid SC, Kauer SD, Hearps SJ, Crooke AH, Khor AS, Sanci LA, et al. A mobile phone application for the assessment and management of youth mental health problems in primary care: health service outcomes from a randomised controlled trial of mobiletype. BMC Fam Pract 2013 Jun 19;14:84 [FREE Full text] [doi: 10.1186/1471-2296-14-84] [Medline: 23782796]
- Lathia N, Pejovic V, Rachuri KK, Mascolo C, Musolesi M, Rentfrow PJ. Smartphones for large scale behavior change interventions. IEEE CS. 2013 URL: <u>http://www.cs.bham.ac.uk/~pejovicv/docs/Lathia2013IEEEPervasive.pdf</u> [accessed 2016-11-02] [WebCite Cache ID 6li61Flsy]
- LiKamWa R, Liu Y, Lane N, Zhong L. MoodScope: building a mood sensor from smartphone usage patterns. In: Proceedings of the 11th Annual International Conference on Mobile Systems, Applications, and Services. 2013 Presented at: MobiSys'13th Annual International Conference on Mobile Systems, Applications, and Services; Jun; 2013; Taipei, Taiwan p. 25-28. [doi: 10.1145/2462456.2464449]
- Asselbergs J, Ruwaard J, Ejdys M, Schrader N, Sijbrandij M, Riper H. Mobile phone-based unobtrusive ecological momentary assessment of day-to-day mood: an explorative study. J Med Internet Res 2016 Mar;18(3):e72 [FREE Full text] [doi: 10.2196/jmir.5505] [Medline: 27025287]
- Nielsen. A Nielsen report on the myths and realities of teen media trends. 2009. URL: <u>http://blog.nielsen.com/nielsenwire/</u> reports/nielsen_howteensusemedia_june09.pdf/ [accessed 2016-06-14] [WebCite Cache ID 6iGxr2fFg]
- DeChoudhury CM, Gamon M, Counts X, Horvitz E. Predicting depression via social media. Microsoft. 2013. URL: <u>http:// /research.microsoft.com/apps/pubs/default.aspx?id=192721</u> [accessed 2016-06-15] [WebCite Cache ID 6iGy6SKTF]
- Kross E, Verduyn P, Demiralp E, Park J, Lee DS, Lin N, et al. Facebook use predicts declines in subjective well-being in young adults. PLoS One 2013;8(8):e69841 [FREE Full text] [doi: 10.1371/journal.pone.0069841] [Medline: 23967061]
- Motl RW, Birnbaum AS, Kubik MY, Dishman RK. Naturally occurring changes in physical activity are inversely related to depressive symptoms during early adolescence. Psychosom Med 2004;66(3):336-342. [Medline: <u>15184692</u>]
- Thayer RE, Newman JR, McClain TM. Self-regulation of mood: strategies for changing a bad mood, raising energy, and reducing tension. J Pers Soc Psychol 1994 Nov;67(5):910-925. [Medline: <u>7983582</u>]
- De Choudhury CM. Predicting postpartum changes in emotion and behavior via social media. In: CHI '13 Proceedings of the SIGCHI Conference on Human Factors in Computing Systems. 2013 Presented at: Conference on Human Factors in Computing Systems; 2013; Paris, France p. 3267-3276. [doi: 10.1145/2470654.2466447]
- Miranda D. Music listening and mental health: variations on internalizing psychopathology. In: Macdonald R, Kreutz G, Mitchell L, editors. Music, Health, and Well-being. Oxford: Oxford University Press; 2012.
- 49. North A, Hargreaves D. The Social and Applied Psychology of Music. Oxford: Oxford University Press; 2008.
- Primack BA, Silk JS, DeLozier CR, Shadel WG, Dillman Carpentier FR, Dahl RE, et al. Using ecological momentary
 assessment to determine media use by individuals with and without major depressive disorder. Arch Pediatr Adolesc Med
 2011 Apr;165(4):360-365 [FREE Full text] [doi: 10.1001/archpediatrics.2011.27] [Medline: 21464384]
- Ooi KE, Lech M, Allen NB. Multichannel weighted speech classification system for prediction of major depression in adolescents. IEEE Trans Biomed Eng 2013 Feb;60(2):497-506. [doi: 10.1109/TBME.2012.2228646] [Medline: 23192475]
- Scherer K, Bänziger T, Roesch EB. Blueprint for Affective Computing: A Sourcebook. Oxford: Oxford University Press; 2010.

http://mental.jmir.org/2016/4/e49/

XSL-FO

JMIR Ment Health 2016 | vol. 3 | iss. 4 | e49 | p.13 (page number not for citation purposes)

- Calvo RA, D'Mello S, Gratch J, Kappas A. The Oxford Handbook of Affective Computing. Oxford: Oxford University Press; 2015.
- Keyes CL. Mental illness and/or mental health? Investigating axioms of the complete state model of health. J Consult Clin Psychol 2005 Jun;73(3):539-548. [doi: 10.1037/0022-006X.73.3.539] [Medline: 15982151]
- Suldo SM, Shaffer EJ. Looking beyond psychopathology: the dual-factor model of mental health in youth. School Psychology Review 2008;37:52-68.
- Huppert FA. Psychological well-being: evidence regarding its causes and consequences. Appl Psychol Health Well-Being 2009;1(2):137-164.
- Scherer K. Ways to study the nature and frequency of our daily emotions: reply to the commentaries on 'Emotions in everyday life'. Soc Sci Inf 2004;43:667-689.
- Kapp KM. The gamification of learning and instruction: Game-Based Methods and Strategies for Training and Education. Ist edition. San Francisco, CA: Pfeiffer; 2012.
- Pennebaker JW, Chung C, Ireland M, Gonzales A, Booth RJ. The LIWC 2007 Application. LIWC. 2016. URL: <u>http://liwc.wpengine.com/</u> [accessed 2016-06-15] [WebCite Cache ID 6iGySPeBZ]
- Kroenke K, Spitzer RL, Williams JB, Löwe B. An ultra-brief screening scale for anxiety and depression: the PHQ-4. Psychosomatics 2009;50(6):613-621. [doi: 10.1176/appi.psy.50.6.613] [Medline: 19996233]
- 62. Mehrabian A, Russell J. An Approach to Environmental Psychology. 1st edition. Cambridge, MA: MIT Press; 1974.
- Russell JA. A circumplex model of affect. J Pers Soc Psychol 1980;39(6):1161-1178.
- Bech P, Kjoller OR, Rasmussen NK. Measuring well-being rather than the absence of distress symptoms: a comparison of the SF-36 Mental Health subscale and the WHO-Five Well-Being Scale. Int J Methods Psychiatr Res 2003;12(2):85-91. [Medline: <u>12830302</u>]
- Kroenke K, Spitzer RL, Williams JB. The PHQ-9: validity of a brief depression severity measure. J Gen Intern Med 2001 Sep;16(9):606-613 [FREE Full text] [Medline: <u>11556941</u>]
- 66. beyondblue. beyondblue: Depression, Anxiety URL: https://www.beyondblue.org.au/ [WebCite Cache ID 6ld91qVDK]
- 67. Headspace. Welcome to headspace. URL: https://headspace.org.au/ [WebCite Cache ID 6ld94wTcH]
- Monash University. Privacy policy. 2016 URL: <u>http://www.privacy.monash.edu/</u> [accessed 2016-06-15] [WebCite Cache ID 6iGxlibho]
- The British Psychological Society. Research guidelines & policy documents. 2016 URL: <u>http://www.bps.org.uk/publications/</u> policy-and-guidelines/research-guidelines-policy-documents/research-guidelines-poli [accessed 2016-06-15] [WebCite Cache ID 6iGxZE8ft]
- Stoyanov S, Hides L, Kavanagh D, Zelenko O, Tjondronegoro D, Mani M. Mobile App Rating Scale: a new tool for assessing the quality of health mobile apps. JMIR mHealth uHealth 2015;3(1):e27. [doi: <u>10.2196/mhealth.3422</u>] [Medline: <u>25760773</u>]
- Picard RW, Vyza E, Healey J. Toward machine emotional intelligence: analysis of affective physiological state. IEEE Trans Pattern Analysis Machine Intell 2001;23(10):1175-1191.
- Young and Well Cooperative Research Centre. Young and Well: researching the role of technology to improve the mental health and wellbeing of young people. 2016. URL: <u>http:// www.youngandwellcrc.org.au [WebCite Cache ID 6iGyrSPAQ]</u>
- Bakker G. Practical CBT: Using Functional Analysis and Standardised Homework in Everyday Therapy. Bowen Hills, Qld: Australian Academic Press; 2008.
- 74. Kazantzis N, MacEwan J, Dattilio FM. A guiding model for practice. In: Kazantzis N, Deane FP, Ronan KR, L'Abate L, editors. Using Homework Assignments in Cognitive Behavior Therapy. New York: Routledge; 2005:359-407.
- Bennett K, Reynolds J, Christensen H, Griffiths KM. e-hub: an online self-help mental health service in the community. Med J Aust 2010;192(11 Suppl):S48-S52. [Medline: 20528710]
- McGorry P, Bates T, Birchwood M. Designing youth mental health services for the 21st century: examples from Australia, Ireland, the UK. Br J Psychol 2013;202(s54):s30-s35. [doi: 10.1192/bjp.bp.112.119214]
- Rickwood DJ, Telford NR, Mazzer KR, Parker AG, Tanti CJ, McGorry PD. The services provided to young people through the headspace centres across Australia. Med J Aust 2015 Jun 1;202(10):533-536. [Medline: <u>26021365</u>]
- Ellison N, Boyd DM. Sociality through social network sites. In: Dutton WF, editor. The Oxford Handbook of Internet Studies. Oxford: Oxford University Press; 2013:151-172.
- Kramer AD, Guillory JE, Hancock JT. Experimental evidence of massive-scale emotional contagion through social networks. In: Proc Natl Acad Sci U S A. 2014 Jun 17 Presented at: Proceedings of the National Academy of Sciences, United States; 2014 Jun 17; p. 8788-8790. [doi: 10.1073/pnas.1320040111]
- Park G, Schwartz HA, Eichstaedt JC, Kern ML, Kosinski M, Stillwell DJ, et al. Automatic personality assessment through social media language. J Pers Soc Psychol 2015;108(6):934-952. [doi: 10.1037/pspp0000020] [Medline: 25365036]

http://mental.jmir.org/2016/4/e49/

XSL-FO

JMIR Ment Health 2016 | vol. 3 | iss. 4 | e49 | p.14 (page number not for citation purposes)

- Davila J, Hershenberg R, Feinstein BA, Gorman K, Bhatia V, Starr LR. Frequency and quality of social networking among young adults: associations with depressive symptoms, rumination, and corumination. Psychol Popular Media Cult 2012;1(2):72-86. [doi: 10.1037/a0027512]
- Feinstein BA, Bhatia V, Hershenberg R, Davila J. Another venue for problematic interpersonal behavior: the effects of depressive and anxious symptoms on social networking experiences. J Soc Clin Psychol 2012 Apr;31(4):356-382. [doi: 10.1521/jscp.2012.31.4.356]
- Valkenburg PM, Peter J, Schouten AP. Friend networking sites and their relationship to adolescents' well-being and social self-esteem. Cyberpsychol Behav 2006 Oct;9(5):584-590. [doi: 10.1089/cpb.2006.9.584] [Medline: 17034326]
- Seabrook E, Kern M, Rickard NS. Social networking sites, depression, and anxiety: a systematic review. JMIR Mental Health 2016 (forthcoming).
- Bonanno GA. Loss, trauma, and human resilience: have we underestimated the human capacity to thrive after extremely aversive events? Am Psychol 2004 Jan;59(1):20-28. [doi: 10.1037/0003-066X.59.1.20] [Medline: 14736317]
- Preotiuc-Pietro D, Eichstaedt JC, Park G, Sap M, Smith L, et al. The role of personality, age, and gender in tweeting about mental illness. 2015 Presented at: Proceedings of the Workshop on Computational Linguistics and Clinical Psychology: From Linguistic Signal to Clinical Reality: NAACL; 2015; Denver Colorado.
- Bonanno GA, Diminich ED. Annual research review: positive adjustment to adversity—trajectories of minimal-impact resilience and emergent resilience. J Child Psychol Psychiatry 2013 Apr;54(4):378-401 [FREE Full text] [doi: 10.1111/jcpp.12021] [Medline: 23215790]
- Dumont M, Provost MA. Resilience in adolescents: protective role of social support, coping strategies, self-esteem, and social activities on experience of stress and depression. Journal of Youth and Adolescence 1999;28:343. [doi: 10.1023/A:1021637011732]
- Herman-Stahl M, Petersen AC. The protective role of coping and social resources for depressive symptoms among young adolescents. J Youth Adolescence 1996;25:733. [doi: 10.1007/BF01537451]
- Noor NM, Alwi A. Stressors and well-being in low socio-economic status Malaysian adolescents: the role of resilience resources. Asian Journal of Social Psychology 2013;16:292-306. [doi: 10.1111/ajsp.12035]
- Bonanno GA, Field NP, Kovacevic A, Kaltman S. Self-enhancement as a buffer against extreme adversity: civil war in Bosnia and traumatic loss in the United States. Personality and Social Psychology Bulletin 2002;28(2):184-196. [doi: 10.1177/0146167202282005]
- Ryff CD, Singer B. From social structure to biology: integrative science in pursuit of human healthwell-being. In: Snyder CR, Lopez SJ, editors. Hanbook of Positive Psychology. New York: Oxford University Press; 2002:541-555.
- Ryff CD, Singer BH, Dienberg Love G. Positive health: connecting well-being with biology. Philos Trans Royal Soc Lond B Biol Sci 2004;359(1449):1383-1394. [doi: 10.1098/rstb.2004.1521] [Medline: 15347530]
- Meeker M, Wu L. Slideshare. 2013. Meeker M, Wu L. KPCB Internet trends 2013. Slideshare. 2013. URL: <u>http://www.slideshare.net/kleinerperkins/kpcb-internet-trends-2013</u> [accessed 2016-06-14] [WebCite Cache ID 6iGyXnxlH]

Abbreviations

APIs: application programming interfaces CBT: cognitive behavioral therapy ESM: experience sampling methodology LIWC: Linguistic Inquiry and Word Count SSH: Secure shell SNS: social network service

Edited by J Strauss; submitted 14.06.16; peer-reviewed by M Prietula, D Rickwood, J Asselbergs, S Rubinelli, TR Soron; comments to author 10.07.16; revised version received 10.08.16; accepted 04.10.16; published 23.11.16

<u>Please cite as:</u> Rickard N, Arjmand HA, Bakker D, Seabrook E Development of a Mobile Phone App to Support Self-Monitoring of Emotional Well-Being: A Mental Health Digital Innovation JMIR Ment Health 2016;3(4):e49 URL: <u>http://mental.jmir.org/2016/4/e49/</u> doi:10.2196/mental.6202 PMID: <u>27881358</u>

©Nikki Rickard, Hussain-Abdulah Arjmand, David Bakker, Elizabeth Seabrook. Originally published in JMIR Mental Health (http://mental.jmir.org), 23.11.2016. This is an open-access article distributed under the terms of the Creative Commons Attribution

http://mental.jmir.org/2016/4/e49/

XSL-FO

JMIR Ment Health 2016 | vol. 3 | iss. 4 | e49 | p.15 (page number not for citation purposes)

Rickard et al

License (http://creativecommons.org/licenses/by/2.0/), which permits unrestricted use, distribution, and reproduction in any medium, provided the original work, first published in JMIR Mental Health, is properly cited. The complete bibliographic information, a link to the original publication on http://mental.jmir.org/, as well as this copyright and license information must be included.

Table 1. Details on 3 forms of data (Automatic, Experience sampling, and Psychological surveys) collected from MoodPrism.

Automat	ed data collection	Experience sampling items		Psychological questionnal	Ires	
Source	Data (downloaded once every 24 h,	Target and questions	Source	Questionnaire	Purpose	Source
	subject to user permissions)					
Facebook	For each post in history (up to last 50):	Emotional well-being	Drawn from PHQ-4	At baseline only		
	- Date and time of post	How were you feeling just before you were	[59];	Demographics	Sample description or	Custom
	- Length of message (characters)	prompted by this app? (rated on a 5-point	3 dimensional		Potential moderator or	developed
	- Number of words in message	scale from "not at all" to "extremely")	models of affect		confounding variable	
	- Number of positive, negative, self-	1 Nervous anvious or on edge	[60,61];	Life event scale	Potential moderator or	[69]
	pronoun, other-pronoun words	2. Not able to stop or control worrying	The WHO"		confounding variable	
	- Number of likes	 Little Interest or pleasure doing things Feeling down, depressed, or hopeless 	emotional well-	Multidimensional scale of	Potential moderator or	[70]
	- Number of comments	5. Active or alert	being scales (WHO-	perceived social support	confounding variable	
	- Number of tags	 regaine or unpreasant Positive or pleasant 	5) [62]; positive	Social desirability scale	Assessment of reliable	[71]
	- City and postcode (if available) of	8. In control of what I'm doing	health literature		responding	1
	where posted	 Socially connected and supported 	[55]; single item			
		Motivated, engaged, and	self-esteem	IPIP	Potential moderator or	[72]
		interested 11. Life is meaningful and with	measures [63]		confounding variable	
		purpose		Rosenberg's self-esteem	Potential moderator or	[73]
		12. Feeling good about myself			confounding variable	
				Barcelona music rewards scale	Potential moderator or	[74]
					confounding variable	
				Technology use survey	Potential moderator or	Custom
					confounding variable	developed
						except for 1
						item drawn
						from [75]
				At baseline and 1 month follow u	đ	

Multimedia Appendix 1

tional [76]	-		-being [77]			-being [78]	1			-being [/9]		tal Vignettes	ange adapted	100] 400 for		Additional	questions	drawn from	beyond blue	website.	ital [81]	ange		ital [82]	ange	
Evaluation of emo	awareness change		Evaluation of well	change		Evaluation of well	change			Evaluation of well.	change	Evaluation of men	health literacy cha								Evaluation of men	health literacy cha		Evaluation of men	health literacy cha	
Emotional self-awareness scale			Warwick Edinburg well-being	scale		PHQ-9 ⁶				GAD-/"		Mental health literacy	questionnaire								Brief resilience scales			Coping self-efficacy scale		
		Items modified	from various	stressor event	questionnaires [64-	68]																				
Event-related experiences		What's the most positive thing that's	happened to you in the past 24 h?	 Nothing positive happened 		 Positive social experience (with 	friends, family, strangers, etc)		 Obtained material item (bought 	or won something, received a	gift, etc)	 Positive exnerience at work or 		201001		 Positive experience outside of 	work or school		 Positive health or fitness 	experience		 A happy occasion (e.g., birthday, 	wedding, holiday)	Other (with ontion to tune in	detail)	1
For each tweet in timeline (up to last	50):	- Date and time of tweet	- Tweet client site or software or app	used	- Length of message (characters)	- Number of words in message	- Number of positive, negative, self-	pronoun, other-pronoun words	- Number of retweets	- Number of favourites		For each song in the user's music library	- Song Title	- Genre	- Artist	- Album	- Last Plaved	- Play count	- Pating	- Release date	- Duration					
Twitter																				Music	library					

happened to you in the past 24 hours?	Feedback questionnaire	Assessment of app	Custom
 Nothing negative happened 		quality	developed
			(although
 Negative social experience (with 			broadly
family, friends, strangers, etc)			consistent
			with factors
 Loss of valued material item 			of MARS ^b).
(misplaced, theft, etc)			
 Negative experience at school or 			
work			
 Negative experience outside of 			
work or school			
Personal health problems			
(illness, injury, etc)			
 Health problems of someone 			
close to you (illness, injury,			
death, etc)			
 Other (with option to type in 			
detail)			
Followed by options for all (except			
"nothing"):			
How positive or negative was it?			
Slightly			
Moderately			

e.
Sca
Rating
plication
Ap
Mobile
^b MARS:

	-	
	Very	
	Extremely	
	Context	
	Where are you? (drop-down selections)	
	At home	
	 At someone else's place 	
	 At work, uni, or school 	
	 At a leisure venue (eg, cinema, 	
	shops, park, sporting venue)	
	Travelling or commuting	
	Other (with option to type in	
	detail)	
	Who's with you? (Drop-down selections)	
	I'm alone	
	Mainly friends	
	Mainly family or my partner	
	Mainly work colleagues	
	Mainly strangers	
	Other	
^a WHO: World Health Organization.		
°PHQ-9: Patient Health Questionnaire-9 ^dGAD-7: General Anxiety Disorder scale-7

Б	
a	
n 8	
Ξ	
su	
5	
Ŭ	
ę	
a	
E	
о	
٦.	
臣	
ea	
Ē	
ď	
e	
В	
Ĕ	
e	
tał	
ŝ	
f	
ĕ	
E	
8	
e	
Þ	
£	
ns	
Б	
X	
60	
.Е	
gat	
Ŧ	
ě	
2	
<u>~·</u>	
臣	œ.
ea	4
Ĕ	Ľ,
tal	ñ
E.	1
Ĕ	
F	$\overline{\mathbb{C}}$
ž	23
Ĕ	5
S	8
es	3
≞	20
:	읒
Jta	Ę
e,	20
2	5
Ĵ.	e,
ن	÷
ŝ	5
Š	0
Ř	
ъ.	

59. Kroenke, K., et al., An ultra-brief screening scale for anxiety and depression: the PHQ-4. Psychosomatics 2009;50(6): p. 613-21. 60. Mehrabian, A., Russell, J.A. An approach to environmental psychology (1 ed.). Cambridge, Mass.: MIT Press, 1974.

Russell, J. A. A circumplex model of affect. Journal of Personality and Social Psychology 1980;39:1161–1178.

62. Bech, P., O.R. Kjoller, and N.K. Rasmussen, Measuring well-being rather than the absence of distress symptoms: a comparison of the SF-36 Mental Health subscale and the WHO-Five Well-Being Scale. International J Methods Psychiatr Res, 2003. 12(2): p. 85-91.

63. Robins, R. W., Hendin, H. M., Trzesniewski, K. H. Measuring global self-esteem: Construct validation of a single-item measure and the Rosenberg Self-Esteem Scale. Personality and Social Psychology Bulletin 2001;27:151-161.

64. Cheng, C. Assessment of major life events for Hong Kong adolescents: The Chinese adolescent life event scale. American Journal of Community Psychology 1997;25(1):17-33. 65. Coddington, R. D. The significance of life events as etiologic factors in the diseases of children—II. A study of a normal population. Journal of Psychosomatic Research 1972; 16(3):205-213.

66. Newcomb, M. D., Huba, G. J., Bentler, P. M. A multidimensional assessment of stressful life events among adolescents: Derivation and correlates. Journal of Health and Social Behavior 1981;22:400-415. 67. Swearingen, E. M., Cohen, L. H. Measurement of adolescents' life events: The junior high life experiences survey. American Journal of Community Psychology 1985;13(1):69-85. 68. Waaktaar, T., Borge, A. I. H., Fundingsrud, H. P., Christie, H. J., Torgersen, S. The role of stressful life events in the development of depressive symptoms in adolescence—A longitudinal community study. Journal of adolescence 2004;27(2):153-163.

Holmes, T.H., Rahe, R.H. The Social Readjustment Rating Scale. J Psychosom Res 1967;11(2): 213–8

71. Reynolds, W.M. Development of reliable and valid short forms of the Marlowe-Crowne social desirability scale. Journal of Clinical 70. Zimet, G.D., et al., The Multidimensional Scale of Perceived Social Support. Journal of Personality Assessment 1988;52(1):30-41.

72. Donnellan, M.B., Oswald, F., Baird, B., Lucas, R.E., et al The Mini-IPIP Scales: Tiny-Yet-Effective Measures of the Big Five Factors of Psychology 2006;38(1):119-125.

73. Rosenberg, M. Society and the adolescent self-image. Princeton, NJ: Princeton University Press, 1965. Personality. Psychological Assessment, 2006 ;18(2), 192-203

Mas-Herrero, E. et al. Individual differences in music reward experiences. Music Perception 2013; 31(2):118-138.

75. Young and Well Cooperative Research Centre. 2016-06-14. URL:http://www.youngandwellcrc.org.au (Accessed: 2016-06-14. (Archived by WebCite^{*} at http://www.webcitation.org/6iGyrSPAQ)

76. Kauer, S.D. et al. Self-monitoring Using Mobile Phones in the Early Stages of Adolescent Depression: Randomized Controlled Trial. Journal of Medical Internet Research 2012;14(3):e67

- 77. Tennant, R. et al The Warwick-Edinburgh Mental Well-being Scale (WEMWBS): Ddevelopment and UK validation. Health and Quality of Life Outcomes 2007;5:63
 - 78. Kroenke K, Spitzer RL, Williams JBW. The PHQ-9: Validity of a brief depression severity measure. J Gen Intern Med 2001;16: 606-613. 79. Spitzer RL, Kroenke K, Williams JBW, Löwe B. A brief measure for assessing generalized anxiety disorder: the GAD-7. Arch Intern Med
- 80. Reavley, N.J., Morgan, A.J, Jorm, F. F., Development of scales to assess mental health literacy relating to recognition of and interventions for 2006;166:1092-1097.
- 81. Smith, B.W. et al The Brief Resilience Scale: Assessing the ability to bounce back. International Journal of Behavioral Medicine 2008;15:194depression, anxiety disorders and schizophrenia/psychosis. Australian and New Zealand Journal of Psychiatry 2014; 48(1):61-9.
- 82. Chesney, M.A., et al. A validity and reliability study of the coping self-efficacy scale. Br J Health Psychol 2006;11(3): 421-437

200.

Multimedia Appendix 2

Feedback generated by the subjects while using MoodPrism

Feedback type	Trigger	Scoring	Sample
Psychological sur	veys		
Normative	When	Scoring based on published guidelines,	■ Tests ♥ 150 pc Survey 1 Survey 2 Survey 5
feedback	block of	and feedback based on published	Eve never deliberately add something that hust some one's feelings.
	surveys	normative data).	Feedback
	completed	For example:	Track you for comparing this block of surveys, which included a sevenhant surveys much included a sevenhant subscript from mark surger type feet you have from these to be surveyed of an one of the seven from the sevenhant an one of the sevenhant sevenhant an one of the sevenhant sevenhant an one of the sevenhant sevenhant and the sevenhant sevenhant sevenhant and the sevenhant sevenh
		(<45) in the lower range of positive	people in your life. Your doore intelleutes that pro-prevative that pro- have and all support acative of the three but net always when you, matchill
		health scores (less than 75% of	ок
		people)	
		(45-50) on the lower end of the	(S) Next (S)
		average range of positive	
		health scores (less than 50% of	
		people)	
		(51-56) on the higher end of the	
		average range of positive	
		health scores (more than 50%	
		of people)	
		(>56) in the higher range of positive	
		health scores (more than 75%	
		of people)	

Feedback type	Trigger	Scoring	Sample
Risk assessment			
Prompt to seek mental health support	Red flag high score	PHQ or GAD score above 15 (as per published recommendations)	Note::::::::::::::::::::::::::::::::::::

type

Experience sampling self-reports

day 1.

Trigger

Visu	al (icon,	On user

colour) and descriptions of emotional state, as well as context information. Reported either in detailed (1 day), brief form (weekly),

or overview (complete log)

format.

function

Description of On user positive request health unlocked

from day 8.





Scoring based on the sum of ESM items reflecting feelings of: positive, control, social connection or support, motivation or engagement, and meaning or purpose

- 5-10: Low score
- 11-19: Medium score
- 20-25: High score

Further information link options (rotating over time) include Smiling Mind, Healthy Habits, and Buddhify apps, and well-being websites such as "Authentic Happiness" and "Soul pancake"

Overview:



Weekly view:



Daily detail:



Feedback	Trigger	Scoring	Sample
type			
Description of	On user	Scoring based on PHQ or GAD	
depression or	request	frequency of behaviors over a 2-week	K Mood History
anxiety levels	unlocked	period (none, less than half the days,	or as accuracy Modelmedia at your majorit works to these Your majorit like to my out one of these appe which are to regeneral antifacting, or choose a website to read model.
	from day 15.	around half the days, every or most of	about this Units (<u>The here for more into</u>) Arekery, Your access fulls into the
		the days)	Denacht Horn, Bridling out mone Abnud false für denä with Breiting Denaci of Harviduk
		Summed to produce	(b) Objectives, including and highly boreful from the dright out-more about those to deal with the refug and or expected at the refused of the tens for more refusion (b) Mare (b) that for more refusion (b)
		• 0-2: low	Contraction of the second
		• 3-4: moderate	
		• 5-6: high	

Further information link options (rotating over time) include

Progress	Frequency	Day 1-7: Counts down to unlocking	247 94 854O
toward	counts and	further mood feedback (positive	Done!
entries into	countdowns	functioning feedback)	
prize draw		Day 8-14: Counts down to unlocking	Great Vice more have: It days before more wood toochack is indepled
		further mood feedback (depression or	E vertiges labor prior draw, and invest 27 water coldies to ge
		anxiety feedback)	
		Every day: Counts up number of days	LOR
		completed to yield number of entries	
		into prize-draw.	

Appendix B

Human Ethics Certificate of Approval



Monash University Human Research Ethics Committee (MUHREC) Research Office

Human Ethics Certificate of Approval

This is to certify that the project below was considered by the Monash University Human Research Ethics Committee. The Committee was satisfied that the proposal meets the requirements of the National Statement on Ethical Conduct in Human Research and has granted approval.

Project Number:	CF14/968 - 2014000398
Project Title:	Monitoring emotional wellbeing via a mobile phone app
Chief Investigator:	Assoc Prof Nikki Rickard
Approved:	From: 10 June 2014 to 10 June 2019

Terms of approval - Failure to comply with the terms below is in breach of your approval and the Australian Code for the Responsible Conduct of Research.

- 1. The Chief investigator is responsible for ensuring that permission letters are obtained, <u>if relevant</u>, before any data collection can occur at the specified organisation.
- 2. Approval is only valid whilst you hold a position at Monash University.
- It is the responsibility of the Chief Investigator to ensure that all investigators are aware of the terms of approval and to
 ensure the project is conducted as approved by MUHREC.
- You should notify MUHREC immediately of any serious or unexpected adverse effects on participants or unforeseen events
 affecting the ethical acceptability of the project.
- The Explanatory Statement must be on Monash University letterhead and the Monash University complaints clause must include your project number.
- Amendments to the approved project (including changes in personnel): Require the submission of a Request for Amendment form to MUHREC and must not begin without written approval from MUHREC. Substantial variations may require a new application.
- 7. Future correspondence: Please quote the project number and project title above in any further correspondence.
- Annual reports: Continued approval of this project is dependent on the submission of an Annual Report. This is determined by the date of your letter of approval.
- Final report: A Final Report should be provided at the conclusion of the project. MUHREC should be notified if the project is discontinued before the expected date of completion.
- 10. Monitoring: Projects may be subject to an audit or any other form of monitoring by MUHREC at any time.
- 11. Retention and storage of data: The Chief Investigator is responsible for the storage and retention of original data pertaining to a project for a minimum period of five years.



Professor Nip Thomson Chair, MUHREC

cc: Assoc Prof Dianne Vella-Brodrick; Mr Hussain-Abdulah Arjmand; Ms Elizabeth Seabrook; Mr David Bakker



Appendix C

Victorian Government Department of Education and Training Research

Permission





2015_002812

Mr Hussain-Abdulah Arjmand

Dear Mr Abdulah-Arjmand

Thank you for your application of 13 July 2015 in which you request permission to conduct research in Victorian government schools titled *Monitoring emotional wellbeing through smartphone technology*.

I am pleased to advise that on the basis of the information you have provided your research proposal is approved in principle subject to the conditions detailed below.

- The research is conducted in accordance with the final documentation you provided to the Department of Education and Training.
- Separate approval for the research needs to be sought from school principals. This is to be supported by the Department of Education and Training approved documentation and, if applicable, the letter of approval from a relevant and formally constituted Human Research Ethics Committee.
- The project is commenced within 12 months of this approval letter and any extensions or variations to your study, including those requested by an ethics committee must be submitted to the Department of Education and Training for its consideration before you proceed.
- 4. As a matter of courtesy, you advise the relevant Regional Director of the schools or governing body of the early childhood settings that you intend to approach. An outline of your research and a copy of this letter should be provided to the Regional Director or governing body.
- You acknowledge the support of the Department of Education Training in any publications arising from the research.
- The Research Agreement conditions, which include the reporting requirements at the conclusion
 of your study, are upheld. A reminder will be sent for reports not submitted by the study's
 indicative completion date.

Your setails will be dealt with in accordance with the Public Records Art 2972 and the Privacy and Data Protection Act 2014. Should you have any quetes or wish to get excess to your personal information held by this department clease context, our Privacy Officer at the accord actives.



I wish you well with your research. Should you have further questions on this matter, please contact Youla Michaels, Project Support Officer, Insights and Evidence Branch, by telephone on or by email at

Yours sincerely

Eleanor Williams Acting Director Insights and Evidence Branch

3_/08/2015



Appendix D

Participant Information Sheet and Consent Process



USER INFORMATION SHEET

Project: Monitoring emotional wellbeing via a mobile phone app

Nikki Rickard Department of Psychological Sciences email:

You are invited to take part in this study. Please read this Explanatory Statement in full before deciding whether or not to participate in this research. If you would like further information regarding any aspect of this project, you are encouraged to contact the researcher via the email address listed above.

What does the research involve?

The aim of this study is to explore whether people's mobile phone behaviours can help us understand or predict their emotional well-being.

If you agree to be in this study, you will be asked to:

- Download the "MoodPrism" app on your smartphone, and leave the app on during the 1 month research period
- (2) Give permission to the researchers to automatically access your mobile phone behaviours (anonymous). The type of data that you will be asked to give researchers access to will include your music use details (e.g., artists and songs, time you listen), social networking information (e.g., which apps you use and for how long, your friend network size, and some anonymous content) and your activity levels and locations. All this information will only be stored ANONYMOUSLY (without any way of linking it to you).
- (3) The app will ask you to complete a set of surveys at the start and end of the 1 month. These surveys will take about an hour in total to complete, but can be broken up and completed at a few different times if you wish. The end surveys will only take about 20 minutes.
- (4) The app will also ask you to answer quick questions (less than 5 minutes each) including a voice recording when prompted at different times on most days during that 1 month
- (5) The app will also deliver some mental health messages to you. This will include information about your moods (which you can access at any time), whether you are flourishing, and after 2 weeks, information about depression and anxiety levels from mental health organizations such as Beyond Blue and Headspace.

Why were you chosen for this research?

We are interested in how people's emotional health (both good and poor) might be reflected in their use of mobile phones. We are therefore seeking all sorts of people (e.g., males and females, living in the country or city, who might be feeling emotionally healthy or might not be feeling that healthy).

This invitation is being sent openly through websites and other means, and is a call for anonymous participation in this research.

Please note that you will not be able to participate in this research if you:

- · Are currently taking any psychotropic medication (e.g., antidepressants, or anti-anxiety drugs)
- do not own your own smartphone

Consenting to participate in the project and withdrawing from the research After reading this, the app will ask you whether you want to be in the research. By clicking on the "Yes" button, you are consenting to being involved in the research and give permission for the researchers to access your mobile phone behaviours during the research (for 1 month).

If you decide during the research that you no longer want to participate, you can withdraw from it at any time. The data already collected will already have been sent to us, and as it is not linked to your name in any way, we will not be able to find and withdraw that data.

Possible benefits and risks to participants

Being involved in this research has a number of benefits for you:

- (1) You can enter a draw for one of four \$AU100 gift cards. Everyone who agrees to be part of this research and completes just 7 days will go into the prize draw. For every week you complete after that, you will be awarded another entry in this draw, and if you complete the final surveys you will receive double entries to the prize draw. The prizes will be drawn in January 2017, and winners will be notified via a push notification on their device. At that time they will be invited to contact the researchers to receive their prize.
- (2) You will receive messages from experts about emotional health. These might include suggestions for improving your emotional wellbeing, or perhaps some suggestions for what you might do if you're feeling down a lot or stressed a lot.
- (3) You will be helping us understand how mobile phones might be used to improve emotional wellbeing of young people like yourself.

Being involved in the research will however involve some regular time from you over this month, and this could be inconvenient at times. Some of the questions we ask are also personal, and about your emotions or mental health – these could cause some upset or concern.

We recommend that if you do feel any distress or concern when you are involved in this research, that you contact your doctor or school/university or work's welfare officer to discuss this, or seek help from help services such as:

Headspace	Kid's Help Line	Lifeline Australia
Headspace provides mental and	Free 24 hour telephone	Lifeline Australia is a 24/7
health wellbeing support,	counselling service for young	phone counselling service.
information and services to	people aged 5-18.	
young people and their families		Phone: 13 11 14
across Australia.	Phone: 1800 551 800	Website:
www.headspace.org.au		http://www.lifeline.org.au/

Confidentiality

All information we collect from the app about you will be stored anonymously (without your name) on our secure and private servers. We will be publishing results of our study but these will there will be no way anyone could identify you and your individual information when we do this.

The only time we will ask for a name and contact details is if you wish to go into the draw to win one of the \$AU100 gift cards for being involved in the research. However, your details will be kept totally separate from the other information about you collected by the app, and the two could never be linked.

Storage and use of data

During data collection, data will be stored confidentially on a secure storage site behind a firewall. Only the research team will be able to access the server, via SSH (encrypted tunnel). All communications with the server will be via HTTPS.

Your information may also be used for future research projects by the researchers, but again only anonymously and as group data.

Results

If you are interested in finding out the results of this study, you can contact us at the end of 2017, when all data will have been analysed. Please contact where you can access our findings.

Complaints

Should you have any concerns or complaints about the conduct of the project, you are welcome to contact the

Executive Officer, Monash University Human Research Ethics (MUHREC):

Executive Officer Monash University Human Research Ethics Committee (MUHREC)

Research Office Monash University VIC 3800



Associate Professor Nikki Rickard



Informed Consent Screens in MoodPrism

Appendix E

Decitivo	Nagativa	Emaiila	Dogiti	uo Emotio		Nagativa Emai	tion Intornat
F USILIVE	negative	Emoji s	Positive Enlotion Internet Slong		Slong		
Emoji s	. 1	. /	Aight i/i huba		Slang		
=D>	:-1	:-/	Aight	J/J	Туке	ADIT	Iomo
=D	:1 	:/ T T	Aightz	JK : /1-	marvy	Acgar	gank
:D	X(Alight	J/K :1	njoy	AllC	grmbi
:-D	X-(1^1	Aite	JKS	qı	Arsed	h8
(:)	0	QQ	Airt	j2f	qool	Awk	h80r
(:-)	b (=0::::	alryt	j4t	qt	Bm	h83r
>:D<	>:-(='(Alol	J4g	qte	Boom	h8ed
:))	>:-)	&.(Aml	J4I	rofl*	Bord	h8r
:-X	(:-&	(;_;)	Aprece8	jfg	*S*	Bovered	h8red
:X	:@	:'(Apreci8	jfl	S'ok	Catwot	h8s h8t
:)	:(:'	Awes	jfk	Snog	Cba	h8t0r
:-)	:-t	:'-(Awsm	jja	Sok	Cbb	h8t3r
	:-11	:*(Awsome	jkz	Sof	Cbf	h8te
:,')	:11	:*-(Bahaha	k	Stm	Cbfa	h8tr
:-")	:t	:_(Besos	kay	Ub3r	Cbfed	hait
:")	:-@	:[Bilu	kewel	W00t	Cmeo	heyt
;)	:{	;_;	Btwilu	kewl	Woot	Cmw	htr
;-)	:-Z	=\	Btwitiailwu	kk	Xellent	Cof	id10d
\ o /	:Z	:1	Byak	kl	Xlnt	Cotf	idjit
;-(??	:(&	Chillin	koo	xtc	Dc	idot
;(@_@	:-E	Ctm	kool		Dbm	idyat
:')	(*)	:-e	Cut3	kss		Dfc	invu
:'-)	(0_0)	DX	Eil	kssd		Dgac	irhy
>-)	b-(/:(Fab	kul		Dgaf	isb
:p	%+l	:-[Fah	kute		Dgara	kmn
:P	:S	:[Fi9	101z		Dgas	kmp
:-P	:-S	:-}	Funee	laff		Eejit	lsr
:-p	(:-	:}	Funy	lal		Eedyat	lzr
:>	:(#-)	*g*	lol		Ef	nv
:->	:-(#)	Gtm	lols		Ef-ing	ofcol
1 1	1-0	:-6	Gud	lel		Effed	oh noes
^.^	\-o	:6	Gudd	liek*		Effin	oh noez
^ <u>?</u> ^	:-0	:-@	Heh	lik		Effing	p/oed
^_^	=(((:(Hppy	lk		F#cking	poed
%)	+-((:-<	Hpy	lke		F'n	ph33r
%-)	3</th <th>><</th> <th>Haha</th> <th>llc</th> <th></th> <th>f-ing</th> <th>phavl</th>	><	Haha	llc		f-ing	phavl
<3	(U)	://	18	11f		fml	phail
<u3< th=""><th>:-c</th><th></th><th>Iatb</th><th>lmao</th><th></th><th>f.m.l</th><th>r8p</th></u3<>	:-c		Iatb	lmao		f.m.l	r8p
	:c		light	lmfao		farg	r8pist
	0.0		ligh	rofl		fck	stupd
	%)		Iidl	lml		fcked	suk
	%-)		Ili	lov		fckin	sukz
	(@ @)		Ilml	luv		fcking	SUXX
	·s		Ilms	luff		feck	wrdo
	·-0		Ilu*	luvv		fk	wtf
	· •		Ilshinmp	lurve		fkd	** 11
	· <u>×</u> · <u>-</u> \$		Irly	lv		fker	
	·\$		Ite	lve		fkin	
	•Ψ		Ji	lvk		fking	

Supplementary Positive and Negative Emojis and Internet Slang for the LIWC 2007